LOCAL EXCHANGE SERVICES TARIFF

ADOPTION NOTICE

MCImetro Access Transmission Services Corp. d/b/a Verizon Access Transmission Services hereby adopts, ratifies, and makes it own, in every respect as if the same had been originally filed by it, all tariffs heretofore filed with the Public Service Commission, State of Missouri, by MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Issued: November 14, 2016 Issued by: Edwin Reese, Tariff Administrator 1300 I Street NW, Suite 400w Washington, DC 20005 Effective: November 15, 2016

FILED Missouri Public Service Commission LN-2017-0142; YL-2017-0071

ADOPTION NOTICE

MCImetro Access Transmission Services, LLC d/b/a Verizon Access Transmission Services, hereby adopts, ratifies, and makes its own, in every respect as if the same had been originally filed by it, all tariffs, schedules, rules, notices, concurrences, schedule agreements, divisions, authorities, or other instruments whatsoever filed with the Public Service Commission, State of Missouri, by MCImetro Access Transmission Services, LLC d/b/a Verizon December 23, 2005. By this notice, MCImetro Access Transmission Services, LLC d/b/a Verizon Access Transmission Services, also adopts and ratifies all supplements or amendments to any of the above tariffs etc., which MCImetro Access Transmission Services, LLC has heretofore filed with said Commission.

Issued: December 23, 2005

Carmen L. Feliciano 205 N. Michigan Avenue, Suite 1100 Chicago, Illinois 60601 Effective: January 23, 2006

Filed

Missouri Public

Service Commission

CANCELLED November 15, 2016 Missouri Public Service Commission LN-2017-0142; YL-2017-0071

LN-2006-0276

LOCAL EXCHANGE SERVICE

3rd Revised Page No. 1 Cancels 2nd Revised Page No. 1

> T T

MCImetro ACCESS TRANSMISSION SERVICES CORP. d/b/a VERIZON ACCESS TRANSMISSION SERVICES

REGULATIONS AND SCHEDULE OF INTRASTATE CHARGES APPLYING TO LOCAL EXCHANGE SERVICE WITHIN THE STATE OF MISSOURI

ISSUED: November 14, 2016

Edwin Reese Tariff Administrator 1300 I Street NW, Suite 400w Washington, DC 20**005** EFFECTIVE: November 15, 2016

FILED Missouri Public Service Commission LN-2017-0142; YL-2017-0071 MCImetro Access Transmission Services, LLC d/b/a Verizon Access Transmission Services(N)

Missouri P.S.C. Tariff No. 1 2nd Revised Page No.1 Cancels 1st Revised Page No.1

_

LOCAL EXCHANGE SERVICE

MCImetro Access Transmission Services, LLC d/b/a Verizon Access Transmission Services

REGULATIONS AND SCHEDULE OF INTRASTATE CHARGES APPLYING TO LOCAL EXCHANGE SERVICE WITHIN THE STATE OF MISSOURI

Issued: December 23, 2005 CANCELLED November 15, 2016 Missouri Public Service Commission LN-2017-0142; YL-2017-0071

h

ļ

Carmen L. Feliciano 205 N. Michigan Ave. Chicago, IL 60601



Effective: January 23, 2006

LN-2006-0276



LOCAL EXCHANGE SERVICE

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 1 CANCELS ORIGINAL PAGE NO. 1

Missouri Public Sorvico Commination

RECT) MAY 28 1999

CANCELLED

January 23, 2006

MISSOURI PUBLIC SERVICE COMMISSION

Т

MCImetro ACCESS TRANSMISSION SERVICES, LLC

REGULATIONS AND SCHEDULE OF INTRASTATE CHARGES APPLYING TO LOCAL EXCHANGE SERVICE WITHIN THE STATE OF MISSOURI

> Sorvier Commission FILED JUN 27 1999

> > EFFECTIVE: June 27, 1999

ISSUED : May 28 , 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ş



MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 1

LOCAL EXCHANGE SERVICE

RECEIVED

SEP 17 1998

MO. FUBLIC SERVICE COMM

MCImetro ACCESS TRANSMISSION SERVICES, INC.

REGULATIONS AND SCHEDULE OF INTRASTATE CHARGES APPLYING TO LOCAL EXCHANGE SERVICE WITHIN THE STATE OF MISSOURI

CANCELLED

JUN 2.7 1999 By (St NU) Park ND . | Public Service Commission MISSOURI

> Missouri Public Sarvica Commission

FILED MIAY 1 1 1999

EFFECTIVE: May 14, 1999

ISSUED : March 2, 1999

Sandy Chandler Teriff Manager Six Concourse Perkway, Suite 3200 Atlanta, GA 30328 96-355

LOCAL EXCHANGE SERVICE

TABLE OF CONTENTS

SEP 17 1998

MO. PUBLIC SERVIC PAGENRA

TABLE OF CONTEN	NTS			2
EXPLANATION OF SYMBOLS				4
APPLICATION OF	TARIFF			5
LIST OF WAIVED S	STATUTES A	ND REGUL	ATIONS	6
1.	DEFINIT	IONS		7
2.	REGULA	TIONS		
	2.1	Undertak	ing of the Company	12
		2.1.1	Scope	12
		2.1.2	Shortage of Equipment or Facilities	11
		2.1.3	Terms and Conditions	12
		2.1.4	Liability of the Company	14
		2.1.5	Notification of Service-Affecting Activities	19
	2.1.6	Provision	of Equipment & Facilities	19
		2.1.7	Non-routine Installation	20
		2.1.8	Ownership of Facilities CANCELLED	20
	2.2	Prohibite	d Uses	21
	2.3	Obligatio	ins of the Customer	
		2.3.1	General AUG 2 0 1999	22
	2.4	Custome	r Equipment and Channels	
		2.4.1	$\mathbf{B}\mathbf{V}$ \mathbf{A} $-\pi \mathbf{A}$ \mathbf{C}	25
		2.4.2	Station Equipment Public Service Commission Interconnection of Facilities	25
		2.4.3	Interconnection of Facilities MISSOUR	25
		2.4.4	Inspections	26
	2.5		t Arrangements	
		2.5.1	Payment for Service	27
		2.5.2	Billing and Collection of Charges	28
		2.5.3	Disputed Bills	28
		2.5.4	Advance Payments	28
		2.5.5	Deposits	29
		2.5.6	Discontinuance of Service	30
	2.6	Allowand	ces for Interruptions in Service	
		2.6.1	Credit for Interruptions	33
		2.6.2	Limitations on Allowances	33
		2.6.3	Use of Alternative Service Provided by the	34
			Company _ Misso	
			Sorvico	Commicgion

FILED MAY 1 & 1999

EFFECTIVE: May 14, 1999

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

LOCAL EXCHANGE SERVICE TABLE OF CONTENTS (Cont'd)

2.	DECUI	A T T ON C	PAGE NO.
Ζ.	REGUI	LATIONS	
	2.8	Cancellation of Service 2.7.1 Cancellation of Application for Service 2.7.2 Cancellation of Service by the Customer Transfers and Assignments Notices and Communications	34 34 35 35
3.	SERVI	ICE DESCRIPTIONS	
		Reserved For Future Use	D D D
3.2	Direc	ctory Assistance	56
3.3		Reserved For Future Use 3.3.1 3.3.2	D D
	3.9	Services (Enhanced 911) Telecommunication Relay Service (TRS) Residential Service Small Business Service	D D 63 63.1 63.18
4.	Promo	otional Offerings	64
5.	Speci	al Construction	65
6.	Misce	ellaneous Services	66

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

FILED Missouri Public Service Commission

.

MISSOURI P.S.C. TARIFF NO. 1 1st REVISED PAGE NO. 3 CANCELS ORIGINAL PAGE NO. 3

LOCAL EXCHANGE SERVICE TABLE OF CONTENTS (Cont.)

2.	REGUL	ATIONS	PAGE NO.
	2.7	Cancellation of Service 2.7.1 Cancellation of Application for Service 2.7.2 Cancellation of Service by the Customer Transfers and Assignments	34 34 35
	2.9	Notices and Communications	35
3.	SERVI	CE DESCRIPTIONS	
	3.1	Local Exchange Service 3.1.1 Service Area 3.1.2 Local Line 3.1.3 Local Trunk 3.1.4 Usage Rates 3.1.5 Local ISDN-PRI 3.1.6 Digital T-1 Service	36 38 41 49 50 55
3.2	Direc	tory Assistance	56
	3.3	Operator Assistance 3.3.1 Operator Assisted Surcharges 3.3.2 Busy Line Verification and Interrupt Service	57 57 57
	3.4	Directory Listings	60
	3.5 3.6 3.7 3.8 3.9 3.10		62 62 63 63.1 63.18
4.	Promo	tional Offerings	64
5.	Speci	al Construction	65
6.	Misce	ellaneous Services	66

T T

Issued: May 1, 2003

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601



REGULATIONS

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 3

LOCAL EXCHANGE SERVICE

TABLE OF CONTENTS (Cont.)

SEP 17 1998

RECEIVED

			MO. PUBLIC	SERVICE COMM
	2.7	Cancella	tion of Service	•=••••
		2.7.1	Cancellation of Application for Service	34
		2.7.2	Cancellation of Service by the Customer	34
	2.8	Transfer	s and Assignments	35
	2.9	Notices	and Communications	35
3.	SERVIC	e descrif	TIONS	
	3.1	Local Ex	change Service	
		3.1.1	Service Area	36
		3.1.2	Local Line	38
		3.1.3	Local Trunk	41
		3.1.4	Usage Rates	49
		3.1.5	Local ISDN-PRI	50
		3.1.6	Digital T-1 Service	55
	3.2	Directory	Assistance	56
	3.3	Operato	Assistance	57
		3.3.1	Operator Assisted Surcharges	57
		3.3.2	Busy Line Verification and Interrupt Service	57
	3.4	Director	/ Listings	60
	3.5	Emerge	ncy Services (Enhanced 911)	62
	3.6	Presubs	cription	62
	3.7	Vanity T	elephone Numbers CANCELLED	63
	3.8	Telecom	munication Relay Service (TRS)	63
4.	Promoti	ional Offering	JUN 0 1 2003	64
5.	Special	Construction	Public Service Commission	DA 65
6.	Miscella	aneous Servi	MISSOUHI	66

CANCELLED BA AUG 2 0 1999 By A-98-515 Public Service Commission MISSOURI

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Sorvice Commission

FILFD MAY 1939

(SSUED : March 2, 1999

i

l

ł

.

2.

RECEIVED

SEP 17 1998

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MO. PUBLIC SEMISSOURIPS: TARIFF NO. 1 ORIGINAL PAGE NO. 4

LOCAL EXCHANGE SERVICE

EXPLANATION OF SYMBOLS, REFERENCE MARKS, AND ABBREVIATIONS OF TECHNICAL TERMS USED IN THIS TARIFF

The following symbols shall be used in this tariff for the purpose indicated below:

C	-	To signify changed regulation.
D	-	To signify discontinued rate or regulation.
ł	-	To signify increased rate.
м	-	To signify a move in the location of text.
N	-	To signify new rate or regulation.
R	-	To signify reduced rate.
S	-	To signify reissued matter.
т	-	To signify a change in text but no change in rate or regulation.

CANCELIZED DE AUG 2 1999 TA-98-575 Public Service Commission MISSOURI

ISSUED : March 2, 1999

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

FILED MAY 18 1999

Missouri Public Sorvieo Commission

96-355



٤

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 5 CANCELS ORIGINAL PAGE NO. 5 MISSOURI PUBLIC BOMICO COMMISSION

Т

REC'D MAY 28 1999

LOCAL EXCHANGE SERVICE

APPLICATION OF TARIFF

This tariff sets forth the service offerings, rates, terms and conditions applicable to the furnishing of intrastate facilities based business communications services by MCImetro Access Transmission Services, LLC, to business Customers within the local exchange service area defined herein.

CANCELLED AUG 2 0 1999 By TAX98-575 Public Service Commission MISSOUR

Miceouri Public Somico Commiceica

FILED JUN 27 1999

ISSUED : May 28, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: June 27, 1999

RECEIVED MISSOURI P.S.C. TARIFF NO. 1 SEP 17 1998 ORIGINAL PAGE NO. 5

LOCAL EXCHANGE SERVICE

APPLICATION OF TARIFF

MO. PUBLIC SERVICE COMM

This tariff sets forth the service offerings, rates, terms and conditions applicable to the furnishing of intrastate facilities based business communications services by MCImetro Access Transmission Services, Inc., to business Customers within the local exchange service area defined herein.

CANCELLED JUN 27 1999 By St AW Commission Public Service Commission MISSOURI

96-355

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILED MAY 1 & 1999

LOCAL EXCHANGE SERVICE

LIST OF WAIVED STATUTES AND REGULATIONS

The Missouri Public Service Commission in its orders granting the Application of MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services for a Certificate of Authority to Provide Basic Local Services within the State of Missouri, Case No. TA-96-355 and in the Matter of the Application of MCImetro Access Transmission Services, LLC, d/b/a Verizon Access Transmission Services, to Expand its Certificate of Basic Local Service Authority to Include Provision of Basic Local Exchange Telecommunications Service Throughout the State of Missouri, and to Continue to Classify the Company and its Services as Competitive, Case No. TA-2009-0083, waived the following statutes and regulations regarding basic local telecommunications service:

STATUTES

Section 392.210.2	-	uniform system of accounts
Section 392,140 (1)	-	just and reasonable rates
Section 392,270	-	valuation of property (ratemaking)
Section 392.280	-	Depreciation accounts
Section 392.290.1	-	issuance of securities
Section 392,300.2	-	acquisition of stock
Section 392.310	-	stock and debt issuance
Section 392.320	-	stock dividend payment
Section 392.330	-	issuance of securities; debts and notes
Section 392,340	-	reorganizations

COMMISSION RULES

4 CSR 240-10.020 4 CSR 240-30.040	-	depreciation fund income uniform system of accounts
4 CSR 240-3.550(4)	-	records of applications
4 CSR 240-3.550(5) (A)	-	quality of service quarterly report
4 CSR 240-3.550(5) (C)) -	exchange boundary maps
4 CSR 240-32.070		engineering and maintenance
4 CSR 240-32.070	-	quality of service
4 CSR 240-32.080	-	service objectives and surveillance levels
4 CSR 240-33.040(1)-(3	3)	
and (5)-(10)	-	billing and payment standards
4 CSR 240-33.045	-	customer bills
4 CSR 240-33.080(1)	-	company name for billing disputes
4 CSR 240-33.130(1),		
(4), and (5)	-	operator service

ALL MATERIAL ON THIS PAGE IS NEW

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 10, 2008

FILED Missouri Public Service Commission MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

MISSOURI. P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 6 CANCELS ORIGINAL PAGE NO. 6

Ţ

LOCAL EXCHANGE SERVICE

LIST OF WAIVED STATUTES AND REGULATIONS

Sorviso Commission RECTO MAY 28 1999

The Missouri Public Service Commission in its orders granting the Application of MCImetro Access Transmission Services, LLC for a Certificate of Authority to Provide Basic Local Services within the State of Missouri, Case No. TA-96-355 waived the following statutes and regulations regarding basic local telecommunications service:

STATUTES

Section 392.210.2	-	uniform system of accounts
Section 392.270	-	valuation of property (ratemaking)
Section 392.280	-	Depreciation accounts
Section 392.290.1	-	issuance of securities
Section 392.300.2	-	acquisition of stock
Section 392.310	-	stock and debt issuance
Section 392.320	-	stock dividend payment
Section 392.330	-	issuance of securities; debts and notes
Section 392.340	-	reorganizations

COMMISSION RULES

4 CSR 240-10.020	-	depreciation fund income
4 CSR 240-30.040	-	uniform system of accounts
4 CSR 240-35	-	reporting of bypass and customer
		specific arrangements



ISSUED : May 28 , 1999

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: June 27, 1999

RECEIVED

MISSOURI, P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 6

SEP 17 1998

LOCAL EXCHANGE SERVICE

MO. PUBLIC SERVICE COMM

LIST OF WAIVED STATUTES AND REGULATIONS

The Missouri Public Service Commission in its orders granting the Application of MCImetro Access Transmission Services, Inc. for a Certificate of Authority to Provide Basic Local Services within the State of Missouri, Case No. TA-96-355 waived the following statutes and regulations regarding basic local telecommunications service:

STATUTES

Section 392.210.2	-	uniform system of accounts
Section 392.270	-	valuation of property (ratemaking)
Section 392.280	-	Depreciation accounts
Section 392.290.1	-	issuance of securities
Section 392.300.2	-	acquisition of stock
Section 392.310	-	stock and debt issuance
Section 392.320	•	stock dividend payment
Section 392.330	-	issuance of securities; debts and notes
Section 392.340	-	reorganizations

COMMISSION RULES

4 CSR 240-10,020	-	depreciation fund income
4 CSR 240-30.040	-	uniform system of accounts
4 CSR 240-35	-	reporting of bypass and customer
		specific arrangements

CANCELLED

JUN 2 7 1999 By Str. Pg. D. C. Public Service Commission MISSOURI

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 96-355

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILED MAY 1 & 1999

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 7 CANCELS ORIGINAL PAGE NO. 7

Missouri Public Service Commission

RECD MAR 0 2 2001

LOCAL EXCHANGE SERVICE

1. Definitions

Certain terms used generally throughout this tariff are defined below.

Advance Payment: Payment of all or part of a charge required before the start of service.

Authorized User: A person, firm, corporation or other entity that either is authorized by the Customer to use local exchange telephone service or is placed in a position by the Customer, either through acts or omissions, to use local exchange telephone service.

Business Communications Service: A switched network service which provides for dial Station origination for which the subscriber pays a rate that is described as a business or commercial rate.

Call Forward Busy: Automatically routes incoming calls to a designated answering point when the called line is busy.

Call Forward No Answer. Automatically routes incoming calls to a designated answering point when the called line does not answer within a pre-specified number of rings.

<u>Call Forward Variable</u>: Automatically routes incoming calls to a designated answering point, regardless of whether the user's Station is idle or busy.

Call Hold: Allows the User to hold one call for any length of time provided that neither party goes On Hook.

Call Park: Allows a User to "park" a call against their directory number within the business group and "unpark" the call from any other directory number. A business group consists of a series of Customer-defined telephone numbers.

<u>Call Pickup</u>: Allows a User to answer incoming calls to another Station line within a defined call pickup group. Call Pickup is provided as either Group Call Pickup, where predesignated groups can pickup each other's calls by activating an access code or a feature key, or Directed Call Pickup, where any call can be retrieved by dialing a different access code followed by the extension number.

Call Transfer/Consultation/Conference: Provides the capability to transfer or add a third party, using the same line.

<u>Call Waiting</u>: Provides the User with a burst of tone to indicate that another call is waiting. The second call can either be answered by flashing the switchhook or hanging up the phone and being rung back by the caller.

Call Waiting Cancel: Allows a User to cancel the Call Waiting feature on a per call basis by diating a specific two digit code.

Caller ID with Name and Number. Allows the subscriber to view the name and phone number of the calling party before the phone is answered.



Ν

FLED APR 01 2001

EFFECTIVE: April 1, 2001

ISSUED : March 2, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328



N

RECEIVED

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 7

LOCAL EXCHANGE SERVICE SEP 17 1998

1. Definitions

MO. PUBLIC SERVICE COMM

Advance Payment: Payment of all or part of a charge required before the start of service.

Certain terms used generally throughout this tariff are defined below

<u>Authorized User</u>: A person, firm, corporation or other entity that either is authorized by the Customer to use local exchange telephone service or is placed in a position by the Customer, either through acts or omissions, to use local exchange telephone service.

Business Communications Service: A switched network service which provides for dial Station origination for which the subscriber pays a rate that is described as a business or commercial rate.

Call Forward Busy: Automatically routes incoming calls to a designated answering point when the called line is busy.

<u>Call Forward No Answer</u>: Automatically routes incoming calls to a designated answering point when the called line does not answer within a pre-specified number of rings.

<u>Call Forward Variable</u>: Automatically routes incoming calls to a designated answering point, regardless of whether the user's Station is idle or busy.

Call Hold: Allows the User to hold one call for any length of time provided that neither party goes On Hook.

Call Park: Allows a User to "park" a call against their directory number within the business group and "unpark" the call from any other directory number. A business group consists of a series of Customer-defined telephone numbers.

<u>Call Pickup</u>: Allows a User to answer incoming calls to another Station line within a defined call pickup group. Call Pickup is provided as either Group Call Pickup, where predesignated groups can pickup each other's calls by activating an access code or a feature key, or Directed Call Pickup, where any call can be retrieved by dialing a different access code followed by the extension number.

Call Transfer/Consultation/Conference: Provides the capability to transfer or add a third party, using the same line.

<u>Call Waiting</u>: Provides the User with a burst of tone to indicate that another call is waiting. The second call can either be answered by flashing the switchhook or hanging up the phone and being rung back by the caller.

Call Waiting Cancel: Allows a User to cancel the Call Waiting feature on a per call basis by dialing a specific two digit code.



95-355

EFFECTIVE: May 14, 1999

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Sorvico Commission

FILED MAY 14 1999

RECEIVED

SEP 17 1998 ORIGINAL PAGE NO. 1

LOCAL EXCHANGE SERVICE MO. FUBLIC SERVICE CUNIVI

1. Definitions (Cont'd)

<u>Calling Number Delivery</u>: This feature enables the customer to view on a display unit the calling party directory name and/or number (CPN) on incoming telephone calls. When a Caller ID is activated on a customer's line, the CPN of incoming calls are displayed at the called CPE during the first, long silent interval of the ringing cycle.

Per line blocking for the blocking of CPN will be available upon request, at no charge, to the following entities for lines over which the official business of the agency is conducted, including those at the residences of employees/volunteers, where an executive officer of the agency registers a need for blocking and provides the required certification to MCImetro a) private, nonprofit, tax exempt, domestic violence intervention agencies and b) federal, state, and local law enforcement agencies. The CPN will not be transmitted from a line equipped with this capability. Per line blocking is operational on a continuous basis but can be deactivated by the customer by dialing an access code immediately prior to placing a call. Line blocking customers can unblock their CPN information on a per call basis, at no charge, by dialing an access code (*82 on their touch tone pad or 1182 from a rotary phone) immediately prior to placing a call.

A customer may prevent the delivery of their calling name and/or number to the called party by dialing an access code (*67 on their touch tone pad or 1167 from a rotary phone) immediately prior to placing a call. The access code will activate per call blocking, which is available at no charge. If the calling party activated blocking, the CPN will not be transmitted across the line to the called party. Instead, Calling Line Identification customers will receive an anonymous indicator. This anonymous indicator notifies the Caller ID customer that the calling party has elected to block the delivery of their name and telephone number. The blocking of CPN will not be provided on calls originating from Customer Owned Pay Telephones. If the Caller ID customer also subscribes to Anonymous Call Rejection, the calling party will be routed to a telephone company recording advising the caller that the caller that the calling party will be routed to a telephone company recording advising the caller that the caller that the caller party will be routed to a telephone company recording advising the caller that the caller that the caller party will be routed to a telephone company recording advising the caller that the caller party will not accept calls whose CPN has been blocked.

Any customer subscribing to Caller ID will be responsible for the provision of a display device, which will be located on the customer's premises. The installation, repair and technical capability of that equipment to function in conjunction with the feature specified herein will be the responsibility of the customer. MCImetro assumes no liability and will be held harmless for any incompatibility of this equipment to perform satisfactorily with the network features described herein.

Telephone CPN information transmitted via Caller ID is intended solely for the use of the Caller ID subscriber. Resale of this information is prohibited by this tariff. CPN will not be displayed if the called party is off-hook or if the called party answers during the first ring interval. CPN will be displayed for calls made from another central office only if it is linked by appropriate facilities. Caller ID is not available on operator handled calls.

CANCELLED
AUG 2-0 1999
AUG 20 1333 D: TAX-98-575
By TA 98-575 Public Service Commission MISSOURI

96-355

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missouri Public Sorvico Commission

FILER MAY 1 4 1999

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 9 CANCELS ORIGINAL PAGE NO. 9 Missouri Public Borvico Commissión

RECT) MAY 28 1999

т

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

Calling Number Delivery Blocking: An optional feature which allows a customer to block the delivery of their telephone number to the called party's destination display on a selective or complete basis.

Selective Blocking allows a customer to activate and deactivate on a per-call basis Calling Number Delivery Blocking using a feature access code prior to placing an outgoing call. Selective Blocking does not prevent the delivery of telephone numbers to 911 emergency providers.

Complete Blocking allows a customer to request Calling Number Delivery Blocking on the customer's line or trunk which automatically blocks the delivery of the caller's number on all outgoing calls. Complete Blocking does not prevent the delivery of telephone numbers to 911 emergency providers.

Class of Service (COS): Used to prevent a Station from dialing certain codes and numbers.

Company: MCImetro Access Transmission Services, LLC, a Delaware Limited Liability Corporation, which is the issuer of this tariff.

Conference/Six-Way: The User can sequentially call up to five other people and add them together to make up a six-way call.

Customer: The person, firm, corporation or other entity which orders service and is responsible for the payment of charges and for compliance with the Company's tariff regulations.

Customer Group Dialing Plan: A dialing scheme shared by the members of a Customer group, such as 4 digit internal dialing.

Dial Pulse (DP): The pulse type employed by rotary dial Station sets.

Direct Inward Dialing (DID): A service attribute that routes incoming calls directly to Stations, by-passing a central answering point.

Do Not Disturb: Allows the User to prevent incoming calls from ringing its line by diverting them to a tone or a recorded announcement that informs the caller that the User is not accepting calls at this time.

Dual Tone Multi-Frequency ("DTMF"): The pulse type employed by tone dial Station sets.

Exchange Carrier: Any individual, partnership, association, joint-stock company, trust, governmental entity or corporation engaged in the provision of local exchange telephone service.

High Capacity: A High Capacity Inbound Line or Trunk is any inbound line or trunk for which, during any monthly billing period and at any location or individual building address of a customer, the following two conditions are met: 1)more than 70 percent of the traffic carried is inbound local and 2)the average off-hook time per call is more than 10 minutes. Monthly charges shall apply to each High Capacity Inbound Line or Trunk used by the Customer. These charges are in lieu of other monthly recurring local line, Local Trunk-Basic (Per Call/Per Minute options/ Unlimited options), and Local Trunk-DID and are in addition to non-recurring and methods a

FILED JUN 27 1999

EFFECTIVE: June 27, 1999

ISSUED : May 28, 1999

ISSION

Sérvice (MISSO

.2.86 1999

0

2

AUG

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

RECEIVED

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 9**

SEP 17 1998

LOCAL EXCHANGE SERVICE MO. PUBLIC SERVICE CUMM

1. Definitions (Cont'd)

Calling Number Delivery Blocking: An optional feature which allows a customer to block the delivery of their telephone number to the called party's destination display on a selective or complete basis.

Selective Blocking allows a customer to activate and deactivate on a per-call basis Calling Number Delivery Blocking using a feature access code prior to placing an outgoing call. Selective Blocking does not prevent the delivery of telephone numbers to 911 emergency providers.

Complete Blocking allows a customer to request Calling Number Delivery Blocking on the customer's line or trunk which automatically blocks the delivery of the caller's number on all outgoing calls. Complete Blocking does not prevent the delivery of telephone numbers to 911 emergency providers.

Class of Service (COS): Used to prevent a Station from dialing certain codes and numbers.

Company: MCImetro Access Transmission Services, Inc., a Delaware corporation, which is the issuer of this tariff.

Conference/Six-Way: The User can sequentially call up to five other people and edd them together to make up a six-way call.

Customer: The person, firm, corporation or other entity which orders service and is responsible for the payment of charges and for compliance with the Company's tariff regulations.

Customer Group Dialing Plan: A dialing scheme shared by the members of a Customer group, such as 4 digit internal dialing.

Dial Pulse (DP): The pulse type employed by rotary dial Station sets.

Direct Inward Dialing (DID): A service attribute that routes incoming calls directly to Stations, by-passing a central answering point.

Do Not Disturb: Allows the User to prevent incoming cells from ringing its fine by diverting them to a tone or a recorded.

Dual Tone Multi-Frequency ("DTMF"): The pulse type employed by tone dial Station sets.

Exchange Carrier: Any individual, partnership, association, joint-stock company, trust, governmental entity or corporation from the provision of local exchange telephone service. . ublic Service Commission

High Capacity: A High Capacity Inbound Line or Trunk is any inbound line or trunk for which, during any monthly billing period and at any location or individual building period and at any location or individual building address of a customer, the following two conditions are met: 1)more than 70 percent of the traffic carried is inbound local and 2) the average off-hook time per call is more than 10 minutes. Monthly charges shall apply to each High Capacity Inbound Line or Trunk used by the Customer. These charges are in lieu of other monthly recurring local line, Local Trunk-Basic (Per Call/Per Minute options/ Unlimited options), and Local Trunk-DiD and are in addition to non-recurring and per minute usage charges specified elsewhere in this tariff.

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECTIVE: May 14, 1999

Missouri Public Sorvice Commission

96-355

JUN 27 1999

FILED MAY 1 4 1999

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 2nd Revised Page No. 10 Cancels 1st Revised Page No. 10

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

Hunting: Routes a call to an idle Station line. With Serial Hunting, calls to a member of a hunt group will search from that point to the end of the group and stop. With Circular Hunting, calls directed to busy Station in the middle of a hunt group will "wrap around" to the beginning of the hunt list and search for an idle Station. The caller is connected to the first idle Station encountered.

Joint User: A person, firm or corporation designated by the Customer as a user of local exchange service furnished to the Customer by the Company, and to whom a portion of the charges for such facilities are billed under a joint use arrangement.

LATA: A local access and transport area established pursuant to the Modification of Final Judgment entered by the United States District Court for Missouri in Civil Action No. 82-0192 for the provision and administration of communications services.

D

Local Calling: A completed call or telephonic communication between a calling Station and any other Station within the local service area of the calling Station.

Local Exchange Carrier: A company which furnishes exchange telephone service.

Mbps: Megabits, or millions of bits, per second.

<u>Message Waiting</u>: This feature provides an indication to a Station User that a message is waiting. Indications may be visual (lamp) or audible (stuttered dialtone).

Most Idle Trunk Selection (MIDL): MIDL Trunk selection occurs when a switching unit selects from a Trunk group the Trunk that has been idle for the longest period of time.

<u>Multiple Appearance Directory Numbers</u>: A directory number that is assigned more than once to one or more Proprietary Business Sets.

<u>Non-Qualified Commercial Affinity Group</u>: an association or affiliation of business entities whose members/individuals are not organized for the sole purpose to qualify for the discounts described herein.

Non-Qualified Residential Affinity Group: an association or affiliation of residential users whose members/individuals are not organized for the sole purpose to qualify for the discounts described herein.

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 10 Cancels Original Page No. 10

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

<u>Hunting</u>: Routes a call to an idle Station line. With Serial Hunting, calls to a member of a hunt group will search from that point to the end of the group and stop. With Circular Hunting, calls directed to busy Station in the middle of a hunt group will "wrap around" to the beginning of the hunt list and search for an idle Station. The caller is connected to the first idle Station encountered.

<u>Individual Case Basis:</u> Rates for Special Access services will be determined on an Individual Case Basis (ICB). ICB rates will be structured to recover the Company's cost of providing the services and will be made available to customers in a non-discriminatory manner. Terms of specific ICB contracts will be made available to the Missouri Public Service Commission Staff upon request on a proprietary basis.

Interim Local Number Portability (ILNP) (via Remote Call Forwarding (RCF)): Remote Call Forwarding allows a Local Line Customer to retain its existing Local Exchange Carrier telephone number by using this feature to port its telephone number to MCI Local Service. Monthly charges will apply per telephone number. A number of paths can be associated with this number to permit multiple simultaneous calls to be received. A maximum of 99 paths can be associated with a single ported number.

<u>Joint User</u>: A person, firm or corporation designated by the Customer as a user of local exchange service furnished to the Customer by the Company, and to whom a portion of the charges for such facilities are billed under a joint use arrangement.

<u>LATA</u>: A local access and transport area established pursuant to the Modification of Final Judgment entered by the United States District Court for Missouri in Civil Action No. 82-0192 for the provision and administration of communications services.

Least Idle Trunk Selection (LIDL): LIDL trunk selection occurs when a switching unit selects from a Trunk group the Trunk that has been idle for the shortest period of time.

Local Calling: A completed call or telephonic communication between a calling Station and any other Station within the local service area of the calling Station.

Local Exchange Carrier: A company which furnishes exchange telephone service.

<u>Mbps:</u> Megabits, or millions of bits, per second.

<u>Message Waiting</u>: This feature provides an indication to a Station User that a message is waiting. Indications may be visual (lamp) or audible (stuttered dialtone).

<u>Most Idle Trunk Selection (MIDL)</u>: MIDL Trunk selection occurs when a switching unit selects from a Trunk group the Trunk that has been idle for the longest period of time.

<u>Multiple Appearance Directory Numbers</u>: A directory number that is assigned more than once to one or more Proprietary Business Sets.

Non-Qualified Commercial Affinity Group: an association or affiliation N of business entities whose members/individuals are not organized for the sole purpose to qualify for the discounts described herein.

Non-Qualified Residential Affinity Group: an association or affiliation of residential users whose members/individuals are not organized for the sole purpose to qualify for the discounts described herein.



N



1.

RECEIVED MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 10**

mmission LOCAL EXCHANGE SERVICE SEP 17 1998

MO, PUBLIC SERVICE CUMM

Hunting: Routes a call to an idle Station line. With Serial Hunting, calls to a member of a hunt group will search from that point to the end of the group and stop. With Circular Hunting, calls directed to busy Station in the middle of a hunt group will "wrap around" to the beginning of the hunt list and search for an idle Station. The caller is connected to the first idle Station encountered.

Individual Case Basis: Rates for Special Access services will be determined on an Individual Case Basis (ICB). ICB rates will be structured to recover the Company's cost of providing the services and will be made available to customers in a non-discriminatory manner. Terms of specific ICB contracts will be made available to the Missouri Public Service Commission Staff upon request on a proprietary basis.

Interim Local Number Portability (ILNP) (via Remote Call Forwarding (RCF)): Remote Call Forwarding allows a Local Line Customer to retain its existing Local Exchange Carrier telephone number by using this feature to port its telephone number to MCI Local Service. Monthly charges will apply per telephone number. A number of paths can be associated with this number to permit multiple simultaneous calls to be received. A maximum of 99 paths can be associated with a single ported number.

Joint User: A person, firm or corporation designated by the Customer as a user of local exchange service furnished to the Customer by the Company, and to whom a portion of the charges for such facilities are billed under a joint use arrangement.

LATA: A local access and transport area established pursuant to the Modification of Final Judgment entered by the United States District Court for Missouri in Civil Action No. 82-0192 for the provision and administration of communications services.

Least Idle Trunk Selection (LIDL): LIDL trunk selection occurs when a switching unit selects from a Trunk group the Trunk that has been idle for the shortest period of time.

Local Calling: A completed call or telephonic communication between a calling Station and any other Station within the local service area of the calling Station.

Local Exchange Carrier: A company which furnishes exchange telephone service.

Mbps: Megabits, or millions of bits, per second.

Message Waiting: This feature provides an indication to a Station User that a message is waiting. Indications may be visual (lamp) or audible (stuttered dialtone).

Most Idle Trunk Selection (MIDL): MIDL Trunk selection occurs when a switching unit selects from a Trunk group the Trunk that has been idle for the longest period of time.

Multiple Appearance Directory Numbers: A directory number that is assigned more than once to one or more Proprietary Business Sets.



MISSOURI

96-355

EFFECTIVE: May 14, 1999

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Sorvies Commission FILED MAY 1 & 1999

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

<u>Multi-Frequency ("MF")</u>: An inter-machine pulse-type used for signaling between telephone switches, or between telephone switches and PBX/key systems.

Non-Recurring Charges: The one-time initial charges for services or facilities, including but not limited to charges for construction, installation, or special fees, for which the Customer becomes liable at the time the Service Order is executed.

Off-Hook: The term "off-hook" denotes the active condition of a telephone exchange service line.

 $\underline{\text{On-Hook:}}$ The term "on-hook" denotes the idle condition of a telephone exchange service line.

<u>Presubscription</u>: An arrangement whereby a Customer may select and designate to the Company an Exchange Carrier it wishes to access, without an access code, for completing intraLATA toll calls. The selected Exchange Carrier is referred to as the End User's Primary Interexchange Carrier (PIC-2).

<u>Recurring Charges:</u> The monthly charges to the Customer for services, facilities and equipment, which continue for the agreed upon duration of the service.

D D D D D D D D D

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission ÷

LOCAL EXCHANGE DEMAR 1 1 2002

Definitions (Cont'd)

Service Commission

<u>Multi-Frequency ("MF")</u>: An inter-machine pulse-type used for signaling between telephone switches, or between telephone switches and PBX/key systems.

<u>Non-Recurring Charges</u>: The one-time initial charges for services or facilities, including but not limited to charges for construction, installation, or special fees, for which the Customer becomes liable at the time the Service Order is executed.

<u>Off-Hook</u>: The term "off-hook" denotes the active condition of a telephone exchange service line.

<u>On-Hook:</u> The term "on-hook" denotes the idle condition of a telephone exchange service line.

<u>Presubscription</u>: An arrangement whereby a Customer may select and designate to the Company an Exchange Carrier it wishes to access, without an access code, for completing intraLATA toll calls. The selected Exchange Carrier is referred to as the End User's Primary Interexchange Carrier (PIC-2).

<u>Recurring Charges:</u> The monthly charges to the Customer for services, facilities and equipment, which continue for the agreed upon duration of the service.

<u>Remote Call Forwarding (RCF):</u> RCF allows all calls dialed to a telephone number equipped for RCF service to be automatically forwarded to another dialable telephone number. This service enables a customer to list a local directory number that is forwarded to a different city or exchange. Each RCF service allows for the forwarding of one call at a given time. The RCF customer is responsible for any applicable usage rates/charges between the RCF number and the terminating number.

Conditions:

- 1. RCF service is not offered when the answering location for a forwarded call is a coin/coinless, Semi-public/public telephone service.
- 2. The Company cannot guarantee the grade of transmission on remotely forwarded calls. Normal grade end-to-end transmission is not guaranteed because transmission characteristics may vary depending on distance and routing required to complete the forwarded portion of the call.
- 3. RCF service will only be provided when, the in the judgement of the Company, the customer subscribes to sufficient RCF facilities at the terminating (answering) location to adequately handle calls without impairing, disrupting or deteriorating any services offered by the Company. In the event that the use of RCF service causes impairment, disruption or deterioration, the Company shall have the right to discontinue the RCF service.

RCF service is required for each directory number being forwarded. A customer can request additional RCF service (call paths) provided the customer has a receiving group of lines equal to the number of RCF services requested, (i.e., directory number is forwarded to a remote group of .5 lines - the customer can have up to 5 RCF services.).

Missouri Public

FILED APR 1 1 2002

CERTAIN MATERIAL WAS MOVED TO PAGE NO. 11.1

Service Commission

Issued: March 11, 2002 CANCELLED September 1, 2008 Missouri Public Service Commission

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 11, 2002

Ν

Ν

RECEIVED MISSOURI P.S.C. TARIFF NO. 1

ORIGINAL PAGE NO. 11

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

MO. PUBLIC SERVICE CUMM

<u>Multi-Frequency ("MF")</u>: An inter-machine pulse-type used for signaling between telephone switches, or between telephone switches and PBX/key systems.

Non-Recurring Charges: The one-time initial charges for services or facilities, including but not limited to charges for construction, installation, or special fees, for which the Customer becomes liable at the time the Service Order is executed.

Off-Hook: The term "off-hook" denotes the active condition of a telephone exchange service line.

On-Hook: The term "on-hook" denotes the idle condition of a telephone exchange service line.

<u>Presubscription</u>: An arrangement whereby a Customer may select and designate to the Company an Exchange Carrier it wishes to access, without an access code, for completing intraLATA toll calls. The selected Exchange Carrier is referred to as the End User's Primary Interexchange Carrier (PIC-2).

<u>Recurring Charges:</u> The monthly charges to the Customer for services, facilities and equipment, which continue for the agreed upon duration of the service.

Service Commencement Date: The first day following the date on which the Company notifies the Customer that the requested service or facility is available for use, unless extended by the Customer's refusal to accept service which does not conform to standards set forth in the Service Order or this tariff, in which case the Service Commencement Date is the date of the Customer's acceptance of service. The parties may mutually agree on a substitute Service Commencement Date.

<u>Service Order</u>: The written request for local exchange services executed by the Customer and the Company in a format specified by the Company. The signing of a Service Order by the Customer and acceptance thereof by the Company initiates the respective obligations of the parties as set forth therein and pursuant to this tariff, but the duration of the service is calculated from the Service Commencement Date.

Services: The Company's telecommunications services offered on the Company's network.

Shared Facilities: A facility or equipment system or subsystem which can be used simultaneously by several Customers.

Speed Call: Provides a User with the option to call selected directory numbers by dialing a one or two-digit code.

Station: Telephone equipment from or to which calls are placed.

<u>Trunk:</u> A communications path connecting two switching systems in a network, used in the establishment of an end-to-end connection.

User: A Customer or any other person authorized by the Customer to use service provided under this tariff.

CANCELLED	CANCELLED	
APR 1 1 2002	AUG 2 0 1999 TA- 13-575	96-355
MISSOURI	Public Service Commissio MISSOURI Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328	Missouri Public Sorvice Commission
		FILED MAY 1 & 1999

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

Remote Call Forwarding (RCF): (Cont'd)

<u>Service Commencement Date</u>: The first day following the date on which the Company notifies the Customer that the requested service or facility is available for use, unless extended by the Customer's refusal to accept service which does not conform to standards set forth in the Service Order or this tariff, in which case the Service Commencement Date is the date of the Customer's acceptance of service. The parties may mutually agree on a substitute Service Commencement Date.

<u>Service Order</u>: The written request for local exchange services executed by the Customer and the Company in a format specified by the Company. The signing of a Service Order by the Customer and acceptance thereof by the Company initiates the respective obligations of the parties as set forth therein and pursuant to this tariff, but the duration of the service is calculated from the Service Commencement Date.

Services: The Company's telecommunications services offered on the Company's network.

<u>Shared Facilities</u>: A facility or equipment system or subsystem which can be used simultaneously by several Customers.

<u>Speed Call</u>: Provides a User with the option to call selected directory numbers by dialing a one or two-digit code.

Station: Telephone equipment from or to which calls are placed.

<u>Trunk:</u> A communications path connecting two switching systems in a network, used in the establishment of an end-to-end connection.

<u>User:</u> A Customer or any other person authorized by the Customer to use service provided under this tariff.

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

LOCAL EXCHANGE SERVICE

1. Definitions (Cont'd)

<u>Remote Call Forwarding (RCF)</u>: (Cont'd)

<u>Service Commencement Date</u>: The first day following the date on which the Company notifies the Customer that the requested service or facility is available for use, unless extended by the Customer's refusal to accept service which does not conform to standards set forth in the Service Order or this tariff, in which case the Service Commencement Date is the date of the Customer's acceptance of service. The parties may mutually agree on a substitute Service Commencement Date.

<u>Service Order</u>: The written request for local exchange services executed by the Customer and the Company in a format specified by the Company. The signing of a Service Order by the Customer and acceptance thereof by the Company initiates the respective obligations of the parties as set forth therein and pursuant to this tariff, but the duration of the service is calculated from the Service Commencement Date.

Services: The Company's telecommunications services offered on the Company's network.

<u>Shared Facilities</u>: A facility or equipment system or subsystem which can be used simultaneously by several Customers.

<u>Speed Call</u>: Provides a User with the option to call selected directory numbers by dialing a one or two-digit code.

Station: Telephone equipment from or to which calls are placed.

<u>Trunk:</u> A communications path connecting two switching systems in a network, used in the establishment of an end-to-end connection.

<u>User:</u> A Customer or any other person authorized by the Customer to use service provided under this tariff.

Overflow Routing: Where technical capabilities exist, overflow routing allowsNthe redirection of incoming calls based on customer conditions of either "all|trunks busy" or disaster-based service outages. The redirection (Overflow) will|route to an alternate number designation determined by the customer.|Standard simultaneous call "path" allocations are usually (1) per number. However,|a customer may opt to increase their "path" quantities per number based on their|ability to process simultaneous call volumes during overflow conditions. A|monthly recurring charge will apply based on a "per path/per number" scenario".|In addition, toll charges may be assessed if the alternate number designated by|the customer is not toll free and is outside of their service address rate center.N

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Service Commission

Effective: December 1, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 11.1

Ν

M

LOCAL EXCHANGE SERVICE Missouri Public

1. Definitions (Cont'd)

REC'D MAR 1 1 2002

<u>Remote Call Forwarding (RCF)</u>: (Cont'd)

Service Commission <u>Service Commencement Date</u>: The first day following the date on which the Company notifies the Customer that the requested service or facility is available for use, unless extended by the Customer's refusal to accept service which does not conform to standards set forth in the Service Order or this tariff, in which case the Service Commencement Date is the date of the Customer's acceptance of service. The parties may mutually agree on a substitute Service Commencement Date.

<u>Service Order</u>: The written request for local exchange services executed by the Customer and the Company in a format specified by the Company. The signing of a Service Order by the Customer and acceptance thereof by the Company initiates the respective obligations of the parties as set forth therein and pursuant to this tariff, but the duration of the service is calculated from the Service Commencement Date.

Services: The Company's telecommunications services offered on the Company's network.

<u>Shared Facilities</u>: A facility or equipment system or subsystem which can be used simultaneously by several Customers.

<u>Speed Call</u>: Provides a User with the option to call selected directory numbers by dialing a one or two-digit code.

Station: Telephone equipment from or to which calls are placed.

<u>Trunk</u>: A communications path connecting two switching systems in a network, used in the establishment of an end-to-end connection.

<u>User:</u> A Customer or any other person authorized by the Customer to use service provided under this tariff.

CANCELLED

DEC 0 1 2002 mssion oublic

Missouri Public

2711	PR		n	1	1	າດດາ
	FI)	A٢	ĸ	L	T	2002

CERTAIN MATERIAL WAS PREVIOUSLY LOCATED ON PAGE NO. 11.

Service Commission

Issued: March 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 11, 2002

RECEIVED

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 12

ł

i

SEP 17 1999

LOCAL EXCHANGE SERVICE

2. <u>Regulations</u>

MO. PUBLIC SERVICE COMM

2.1 Undertaking of the Company

2.1.1 <u>Scope</u>

The Company undertakes to furnish communications service in connection with one-way and/or two-way information transmission between points within the State of Missouri under the terms of this tariff.

Customers may use services and facilities provided under this tariff to obtain access to services offered by other service providers. The Company is responsible under this tariff only for the services and facilities provided herein, and it assumes no responsibility for any service provided by any other entity that purchases access to the Company network in order to originate or terminate its own services, or to communicate with its own Customers.

Applications for initial or additional service made verbally or in writing become a contract upon the establishment of the service or facility.

2.1.2 Shortage of Equipment or Facilities

- 2.1.2.1 The Company reserves the right to limit or allocate the use of existing facilities, or of additional facilities offered by the Company when necessary because of lack of facilities or due to some other cause beyond the Company's control.
- 2.1.2.2 The furnishing of service under this tariff is subject to the availability on a continuing basis of all the necessary facilities and is limited to the capacity of the Company's fiber optic cable facilities as well as facilities the Company may obtain from other carriers, from time to time, to furnish service as required at the sole discretion of the Company.

2.1.3 Terms and Conditions

- 2.1.3.1 Except as otherwise provided herein, service is provided and billed on the basis of a minimum period of at least one month, and shall continue to be provided until cancelled by the Customer, orally or in writing, on not less than 30 days notice. Unless otherwise specified herein, for the purpose of computing charges in this tariff, a month is considered to have 30 days. All calculations of dates set forth in this tariff shall be based on calendar days, unless otherwise specified herein.
- 2.1.3.2 Customers may be required to enter into written Service Orders which shall contain or reference the name of the Customer, a specific description of the service ordered, the rates to be charged, the duration of the services, and the terms and conditions in this tariff.



ISSUED : March 2, 1999

L

By Public Service Commission Manay Hunder Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta. GA 30328

EFFECTIVE: May 14, 1999

Missouri Public Sorvice Commission

96-355

FILED MAY 1 4 1999

MCImetro A	CCESS	TRANSMISSION
SERVICES,	LLC	

RECEIVED

MISSOURI P.S.C. TARIFF NO. 1 SFP 17 1998 ORIGINAL PAGE NO. 13

İ

LOCAL EXCHANGE SERVICE

MO. PUBLIC SERVICE COMM

2. <u>Regulations (Cont'd)</u>

2.1 Undertaking of the Company (Cont'd)

- 2.1.3 Terms and Conditions (Cont'd)
 - 2.1.3.3 At the expiration of the initial term specified in each Service Order, or in any extension thereof, service shall continue on a month to month basis at the then current rates unless terminated by either party upon 30 days written notice. Any termination shall not relieve Customer of its obligation to pay any charges incurred under the Service Order and this tariff prior to termination. The rights and obligations which by their nature extend beyond the termination of the term of the Service Order shall survive such termination.
 - 2.1.3.4 This tariff shall be interpreted and governed by the laws of the State of Missouri.
 - 2.1.3.5 Another telephone company must not interfere with the right of any person or entity to obtain service directly from the Company.
 - 2.1.3.6 The Customer has no property right to the telephone number or any other call number designation associated with services furnished by the Company. The Company reserves the right to change such numbers, or the central office designation associated with such numbers, or both, assigned to the Customer, whenever the Company deems it necessary to do so in the conduct of its business.
 - 2.1.3.7 The Customer agrees to operate Company-provided equipment in accordance with instructions of the Company or the Company's agent. Failure to do so will void Company isability for interruption of service and may make the Customer responsible for damage to equipment pursuant to section 2.1.3.8 below.
 - 2.1.3.8 The Customer egrees to return to the Company, or allow the Company access to retrieve, all Company-provided equipment delivered to Customer within five (5) days of termination of the service in connection with which the equipment was used. Said equipment shall be in the same condition as when delivered to Customer, normal wear and tear only excepted. Customer shall reimburse the Company, upon demand, for any costs incurred by the Company due to Customer's failure to comply with this provision.
 - 2.1.3.9 The Company will provide a number intercept service for customers who change their telephone number. Service will be provided to customers for a minimum of 30 days at no charge for customers who request service.



ISSUED : March 2, 1999

Ļ

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missouri Public Service Commission

96-355

FILFD MAY 1 4 1999

RECEIVED

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 14

SEP 17 1998

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

L

CANCELLED

AUG 2/0 1999

Public Service Commission

MISSOURI

/ISSUED : March 2, 1999

MO. FUBLIC SERVICE COMM

2.1 Undertaking of the Company (Cont'd)

2.1.4 Liability of the Company

- 2.1.4.1 The liability of the Company for damages arising out of the furnishing of its services, including but not limited to mistakes, omissions, interruptions, delays, or errors, other defects, or representations by the Company, or use of these services or damages arising out of the failure to furnish the service whether caused by acts or omission, shall be limited to the extension of allowances for interruption as set forth in Section 2.6, below. The extension of such allowances for interruption shall be the sole remedy of the Customer and the sole liability of the Company. The Company will not be liable for any direct, indirect, incidental, special, consequential, exemplary or punitive damages to Customer as a result of any Company service, equipment or facilities, or the acts or omissions or ordinary negligence of the Company's employees or agents.
- 2.1.4.2 The Company's liability for willful misconduct, if established as a result of judicial or administrative proceedings, is not limited by this tariff. Company's liability, if any, with regard to delayed installation of Company facilities or commencement of service, shall not exceed \$1,000. With respect to any other claim or suit, by a Customer or by any others, for damages associated with the ordering (including the reservation of any specific number for use with a service), installation (including delays thereof), provision, termination, maintenance, repair, interruption or restoration of any service or facilities offered under this tariff, and subject to the provisions of Section 2.6, the Company's liability, if any, shall be limited as provided herein.
- 2.1.4.3 The Company shall not be liable for any delay or failure of performance or equipment due to causes beyond its control, including but not limited to: acts of God, fire, flood, explosion or other catastrophes; any law, order, regulation, direction, action or request of the United States government or of any other government, including state and local governments having or claiming jurisdiction over the Company, or of any department, agency, commission, bureau, corporation or other instrumentality of any one or more of these federal, state, or local governments, or of any military authority; preemption of existing service in compliance with national emergencies; insurrections; riots; wars; unavailability of rights-of-way or materials, or strikes, lockouts, work stoppages, or other labor difficulties.
- 2.1.4.4 The Company shall not be liable for: (a) any act or omission of any entity furnishing the Company or the Company's Customers facilities or equipment used for or with the services the Company offers; or (b) for the acts or omissions of other common carriers or warehousemen, except where contracted by the Company.
- 2.1.4.5 The Company shall not be liable for any damages or losses due to the fault or negligence of the Customer or due to the failure or malfunction of Customer-provided equipment or facilities.

96-355

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missourl Public Service Commission

FILED MAY 1.4 1999

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

RECEIVED

MISSOURI P.S.C. TARIFF NO. 1 SFP 17 1999 ORIGINAL PAGE NO. 15

2. Regulations (Cont'd)

LOCAL EXCHANGE SERVICE MO. PUBLIC SERVICE COMM

2.1 Undertaking of the Company (Cont'd)

2.1.4 Liability of the Company (Cont'd)

2.1.4.6 The Customer shall indemnify and hold the Company harmless from any and all loss, claims, demands, suits or other action, or an liability whatsoever, whether suffered, made, instituted, or asserted by any other party or person(s), and for any loss, damage, or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused directly or indirectly by the installation, removal, condition, location or use of any installation provided by the Company. The Company reserves the right to require each Customer to sign an agreement acknowledging acceptance of the provisions of this section as a condition precedent to such installation.

- The Company shall not be liable for any defacement of or damage to the Customer's premises 2.1.4.7 resulting from the furnishing of services or equipment on such premises or the installation or removal thereof, unless such defacement or damage is caused by gross negligence or willful misconduct of the Company's agents or employees. No agents or employees of other participating carriers shall be deemed to be agents or employees of the Company, except where contracted by the Company.
- 2.1.4.8 Where misuse of service is claimed notwithstanding the Customer's obligations as set forth in Section 2.3.2, the Company shall be indemnified, defended, and held harmless by the Customer or by others authorized by it to use the service against any claim, loss or damage arising from Customer's use of services furnished under this tariff, including; claims for libel, slander, invasion of privacy or infringement of copyright arising from the material, data, information, or other content transmitted via the Company's service; and patent infringement claims arising from combining or connecting the service offered by the Company with apparatus and systems of the Customer or others all other claims arising out of any act or omission of the Customer or others, in connection with any service provided by the Company pursuant to this tariff,
- 2.1.4.9 The entire liability of the Company, resulting from ordinary negligence of ordinary default of the Company, for any claim, loss, damage or expense from any cause shall in no event exceed sums actually paid to the Company by the Customer for the specific services giving rise to the claim, and no action or proceeding against the Company shall be commenced more than one year after the service is rendered.



ISSUED : March 2, 1999

L

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Servico Commission

FILED MAY 1 4 1999

RECENTO MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 16

SEP 17 1998

LOCAL EXCHANGE SERVICE

2. <u>Regulations (Cont'd)</u>

MO. FUBLIC SERVICE COMM

2.1 Undertaking of the Company (Cont'd)

2.1.4 Liability of the Company (Cont'd)

- 2.1.4.10 The Company makes no warranties or representations, express or implied, [either in fact or by operation of the law, statutory or otherwise,] including warranties of merchantability or fitness for a particular use, except those expressly set forth herein.
- 2.1.4.11 The Company shall not be liable for any act or omission of any other company or companies furnishing a portion of the service, or for damages associated with service, channels, or equipment which it does not furnish, except where contracted by the Company, or for damages which result from the operation of Customer-provided systems, equipment, facilities or services which are interconnected with Company services.
- 2.1.4.12 The Company does not guarantee nor make any warranty with respect to service installations at locations at which there is present an atmosphere that is explosive, prone to fire, dangerous or otherwise unsuitable for such installations. The Customer shall indemnify and hold the Company harmless from any and all loss, claims, demands, suits or other action, or any liability whatsoever, whether suffered, made, instituted or asserted by the Customer or by any other party, for any personal injury to, or death of, any person or persons, or for any loss, damage or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused, directly or indirectly, by the installation, operation, failure to operate, maintenance, removal, presence, condition, locations or use of service furnished by the Company at such locations.

CANCELLED AUG 2/0 1999 XA-98-575 Public Service Commission MISSOURI

Sendy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 2, 1999

4

EFFECTIVE: May 14, 1999

Missouri Public Servico Commission

96-355

i

FILED MAY 14 1999
RECEIVE MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 17**

LOCAL EXCHANGE SERVICE

SEP 17 1999

2. Regulations (Cont'd)

MO. FUBLIC SERVICE COMM

2.1 Undertaking of the Company (Cont'd)

2.1.4 Liability of the Company (Cont'd)

2.1.4.13 The Company shall not be liable for the Customer's failure to fulfill its obligations to take all necessary steps including, without limitation, obtaining, installing and maintaining all necessary equipment, materials and supplies, for interconnecting the terminal equipment or communications system of the Customer, or any third party acting as its agent, to the Company's network. In instances when the Customer is reasonably required, the Customer shall secure all licenses, permits, rights-of-way, and other arrangements necessary for such interconnection. In addition, the Customer shall ensure that its equipment and/or system or that of its agent is properly interfaced with the Company's service, that the signals emitted into the Company's network are of the proper mode, band-width, power, data speed, and signal level for the intended use of the Customer and in compliance with the criteria set forth in Section 2.1.6 following, and that the signals do not damage Company equipment, injure its personnel or degrade service to other Customers. If the Customer or its agent fails to maintain and operate its equipment and/or system or that of its agent properly, with resulting imminent harm to Company equipment, personnel, or the quality of service to other Customers, the Company may, upon written notice, require the use of protective equipment at the Customer's expense. If this fails to produce satisfactory quality and safety, the Company may, upon written notice, terminate the Customer's service without liability.

2.1.4.14 With respect to Emergency Number 911 Service:

(a)

This service is offered solely as an aid in handling assistance calls in connection with fire, police and other emergencies. The Company is not responsible for any losses, claims, demands, suits or any liability whatsoever, whether suffered, made, instituted or asserted by the Customer or by any other party or person for any personal injury to or death of any person or persons, and for any loss, damage or destruction of any property, whether owned by the Customer or others, caused or claimed to have been caused by: (1) mistakes, omissions, interruptions, delays, errors or other defects in the provision of this service, or (2) installation, operation, failure to operate, maintenance, removal, presence, condition, location or use of any equipment and facilities furnishing this service.

(b)

Neither is the Company responsible for any infringement or invasion of the right of privacy of any person or persons, caused or claimed to have been caused, directly or indirectly, by the installation, operation, failure to operate, maintenance, removal, presence, condition, occasion or use of enhanced 911 service features and the equipment associated therewith, or by any services furnished by the Company including, but not limited to, the identification of the telephone number, address or name associated with the telephone used by the party or parties accessing enhanced 911 service, and which arise out of the negligence or other wrongful act of the Company, the Customer, its Users, agencies or municipalities, 9 6 - 3 5 5 EFFECTIVE: May 14, 1999 or the employees or agents of any one of them.

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missourl Public Service Commission

FILFD MAY 1 4 1999



L

ISSUED : March 2, 1999

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 18

LOCAL EXCHANGE SERVICE

RECEIVED

2. <u>Regulations (Cont'd)</u>

- 2.1 Undertaking of the Company (Cont'd)
 - 2.1.4 Liability of the Company (Cont'd)

MÜ. PUBLIG SERVICE OUMM

SFP 17 1998

- 2.1.4.15 The Company's liability arising from errors or omissions in Directory Listings, other than charged listings, shall be limited to the amount of actual impairment to the Customer's service and in no event shall exceed one-half the amount of the fixed monthly charges applicable to exchange service affected during the period covered by the directory in which the error or omission occurs. In cases of charged Directory Listings, the liability of the Company shall be limited to an amount not exceeding the amount of charges for the charged listings involved during the period covered by the directory in which the error or omission occurs.
- 2.1.4.16 In conjunction with a nonpublished telephone number, as described in Section 3.4.5.3, the Company will not be liable for failure or refusal to complete any call to such telephone when the call is not placed by number. The Company will try to prevent the disclosure of the number of such telephone, but will not be liable should such number be divulged.
- 2.1.4.17 When a Customer with a nonpublished telephone number, as defined herein, places a call to the Emergency 911 Service, the Company will release the name and address of the calling party, where such information can be determined, to the appropriate local governmental authority responsible for the Emergency 911 Service upon request of such governmental authority. By subscribing to service under this tariff, Customer acknowledges and agrees with the release of information as described above.
- 2.1.4.18 In conjunction with the Busy Line Verification and Interrupt Service as described in Section 3.3.2, the Company shall not be liable for any claim that may arise from either party to the interrupted call or any person.
- 2.1.4.19 The Company shall not be liable for any act or omission concerning the implementation of presubscription, as defined herein.

CANCELLE AUG 2 1999 TA-98-575 By Public Seprice Commission MISSOURI



L

<u>96-35</u>5 EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missourl Public Service Commission

FILED MAY 1 & 1999

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 19

ł

ł

LOCAL EXCHANGE SERVICE

SEP 17 1998

MO. YUDLIU OCHAIOL UUMM

2. Regulations (Cont d)

2.1 Undertaking of the Company (Cont'd)

2.1.5 Notification of Service-Affecting Activities

The Company will provide the Customer reasonable notification of service-affecting activities that may occur in normal operation of its business. Such activities may include, but are not limited to, equipment or facilities additions, removals or rearrangements and routine preventative maintenance. Generally, such activities are not specific to an individual Customer but affect many Customers' services. No specific advance notification period is applicable to all service activities. The Company will work cooperatively with the Customer to determine the reasonable notifications requirements. With some emergency or unplanned service-affecting conditions, such as outage resulting from cable damage, notification to the Customer may not be possible.

2.1.6 Provision of Equipment and Facilities

- 2.1.6.1 The Company shall use reasonable efforts to make available services to a Customer on or before a particular date, subject to the provisions of and compliance by the Customer with, the regulations contained in this tariff. The Company does not guarantee availability by any such date and shall not be liable for any delays in commencing service to any Customer.
- 2.1.6.2 The Company shall use reasonable efforts to maintain facilities that it furnishes to the Customer. The Customer may not, nor may the Customer permit others to, rearrange, disconnect, remove, attempt to repair or otherwise interfere with any of the facilities installed by the Company, except upon the written consent of the Company.
- 2.1.6.3 Equipment installed at the Customer Premises for use in connections with the services the Company offers shall not be used for any purpose other than that for which the Company provided it.

Sandy Chandler

Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

2.1.6.4 The Customer shall be responsible for the payment of service charges as set forth herein for visits by the Company's agents or employees to the premises of the Customer when the service difficulty or trouble report results from the use of equipment or facilities provided by any party other than the Company, including but not limited to the Customer.

ANCELIER AUG 2 0 1999 By 7A 78-575 Public Service Commission MISSOURI

ISSUED : March 2, 1999

4

EFFECTIVE: May 14, 1999

Missouri Public Sorvico Commission

96-355

FILFD MAY 1 3 1999

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 20** RECEIVED

SEP 17 1998

MU. FUDLIC SERVICE COMM

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

2.1 Undertaking of the Company (Cont'd)

2.1.6 Provision of Equipment and Facilities (Cont'd)

- 2.1.6.5 The Company shall not be responsible for the installation, operation, or maintenance of any Customer provided communications equipment. Where such equipment is connected to the facilities furnished pursuant to this tariff, the responsibility of the Company shall be limited to the furnishing of facilities offered under this tariff and to the maintenance and operation of such facilities. Beyond this responsibility, the Company shall not be responsible for:
 - (a) the transmission of signals by Customer provided equipment or for the quality of, or defects in, such transmission; or
 - (b) the reception of signals by Customer provided equipment; or
 - network control signaling where such signaling is performed by Customer-provided (c) network control signaling equipment.

2.1.7 Non-routine Installation

At the Customer's request, installation and/or maintenance may be performed outside the Company's regular business hours or in hazardous locations. In such cases, charges based on cost of the actual labor, material, or other costs incurred by or charged to the Company will apply. If installation is started during regular business hours but, at the Customer's request, extends beyond regular business hours into time periods including, but not limited to, weekends, holidays, and/or night hours, additional charges may apply.

2.1.8 **Ownership of Facilities**

Ł

ISSUED : March 2, 1999

Title to all facilities provided in accordance with this tariff remains in the Company, its agents or contractors.

By

Sandy Chandler

Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECTIVE: May 14, 1999

AUG 2 0 1999

Public Service Commission MISSOURL

TAL98-575

Missouri Public Service Commission

96-355

FILED MAY 1 4 1999



eg Missell	s Pupligor
------------	------------

REC'D FEB 29 2000

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 21 CANCELS ORIGINAL PAGE NO. 21

LOCAL EXCHANGE SERVICE

- 2. Regulations (Cont'd)
 - 2.2 <u>Prohibited Uses</u>
 - 2.2.1 The services the Company offers shall not be used for any unlawful purpose or for any use as to which the Customer has not obtained all required governmental approvals, authorizations, licenses, consents and permits.
 - 2.2.2 The Company may require a Customer to immediately shut down its transmission of signals if said transmission is causing interference to others.
 - 2.2.3 Customer may not use service furnished under this local exchange tariff, directly or indirectly, to provide a service that constitutes exchange access and/or is subject to the application of access charges under applicable law. The Company reserves the right to:
 - (1) request that Customer provide written certification that it is using service in compliance with this requirement; and/or
 - (2) conduct a site survey of Customer premises or an audit of Customer books and records upon reasonable notice or take other reasonable measures to satisfy itself that Customer is using service in compliance with this tariff.

In the event the Customer is found to be using service in violation of this requirement, the Company may discontinue the provision of service without notice, any other provision of this tariff to the contrary notwithstanding. Customer shall indemnify the Company for any liability, losses penalties or payments (including without limitation access charges and the Company's attorneys' fees) incurred due to Customer's misuse of the Company's services obtained under this tariff.

2.2.4 The Company's services are not allowed to be resold.



ISSUED : February 29, 2000

EFFECTIVE: March 30, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

2.

P T C T Missouri P.S.C. Tariff No. 1 Original Page No. 21

LOCAL EXCHANGE SERVICE

SEP 17 1999

Regutations (Cont'd)
2.2 Prohibited Uses

MO. FUDLIC SERVICE UCIVIM

- 2.2.1 The services the Company offers shall not be used for any unlawful purpose or for any use as to which the Customer has not obtained all required governmental approvals, authorizations, licenses, consents and permits.
- 2.2.2 The Company may require a Customer to immediately shut down its transmission of signals if said transmission is causing interference to others.
- 2.2.3 Service furnished under this local exchange tariff may not be used, directly or indirectly, by a telecommunications carrier for the purpose of terminating interLATA services. The Company reserves the right to: (1) request that Customer provide written certification that it is using service in compliance with this requirement; and/or (2) conduct a site survey of Customer premises upon reasonable notice or take other reasonable measures to satisfy itself that Customer is using service in compliance with this tariff. In the event Customer is found to be using service in violation of this requirement, the Company may discontinue the provision of service without notice, any other provision of this tariff to the contrary notwithstanding. Customer shall indemnify the Company for any losses or penalties incurred due to Customer's misuse of the Company's services obtained under this tariff.
- 2.2.4 The Company's services are not allowed to be resold.

CANCELLED MAR 3 0 2000 By 151 RP 21 Public Service Commission MISSOURI

ISSUED : March 2, 1999

£

96-355

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILED MAY 1 4 1999

RECEIVED

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 22

SEP 17 1999

LOCAL EXCHANGE SERVICE

MO. Public Schwich ComM

2. <u>Regulations (Cont'd)</u>

2.3 Obligations of the Customer

2.3.1 General

The Customer shall be responsible for:

- (a) the payment of all applicable charges pursuant to this tariff;
- (b) reimbursing the Company for damage to, or loss of, the Company's facilities or equipment caused by the acts or omissions of the Customer; or the noncompliance by the Customer, with these regulations; or by fire or theft or other casualty on the Customer's premises, unless caused by the negligence or willful misconduct of the employees or agents of the Company. The Company will, upon reimbursement for damages, cooperate with the Customer in prosecuting a claim against the person causing such damage and the Customer shall be subrogated to the Company's right of recovery of damages to the extent of such payment.
- (c) providing at no charge, as specified from time to time by the Company, any needed equipment space and power to operate Company facilities and equipment installed on the premises of the Customer, and the level of heating and air conditioning necessary to maintain the proper operating environment on such premises;
- (d) obtaining, maintaining, and otherwise having full responsibility for all rights-of-way and conduit necessary for installation of fiber optic cable and associated equipment used to provide local exchange service to the Customer from the cable building entrance or property line to the location of the equipment space described in 2.3.1(c). Any costs associated with the obtaining and maintaining the rights-of-way described herein, including the costs of altering the structure to permit installation of the Company-provided facilities, shall be borne entirely by, or may be charged by the Company to, the Customer. The Company may require the Customer to demonstrate its compliance with this section prior to accepting an order for service;
 - providing a safe place to work and complying with all laws and regulations regarding the working conditions on the premises at which Company employees and agents shall be installing or maintaining the Company's facilities and equipment. The Customer may be required to install and maintain Company facilities and equipment within a hazardous area if, in the Company's opinion, injury or damage to the Company's employees or property might result from installation or maintenance by the Company. The Customer shall be responsible for identifying, monitoring, removing and disposing of any hazardous material (e.g. friable asbestos) prior to any construction or installation work;

CANCELLÉD AUG 2 0 1999 98-575 Bv Public Service Commission MISSOURI

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 96-355

Ĩ

÷

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILED MAY 1 & 1999



L

(e)

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 23** RECEIVED

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

4

SEP 17 1998

MO. PUBLIC SERVICE CONIN

1

ł

ł

2.3 Obligations of the Customer (Cont'd)

2.3.1 General (Cont'd)

- (1) complying with all laws and regulations applicable to, and obtaining all consents, approvals, licenses and permits as may be required with respect to, the location of Company facilities and equipment in any Customer premises or the rights-of-way for which Customer is responsible under Section 2.3.1(d) above; and granting or obtaining permission for Company agents or employees to enter the premises of the Customer at any time for the purpose of installing, inspecting, maintaining, repairing, or upon termination of service as stated herein, removing the facilities or equipment of the Company;
- (g) not creating or allowing to be placed or maintained any liens or other encumbrances on the Company's equipment or facilities; and
- (h) making Company facilities and equipment available periodically for maintenance purposes at a time agreeable to both the Company and the Customer. No allowance for interruptions in service will be made for the period during which service is interrupted for such purposes.

CANCELLED AUG 2 0 1999 98-575 By TA- 8-575 Public Service Commission 96-355 MISSOURI ISSUED : March 2, 1999 EFFECTIVE: May 14, 1999 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200

FILFD MAY 1 & 1999



Atlanta, GA 30328

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 24

LOCAL EXCHANGE SERVICE

DECEMEN

SFP 17 1999

MO. FUDER DERVICE OUNIN

2. Regulations (Cont'd)

2.3 Obligations of the Customer (Cont'd)

2.3.2 <u>Claims</u>

With respect to any service or facility provided by the Company, Customer shall indemnify, defend and hold harmless the Company from all claims, actions, damages, liabilities, costs and expenses, including reasonable attorneys' fees for:

- (a) any loss, destruction or damage to property of the Company or any third party, or the death of or injury to persons, including, but not limited to, employees or invitees of either the Company or the Customer, to the extent caused by or resulting from the negligent or intentional act or omission of the Customer, its employees, agents, representatives or invitees; or
- (b) any claim, loss damage, expense or liability for infringement of any copyright, patent, trade secret, or any proprietary or intellectual property right of any third party, arising from any act or omission by the Customer, including, without limitation, use of the Company's services and facilities in a manner not contemplated by the agreement between the Customer and the Company.

ÀUG 2 0 1999 TX-98-575 By Public Service Commission MISSOURI

ISSUED : March 2, 1999

L

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200

Atlanta, GA 30328

9 6 - 3 5 5 EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILFD MAY 1 4 1999

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 25

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

- 2.4 Customer Equipment and Channels
 - 2.4.1 General

MO. PUBLIC SERVICE CONNIN

SEP 17 1999

RECEIVED

A Customer may transmit or receive information or signals via the facilities of the Company.

2.4.2 Station Equipment

- 2.4.2.1 The Customer is responsible for providing and maintaining any terminal equipment on the Customer premises. The electric power consumed by such equipment shall be provided by, and maintained at the expense of, the Customer. All such terminal equipment must be registered with the FCC under 47 C.F.R., Part 68 and all wiring must be installed and maintained in compliance with those regulations. The Company will, where practicable, notify the Customer that temporary discontinuance of the use of a service may be required; however, where prior notice is not practicable, nothing contained herein shall be deemed to impair the Company's right to discontinue forthwith the use of a service temporarily if such action is reasonable under the circumstances. In case of such temporary discontinuance, the Customer will be promptly notified and afforded the opportunity to correct the condition which gave rise to the temporary discontinuance. During such period of temporary discontinuance, credit allowance for service interruptions as set forth in Section 2.6 following is not applicable.
- 2.4.2.2 The Customer is responsible for ensuring that Customer-provided equipment connected to Company equipment and facilities is compatible with such equipment and facilities. The magnitude and character of the voltages and currents impressed on Company-provided equipment and wiring by the connection, operation, or maintenance of such equipment and wiring shall be such as not to cause damage to the Company-provided equipment and wiring or injury to the Company's employees or other persons. Any additional protective equipment required to prevent such damage or injury shall be provided by the Company with the approval of the Customer at the Customer's expense.
- 2.4.3 Interconnection of Facilities
- CANCELLED AUG 2 0 1999 By TA-98-575 Public Service Commission 2.4.3.3 MISSOUR

L

- 2.4.3.1 Any special interface equipment necessary to achieve compatibility between the facilities and equipment of the Company used for furnishing local exchange service and the channels, facilities, or equipment of others may be provided at the Customer's expense.
 - .3.2 Local Services may be connected to the services or facilities of other communications carriers only when authorized by, and in accordance with, the terms and conditions of the tariffs of the other communications carriers which are applicable to such connections.
 - .3.3 Facilities furnished under this tariff may be connected to Customer provided terminal equipment in accordance with the provisions of this tariff.

ISSUED : March 2, 1999

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission FILED MAY 1. 4. 1909

<u>96-35</u>5

i

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 26 RECEIVED

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

SEP 17 1998

2.4 Customer Equipment and Channels (Cont'd)

MO. FUDLIC SERVICE UNIM

2.4.4 Inspections

- 2.4.4.1 Upon reasonable notification to the Customer, and at a reasonable time, the Company may make such tests and inspections as may be necessary to determine that the Customer is complying with the requirements set forth in Section 2.4.2.2 for the installation, operation, and maintenance of Customer-provided facilities and equipment to Company-owned facilities and equipment. No credit will be allowed for any interruptions occurring during such inspections.
- 2.4.4.2 If the protective requirements for Customer-provided equipment are not being complied with, the Company may take such action as it deems necessary to protect its facilities, equipment, and personnel. The Company will notify the Customer promptly if there is any need for further corrective action. Within ten days of receiving this notice the Customer must take this corrective action and notify the Company of the action taken. If the Customer fails to do this, the Company may take whatever additional action is deemed necessary, including the suspension of service, to protect its facilities, equipment and personnel from harm. The Company will, upon request 24 hours in advance, provide the Customer with a statement of technical parameters that the Customer's equipment must meet.

CANCELL AUG 2 0/ 1999 By TA X18-575 Public Service Commission MISSOURI

9 6 - 3 5 5 EFFECTIVE: May 14, 1999

ISSUED : March 2, 1999

£

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission

FILFD MAY 3 A 1999

Regulations (Cont'd)

2.

LOCAL EXCHANGE SERVICE

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 27

RECEIVED

SEP 17 1998

- 2.5 Payment Arrangements
 - 2.5.1 Payment for Service

MO. PUBLIC SERVICE COMM

L

The Customer is responsible for payment of all charges for service and facilities furnished by the Company to the Customer or its Joint or Authorized Users. Objections must be received by the Company within 30 days after statement of account is rendered, or the charges shall be deemed correct and binding upon the Customer.

2.5.1.1 <u>Taxes</u>: The Customer is responsible for the payment of any sales, use, gross receipts, excise, access or other local, state and federal taxes, (however designated) excluding taxes on the Company's net income imposed on or based upon the provision of Local Exchange Service, all of which shall be separately designated on the Company's invoices. Any taxes imposed by a local jurisdiction (e.g., County and municipal taxes) will only be recovered from those Customers located in the affected jurisdictions. It shall be the responsibility of the Customer to pay any such taxes that subsequently become applicable retroactively.

ISSUED : March 2, 1999

L

EFFECTIVE: May 14, 1999

CANCELLED

AUG 2 0 1999

By TA-98-575 Public Service Commission MISSOURI

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328



Missouri Public Service Commission

96-355

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 28 Cancels Original Page No. 28

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

- 2.5 <u>Payment Arrangements (Cont'd)</u>
 - 2.5.2 Billing and Collection of Charges

Bills will be rendered monthly to Customer.

- 2.5.2.1 All service, installation and monthly Recurring Charges and Non-Recurring Charges are due and payable upon receipt.
- 2.5.2.2 The Company shall present bills for Recurring Charges monthly to the Customer, in advance of the month which service is provided.
- 2.5.2.3 When service does not begin on the first day of the month, or end on the last day of the month, the charge for the fraction of the month in which service was furnished will be calculated on a pro rata basis. For this purpose, every month is considered to have 30 days.
- 2.5.2.4 Amounts not paid within 30 days after the date of invoice are considered past due.
- 2.5.2.5 A \$10.00 charge will be assessed for checks with insufficient funds or non-existing accounts.
- 2.5.3
- 2.5.4 Advance Payments

To safeguard its interests, the Company may require a Customer to make an Advance Payment before services and facilities are furnished. The Advance Payment will not exceed an amount equal to the Non-Recurring Charge(s) and one month's charges for the service or facility. In addition, where special construction is involved, the Advance Payment may also include an amount equal to the estimated Non-Recurring Charges for the special construction and Recurring Charges (if any) for a period to be set between the Company and the Customer. The Advance Payment will be credited to the Customer's initial bill. An Advance Payment may be required in addition to a deposit.

+

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

FILED Missouri Public Service Commission

D

LOCAL EXCHANGE SERVICE

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 28**

RECEIVED

SEP 17 199A

MO. PUBLIC SERVICE COMM

2. Regulations (Cont'd)

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.2 Billing and Collection of Charges

Bills will be rendered monthly to Customer.

- 2.5.2.1 All service, installation and monthly Recurring Charges and Non-Recurring Charges are due and payable upon receipt.
- 2.5.2.2 The Company shall present bills for Recurring Charges monthly to the Customer, in advance of the month which service is provided.
- 2.5.2.3 When service does not begin on the first day of the month, or end on the last day of the month, the charge for the fraction of the month in which service was furnished will be calculated on a pro rata basis. For this purpose, every month is considered to have 30 days.
- 2.5.2.4 Amounts not paid within 30 days after the date of invoice are considered past due.
- 2.5.2.5 A \$10.00 charge will be assessed for checks with insufficient funds or non-existing accounts,

2,5.3 **Disputed Bills**

The Customer shall notify the Company of any disputed items on a bill within 30 days of receipt of the bill. If the Customer and the Company are unable to resolve the dispute to their mutual satisfaction, the Customer may file a complaint with the Missouri Public Service Commission in accordance with the Commission's rules of procedure.

2.5.3.1 The date of the dispute shall be the date the Company receives sufficient documentation to enable it to investigate the dispute.

> The date of the resolution is the date the Company completes its investigation and notifies the Customer of the disposition of the dispute.

2.5.4 Advance Payments

To safeguard its interests, the Company may require a Customer to make an Advance Payment before services and facilities are furnished. The Advance Payment will not exceed an amount equal to the Non-Recurring Charge(s) and one month's charges for the service or facility. In addition, where special construction is involved, the Advance Payment may also include an amount equal to the estimated Non-Recurring Charges for the special construction and Recurring Charges (if any) for a period to be set between the Company and the Customer. The Advance Payment will be credited to the Customer's initial bill. An Advance Payment may be required in addition to a deposit,

ISSUED : March 2, 1999

EFFECTIVE: May 14, 1999

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission

9<u>6-35</u>5

FILFD WAY 1.4 1999

CANCELL/EI ′ 1999 88-515 Public Service Commission MISSOURI



Ł

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 29 Cancels Original Page No. 29 Missouri Public

N

N

Ν

N

т

 \mathbf{T}

т

Ť

2002

15-1

2002

Missouri Public

Service Commission

FILED APR 30

LOCAL EXCHANGE SERVICE

Regulations (Cont'd)

REC'D MAR 1 5 2002

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.5 <u>Deposits</u>

Service Commission

- 2.5.5.1 Applicants for service or existing Customer's whose financial condition is not acceptable to the Company, or is not a matter of general knowledge, may be required at any time to provide the Company a security deposit. The Deposit requested will be in cash or the equivalent of cash, and will be held as a guarantee for the payment of charges. A deposit does not relieve the customer of the responsibility for the prompt payment of bills on representation.
- 2.5.5.2 For residential Service Customers, the deposit will not exceed an amount equal to:
 - .Estimated charges of two month's service for a service a) or facility which has a minimum payment period of one month)
 - b) The charges that would apply for the minimum payment Period for a service or facility which has a minimum payment period of more than one month; except that the deposit may include an additional amount in event that a termination charge is applicable. In addition, the Company shall be entitled to require such an applicant or Customer to pay all its bills within a specified period of time, and to make such payments in cash or the equivalent of cash. At the Company's option, such deposit may be refunded to the Customer's account at any time. Also, the Company reserves the right to cease accepting and processing Service Orders after it has requested a security deposit and prior to the Customer's compliance with this request.
- 2.5.5.3 A deposit may be required in addition to advanced payment.
- 2.5.5.4 When a service or facility is discontinued, the amount of a deposit, if any, will be applied to the Customer's account and any credit balance remaining will be refunded. Before the service or facility is discontinued, the Company may, at its option, return the deposit or credit it to the Customer's account.
- 2.5.5.5 Deposits held for Residential Service customers will accrue interest at a rate of one (1) percent above the prime lending rate as published in the Wall Street Journal on the last business day of September (rate adjusted annually on December 1).

Issued: March 15, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 29

2. Regulations (Cont'd)

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.5 Deposits

REC	Eľ	VĽ.)
SEP	17	1998	}

MO. PUBLIC SERVICE CONM

- 2.5.5.1 Applicants for service or existing Customer's whose financial condition is not acceptable to the Company, or is not a matter of general knowledge, may be required at any time to provide the Company a security deposit in accordance with the Code of Missouri Regulations Title 20, Subtitle 30. The deposit requested will be in cash or the equivalent of cash, and will be held as a guarantee for the payment of charges. A deposit does not relieve the Customer of the responsibility for the prompt payment of bills on presentation. The deposit will not exceed an amount equal to:
 - two month's charges for a service or facility which has a minimum payment period of one month; or
 - (b) the charges, up to three months, that would apply for the minimum payment period for a service or facility which has a minimum payment period of more than one month. In addition, the Company shall be entitled to require such an applicant or Customer to pay all its bills within a specified period of time, and to make such payments in cash or the equivalent of cash. Also, the Company reserves the right to cease accepting and processing Service Orders after it has requested a security deposit and prior to the Customer's compliance with this request.
- 2.5.5.2 A deposit may be required in addition to an advance payment.

L

When a service or facility is discontinued, the amount of a deposit, if any, will be applied to the Customer's account and any credit balance remaining will be refunded. Before the service or facility is discontinued, the Company may, at its option and in accordance with Code of Missouri Regulations, Title 20, Subtitle 30, return the deposit or credit it to the Customer's account.



2.5.5.3

Deposits held will accrue interest at a rate specified by the Missouri Public Service Commission.

APR 3 0 2002 ISTRS 30 Public Service Commission MISSOURI

AUG 2 0 1999 Public Service Commission MISSOURI 98-575

96-355

EFFECTIVE: May 14, 1999 MISSOURI Public Service Curtifitission

FILED MAY 1 & 1999

ISSUED : March 2, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 30 RECENTO

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

SEP 17 1999

MO. FUDLIC SERVICE COMM

2.5 Payment Arrangements (Cont'd)

2.5.6 Discontinuance of Service

- 2.5.6.1 Upon nonpayment of any amounts owing to the Company, the Company may, by giving requisite prior written notice to the Customer in accordance with Code of Missouri Regulations, Title 20, Subtitle 45, discontinue or suspend service without incurring any liability.
- 2.5.6.2 Upon violation of any of the other material terms or conditions for furnishing service the Company may, by giving 30 days' prior notice in writing to the Customer, discontinue or suspend service without incurring any liability if such violation continues during that period.
- 2.5.6.3 Upon condemnation of any material portion of the facilities used by the Company to provide service to a Customer or if a casualty renders all or any material portion of such facilities inoperable beyond feasible repair, the Company, by notice to the Customer, may discontinue or suspend service without incurring any liability.
- 2.5.6.4 Upon the Customer's insolvency, assignment for the benefit of creditors, filing for bankruptcy or reorganization, failing to discharge an involuntary petition within the time permitted by law, or abandonment of service, the Company may, with prior notice to the customer, immediately discontinue or suspend service without incurring any liability.
- 2.5.6.5 Upon any governmental directive or any violation of any applicable law or regulation, the Company may immediately discontinue or suspend service without incurring any liability.
- 2.5.6.6 The Company may discontinue the furnishings of any and/or all service(s) to a Customer, without incurring any liability:
 - 2.5.6.6.1 Immediately and without notice if the Company deems that such action is necessary to prevent or to protect against fraud or to otherwise protect its personnel, agents, facilities or services. The Company may discontinue service pursuant to this sub-section.2.5.6.6.1 (a-f) if:
 - (a) The Customer refuses to furnish information to the Company regarding the Customer's current use of common carrier communications services or its planned use of service(s); or
 - The Customer provides false information to the Company regarding the Customer's identity, address, credit-worthiness, past or current use of common carrier communications services, or its planned use of the Company's service(s); or

ISSUED : March 2, 1999

CANCELLED

AUG 2 0 1999

By TX-F18-575 Public Service Commission (b)

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200

Atlanta GA 30328

L

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

96-355

FILED MAY 1.4 1999

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

LOCAL EXCHANGE SERVICE

- 2. Regulations (Cont'd)
 - 2.5 Payment Arrangements (Cont'd)
 - 2.5.6 Discontinuance of Service (Cont'd)

(c)

(d)

2.5.6.6.1 (Cont'd)

ORIGINAL PAGE NO. 31

MISSOURI P.S.C. TARIFF NO. 1

REC	Eľ	VE	5
SEP	17	199	9

MO. FUBLIC SERVICE CONIM

The Customer has been given ten (10) day written notice by the Company of any past due amount (which remains unpaid in whole or in part) for any of the Company's other common carrier communications services to which the Customer either subscribes or had subscribed or used; or

The Customer uses, or attempts to use, service with the intent to avoid the payment, either in whole or in part, of the tariffed charges for the service by:

- (d.1) Using or attempting to use service by rearranging, tampering with, or making connections to the Company's service not authorized by this tariff; or
- (d.2) Using tricks, schemes, false or invalid numbers, false credit devices, electronic devices; or
- (d.3) Any other fraudulent means or devices; or
- Use of service in such a manner as to interfere with the service of other users; or
- (f) Use of service for unlawful purposes.
- 2.5.6.6.2 Upon ten (10) days written notice to a Customer presently receiving service, after failure of the Customer to comply with a request made by the Company in accordance with Section 2.5.5 for increased security for the payment of service, pursuant to the applicable provisions of the Code of Missouri Regulations, Title 20, Subtitle 45, Chapter .04.
- 2.5.6.6.3 Ten (10) days after sending the Customer written notice of noncompliance with any provision of this tariff if the noncompliance is not corrected within that ten (10) day period; or

2.5.6.6.4 Upon five (5) days written notice, excluding Sundays and holidays, for nonpayment of a bill for service.

AUG 20 1999 By A 98.575 Public Service Commission MISSOURI

4

ISSUED : March 2, 1999

<u>96-355</u> EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission

FILFD MAY 1 & 1999

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.6 Discontinuance of Service (Cont'd)
 - 2.5.6.7 The suspension or discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for service(s) furnished during the time of or up to suspension or discontinuance.
 - 2.5.6.8 Upon the Company's discontinuance of service to the Customer under Section 2.5.6.1 or 2.5.6.2, all applicable charges, including termination charges, shall become due. This is in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff.

2.5.7

2.5.8 Late Fee

For Residential and Small Business Service customers, a late payment charge of 1.5% may be assessed on payments not received within 30 days from the invoice date, where capabilities exist. The late payment charge will be applied to the entire unpaid balance of the customer's monthly invoice including taxes. The late payment charge will not be applied to any disputed portion of the unpaid balance unless the dispute is resolved against the customer. The late payment charge of1.5% is not applicable to subsequent rebilling of any amount to which a late payment charge has already been applied. Late charges are to be applied without discrimination.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

D I I D

5

MISSOURI P.S.C. TARIFF NO. 1 4th REVISED PAGE NO. 32 CANCELS 3rd REVISED PAGE NO. 32

LOCAL EXCHANGE SERVICE

2. <u>Regulations (Cont'd)</u>

- 2.5 <u>Payment Arrangements</u> (Cont'd)
 - 2.5.6 <u>Discontinuance of Service (Cont'd)</u>
 - 2.5.6.7 The suspension or discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for service(s) furnished during the time of or up to suspension or discontinuance.
 - 2.5.6.8 Upon the Company's discontinuance of service to the Customer under Section 2.5.6.1 or 2.5.6.2, all applicable charges, including termination charges, shall become due. This is in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff.

2.5.7 Use Of Service

2.5.8 <u>Late Fee</u>

For Residential and Small Business Service customers, a late payment charge of 1.5% may be assessed on payments not received within 30 days from the invoice date, where capabilities exist. The late payment charge will be applied to the entire unpaid balance of the customer's monthly invoice including taxes. The late payment charge will not be applied to any disputed portion of the unpaid balance unless the dispute is resolved against the customer. The late payment charge of1.5% is not applicable to subsequent rebilling of any amount to which a late payment charge has already been applied. Late charges are to be applied without discrimination.

Issued: October 11, 2005

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL, 60601 Effective: November 11, 2005



D | D

LOCAL EXCHANGE SERVICE

2. <u>Regulations (Cont'd)</u>

- 2.5 <u>Payment Arrangements</u> (Cont'd)
 - 2.5.6 Discontinuance of Service (Cont'd)
 - 2.5.6.7 The suspension or discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for service(s) furnished during the time of or up to suspension or discontinuance.
 - 2.5.6.8 Upon the Company's discontinuance of service to the Customer under Section 2.5.6.1 or 2.5.6.2, all applicable charges, including termination charges, shall become due. This is in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff.
 - 2.5.7 Use Of Service
 - 2.5.7.1 Paper Invoices

For business Customers who receive notification that invoicing will change to E-Billing and who do not elect to use E-Billing, but continue to receive paper invoices, the following monthly recurring charge will apply per invoice based on the number of sheets in the paper invoice:

Monthly Recurring Charge: 1-55 sheets of paper \$5.00 56 or more sheets of paper \$25.00

2.5.8 Late Fee

For Residential and Small Business Service customers, a late N payment charge of 1.5% may be assessed on payments not received within 30 days from the invoice date, where capabilities exist. The late payment charge will be applied to the entire unpaid balance of the customer's monthly invoice including taxes. The late payment charge will not be applied to any disputed portion of the unpaid balance unless the dispute is resolved against the customer. The late payment charge of1.5% is not applicable to subsequent rebilling of any amount to which a late payment charge has already been applied. Late charges are to be applied without discrimination.

CANCELLED

November 11, 2005

MISSOURI PUBLIC SERVICE COMMISSION

Issued: May 27, 2003

Effective: June 27, 2003

MISSOURI P.S.C. TARIFF NO. 1 2nd REVISED PAGE NO. 32 CANCELS 1st REVISED PAGE NO. 32

LOCAL EXCHANGE SERVICE

2. <u>Regulations (Cont'd)</u>

Missouri Public Service Commission

RFCD FEB 06 2003

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.6 <u>Discontinuance of Service (Cont'd)</u>
 - 2.5.6.7 The suspension or discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for service(s) furnished during the time of or up to suspension or discontinuance.
 - 2.5.6.8 Upon the Company's discontinuance of service to the Customer under Section 2.5.6.1 or 2.5.6.2, all applicable charges, including termination charges, shall become due. This is in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff.
 - 2.5.7 Use Of Service
 - 2.5.7.1 Paper Invoices

For business Customers who receive notification that invoicing will change to E-Billing and who do not elect to use E-Billing, but continue to receive paper invoices, the following monthly recurring charge will apply per invoice based on the number of sheets in the paper invoice:

Monthly Recurring Charge: 1-55 sheets of paper \$5.00 56 or more sheets of paper \$25.00

2.5.8 <u>Late Fee</u>

For residential service customers, a late payment charge of 1.5% may be assessed on payments not received within 30 days from the invoice date, where capabilities exist. The late payment charge will be applied to the entire unpaid balance of the customer's monthly invoice including taxes. The late payment charge will not be applied to any disputed portion of the unpaid balance unless the dispute is resolved against the customer. The late payment charge of1.5% is not applicable to subsequent rebilling of any amount to which a late payment charge has already been applied. Late charges are to be applied without discrimination.





Missouri Public Service Commission

Ν

Ν

FILED MAR 06 2003

Effective: March 6, 2003

Issued: February 6, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

MCI SER

MISSOURI P.S.C. TARIFF NO. 1 1st REVISED PAGE NO. 32 CANCELS ORIGINAL PAGE NO. 32

Missouri Publio

RECD NOV 01 2002

N

Ν

LOCAL EXCHANGE SERVICE

<u>Regulations (Cont'd)</u>

2.5 Payment Arrangements (Cont'd)

2.5.6.7

2.5.6 <u>Discontinuance of Service (Cont'd)</u>

or discontinuance.

Service Commission The suspension or discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for

service(s) furnished during the time of or up to suspension

- 2.5.6.8 Upon the Company's discontinuance of service to the Customer under Section 2.5.6.1 or 2.5.6.2, all applicable charges, including termination charges, shall become due. This is in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff.
- 2.5.7 Use Of Service

2.5.7.1 Paper Invoices

For business Customers who receive notification that invoicing will change to E-Billing and who do not elect to use E-Billing, but continue to receive paper invoices, the following monthly recurring charge will apply per invoice based on the number of sheets in the paper invoice:

Monthly Recurring Charge:

1-55 sheets of paper \$5.00 56 or more sheets of paper \$25.00

CANCELLED

MAR 0 8 2003

mssion

Issued: November 1, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

FILED DEC 01 2002

Effective: December 1, 2002 Missouri Public Service Commission



MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 32

LOCAL EXCHANGE SERVICE

- 2. Regulations (Cont'd)
 - 2.5 Payment Arrangements (Cont'd)
 - 2.5.6 Discontinuance of Service (Cont'd)

MO. FUBLIC SERVICE COMM

RECEIVED

- 2.5.6.7 The suspension or discontinuance of service(s) by the Company pursuant to this Section does not relieve the Customer of any obligation to pay the Company for charges due and owing for service(s) furnished during the time of or up to suspension or discontinuance.
- 2.5.6.8 Upon the Company's discontinuance of service to the Customer under Section 2.5.6.1 or 2.5.6.2, all applicable charges, including termination charges, shall become due. This is in addition to all other remedies that may be available to the Company at law or in equity or under any other provision of this tariff.

CANCELLED

DEC_0 1 2002 ission

199 By Tree Commission Public Service Commission MISSOURI 575

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200

Atlanta, GA 30328

ISSUED : March 2, 1999

4

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

96-355

FILFD MAY 1 4 1999

SEP 17 1998

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 2nd Revised Page No. 32.1 Cancels 1st Revised Page No. 32.1

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.7 Use of Service (Cont'd)

2.5.7.2 No Fault Found Dispatch Charge for Business Customers A non-recurring charge applies when a Company representative is dispatched to the Customer's premises at the request of the Customer to investigate a suspected issue with Company service, and the Company representative responds to the dispatch and confirms the proper functioning of Company service. This can include, but is not limited to, dispatches requested to: assist with identifying a problem which turns out to be within the scope of the Customer/vendor-maintained equipment or network; provide technical assistance with Customer- or vendor-maintained network and equipment issues that are outside the scope of the Company's responsibility; or tag Company's demarcation point and, in doing so, the Company representative finds the circuit is clearly marked.

> The following charges apply per visit to the Customer's Premises, based on time of the visit

Time of Day/Charge

Normal Working Hours Outside of Normal Working Hours

\$265 \$400

For purposes of this provision, "Normal Working Hours" are defined as Monday to Friday, excluding New Year's Day, Martin Luther King Jr. Day, Presidents' Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day, Day after Thanksgiving and Christmas Day, 7 AM to 7 PM in the time zone of the Customer's premises; a visit to Customer premises which begins or ends outside of Normal Working Hours is "Outside of Normal Working Hours."

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: April 8, 2008

-

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 32.1 Cancels Original Page No. 32.1

LOCAL EXCHANGE SERVICE

- 2. <u>Regulations (Cont'd)</u> 2.5 <u>Payment Arrangements</u> (Cont'd)
 - 2.5.7 <u>Use Of Service</u> (cont'd)
 - 2.5.7.2 <u>No Fault Found Dispatch Charge for Business Customers</u> The Customer is responsible for the payment of a No Fault Found Dispatch Charge when:
 - -when requested by the Customer, maintenance personnel visit the Customer's premises, and
 - as a result of the visit, the proper functioning of the WorldCom service is confirmed (i.e., the cause of the trouble condition was other than a malfunction of a WorldCom service or of WorldCom maintained equipment). This can include, but not be limited to, customer requested dispatches
 - Where the root cause of a trouble is proven to be within the scope of the customer's or customer's vendor-owned equipment not maintained by WorldCom.
 - Where the root cause of a trouble has been proven to be within the scope of the customers or customer's vendor-owned inside wiring. To provide WorldCom technical assistance to the customer or customer's vendor in performing specific testing to isolate a problem which has been proved off the WorldCom network and is not within any WorldCom contract supported area.
 - To provide WorldCom technical assistance to the customer or customer's vendor in isolating or repairing a fault or installation support for areas not within WorldCom contract supported equipment, network or services.
 - In which the root cause of a trouble has been proven to be off the WorldCom network and is not within any WorldCom contract supported area and proves to be within the scope of the customer's or customer's vendor-owned network.

The charges are non-recurring, and are charged per visit as follows:

Normal Working Hours: \$265 per visit Outside of Normal Working Hours : \$400 per visit

Normal Working Hours are defined as Monday to Friday, 7am to 7pm in the time zone of the customer's location of the dispatch. If a visit begins and/or ends outside this period, it is considered Outside of Normal Working Hours.

Any Dispatch that begins or ends from 12:01 am to 12:00 am the following day the time zone of the customers Premises on these holidays will also be considered 'Outside of Normal Working Hours':

New Years Day	Labor
Martin Luther King Jr. Day	Thank
Presidents Day	Day a
Memorial Day	Chris
Independence Day	

Labor Day Thanksgiving Day Day after Thanksgiving Christmas Day

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: May 15, 2003



Ν

1

| N

MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 32.1

LOCAL EXCHANGE SERVICE

<u>Regulations (Cont'd)</u>

2.

2.5 <u>Payment_Arrangements</u> (Cont'd)

Original Page No. 32

Missouri Public Service Commission

RECTO FEB 28 2003

- 2.5.7 <u>Use Of Service</u> (cont'd)
 - 2.5.7.2 <u>No Fault_Found Dispatch Charge for Business Customers</u> The Customer is responsible for the payment of a No Fault Found Dispatch Charge when:
 - -when requested by the Customer, maintenance personnel visit the Customer's premises, and
 - as a result of the visit, the proper functioning of the WorldCom service is confirmed (i.e., the cause of the trouble condition was other than a malfunction of a WorldCom service or of WorldCom maintained equipment). This can include, but not be limited to, customer requested dispatches
 - Where the root cause of a trouble is proven to be within the scope of the customer's or customer's vendor-owned equipment not maintained by WorldCom..
 - Where the root cause of a trouble has been proven to be within the scope of the customers or customer's vendor-owned inside wiring. To provide WorldCom technical assistance to the customer or customer's vendor in performing specific testing to isolate a problem which has been proved off the WorldCom network and is not within any WorldCom contract supported area.
 - To provide WorldCom technical assistance to the customer or customer's vendor in isolating or repairing a fault or installation support for areas not within WorldCom contract supported equipment, network or services.
 - In which the root cause of a trouble has been proven to be off the WorldCom network and is not within any WorldCom contract supported area and proves to be within the scope of the customer's or customer's vendor-owned network.

The charges are non-recurring, and are charged per visit as follows:

Normal Working Hours: \$265 per visit Outside of Normal Working Hours : \$400 per visit

Normal Working Hours are defined as Monday to Friday, 7am to 7pm in the time zone of the customer's location of the dispatch. If a visit begins and/or ends outside this period, it is considered Outside of Normal Working Hours.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: February 28, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 1, 2003

Missouri Public Servico Commicelon

FILFD APR 01 2003



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 32.2 Cancels Original Page No. 32.2

LOCAL EXCHANGE SERVICE

2.

2

-

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

LOCAL EXCHANGE SERVICE

- 2. <u>Regulations (Cont'd)</u>
 - 2.5 <u>Payment Arrangements</u> (Cont'd)
 - 2.5.7 Use Of Service (cont'd)
 - 2.5.7.3 Telecommunications Service Priority (TSP) The Telecommunications Service Priority (TSP) program is a federally-established program under which the Office of Priority Telecommunications in the Executive Office of the President prioritizes the restoration and provisioning of telecommunications services - including services to private companies and institutions -- that support national security or emergency preparedness (NS/EP). The FCC defines telecommunications services under the TSP program to include the sending and receiving of signals or most any kind, by virtually any means. NS/EP services are those used to maintain a state of readiness or to respond to and manage any event or crisis (local, national, or international) that causes or could cause injury or harm to the population, damage to or loss of property, or that degrades or threatens the NS/EP posture of the United States. For telecommunications services enrolled in the program, the Company will provision and restore TSP-coded circuits, and provide TSP Special Construction services, under the terms set forth in this TSP service product description, and as required by the FCC's TSP regulations (currently at 47 CFR Part 64, Subpart D, Appendix A), and other applicable law. TSP services are in two categories: Priority Provisioning (including Emergency Provisioning and Essential Provisioning) and Priority Restoration.

Customers are subjected to the features, rate and charges as described in the Company's "Service Publication and Price Guide" (The Guide), located on the Company's website at www.mci.com.

ALL MATERIAL ON THIS PAGE IS NEW.

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: October 1, 2003



LOCAL EXCHANGE SERVICE

- 2. <u>Regulations (Cont'd)</u>
 - 2.5 <u>Payment Arrangements</u> (Cont'd)
 - 2.5.7 <u>Use Of Service</u> (cont'd)
 - 2.5.7.4 Local Disaster Recovery

MCI Disaster Recovery Plans Service is an optional feature for customers of MCI Local Line and/or Trunk Services. MCI Disaster Recovery Plans Service is not available for circuits provided via UNE-P.

MCI Disaster Recovery provides MCI Local Customers with pre-established Disaster Recovery Plans to be invoked in case of a local line/trunk outage related to an emergency /disaster. Disaster Recovery can be defined as a collection of actions, procedures, and information that is developed, tested and held in readiness for use in the event of an emergency / disaster. For purposes of this optional feature, an emergency / disaster is defined as any event that may cause a lengthy disruption of the Customer's local line/trunk service. These events include, but are not limited to, natural events, accidents, or events of sabotage. The customer must notify MCI when to invoke these pre-established plans with a secure password. These pre-established plans may consist of specific restoration processes involving the redirection of traffic through Remote Call Forward Feature Service; or Trunk Group Redirection. More complex plans that involve over100 numbers may also be established on an individual case basis. Changes or modifications to these plans can be made as part of the monthly recurring fee.

Local Disaster Recovery: Non Recurring Charge

 1 number to 10 numbers
 \$750

 11 numbers to 40 numbers
 \$850

 41 numbers to 100 numbers
 \$1,000

 101 numbers and above
 \$1,500

Local Disaster Recovery: Recurring Charge

1 number to 10 numbers	\$50
11 numbers to 40 numbers	\$100
41 numbers to 100 numbers	\$200
101 numbers and above	\$200

Filed

Missouri Public Service Commission



MCImetro Access Transmission Services LLC d/b/a Verízon Access Transmission Services Missouri P.S.C. Tariff No. 1 Original Page No. 32.4

2. Regulations (Cont'd)

5

i

LOCAL EXCHANGE SERVICE

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.7 Use of Service (Cont'd)
 - 2.5.7.5 811 Dialing Service

2.5.7.5.1 General

811 Dialing Service (811) is a custom call-routing application utilizing a three-digit local dialing arrangement, terminating to a subscriber-provided number for access to advance excavation notice services. It provides the calling party an easy-to-remember threedigit dialing code with call delivery to established 811 subscribers. The 811 code was assigned for this purpose pursuant to the Sixth Report and Order, released March 14, 2005 by the Federal Communications Commission in CC Docket No. 92-105, which specifies that such calls be delivered to a number provided by the relevant 811 subscriber that is not a toll call for the party dialing the number (i.e., either a toll-free (8XX) or local number). This tariff covers calls originating on lines terminating in a Verizon Access Transmission Services switch (i.e., originating and terminating within the same MSA); it does not cover 1+, 0+,0- operatorassisted, 101XXXX, or inmate calls).

2.5.7.5.2 Conditions

Calls placed using 811 are automatically routed to the 811 subscriber's terminating number, which the subscriber must provide in the form of either a tollfree number or a local number whose local calling area covers all of the locations to which the service is provided. The subscriber shall provide Verizon Access Transmission Services with this number in advance so that Verizon Access Transmission Services may properly translate its central office switches. If charges are required to re-route the call to the terminating number, they will be cared for by the use of a subscriber-provided toll-free number. Verizon Access Transmission Services is not responsible for redirecting or otherwise handling 911 and other calls misdialed or misrouted as 811 calls. The subscriber shall provide sufficient terminating number paths to its toll-free or local terminating number so as to not clog nor impair Verizon Access Transmission Services network.

Verizon Access Transmission Services offering of 811 to the subscriber also is conditioned on the subscriber's representation that it has been authorized by appropriate state authorities to receive and respond to 811 calls from the public within the areas served by Verizon Access Transmission Services and that the subscriber has obtained all licenses, authorizations, and other prerequisites necessary to provide that service, and will at all times comply with all applicable laws and regulations.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: March 19, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: April 20, 2007



LOCAL EXCHANGE SERVICE

2. <u>REGULATIONS (Cont'd)</u>

- 2.5 Payment Arrangements (Cont'd)
 - 2.5.7 Use of Service (Cont'd)
 - 2.5.7.5 <u>811 Dialing Service (Cont'd)</u>

2.5.7.5.3

2.5.7.5.2 Conditions (Cont'd)

The Company reserves the right to discontinue the service, without notice, if interruption of 811 is necessary to prevent or protect against fraud or otherwise protect Verizon Access Transmission Services personnel, facilities or services.

811 is not available for resale.

Limitations on Liability The Company shall be indemnified and saved harmless by the subscriber against claims for libel, slander, or the infringement of copyright arising directly or indirectly from the material transmitted over the facilities or the use thereof; against claims for infringement of patents arising from combining with or using in connection with, facilities furnished by the Company, apparatus and systems of the subscriber; and against all other claims arising out of any act or omission of the subscriber in connection with the facilities provided by the Company. Neither the Company nor any concurring, connecting or other participating carrier shall be liable for any act or omission of another company or companies furnishing a portion of such service. The Company is not responsible to the subscriber, authorized user, joint user, sharer of service, patron of a reseller or any other person for damages arising out of mistakes, omissions, interruptions, delays, errors or defects in transmission, or failures or defects in facilities furnished by the Company occurring in the course of furnishing service or other facilities (Service Problems) or other injury, including but not limited to injuries to persons or property from voltages or currents transmitted over the service of the Company caused by terminal equipment. The Company is not responsible to the subscriber, authorized user, joint user, sharer of service, patron of a reseller or any other person for injuries or damages to persons or property arising from the existence of subscriber-provided power supply.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: March 19, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: April 20, 2007

Filed Missouri Public Service Commission

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 33 Cancels ORIGINAL Page No. 33

LOCAL EXCHANGE SERVICE

2. <u>Regulations (Cont'd)</u>

2.6 <u>Allowances for Interruptions of Service</u>

2.6.1 <u>Credit for Interruptions</u>: When the use of service or facilities furnished by the Company is interrupted due to any cause other than the negligence or willful act of the Customer, or the operation or failure of the facilities or equipment provided by the Customer, a pro rata adjustment of the monthly Recurring Charges subject to interruption will be allowed for the service and facilities rendered useless and inoperative by reason of the interruption whenever said interruption continues for a period of 24 hours or more(or for a period of 8 hours N or more for customers of small business local service) from the time N the interruption is reported to or known to exist by the Company, except as otherwise specified in the Company's tariffs. If the Customer reports a service, facility or circuit to be inoperative but declines to release it for testing and repair, it is considered to be impaired, but not interrupted.

For calculating credit allowances, every month is considered to have 30 days. A credit allowance is applied on a pro rata basis against the monthly Recurring Charges specified hereunder for Local Line or Local Trunk Service and is dependent upon the length of the interruption. Only those facilities on the interrupted portion of the circuit will receive a credit. There will be no credit allowance for service outages less than 24 hours in duration. Credit allowances for service outages that exceed 24 hours in duration will be rounded up to the next whole 24 hours.

2.6.2 Limitations on Allowances

No credit allowance will be made for:

- (a) interruptions due to the negligence of, or noncompliance with the provisions of this tariff by, the Customer, Authorized User, Joint-User, or other common carrier providing service connected to the service of Company except where contracted by the Company;
- (b) interruptions due to the negligence of any person other than the Company including but not limited to the Customer or other common carriers connected to the Company's facilities except where contracted by the Company;
- (c) interruptions due to the failure or malfunction of non-Company equipment;
- (d) interruptions of service during any period in which the Company is not given full access to its facilities and equipment for the purpose of investigating and correcting interruptions;
- (e) interruptions of service during a period in which the Customer continues to use the service on an impaired basis;
- (f) interruptions of service during any period when the Customer has released service to the Company for maintenance purposes or for implementation of a Customer order for a change in service arrangements;
- (g) interruption of service due to circumstances or causes beyond the reasonable control of the Company.



MISSOURI P.S.C. TARIFF NO. 1 RECEIORIGINAL PAGE NO. 33

LOCAL EXCHANGE SERVICE

SFP 17 1998

2. Regulations (Cont'd)

2.6

MO. FUDLIC SERVICE JUMM

2.6.1 Credit for Interruptions: When the use of service or facilities furnished by the Company is interrupted due to any cause other than the negligence or willful act of the Customer, or the operation or failure of the facilities or equipment provided by the Customer, a pro rata adjustment of the monthly Recurring Charges subject to interruption will be allowed for the service and facilities rendered useless and inoperative by reason of the interruption whenever said interruption continues for a period of 24 hours or more from the time the interruption is reported to or known to exist by the Company, except as otherwise specified in the Company's tariffs. If the Customer reports a service, facility or circuit to be inoperative but declines to release it for testing and repair, it is considered to be impaired, but not interrupted.

For calculating credit allowances, every month is considered to have 30 days. A credit allowance is applied on a pro rata basis against the monthly Recurring Charges specified hereunder for Local Line or Local Trunk Service and is dependent upon the length of the interruption. Only those facilities on the interrupted portion of the circuit will receive a credit. There will be no credit allowance for service outages less than 24 hours in duration. Credit allowances for service outages that exceed 24 hours in duration will be rounded up to the next whole 24 hours.

2.6.2 Limitations on Allowances

(f)

Allowances for Interruptions of Service

No credit allowance will be made for:

- (a) interruptions due to the negligence of, or noncompliance with the provisions of this tariff by, the Customer, Authorized User, Joint-User, or other common carrier providing service connected to the service of Company except where contracted by the Company;
- (b) interruptions due to the negligence of any person other than the Company including but not limited to the Customer or other common carriers connected to the Company's facilities except where contracted by the Company;
- {c} interruptions due to the failure or malfunction of non-Company equipment;
- (d) interruptions of service during any period in which the Company is not given full access to its facilities and equipment for the purpose of investigating and correcting interruptions;
- (e) interruptions of service during a period in which the Customer continues to use the service on an impaired basis;

interruptions of service during any period when the Customer has released service to the Company for maintenance purposes or for implementation of a Customer order for a change in service arrangements;

interruption of service due to circumstances or causes beyond the reasonable control of the Company.

Sandy Chandler Tariff Manager Six Concourse Perkway, Suite 3200 Atlanta, GA 30328

<u>96-35</u>5 EFFECTIVE: May 14, 1999

Missouri Public Service Cummission

FILED MAY 1 & 1999

CANCELLED

MAR 1 9 2004

1St RS 33 Public Service Commission MISSOURI 2/0 1999

Public Service Commission (9) MISSOUR

ISSUED : March 2, 1999



Regulations (Cont'd)

2.

RE CHESDURIES. TARIFF NO. 1 **ORIGINAL PAGE NO. 34**

ļ

|

ł

ł

i

<u>96-3</u>55

LOCAL EXCHANGE SERVICE

SEP 17 1998

2.6	Allowance	es for Interr	uptions of Service (Cont'd)	MO. PUBLIC SERVICE COMM
	2.6.3	service p	<u>Iternative Service Provided by the Company</u> : Should rovided by the Company during the period that a serv ad rates and charges for the alternative service used.	
2.7	<u>Cancellati</u>	on of <u>Serv</u>	ice	
	2.7.1	<u>Cancella</u>	tion of Application for Service	
		2.7.1.1	Unless the Company breaches its obligations, appl unless the Company otherwise agrees. Where the application for service prior to the start of service o charges will be imposed except for those specified	e Company permits Customer to cancel an r prior to any special construction, no
		2.7.1.2	Where, prior to cancellation by the Customer, the the service or in preparing to install the service that charge equal to the costs the Company incurred, le shall this charge exceed the sum of the charge for including installation charges, and all charges other have been chargeable to the Customer had service	It it otherwise would not have incurred, a ass net salvage, shall apply, but in no case the minimum period of service ordered, rs levy against the Company that would
		2.7.1.3	The special charges described in 2.7.1.1 and 2.7.1 by-case basis.	.2 will be calculated and applied on a case-
	2.7.2	<u>Cançella</u>	tion of Service by the Customer	
L		before th in 2.6.1 a owing as	e Company breaches its obligation, if a Customer car e completion of the term for any reason whatsoever o bove), Customer agrees to pay to Company the follow of the effective date of the cancellation or termination all costs, fees and expenses reasonable incurred in co	ther than a service interruption (as defined wing sums which shall become due and and be payable within the period set forth
		1)	all Non-Recurring Charges reasonably expended b Customer, plus	y Company to establish service to
CANCELLED		2)	any disconnection, early cancellation or termination third parties by Company on behalf of Customer, p	· · · · ·
AUG 2 0 1999		3)	if service is terminated prior to the completion of the the applicable tariff for the balance of the then curr	
By TAX 98-575 Public Service Commissi MISSOURI	10 n		ges referenced above will be waived if the customer to ess otherwise provided for by contract.	erminates service in order to complete 900
\mathcal{I}				96

ISSUED : March 2, 1999

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission

FILED MAY 1 4 1999

MISSOURI P.S.C. TARIFF NO. 1 RECEIVED ORIGINAL PAGE NO. 35

SEP 17 1990

MŨ. FUBLIU SERVIUL UUMM

LOCAL EXCHANGE SERVICE

2. Regulations (Cont'd)

2.8 Transfer and Assignments

Neither the Company nor the Customer may assign or transfer its rights or duties in connection with the services and facilities provided by the Company without the written consent of the other party, except that the Company may assign its rights and duties (a) to any subsidiary, parent company or affiliate of the Company; (b) pursuant to any sale or transfer of substantially all the assets of the Company; or (c) pursuant to any financing, merger or reorganization of the Company.

2.9 Notices and Communications

- 2.9.1 The Customer shall designate on the Service Order an address to which the Company shall mail or deliver all notices and other communications, except that Customer may also designate a separate address to which the Company's bills for service shall be mailed.
- 2.9.2 The Company shall designate on the Service Order an address to which the Customer shall mail or deliver all notices and other communications, except that Company may designate a separate address on each bill for service to which the Customer shall mail payment on that bill.
- 2.9.3 Notices and other communications of either party, and all bills mailed by the Company, shall be presumed to have been delivered to the other party on the third business day following deposit of the notice, communication or bill with the U.S. Mail or a private delivery service, prepaid and property addressed, or when actually received or refused by the addressee, whichever occurs first.
- 2.9.4 The Company or the Customer shall advise the other party of any changes to the addresses designated for notices, other communications or billing, by following the procedures for giving notice set forth herein.

CANCELLED AUG 2 Ø 1999 /98-575 Public Service Commission MISSOURI

ISSUED : March 2, 1999

Ĺ

<u>96-3</u>55 EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

FILED MAY 1 4 1999

Missouri Public Service Commission
LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

ć

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Accewss Transmission Services LLC d/b/a Verizon Access Transmission Services

MO PSC TARIFF NO. 1 Original Page No. 35.1

INTEREXCHANGE TELECOMMUNICATIONS SERVICES TARIFF

2. <u>REGULATIONS (Cont.</u>)

3.0 Individual Case Basis (ICB) Arrangements

Private line services will be made available to customers in a nondiscriminatory manner. Rates for dedicated access, private line, non-switched and special access services will be determined on an individual Case Basis (ICB). ICB rates will be structured to recover the Company's cost of providing the service and will be made available to the Commission staff upon request on a proprietary basis. Rates may be negotiated and adjusted on individual contract Basis (ICB) Customer Specific Proposals (CSPs) in an exchange in which basic local exchange company has been declared competitive. Terms of ICB/CSP provided to business will be provided on a proprietary basis to the Commission Staff, upon request.

Issued: August 28, 2006

Carmen L. feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: September 29, 2006

Filed Missouri Public Service Commission

CANCELLED September 1, 2008 Missouri Public Service Commission LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

LOCAL EXCHANGE SERVICE

з. Service Descriptions

- 3.1 Local Exchange Service: The Company's Local Telephone Service provides a Customer with the ability to connect to the Company's switching network which enables the Customer to:
 - place or receive calls to any calling Station in the local calling area, as defined herein;
 - access enhanced 911 Emergency Service;
 - access the interexchange carrier selected by the Customer for interLATA, intraLATA, interstate or international calling;
 - access Operator Services;
 - access Directory Assistance for the local calling area;
 - place or receive calls to 800 telephone numbers;
 - access Telecommunication Relay Service.

The Company's service can not be used to originate calls to other telephone companies caller-paid information services (900, 976). Calls to those numbers and other numbers used for caller-paid information services will be blocked by the Company's switch.

3.1.1 Service Area:

<u>Plan 1</u>: Where facilities are available, service areas are Α. defined by the following Exchange or Zone designations:

Kansas City/Springfield Service Area

The Company's service area consists fo the Principal Zone and MCA 1 of the Kansas City and Springfield Metropolitan Exchanges as defined in Southwestern Bell Telephone Company's Local Exchange tariffs. The company's provision of service within said service area is subject to the availability of appropriate facilities.

<u>Louis Service Area</u> Bridgeton, Creve Coeur, Ladue, Overland, St. Louis, Fenton, Manchester, St Charles.

Plan 2: For service provisioned via UNE-Platform (UNE-P), the Β. service are defined below. UNE-P is a service delivery method where Company obtains local exchange facilities via unbundled network elements through the incumbent ILEC.

unbundled network elements through the incumbent ILEC. Adrian, Advance, Agency, Antonia, Archie, Argyle, Armstrong, Ash Grove, Beaufort, Bell City, Benton, Billings, Bismarck, Bloomfield, Bloomsdale, Blue Spring MT, Blytheville, Bonneterre, Boonville, Bowling Green, Brookfield, Camdenton, Campbell, Cape Girard, Carl Jct, Carrollton, Carthage, Caruthersville, Cedar Hill, Center, Chaffee, Charleston, Chillicoth, Clarksville, Clever, Climax Spring, De Kalb, De Soto, Deering, Delta, Dexter, Downing E Atchison, E Prairie Edina, E Fort Scott, Eldon, Elsberry, E Pittsburg, Essex, Eureka, Fair Grove, FArley, Farmington, Fayette, Fisk, Flat River, Frankford, Fredercktn, Freeburg, Fulton, Gideon, Glasgow, Grain Valley, Gravois MI, Gray Summit, Greenwedmet, Hayti, Higbee, High Ridge, Hillsboro, Holcomb, Hornersville, Imperial, Jackson, Jasper, Joplin, Kennett, Kirksville, Knobnoster, La Monte, Lamar, Lancaster, Leadwood, Lilbourn, Linn, Lockwood, Louisiana, Macks Creek, Malden, Marble Hill, Marceline, Marionville, Marshall, Marston, Maxville, Meta, Mexico, Moberly, Monett, Montgory, Morehouse, Neosho, Nevada, New Madrid, New Franklin, Nixa, Oak Ridge, Oakville, Old Appleton, Oran, Pacific, Patton, Paynesville, Perryville, Pierce City, Pocahontas, Pond, Poplar Bluff, Portageville, Portagsiux, Puxico, Qulin, Republic, Richmond, Richwoods, Risco, Rushville, San Antonio, Scott City, Sedalia, Senath, Sikeston, Slater, Smithvlmet, St Clair, St Joseph, St Marys, Stanberry, Stegnvieve, Strafford, Trenton, Tuscumbia, Union, Versailles, Vienna, Walnut Grove, Wardell, Ware, Washington, Webb City, Wellsville, Westphalia, Willard,Wyatt

CERTAIN MATERIAL ON THIS PAGE WAS MOVED TO PAGE NO. 36.1

Issued: November 5, 2003 CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601



Т

Ν

Ň

MISSOURI P.S.C. TARIFF NO. 1 2nd REVISED PAGE NO. 36 Cancels 1st REVISED PAGE NO. 36

Т

ሞ

LOCAL EXCHANGE SERVICE

Service Descriptions

- Local Exchange Service: The Company's Local Telephone Service provides a Customer with the ability to connect to the Company's switching network which 3.1enables the Customer to:
 - place or receive calls to any calling Station in the loca Missouri Public as defined herein;
 - access enhanced 911 Emergency Service;
 - access enhanced 911 Emergency Service, access the interexchange carrier selected by the Customer for interLATA 2003

 - access Operator Services;
 access Directory Assistance for the local calling area;
 Service Commission - access Telecommunication Relay Service.

The Company's service can not be used to originate calls to other telephone companies caller-paid information services (900, 976). Calls to those numbers and other numbers used for caller-paid information services will be blocked by the Company's switch.

3.1.1 <u>Service Area:</u>

Α. <u>Plan 1</u>: Where facilities are available, service areas are defined by the following Exchange or Zone designations: T

Kansas City/Springfield Service Area

The Company's service area consists fo the Principal Zone and MCA 1 of the Kansas City and Springfield Metropolitan Exchanges as defined in Southwestern Bell Telephone Company's Local Exchange tariffs. The company's provision of service within said service area is

subject to the availability of appropriate facilities.

<u>St Louis Service Area</u> Bridgeton, Creve Coeur, Ladue, Overland, St. Louis, Fenton, Manchester, St Charles.

Β. <u>Plan 2:</u> For service provisioned via UNE-Platform (UNE-P), the Ν service areas will mirror the existing ILEC Exchange Services Tariff. UNE-P is a service delivery method where Company obtains local exchange facilities via unbundled network elements through the incumbent ILEC. Ν

3.1.1.1 Local Calling Areas:

A. Plan 1: For company-provided facility based service, where facilities are available, Exchanges or Zones included in the local calling area for each of the Service Areas designations are specified below. Service Areas associated with each particular exchange or zone may be found in the telephone directory published by the dominant exchange service provider in the Customer's exchange area. JOIN

<u>Kansas City/Springfield Local Calling Areas</u> The local calling area for the Company's Missouri customers consists of the geographic area encompassed within the Kansas City and Springfield Metropolitan Exchange (Principal Zone and MCA-1 and MCA-2 Zones) as defined in Southwestern Bell Telephone Company's Local Exchange tariffs.

B. Plan 2: For service provisioned via UNE-Platform (UNE-P), N the Calling areas will mirror the existing ILEC Exchange Services Tariff. UNE-P is a service delivery method where Company obtains local exchange facilities via unbundled L network elements through Public ncumbent ILEC. Service Commission N

Issued: October 3, 2003

CANCELLER

DEC 06 2003

rals 34

FIED NOV 03 2003 Effective: November 3, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

5

MISSOURI P.S.C. TARIFF NO. 1 1st REVISED PAGE NO. 36 Cancels ORIGINAL PAGE NO. 36 Missouri Public

LOCAL EXCHANGE SERVICE

REC'D AUG 13 2003

Service Descriptions

- 3.1 <u>Local Exchange Service</u>: The Company's Local Teleph**Service Gommission** Customer with the ability to connect to the Company's switching network which enables the Customer to:
 - place or receive calls to any calling Station in the local calling area, as defined herein;
 - access enhanced 911 Emergency Service;
 - access the interexchange carrier selected by the Customer for interLATA, intraLATA, interstate or international calling;
 - access Operator Services;
 - access Directory Assistance for the local calling area;
 - place or receive calls to 800 telephone numbers;
 - access Telecommunication Relay Service.

The Company's service can not be used to originate calls to other telephone companies caller-paid information services (900, 976). Calls to those numbers and other numbers used for caller-paid information services will be blocked by the Company's switch.

3.1.1 <u>Service Area:</u> Where facilities are available, service areas are defined by the following Exchange or Zone designations:

Kansas City/Springfield Service Area

The Company's service area consists fo the Principal Zone and MCA 1 of the Kansas City and Springfield Metropolitan Exchanges as defined in Southwestern Bell Telephone Company's Local Exchange tariffs. The company's provision of service within said service area is subject to the availability of appropriate facilities.

<u>St Louis Service Area</u>

Bridgeton, Creve Coeur, Ladue, Overland, St. Louis, Fenton, Manchester, St Charles.

D

-

3.1.1.1 <u>Local Calling Areas:</u> Exchanges or Zones included in the local calling area for each of the Service Areas designations are specified below. Service Areas associated with each particular exchange or zone may be found in the telephone directory published by the dominant exchange service provider in the Customer's exchange area.

orramission

<u>Kansas City/Springfield Local Calling Areas</u> The local calling area for the Company's Missouri customers consists of the geographic area encompassed within the Kansas City and Springfield Metropolitan Exchange (Principal Zone and MCA-1 and MCA-2 Zones) as defined in Southwestern Bell Telephone Company's Local Exchange tariffs.



NOV. 0 3 2003

SCILIAI

Missouri Public Sorvico Commission

FILED SEP 13 2003

Issued: August 13, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Public

Effective: September 13, 2003

RECENT MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 36

LOCAL EXCHANGE SERVICE SEP 17 1999

3. <u>Service Descriptions</u>

MU. FUBLIC SERVICE COMM

- 3.1 Local Exchange Service: The Company's Local Telephone Service provides a Customer with the ability to connect to the Company's switching network which enables the Customer to:
 - place or receive calls to any calling Station in the local calling area, as defined herein;
 - access enhanced 911 Emergency Service;
 - access the interexchange carrier selected by the Customer for interLATA, intraLATA, interstate or international calling;
 - access Operator Services;
 - access Directory Assistance for the local calling area;
 - place or receive calls to 800 telephone numbers;
 - access Telecommunication Relay Service.

The Company's service can not be used to originate calls to other telephone companies caller-paid information services (900, 976). Calls to those numbers and other numbers used for caller-paid information services will be blocked by the Company's switch.

3.1.1 <u>Service Area:</u> Where facilities are available, service areas are defined by the following Exchange or Zone designations:

Kansas City/Springfield Service Area

The Company's service area consists to the Principal Zone and MCA 1 of the Kansas City and Springfield Metropolitan Exchanges as defined in Southwestern Bell Telephone Company's Local Exchange tariffs. The company's provision of service within said service area is subject to the availability of appropriate facilities.

St Louis Service Area

Bridgeton, Creve Coeur, Ladue, Overland, St. Louis, Fenton, Imperial, Manchester, St Charles.

3.1.1.1 Local Calling Areas: Exchanges or Zones included in the local calling area for each of the Service Areas designations are specified below. Service Areas associated with each particular exchange or zone may be found in the telephone directory published by the dominant exchange service provider in the Customer's exchange area.

Kansas City/Springfield Local Calling Areas

The local calling area for the Company's Missouri customers consists of the geographic area encompassed within the Kansas City and Springfield Metropolitan Exchange (Principal Zone and MCA-1 and MCA-2 Zones) as defined in Southwestern Bell Telephone Company's Local Exchange tariffs **ANALLI LED**

2 0 1999 Public Service Commission MISSOUR

EFFECTIVE: May 14, 1999

96-355

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission

FILED MAY 1 & 1999



٤

ISSUED : March 2, 1999



т

Ν

LOCAL EXCHANGE SERVICE

3. Service Descriptions

3.1.1.1 Local Calling Areas:

A. Plan 1: For company-provided facility based service, where facilities are available, Exchanges or Zones included in the local calling area for each of the Service Areas designations are specified below. Service Areas associated with each particular exchange or zone may be found in the telephone directory published by the dominant exchange service provider in the Customer's exchange area.

Kansas City/Springfield Local Calling Areas The local calling area for the Company's Missouri customers consists of the geographic area encompassed within the Kansas City and Springfield Metropolitan Exchange (Principal Zone and MCA-1 and MCA-2 Zones) as defined in Southwestern Bell Telephone Company's Local Exchange tariffs.

B. Plan 2: For service provisioned via UNE-Platform (UNE-P), the Calling areas are defined below. UNE-P is a service delivery method where Company obtains local exchange facilities via unbundled network elements through the incumbent ILEC.

Company obtains local exchange facilities via unbundled network elements through the incumbent ILEC. Adrian, Advance, Agency, Antonia, Archie, Arkadelphia, Armstrong, Ant Grove, Ashdown, Atchison, Atkins, Augusta, Baschor, Batesville, Bauxite, Beaufort, Beebe, Bell Clivy, Bell Flower, Beiton, Benton, Bethel, Billings, Bismarck, Black Rock, Bioomfield, Bloomsdale, Blue Spring Mt, Blytheville, Bodcaw, Bonner Spg, Bonneterre, Boonville, Bowlingreen, Brimson, Brinkley, Borrkfield, Buckner, Cabot, Camden Pt, Camdenton, Campbell, Cape Girard, Cardwell, Carl Jct, Carrollton, Carthage, Caruthrsville, Cash, Cave City, Cedar Hill, Center, Center Ridge, Chaffee, Charleston, Cherry Valley, Chesterfield, Chillicoth, Clarksville, Cleveland, Clever, Climax Spg, Concord, Conway, Crawfordsville, Crevecoeur, Crystal Spg, Dardenne, De Kalb, De Soto, Dearborn, Deering, Defiance, Dell, Delfa, Dexter, Downing, Drexel, E Atchison, E Prairie, Earle, East Lynne, Edgerton, Edina, Effingham, E Fort Scott, Eldon, Elkins, Elkland, Elsberry, Flwood, E Pittsburg, Essex, Bureka, Fair Grove, Farley, Farmington, Fayette, Fenton, Ferndale, Ferrelview, Fisk, Flat River, Fordland, Boristell, Forres City, Harrison, Holden, Alt, Hope, Hornersville, Haltown, Harrisburg, Harrison, Holden, Alt, Hope, Hornersville, Heitspy, Heins, Henrietta, Hickory Ridge, Higbee, High Ridge, Highlandvillage, Hillsboro, Hulcon, Holden, Alt, Hope, Hornersville, Hot Springs, Hughes, Hurley, Imperial, Independence, Jackson, Jasper, Jessieville, Joiner, Johesboro, Jonesmills, Jopin, Kaşkakia, Kearney, Kennett, Kingston, Kingsville, Kirkwood, Knobnoster, La Wonte, Ladue, Lamar, Lancaster, Laredo, Lathrop, Lawson, Leadwood, Leavehlsng, Levenworth, Lees Summit, Liberty, Lilourne, Linn, Little Rock, Lockwood, Lone Jack, Lonoke, Lousialan, Luxora, MacKscreek, Malden, Malvern, Manchester, Marbield, Marston, Maxville, McClure, Mehville, Mena, Meta, Mexice, Midland, Milo, Missouir City, Moberly, Monett, Montory City, Morehouse, Moro, Morrilton, Morrisville, Moscow Mi, Nashville, Ne

Wyatt, Wynne. MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 36

Effective: December 6, 2003 **Filed** Missouri Public

Service Commission

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 8th Revised Page No. 37 Cancels 7th Revised Page No. 37

D

LOCAL EXCHANGE SERVICE

3. Service Descriptions

.

3.1 Local Exchange Service (Cont'd)

- 3.1.1 Service Area (Cont'd)
 - 3.1.1.1 Local Calling Areas (Cont'd)

St Louis Metropolitan Exchange (SME) Local Calling Areas The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Creve Coeur, Kirkwood, Ladue, Mehlville, Oakville, Sappington, St. Louis Principal.

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro</u> <u>Calling (EMC)</u>)

The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Beaufort, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton, Festus-Crystal City, Foley, Foristell, Gray Summit, Harvester, Herculaneum - Pevely, High Ridge, Hillsboro, Maxville, Holstein, Imperial Manchester, Marthasville, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Clair, St. Peters, Troy, Union Valley Park, Ware, Washington, Warrenton, Wentzville, Winfield, Wright City

Exchange	Local Calling Area	Expanded Calling Area
Creve Coeur	All Exchanges of The SME	D All Exchanges of the EMC
Ladue	All Exchanges of The SME	the EMC
St Louis	All Exchanges of The SME	D All Exchanges of the EMC
Fenton	East St. Louis All Exchanges of	Granite City, IL N All Exchanges of
Manchester	The SME All Exchanges of	
St Charles	The SME All Exchanges of The SME	
Mehlville	All Exchanges of The SME	the EMC All Exchanges of the EMC
Harvester	All Exchanges of The SME	All Exchanges of the EMC
Valley Park	All Exchanges of The SME	All Exchanges of the EMC
Chesterfield	All Exchanges of The SME	All Exchanges of the EMC
Kirkwood	All Exchanges of The SME	the EMC -
Sappington	All Exchanges of The SME	the EMC
Maxville	All Exchanges of The SME	All Exchanges of N the EMC

I/Expanded Metropolitan Calling Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA toll charges.

Issued: September 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: October 1, 2007

Missouri P.S.C. Tariff No. 1 7th Revised Page No. 37 Cancels 6th Revised Page No. 37

LOCAL EXCHANGE SERVICE

3. Service Descriptions

3.1 Local Exchange Service (Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

St Louis Metropolitan Exchange (SME) Local Calling Areas The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Creve Coeur, Kirkwood, Ladue, Mehlville, Oakville, Overland, Riverview, Sappington, St. Louis Principal.

St Louis Expanded Metropolitan Exchange 1/ (Expanded MetroCalling (EMC))The Expanded Service Area of the St Louis MetropolitanExchange Area consists of the following exchanges: Antonia,Augusta, Beaufort, Cedar Hill, Chesterfield, Dardenne,Defiance, Desoto, Eureka, Fenton, Festus-Crystal City,Foley, Foristell, Gray Summit, Harvester,Herculaneum - Pevely, High Ridge, Hillsboro, Maxville,Holstein, Imperial Manchester, Marthasville, Moscow Hills,New Melle, O'Fallon, Old Monroe, Orchard Farm, PacificPond, Portage Des Sioux, St. Charles, St. Clair,St. Peters, Troy, Union Valley Park, Ware, Washington,Warrenton, Wentzville, Winfield, Wright City

Exchange	Local Calling Area		Expanded Calling Area	
Bridgeton	All Exchanges The SME	of	All Exchanges of the EMC	
Creve Coeur	All Exchanges The SME	of	All Exchanges of the EMC	
Ladue	All Exchanges The SME	of	All Exchanges of the EMC	
Overland	All Exchanges The SME	of	All Exchanges of the EMC	
St Louis	All Exchanges The SME	of	All Exchanges of the EMC	
Fenton	All Exchanges The SME	of	All Exchanges of the EMC	
Manchester	All Exchanges The SME	of	All Exchanges of the EMC	
St Charles	All Exchanges The SME	of	All Exchanges of the EMC	
Mehlville	All Exchanges The SME	of	All Exchanges of the EMC	
Harvester	All Exchanges The SME	of	All Exchanges of the EMC	
Valley Park	All Exchanges The SME	of	All Exchanges of the EMC	
Chesterfield	All Exchanges The SME	of	All Exchanges of the EMC	
Kirkwood	All Exchanges The SME	of	All Exchanges of the EMC	
Sappington	All Exchanges The SME	of	All Exchanges of the EMC	

1/Expanded Metropolitan Calling Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA toll charges.

Issued: April 20, 2007 CANCELLED Oct. 1, 2007 Missouri Public Service Commission

Carmen L.Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: May 21, 2007

Filed Missouri Public Service Commission MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

MISSOURI P.S.C. TARIFF NO. 1 6th Revised Page No. 37 Cancels 5th Revised Page No. 37

3. <u>Service Descriptions</u>

LOCAL EXCHANGE SERVICE

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.1 <u>Service Area (Cont'd)</u>
 - 3.1.1.1 Local Calling Areas (Cont'd)

<u>St Louis Metropolitan Exchange (SME) Local Calling Areas</u> The Local callings area of the ST. Louis Metropolitan Exchange Area consists of the following exchanges: Creve Coeur, Kirkwood, Ladue, Mehlville, Oakville, Sappington, St. Louis Principal.

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro</u> <u>Calling (EMC)</u>) The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Beaufort, Cedar Hill, Chesterfield, Dardenne, Defiance, N Desoto, Eureka, Fenton, Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Maxville, Imperial Manchester, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Clair, St. Peters, Troy, Union N Valley Park, Ware, Washington, Wentzville, Winfield N

E <u>xchange</u> Creve Coeur	Local <u>Calling Area</u> All Exchanges the SME	of	<u>Call</u>	inded <u>ing Area</u> Exchanges EMC	of
Ladue	All Exchanges	of	All	Exchanges	of
St Louis	All Exchanges the SME East St.Louis		the	Exchanges EMC ite City,	
Fenton	All Exchanges the SME	of	All the	Exchanges EMC	of
Manchester	All Exchanges the SME	of	All the	Exchanges EMC	of
St Charles	All Exchanges the SME	of	All the	Exchanges EMC	of
Mehlville	All Exchanges the SME	of	All the	Exchanges EMC	of
Harvester	All Exchanges the SME	of	All the	Exchanges EMC	of
Valley Park	All Exchanges the SME	of	All the	Exchanges EMC	of
Chesterfield	All Exchanges the SME	of	All the	Exchanges EMC	of
Kirkwood	All Exchanges the SME	of	All the	Exchanges EMC	of
Sappington	All Exchanges the SME	of	All the	Exchanges EMC	of

1/Expanded Metropolitan Calling Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA Toll charges.

Issued: October 10, 2006

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: November 10, 2006



Ν

May 21, 2007 Missouri Public Service Commission

Cancelled

MISSOURI P.S.C. TARIFF NO. 1 5th Revised Page No. 37 Cancels 4th Revised Page No. 37

D

D

3. Service Descriptions

3.1 Local Exchange Service (Cont'd)

- 3.1.1 <u>Service Area (Cont'd)</u>
 - 3.1.1.1 Local Calling Areas (Cont'd)

LOCAL EXCHANGE SERVICE

<u>St Louis Metropolitan Exchange (SME) Local Calling Areas</u> The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Creve Coeur, Kirkwood, Ladue, Mehlville, Oakville, Sappington, St. Louis Principal

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro</u> <u>Calling (EMC))</u>

The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton, Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Maxville, Imperial Manchester, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Peters, Troy, Valley Park, Ware, Wentzville, Winfield

E <u>xchange</u>	Local <u>Calling Area</u>	Expanded <u>Calling Area</u> I	2
Creve Coeur	All Exchanges of The SME	All Exchanges of	,
Ladue	All Exchanges of)
	All Exchanges of The SME	the EMC	
Fenton	All Exchanges of The SME	All Exchanges of the EMC	
Manchester	All Exchanges of The SME	the EMC	
St Charles	All Exchanges of The SME	All Exchanges of the EMC	
Mehlville	All Exchanges of The SME	All Exchanges of the EMC	
Harvester	All Exchanges of The SME	the EMC	
Valley Park	All Exchanges of The SME	All Exchanges of the EMC	
Chesterfield	All Exchanges of The SME	All Exchanges of the EMC	
Kirkwood	All Exchanges of The SME	All Exchanges of the EMC	
Sappington	All Exchanges of The SME	All Exchanges of the EMC	

1/Expanded Metropolitan Calling Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA Toll charges.

Issued: May 10, 2006

Cancelled

November 10, 2006 Missouri Public Service Commission Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: June 11, 2006



с

MISSOURI P.S.C. TARIFF NO. 1 4th REVISED PAGE NO. 37 CANCELS 3rd REVISED PAGE NO. 37

3. <u>Service Descriptions</u>

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.1 <u>Service Area (Cont'd)</u>
 - 3.1.1.1 <u>Local Calling Areas (Cont'd)</u>

LOCAL EXCHANGE SERVICE

<u>St Louis Metropolitan Exchange (SME) Local Calling Areas</u> The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Bridgeton, Creve Coeur, Ferguson, Florissant, Kirkwood, Ladue, Mehlville, Oakville, Overland, Riverview, Sappington, St. Louis Principal, Spanish Lake, Webster Groves

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro</u> <u>Calling (EMC)</u>) The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton, Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Maxville, Imperial Manchester, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Peters, Troy, Valley Park, Ware, Wentzville, Winfield

E <u>xchange</u>	Local <u>Cal</u> ling	g Area			nded ing Area	
Bridgeton	All Exc The SM	changes F	of	All the	Exchanges	of
Creve Coeur		changes	of		Exchanges	of
Ladue		changes	of		Exchanges	of
Overland		changes	of		Exchanges	of
St Louis	All Exe The SM	changes E	of	All the	Exchanges EMC	of
Fenton		changes	of	A11	Exchanges EMC	of
Manchester	All Exe The SM	changes E	of		Exchanges EMC	of
St Charles		changes	of	All	Exchanges EMC	of
Mehlville	All Ex The SM	changes E	of		Exchanges EMC	of
Harvester	All Ex The SM	changes E	of		Exchanges EMC	of
Valley Park		changes	of	All	Exchanges EMC	of
Chesterfield	All Ex The SM	changes E	of		Exchanges EMC	of
Kirkwood	All Ex The SM	changes E	of	All the	Exchanges EMC	of
Sappington	All Ex The SM	changes E	of	All	Exchanges EMC	of

1/Expanded Metropolitan Callin Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA toll charges.

Issued: March 30, 2005

Cancelled

June 11, 2006

Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: May 1, 2005



Т

| T

MISSOURI P.S.C. TARIFF NO. 1 3rd REVISED PAGE NO. 37 CANCELS 2nd REVISED PAGE NO. 37

LOCAL EXCHANGE SERVICE

3. Service Descriptions

٥

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.1 Service Area (Cont'd)
 - 3.1.1.1 Local Calling Areas (Cont'd)

<u>St Louis Metropolitan Exchange (SME) Local Calling Areas</u> The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Bridgeton, Creve Coeur, Ferguson, Florissant, Kirkwood, Ladue, Mehlville, Oakville, Overland, Riverview, Sappington, St. Louis Principal, Spanish Lake, Webster Groves

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro</u> <u>Calling (EMC)</u>) The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton, Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Maxville, Imperial N Manchester, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Peters, Troy, Valley Park, Ware, Wentzville, Winfield

E <u>xchange</u>	Local <u>Calling Area</u>	Expanded <u>Calling Area</u>
Bridgeton	All Exchanges of The SME	All Exchanges of the EMC
Creve Coeur	All Exchanges of The SME	
Ladue	All Exchanges of The SME	
Overland	All Exchanges of The SME	
St Louis	All Exchanges of The SME	All Exchanges of the EMC
Fenton	All Exchanges of The SME	All Exchanges of the EMC
Manchester	All Exchanges of The SME	All Exchanges of the EMC
St Charles	All Exchanges of The SME	
Mehlville	All Exchanges of The SME	All Exchanges of the EMC
Harvester	All Exchanges of The SME	All Exchanges of the EMC
Valley Park	All Exchanges of The SME	
Chesterfield	All Exchanges of The SME	
Kirkwood	All Exchanges of The SME	
Sappington	All Exchanges of The SME	

MISSOURI P.S.C. TARIFF NO. 1 2nd REVISED PAGE NO. 37 CANCELS 1st REVISED PAGE NO. 37

ł

Ν

Ν

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.1 <u>Service Area (Cont'd)</u>
 - 3.1.1.1 Local Calling Areas (Cont'd)

<u>St Louis Metropolitan Exchange (SME) Local Calling Areas</u> The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Bridgeton, Creve Coeur, Ferguson, Florissant, Kirkwood, Ladue, Mehlville, Oakville, Overland, Riverview, Sappington, St. Louis Principal, Spanish Lake, Webster Groves

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro</u> <u>Calling (EMC)</u>) The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton, Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Maxville, Manchester, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Peters, Troy, Valley Park, Ware, Wentzville, Winfield

E <u>xchange</u>	Local <u>Calling Area</u>	Expanded Calling_Area
Bridgeton	All Exchanges of The SME	All Exchanges of the EMC
Creve Coeur	All Exchanges of The SME	All Exchanges of the EMC
Ladue	All Exchanges of The SME	All Exchanges of the EMC
Overland	All Exchanges of The SME	All Exchanges of the EMC
St Louis	All Exchanges of The SME	All Exchanges of the EMC
Fenton	All Exchanges of The SME	All Exchanges of the EMC
Manchester	All Exchanges of The SME	All Exchanges of the EMC
St Charles	All Exchanges of The SME	All Exchanges of the EMC
Mehlville	All Exchanges of The SME	All Exchanges of the EMC
Harvester All		1 Exchanges of the EMC
Valley Park	All Exchanges of The SME	All Exchanges of the EMC
Chesterfield	All Exchanges of The SME	All Exchanges of the EMC
Kirkwood	All Exchanges of The SME	
Sappington	All Exchanges of The SME	All Exchanges of the EMC

1/Expanded Metropolitan Callin Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA toll charges.

MISSOURI P.S.C. TARIFF NO. 1 1st REVISED PAGE NO. 37 CANCELS ORIGINAL PAGE NO. 37

LOCAL EXCHANGE SERVICE

3. Service Descriptions

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.1 <u>Service Area (Cont'd)</u>
 - 3.1.1.1 Local Calling Areas (Cont'd)

<u>St Louis Metropolitan Exchange (SME) Local Calling Areas</u> The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Bridgeton, Creve Coeur, Ferguson, Florissant, Kirkwood, Ladue, Mehlville, Oakville, Overland, Riverview, Sappington, St. Louis Principal, Spanish Lake, Webster Groves

St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro Calling (EMC)) The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton,

Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Maxville, D Manchester, Moscow Hills, New Melle, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Peters, Troy, Valley Park, Ware, Wentzville, Winfield

E <u>xchange</u>	Local <u>Calling Area</u>	Expanded <u>Calling Area</u>
Bridgeton	All Exchanges of The SME	All Exchanges of the EMC
Creve Coeur	All Exchanges of The SME	All Exchanges of the EMC
Ladue	All Exchanges of The SME	All Exchanges of the EMC
Overland	All Exchanges of The SME	All Exchanges of the EMC
St Louis	All Exchanges of The SME	All Exchanges of the EMC
Fenton	All Exchanges of The SME	All Exchanges of the EMC
Manchester	All Exchanges of The SME	All Exchanges of the EMC
St Charles	All Exchanges of The SME	All Exchanges of the EMC

1/ Expanded Metropolitan Callin Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA toll charges.

D

MCImetro ACCESS TRANSMISSION SERVICES, LLC

- 3. Service Descriptions
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.1 Service Area (Cont'd)
 - 3.1.1.1 Local Calling Areas (Cont'd)

St Louis Metropolitan Exchange (SME) Local Calling Areas

LOCAL EXCHANGE SERVICE

The Local callings area of the S. Louis Metropolitan Exchange Area consists of the following exchanges: Bridgeton, Creve Coeur, Ferguson, Florissant, Kirkwood, Ladue, Mehiville, Oakville, Overland, Riverview, Sappington, St. Louis Principal, Spanish Lake, Webster Groves

PEASINEN

SEP 17 1998

MU. FUELIN SERVINE UNIM

MISSOURI P.S.C. TARIFF NO. 1

ORIGINAL PAGE NO. 37

<u>St Louis Expanded Metropolitan Exchange 1/ (Expanded Metro Calling (EMC))</u> The Expanded Service Area of the St Louis Metropolitan Exchange Area consists of the following exchanges: Antonia, Augusta, Cedar Hill, Chesterfield, Dardenne, Defiance, Desoto, Eureka, Fenton, Festus-Crystal City, Foristell, Gray Summit, Harvester, Herculaneum-Pevely, High Ridge, Hillsboro, Imperial, Maxville, Manchester, Moscow Hills, New Meile, O'Fallon, Old Monroe, Orchard Farm, Pacific Pond, Portage Des Sioux, St. Charles, St. Peters, Troy, Valley Park, Ware, Wentzville, Winfield

	Local	Expanded
E <u>xchange</u>	Calling Area	Calling Area
Bridgeton	All Exchanges of	All Exchanges of
	The SME	the EMC
Creve Coeur	All Exchanges of	All Exchanges of
	The SME	the EMC
Ladue	All Exchanges of	All Exchanges of
	The SME	the EMC
Overland	All Exchanges of	All Exchanges of
	The SME	the EMC
St Louis	All Exchanges of	All Exchanges of
	The SME	the EMC
Fenton	All Exchanges of	All Exchanges of
	The SME	the EMC
Imperial	All Exchanges of	All Exchanges of
	The SME	the EMC
Manchester	All Exchanges of	All Exchanges of
	The SME	the EMC
St Charles	All Exchanges of	All Exchanges of
	The SME	the EMC

1/

Public Service

Expanded Metropolitan Callin Area is defined as those exchanges where customers in the St. Louis Metropolitan Area can call without incurring intraLATA toll charges.

96-355

ISSUED : March 2, 1999

MISSOUR

ommission

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missouri Public Sorvico Commission

FILED MAY 1 & 1999

CANCELLED

SEP 1 8 200 -siDA

L

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 1st Revised Page No. 37.1 Cancels Original Page No. 37.1

LOCAL EXCHANGE SERVICE

Terminating Rate Center

3. Service Description (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center KS BASEHOR

MO BELTON

1

• -

KS BONNER SPG

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO CAMDEN PT, MO CLEVELAND, MO DEARBORN, MO DREXEL, MO EAST LYNNE MO EDGERTON, MO EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEE'S SUMMIT, MO LIBERTY, MO LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

Issued: November 7, 2006

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: December 8, 2006



Т

Τ

Т

Τ

MCImetro Access Transmission Services LLC d/b/a Verizon Access transmission Services

Missourri P.S.C. Tariff No. 1 Original Page No. 37.1

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1 Local Exchange Service(Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Terminating Rate Center

<u>Originating Rate Center</u> KS BASEHOR

MO BELTON

٠

KS BONNER SPG

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO CAMDEN PT, MO CLEVELAND, MO DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEE'S SUMMIT, MO LIBERTY, MO LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MOSTRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MOPLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO NER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MOPLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

ALL MATERIAL ON THIS PAGE IS NEW

Issued: May 10, 2006

Cancelled December 8, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: June 11, 2006



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 lst Revised Page No. 37.1.1 Cancels Original Page No. 37.1.1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.1 <u>Service Area (Cont'd)</u>
 - 3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate CenterTermination Rate CenterKSLeavenworth - Belton MO, Blue Springs MO,
Bonner Springs KS, De Soto KS, Greenwood
MO, Kansas City KS, Kansas City MO,
Leavehlsng, Leavenworth KS, Lee's Summit
MO, Liberty MO, Olathe, KS, Paola KS,
Tonganoxie KS

Joplin

Carl Junction, Joplin, Webb City

ALL MATERIAL ON THIS PAGE IS NEW

Issued: September 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: October 1, 2007

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

2

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 37.2 Cancels Original Page No. 37.2

T

T

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1 Local Exchange Service(Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Terminating Rate Center

Originating Rate Center KS KANSAS CITY

MO

KANSAS CITYINTERDECEMENTATIONKANSAS CITYARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS,
MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS
CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN,
MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS
EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN,
MO GARDENCITY, MO GARDNER, KS GRAIN VLY,
MO GREENWOOD, MO HARRISONVL, MO HENRIETTA,
MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY,
MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON,
MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD,
KS LKLOTAWANA, MO LONEJACK, MO MISSOURI CY, MO OAK GROVE,
MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA,
KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL,
MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG,
MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST
CLEVELAND, KS WEST DREXEL, KS WESTON, MOKANSAS CITYARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS,
O BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT,
MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO
EAST LYNNE, MO EDGERTON, MO FREEMAN, MO GARDENCITY,

O BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

Issued: November 7, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: December 8, 2006



MCImetro Access Transmission Services LLC d/b/a Verizon Access transmission Services

Missouri P.S.C. Tariff No. 1 Original Page No. 37.2

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1 Local Exchange Service(Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center KS KANSAS CITY Terminating Rate Center ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MOGREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

MO KANSAS CITY ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, O BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MOPLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

ALL MATERIAL ON THIS PAGE IS NEW

Issued: May 10, 2006



Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: June 11, 2006



Missouri P.S.C. Tariff No. 1 2nd Revised Page No. 37.3 Cancels 1st Revised Page No. 37.3

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Т

'n

ጥ

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center MO LEES SUMMIT Terminating Rate Center ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO FLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO

MO LIBERTY ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, T MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO T LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKOTAWANA, MO LONEJACK, T MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, KS WESTON, MO T ABCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS.

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO GREENWOOD, MO HARRISONVL, T MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, T MO LAWSON, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, T MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS WEST DREXEL, T KS WESTON, MO

Issued: November 7, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: December 8, 2006



MCImetro Access Transmission Services LLC d/b/a Verizon Access transmission Services Missouri P.S.C. Tariff No. 1 Original Page No. 37.3

LOCAL EXCHANGE SERVICE

......

3. Service Description (Cont)

Local Exchange Service (Cont'd) 3.1

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center MO LEES SUMMIT

MO

Terminating Rate Center ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO REENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS ANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS KLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS REXEL, KS WESTON, MO

ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO LIBERTY BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MO REENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KS KNSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWORTH LANSING, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS KLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO PLATTSBURG, MO PLEASANTHL, MO DICUMUND MO CONTUNITIES OF DECUSION RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS EST DREXEL, KS WESTON, MO

KS OLATHE ARCHIE, MO BASEHOR, KS BELTON, MO BLUE SPRINGS, MO ARCHIE, MO DASEHOR, AS BELION, MO BLOE SPRINGS, MO BONNER SPRINGS, KS BUCKNER, MO BUCYRUS, KS CAMDEN PT, MO CLEVELAND, MO DE SOTO, KS DEARBORN, MO DREXEL, MO EAST LYNNE, MO EDGERTON, MO EDGERTON, KS EXCELSRSPG, MO FARLEY, MO FERRELVIEW, MO FREEMAN, MO GARDENCITY, MO GARDNER, KS GRAIN VLY, MOREENWOOD, MO HARRISONVL, MO HENRIETTA, MO HOLDEN, MO HOLT, MO KANSAS CITY, KSKANSAS CITY, MO KEARNEY, MO KINGSVILLE, MO LATHROP, MO LAWSON, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO LINWOOD, KS LKLOTAWANA, MO LONEJACK, MO MISSOURICY, MO OAK GROVE, MO ODESSA, MO OLATHE, KS ORRICK, MO OSAWATOMIE, KS PAOLA, KS PECULIAR, MO PLATTECITY, MO LATTSBURG, MO PLEASANTHL, MO RICHMOND, MO SMITHVILLE, MO SPRINGHILL, KS STRASBURG, MO TONGANOXIE, KS TRIMBLE, MO WELLINGTON, MO WEST CLEVELAND, KS EST DREXEL, KS WESTON, MO

ALL MATERIAL ON THIS PAGE IS NEW

Issued: May 10, 2006

Cancelled December 8, 2006 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601

Effective: June 11, 2006



.

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

- 3.1 Local Exchange Service(Cont'd)
 - 3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center KS PAOLA	Terminating Rate Center BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS KANSAS CITY, KS KANSAS CITY, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO OLATHE, KS OSAWATOMIE, KS PAOLA, KS TONGANOXIE, KS
Springfield	Ash, Grove, Billings, Clever, Elkland, FairGrove, Fordland, Halltown, Highlandville, Hurley, Marionville, Marshfield, Morrisville, Nixa, Ozark, Pleasant Hope, Republic, Rogersville, Sparta, Springfield, Strafford, Walnut Grove, Willard.
KS Witchita	Anadala KS, Benton KS, Cheney KS, Clearwater KS, Douglas KS, El Dorado KS, Graden Plain KS, Goddard KS, Halstead Ks, Leon KS, Mount Hope KS, Newton KS, Towanda KS, White Water KS, Whitchata KS
KS Topeka	Lecompton KS, Rossville KS, Silverlake KS, Topeka KS
Nixa	Ash Grove, Billings, Clever, Elkland, N Fair Grove, Fordland, Halltown, Highlandville, Hurley, Marionville, Marshfield, Morrisville, Nixa, Ozark, Pleasant Hope, Republic, Rogersville, Sparta, Springfield, Stafford, Walnut Grove, Willard
Blue Springs, MO	Archie, Basehor KS, Belton, Blue Springs, Bonner Springs KS, Buckner, Buckyrus KS, Camden Pt, Cleveland, De Soto KS, Dearborn, Drexel, East Lynne, Edgerton KS, Edgerton, Excelsrspg, Farley, Ferrelyview, Freeman, Gardencity, Gardner KS, Grain Valley, Greenwood, Harrisonville, Henrietta, Holden, Holt, Kansas City KS, Kansas City, Kearney, Kingsville, Lathrop, Lawson, Leavehlsng, Leavenworth KS, Lee's Summit, Liberty, Linwood KS, Lklotawana, Lonejack, Missouri City, Oak Grove, Odessa, Olathe KS, Orrick, Osawatomie KS, Paola KS, Peculiar, Plattecity, Plattsburg, Pleasant Hill, Richmond, Smithville, Springhill KS, Strasburg, Tonganoxie KS, Trimble, Wellington, West Cleveland KS, West Drexel KS,

Issued: September 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: October 1, 2007

Missouri P.S.C Tariff No.1 2nd Revised Page No. 37.4 Cancels 1st Revised Page No. 37.4

Local Exchange Service

3. Service Descriptions and Rates (Cont'd)

I

I

3.1 Local Exchange Service (Cont'd)

3.1.1.1 Local calling Areas (Cont'd)

Origination Rate Center KS PAOLA	Termination Rate Center Belton, MO Blue Springs, MO Bonner Springs, KS Kansa City, MO Leavenwth, KS Lee's Summit, MO Liberty, MO Olathe, KS Oswatomie, KS Paola, KS Tonganoxie, KS
Springfield	Ash Grove, Billings, Clever, Elkland, Fair Grove, Fordland, Halltown, Highlandville, Hurley, Marionville, Marshfield, Morrisville,Nixa,Ozark, Pleasant Hope, Republic, Rogersville,Sparta, Springfield, Strafford, Walnut Grove, Willard
KS Wichita	Anadale KS, Benton KS, N Cheney KS, Clearwater KS, Douglas KS, El Dorado KS, Garden Plain KS, Goddard KS, Halstead KS, Leon KS, Mount Hope KS, Newton KS, Towanda KS, White Water KS, Wichita KS
KS Topeka	Lecompton KS, Rossville KS, Silverlake KS, Topeka KS N

Issued: April 20, 2007 CANCELLED Oct. 1, 2007 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago IL, 60601 Effective: May 21, 2007



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 1st Revised Page No. 37.4 Cancels Original Page No. 37.4

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3

- 3.1 Local Exchange Service(Cont'd)
 - 3.1.1 <u>Service Area (Cont'd)</u>

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate CenterTerminating Rate CenterKS PAOLABELTON, MO BLUE SPRINGS, MO BONNER SPRINGS,
KS KANSAS CITY, KS KANSAS CITY, MO
LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO
OLATHE, KS OSAWATOMIE, KS PAOLA, KS
TONGANOXIE, KS

SpringfieldAsh Grove, Billings, Clever, Elkland,NFair Grove, Fordland,|Halltown, Highlandville, Hurley,|Marionville, Marshfield, Morrisville,|Nixa, Ozark, Pleasant Hope,|Republic, Rogersville, Sparta,|Springfield, Strafford, Walnut|Grove, Willard.N

Issued: August 1, 2006



Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: September 1, 2006



MCImetro Access Transmission Services LLC d/b/a Verizon Access transmission Services

Missouri P.S.C. Tariff No. 1 Original Page No. 37.4

----

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1 Local Exchange Service(Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center

KS PAOLA

•

<u>Terminating Rate Center</u>

BELTON, MO BLUE SPRINGS, MO BONNER SPRINGS, KS KANSAS CITY, KS KANSAS CITY, MO LEAVENWTH, KS LEE'S SUMMIT, MO LIBERTY, MO OLATHE, KS OSAWATOMIE, KS PAOLA, KS TONGANOXIE, KS

.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: May 10, 2006

Cancelled September 1, 2006

Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: June 11, 2006



LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 Original Page No. 37.4.1

LOCAL EXCHANGE SERVICE

Terminating Rate Center

3. Service Description (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center

Dardenne

O'Fallon

Ozark

St Peters

Antonia, Augusta, Beaufort, Cedar N Hill, Chesterfield, Creve Coeur, I Dardenne, De Soto, Defiance, Eureka, I Fenton, Festus Crystal City, Foley, I Foristell, Gary Summit, Harvester, I Herchulaneum-Pewely, High Ridge, I Hillsboro, Holstein, Imperial, Kirkwood, I Ladue, Manchester, Marthasville, Maxville, I Mehlville, Moscow Mills, New Melle, I Oakville, OFallon, Old Monroe, Orchard I Farm, Pacific, Pond, Portage Des Sioux, I Sappinton, St. Charles, St. Clair, St. I Louis, St. Peters, Troy, Union, Valley Park, Ware, Warrington, Washington, I Wentlzville, Winfield, Wright City N

Antonia, Augusta, Beaufort, Cedar Hill, N Chesterfield, Creve Coeur, Dardenne, De Soto, Defiance, Eureka, Fenton, Festus Crystal City, Foley, Foristell, Gary Summit, Harvester, Herchulaneum-Pewely, High Ridge, Hillsboro, Holstein, Imperial, Kirkwood, Ladue, Manchester, Marthasville, Maxville, Mehlville, Moscow Mills, New Melle, Oakville, O'Fallon, Old Monroe, Crchard Farm, Pacific, Pond, Portage Des Sioux, Sappinton, St. Charles, St. Clair, St. Louis, St. Peters, Troy, Union, Valley Park, Ware, Warrington, Washington, Wentlzville, Winfield, Wright City

Ash Grove, Billings, Clever, Elkland, Fair N Grove, Fordland, Halltown, Highlandville, | Hurley, Marionville, Marshfield, | Morrisville, Nixa, Ozark, Pleasant Hope, | Republic, Rogersville, Sparta, | Springfield, Strafford, Walnut Grove, | Willard N

Antonia, Augusta, Beaufort, Cedar Hill, N Chesterfield, Creve Coeur, Dardenne, De Soto, Defiance, Eureka, Fenton, Festus Crystal City, Foley, Foristell, Gary Summit, Harvester, Herchulaneum-Pewely, High Ridge, Hillsboro, Holstein, Imperial, Kirkwood, Ladue, Manchester, Marthasville, Maxville, Mehlville, Moscow Mills, New Melle, Oakville, OFallon, Old Monroe, Orchard Farm, Pacific, Pond, Portage Des Sioux, Sappinton, St. Charles, St. Clair, St. Louis, St. Peters, Troy, Union, Valley Park, Ware, Warrington, Washington, Wentlzville, Winfield, Wright City

Issued: January 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission Effective: March 1, 2008
LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 Original Page No. 37.4.1.1

LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.1 Service Area (Cont'd)

3.1.1.1 Local Calling Areas (Cont'd)

Originating Rate Center Wentzville Terminating Rate Center

Antonia, Augusta, Beaufort, Cedar Hill, Chesterfield, Creve Coeur, Dardenne, De Soto, Defiance, Eureka, Fenton, Festus Crystal City. Foley, Foristell. Gray Summit, Harvester, Heculaneum Pevely, High Ridge, Hillsboro, Holstein, Imperial, Kirkwood, Ladue, Manchester, Mathasville, Maxville, Mehlville, Moscow Mills, New Melle, Oakville, OFallon, Old Monroe, Orchard Farm, Pacific, Pond, Portage Des Sioux, Sappington, St. Charles, St, Clair, St, Louis, St. Peters, Troy, Union, Valley Park, Ware, Warrneton, Washington, Wentzville and Winfield.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: January 31, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 1, 2008

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

2

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 38 Cancels 1st Revised Page No. 38 Missouri Public

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

REC'D MAR 11 2002

3.1 Local Exchange Service (Cont'd)

- Service Commission
- 3.1.2 Local Line: Local Line provides the Customer with a single, voice-grade communications channel. Each Local Line will include a telephone number.
 - 3.1.2.1 <u>Standard Features</u>: Each Local Line Customer is provided with the following standard features:

Automatic Dial1 Call Forward Busy, No Answer and Variable Call Hold Call Park Call Pickup (Group Call Pickup or Directed Call Pickup) Call Transfer, Consultation and Conference Call Waiting Call Waiting-Cancel Class of Service Conference Six-Way Customer Group Dialing Plan Digit Display¹ Do Not Disturb Group Intercom¹ Intercom¹ Key Short Hunt¹ Last Number Redial¹ Message Waiting Multiple Appearance Directory Numbers¹ Privacy Release¹ Query Time & Date¹ Serial Hunting Speed Call Optional Features: A Local Line Customer may order 3.1.2.2 the following optional features, at the rates specified in

> Caller ID with Name and Number Calling Number Delivery Calling Number Delivery Blocking Interim Local Number Portability (ILNP) Remote Call Forwarding (RCF)

3.1.2.3 <u>Local Line Rates and Charges:</u> A Local Line Customer will be charged applicable Non-Recurring Charges, monthly Recurring Charges and usage charges as specified in Sections 3.1.2.3.1, 3.1.2.3.2, 3.1.2.3.3 and 3.1.2.3.4, respectively.

Missouri Public

Ν

These features are only available in conjunction with Proprietary Business Sets. ELED APR 11 2002

Section 3.1.2.3.3:

Service Commission

Effective: April 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission

Issued: March 11, 2002

. .*****

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 38 CANCELS ORIGINAL PAGE NO. 38 MISSOURI PUBLIC Servise Semmission

RECD MAR 02 2001

CANCELLED

Public

APR 1 1 2002 2 ARS 38 Service Commission MISSOURI

Ν

FILED APR 01 2001

LOCAL EXCHANGE SERVICE

- Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.2 Local Line: Local Line provides the Customer with a single, voice-grade communications channel. Each Local Line will include a telephone number.
 - 3.1.2.1 <u>Standard Features</u>: Each Local Line Customer is provided with the following standard features:
 - Automatic Dial¹
 - Call Forward Busy, No Answer and Variable Call Hold
 - Call Park

Call Pickup (Group Call Pickup or Directed Call Pickup)

Call Transfer, Consultation and Conference

- Call Waiting
- Call Waiting-Cancel
- Class of Service Conference Six-Way

Customer Group Dialing Plan Digit Display¹ Do Not Disturb Group Intercom¹

Intercom¹

Key Short Hunt¹ Last Number Redial¹ Message Walting Multiple Appearance Directory Numbers¹ Privacy Release¹

Privacy Release¹ Query Time & Date¹ Serial Hunting

Speed Call

3.1.2.2 <u>Optional Features</u>: A Local Line Customer may order the following optional features, at the rates specified in Section 3.1.2.3.3:

Caller ID with Name and Number Calling Number Delivery Calling Number Delivery Blocking Interim Local Number Portability (ILNP)

3.1.2.3 Local Line Rates and Charges: A Local Line Customer will be charged applicable Non-Recurring Charges, monthly Recurring Charges and usage charges as specified in Sections 3.124153432471 Public 3.1.2.3.3 and 3.1.2.3.4, respectively.

These features are only available in conjunction with Proprietary Business Sets.

ISSUED : March 2, 2001

EFFECTIVE: April 1, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

RECENT MISSOURI P.S.C. TARIFF NO. 1

ORIGINAL PAGE NO. 38

SEP 17 1998

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd) 3

MO. FUBLIC SERVICE CUMM

- 3.1 Local Exchange Service (Cont'd)
 - Local Line: Local Line provides the Customer with a single, voice-grade communications channel. Each 3.1.2 Local Line will include a telephone number.
 - Standard Features: Each Local Line Customer is provided with the following standard 3.1.2.1

features:

Automatic Dial¹ Call Forward Busy, No Answer and Variable Call Hold Call Park Call Pickup (Group Call Pickup or Directed Call Pickup) Call Transfer, Consultation and Conference CANCELLED **Call Waiting Call Waiting-Cancel** Class of Service APR 0 1 2001 Conference Six-Way 1STRP38 **Customer Group Dialing Plan** Bγ Public Service Commission Digit Display¹ Do Not Disturb MISSOURI Group Intercom¹ Intercom¹ CANCELLED Key Short Hunt¹ Last Number Redial¹ Message Waiting 1999 Multiple Appearance Directory Numbers¹ XA- 98-575 Privacy Release¹ B۷ Public Service Commission Query Time & Date1 Serial Hunting MISSOURI Speed Call

Optional Features: A Local Line Customer may order the following optional features, at the 3.1.2.2 rates specified in Section 3.1.2.3.3:

> Calling Number Delivery **Calling Number Delivery Blocking** Interim Local Number Portability (ILNP)

Local Line Rates and Charges: A Local Line Customer will be charged applicable Non-3.1.2.3 Recurring Charges, monthly Recurring Charges and usage charges as specified in Sections 3.1.2.3.1, 3.1.2.3.2, 3.1.2.3.3 and 3.1.2.3.4, respectively.

These features are only available in conjunction with Proprietary Business Sets.

<u>96-355</u>

ISSUED : March 2, 1999

Ł

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta GA 30328

Missouri Public Service Commission FILED MAY 1 & 1999

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 39 Cancels 1st Revised Page NO. 39

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.2 <u>Local Line</u>

3.1.2.3 Local Line Rates and Charges (Cont'd):

3.1.2.3.1 <u>Non-Recurring Charges</u> Line Connection Charge (per line)	<u>St Louis/Kansas City</u> \$15.00	<u>Springfield</u> \$15.00	
Account Setup (per account)	\$00.00	\$0.00	
Account Changes Moves, Changes, Additio (per change)	\$10.50 ns	\$10.50	
Account Changes (per billing record cha	\$7.75 nge)	\$7.75	
Line Restoral Charge	\$20.00	\$20.00 I	

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

Suspension of Service \$20.50 \$20.50 Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer-initiated suspension.)

- Order Expedite Charge \$25.00 \$25.00 N (per line or per trunk)
- Due Date change \$10.00 \$10.00 N (per occurrence)

Issued: March 3, 2003

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 2, 2003



MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 39 Cancels Original Page NO. 39

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.2 Local Line

3.1.2.3 Local Line Rates and Charges (Cont'd):

3.1.2.3.1 <u>Non-Recurring Charges</u> Line Connection Charge (per line)	<u>St_Louis/Kansas_City</u> \$15.00	<u>Springfield</u> \$15.00	
Account Setup (per account)	\$00.00	\$0.00	
Account Changes Moves, Changes, Additio (per change)	\$10.50 ns	\$10.50	
Account Changes (per billing record cha	\$7.75 nge)	\$7.75	
Line Restoral Charge	\$20.00	\$20.00	Ε

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

Suspension of Service \$20.50 \$20.50 Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer-initiated suspension.)



Missouri Public Service Commission

FILED MAR 06 2003

Effective

Issued: February 24, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

MAR 0 6 2003

Missouri Public Service Commission

RECT) FEB 24 2003

3. Service Descriptions (Cont'd)

SEP 17 1998

MO. FUDLIU DERVICE UUMM

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.2 Local Line

3.1.2.3 Local Line Rates and Charges (Cont'd);

LOCAL EXCHANGE SERVICE

3.1.2.3.1	<u>Non-Recurring Charges</u> Line Connection Charge (per line)	<u>St Louis/Kansas City</u> \$52.25	Springfield \$52.25
	Account Setup (per account)	\$00.00	\$0.00
	Account Changes Moves, Changes, Additions (per change)	\$10.50	\$10.50
	Account Changes (per billing record change)	\$ 7.75	\$7 .75
	Line Restoral Charge	\$15.75	\$15.75

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

\$20.50 \$20,50 Suspension of Service Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer-initiated suspension.)

×2 0 1999 98-575 By Public Service Commission MISSOURI 96-355

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILED MAY 1 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 2, 1999



L



MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 39** RECEIVED

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

З.

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 7th Revised Page No. 40 Cancels 6th Revised Page No. 40

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

- 3.1.2 Local Line (Cont'd)
 - 3.1.2.3 Local Line Rates and Charges (Cont'd)

3.1.2.3.2 Monthly Recurring Charges

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Local Line - Line Charge (per line)

Pl<u>an 1:</u>

Per Call \$18.37 \$21.73 I Per Minute ¹ \$18.37 \$21.73 I		Kansas City		
	Per Call	<u>St. Louis</u> \$18,37	Springfield \$21.73	I
	Per Minute ¹ Unlimited	\$18.37 \$36.91	\$21.73 \$30.51	 T

<u>Plan 2</u>

Per Call Per Minute	Kansas City St. Louis \$18.37 \$18.37	Springfield \$18.37 \$18.37	Ī
Unlimited	\$44.00	\$44.00	I

3.1.2.3.3	Optional Features:		
	Caller ID with Name		
	and Number	\$5.00	\$5.00
	Calling Number	40.00	40.00
	Delivery	\$8.50	
		90.JU	
	Calling Number		
	Delivery		***
	Blocking Selective Complete ²	\$0.00	\$0.00
	Complete ²	\$0.00	\$0.00
	Interim Local Number		
	Portability	\$0.00	\$0.00
	Remote Calĺ³		,
	Forwarding RCF)	\$20.00	\$20.00
	(per each line path)	~20.00	+20.00
	(per each rrie pach)		

3.1.2.3.4 Usage <u>Rates</u>: The rates in Section 3.1.4 will apply.

¹Effective April 1, 2002 this calling option will not be available to new subscribers. ² available to Social Service and Law Enforcement Agencies (See Section 1, Calling Numbers Delivery).

³ Customer who subscribe to MCI Local Line Service Via UNEP are not eligible to Utilize Remote Call Forwarding (RCF).

Issued: June 30, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 6th Revised Page No. 40 CANCELS 5th Revised Page No. 40

Ν

1

Ţ

Ν

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.2 Local Line (Cont'd)
 - 3.1.2.3 Local Line Rates and Charges (Cont'd)

3.1.2.3.2 Monthly Recurring Charges

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Local Line - Line Charge (per line)

	<u>Plan 1:</u>	Kansas City		
	Per Call Per Minute ¹ Unlimited	St. Louis \$16.70 \$16.70 \$33.55	Springfield \$13.75 \$13.75 \$27.74	
	<u>Plan 2</u>	Kansas City St. Louis	Springfield	
	Per Call	\$16.70	\$16.70	D
	Per Minute	\$16.70	\$16.70	ס
	Unlimited	\$40.00	\$40.00	D
3.1.2.3.3	Optional Features: Caller ID with Name and Number	\$5.00	\$5.00	
	Calling Number Delivery Calling Number	\$8.50	9 3.00	
	Delivery Blocking Selective Complete ²	\$0.00 \$0.00	\$0.00 \$0.00	
	Interim Local Number Portability Remote Call	\$0.00	\$0.00	
	Forwarding ¹ (RCF) (per each line path)	\$20.00	\$20.00	

3.1.2.3.4 <u>Usage Rates</u>: The rates in Section 3.1.4 will apply.

¹Effective April 1, 2002 this calling option will not be available to new subscribers.

³Available only to Social Service and Law Enforcement Agencies(See Section 1, Calling Number Delivery)

l/Customers who subscribe to MCI Local Line Service via UNEP are not eligible to utilize Remote Call Forwrding (RCF).

January 20, 2004

Effective: January 16, 2004

N

Ν

Issued: December 16, 2003 CANCELLED August 1, 2008 Missouri Public Service Commission

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Filed

İ

MISSOURI P.S.C. TARIFF NO. 1 5th Revised Page No. 40 CANCELS 4th Revised Page No. 40

Missouri Public

REC'D OCT 03 2003

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.2 Local Line (Cont'd)

3.1.2.3 Local Line Rates and Charges (Cont'd) Service Commission

3.1.2.3.2 Monthly Recurring Charges Local Line - Line Charge (per line)

		-90 (por rine,		
	<u>Plan 1:</u> Per Call Per Minute ¹ Unlimited		Kansas City St. Louis \$16.70 \$16.70 \$33.55	Springfield \$13.75 \$13.75	Т Т
	<u>Plan 2</u>		Kansas City		N
	Per Call a) for 1-3 lines: b) for 4 or more lin (rate applies to lines)	nes:	St. Louis \$16.70	Springfield \$16.70	
	Per Minute a) for 1-3 lines: b) for 4 or more lin (rate applies to lines) Unlimited		\$16.70	\$16.70	
	a) for 1-3 lines: b) for 4 or more lin (rate applies to lines)	nes:	\$40.00 \$40.00	\$40.00 \$40.00	N
3.1.2.3.3	Optional Features:				
LLED	Caller ID with Name and Number Calling Number Delivery Calling Number	\$5.0 \$8.5	-	\$5.00	
2004	Delivery Blocking Selective Complete ²	\$0.0 \$0.0		\$0.00 \$0.00	
40 Commission	Interim Local Number Portability Remote Call	\$0.0		\$0.00	
URI	Forwarding (RCF) (per each line path)	\$20.	00	\$20.00	
3.1.2.3.4	Usage Rates: The rat	es in	Section 3.	1.4 will apply	,

CANCEL

JAN 2 0 Public S

3.1.2.3.4 <u>Usage Rates</u>: The rates in Section 3.1.4 will apply Miceouri Public Service Commission

NOV 08 2003

'Effective April 1, 2002 this calling option will not be available to new subscribers

Available only to Social Service and Law Enforcement Agencies(See Section 1, Calling Number Delivery)

Issued: October 3, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: November 3, 2003

MISSOURI P.S.C. TARIFF NO. 1 4th Revised Page No. 40 CANCELS 3rd Revised Page No. 40

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.2 Local Line (Cont'd)

3.1.2.3 Local Line Rates and Charges (Cont'd)

3.1.2.3.2		nthly Louis/Kansas City	<u>Springfield</u>
	Per Call/Per Minute Option ¹ Unlimited Option	\$16.70 \$33.55	\$13.75 ² \$27.74 (I)
3.1.2.3.3	Optional Features:		
	Caller ID with Name and Number Calling Number	\$5.00	\$5.00
	Delivery Calling Number Delivery	\$8.50	
	Blocking Selective Complete ³	\$0.00 \$0.00	\$0.00 \$0.00
	Interim Local Number Portability	\$0.00	\$0.00
	Remote Call Forwarding (RCF) (per each line path)	\$20.00	\$20.00

3.1.2.3.4 <u>Usage Rates</u>: The rates in Section 3.1.4 will apply.

CANCELLED

0 8 2003 NOV -'ublic Se amission

¹Effective April 1, 2002 this calling option will not be available to new subscribers.

²The Per Call Option for Springfield includes 100 free calls per month per line.

³Available only to Social Service and Law Enforcement Agencies(See Section 1, Calling Number Delivery)

Miccouri Publie

Effective: September 1, 2002

issued: August 21, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Missouri Public

REC'D AUG 21 2002

Service Commission

Service Commission

FILED SEP 01 2002

i

P.* ...

MCImetro ACCESS TRANSMISSION

MISSOURI P.S.C. TARIFF NO. 1 3rd Revised Page No. 40 CANCELS 2nd Revised Page No. 40

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.2 Local Line (Cont'd)

Service Commission

3.1.2.3 Local Line Rates and Charges (Cont'd)

3.1.2.3.2		onthly Louis/Kansas_City	<u>Springfield</u>
	Per Call/Per Minute Option ¹ Unlimited Option	\$16.70 \$33.55	\$13.75² \$25.70
3.1.2.3.3	Optional Features:		
_	Caller ID with Name and Number Calling Number Delivery Calling Number Delivery	\$5.00 \$8.50	\$5.00
SEP 0 1 2002	Blocking Selective Complete ³	\$0.00 \$0.00	\$0.00 \$0.00
AUM PS 40 Public Series Continission MISSOURI	Interim Local Number Portability Remote Call Forwarding (RCF) (per each line path)	\$0.00 \$20.00	\$0.00 \$20.00 N
3.1.2.3.4	<u>Usage Rates</u> : The ra	tes in Section 3.1.4	will apply.

Missouri Public

FILED APR 1 1 2002

Service Commission

¹Effective April 1, 2002 this calling option will not be available to new subscribers.

²The Per Call Option for Springfield includes 100 free calls per month per line.

³Available only to Social Service and Law Enforcement Agencies(See Section 1, Calling Number Delivery)

Issued: March 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: April 11, 2002

REC'D MAR 11 2002

Missouri Public

SERVICES, LLC

3.

.

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 40 CANCELS 1ST REVISED PAGE NO. 40

Missouri Public Service Commission

RECT MAR 0 2 2001

3.1 Local Exchange Service (Cont'd)

Service Descriptions (Cont'd)

3.1.2 Local Line (Cont'd)

3.1.2.3 Local Line Rates and Charges (Cont'd)

LOCAL EXCHANGE SERVICE

3.1.2,3.2	Recurring Charges Local Line - Line Charge	<u>Monthly</u> <u>St Louis/Kansas City</u>	Springfield	
	(per line) Per Call/Per Minute Option 1/	\$16.70	\$13.75 <u>2</u> /	'N
	Unlimited Option	\$33.55	\$25.70	
		400.00	*-0.1 *	Ð
				D
3.1.2.3.3	Optional Features:			
	Caller ID with Name and Number	\$5.00	\$5.00	N
	Calling Number Delivery	\$8.50		
	Calling Number Delivery		• • •	
•	Blocking Selective	\$0.00	\$0.00	
	Complete 3/	\$0.00	\$0.00	T
	Interim Local Number Portability	\$0.00	\$0.00	

3.1.2.3.4 Usage Rates: The rates in Section 3.1.4 will apply.

CANCELLED

APR 1 1 2002 ce Commission

<u>1</u> /	Effective April 1, 2001, this calling option will not be available to new subscribers.	Missouri Public	N
<u>2</u> /	The Per Call Option for Springfield includes 100 free calls per month per line.	Missouri Public Service Commission	N
<u>3</u> /	Available only to Social Service and Law Enforcement Agencies (See Section 1, Calling Number Deliv	_{ery).} FILED APR 01 2001	ĩ

ISSUED : March 2, 2001

EFFECTIVE: April 1, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 40 CANCELS ORIGINAL PAGE NO. 40 MICCOMIT PURIO SCHWICH COMMING

RFCD NOV 01 1999

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.2 Local Line (Cont'd)

3.1.2.3 Local Line Rates and Charges (Cont'd)

3.1.2.3.2	Recurring Charges	<u>Monthly</u>			
	Local Line - Line Charge	St Louis/Kansas City	Springfield		
	(per line)				
	Per Call/Per Minute Option	\$16.70	\$13.75		
	Unlimited Option	\$33.55	\$25.70		
	High Capacity Inbound Service				
	Analog	\$125.00	\$125.00		
3.1.2.3.3	Optional Features:				
		a a <i>c</i> a			
	Calling Number Delivery	\$8.50			
	Calling Number Delivery				
	Blocking Selective	\$0.00	\$0.00		
	Complete 1/	\$0.00	\$0.00		
	Interim Local Number Portability	\$0.00	\$0.00		

3.1.2.3.4 Usage Rates: The rates in Section 3.1.4 will apply.

CANCELLED

APR 01 2001 L, 2ND R P 40 Public Service Commission MISSOURI



1/

Available only to Social Service and Law Enforcement Agencies (See Section 1, Calling Number Delivery)_ED JAN 1 0 2008

ISSUED : November 1, 1999

EFFECT

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

JAN 1 0 2000

Ν

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 40**

LOCAL EXCHANGE SERVICE

Local Line Rates and Charges (Cont'd)

3. Service Descriptions (Cont'd)

SEP 17 1998

RECEIVED

3.1 Local Exchange Service (Cont'd)

Local Line (Cont'd)

3.1.2.3

3.1.2

MO. PUBLIC SERVICE CONIM

3.1.2.3.2	Recurring Charges	Monthly	
	Local Line - Line Charge (per line)	St Louis/Kansas City	<u>Springfield</u>
	Per Call/Per Minute Option	\$16.70	\$13.75
	Unlimited Option	\$33.55	\$25.70
	High Capacity Inbound Service		
	Analog	\$125.00	\$125.00
3.1.2.3.3	Optional Features:		
	Calling Number Delivery Calling Number Delivery	\$8.50	
	Blocking Selective	\$0,00	\$0.00
	Complete	\$0.00	\$0.00
	Interim Local Number Portability	\$0.00	\$0.00

3.1.2.3.4 Usage Rates: The rates in Section 3.1.4 will apply.

CANCELLED

JAN 1 0 2000 By 195 RP 40 Public Service Commission MISSOURI

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 2, 1999

Ļ



Missouri Public Servico Commission

FILED WAY 1 & 1999

Missouri P.S.C. Tariff No. 1 2nd Revised Page No. 41 Cancels 1st Revised Page No. 41

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 41 Cancels Original Page No. 41

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

Missouri Public REC'D MAR 1 1 2002

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 <u>Local Trunk</u>: Local Trunk(s) provide Customer with Voice Grade Monification channel(s) to the Customer-provided wiring. Customer's Private Branch Exchange (PBX) or Hybrid Key System. Local Trunks can be provisioned as either analog or digital and will be provided in the following manner:
 - 3.1 <u>Local Trunk-Basic</u>: Local Trunk- Basic can be used to carry oneway outbound traffic, one-way inbound or two-way traffic.
 - 3.1.3.1.1 <u>One-Way Outbound</u>: Provides the Customer with a single analog or digital connection which is restricted to carry outbound traffic only.
 - 3.1.3.1.2 <u>One-Way Inbound or Two-Way</u>: Provides the Customer with a single analog or digital connection which can carry one-way inbound or two-way traffic.

3.1.3.1.2.1 <u>Features:</u> The following features are available:

Multiline Hunting Serial Hunting Distributed Line Hunting Interim Local Number portability (ILNP) Remote Call Forwarding(RCF)

3.1.3.1.2.2 Optional Features

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System. Digital Interface can be used to carry one-way outbound traffic, one-way Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.1.3.1 and 3.1.3.1.3.2.

Missouri Public

Ν

٣	11	PB.	ለሮ		4	4	2002
Þ.	И.	114	- LA 1-	' H '			
•	п.,		1.1.1		- L	- A.	LUVL

Service Commission

Issued: March 11, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 11, 2002

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO 41

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

MU. TUDLIV OLININUL UVININ

SFP 17 1998

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 <u>Local Trunk</u>: Local Trunk(s) provide Customer with voice-grade communication channel(s) to the Customer-provided wiring. Customer's Private Branch Exchange (PBX) or Hybrid Key System. Local Trunks can be provisioned as either analog or digital and will be provided in the following manner:
 - 3.1.3.1 <u>Local Trunk-Basic</u>: Local Trunk- Basic can be used to carry one-way outbound traffic, oneway inbound or two-way traffic.
 - 3.1.3.1.1 <u>One-Way Outbound</u>: Provides the Customer with a single analog or digital connection which is restricted to carry outbound traffic only.

CANCELLED

APR 1 1 2002

CService Commission MISSOURI

3.1.3.1.2 <u>One-Way Inbound or Two-Way</u>: Provides the Customer with a single analog or digital connection which can carry one-way inbound or two-way traffic.

3.1.3.1.2.1

Multiline Hunting Serial Hunting Distributed Line Hunting Interim Local Number portability (ILNP)

Features: The following features are available:

3,1.3.1.2.2

Optional Features

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System. Digital Interface can be used to carry oneway outbound traffic, one-way Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.1.3.1 and 3.1.3.1.3.2.

CANCELLED DH AUG 2 0 1999 By 7A -98-575 Public Service Commission MISSOURI

ISSUED : March 2, 1999

٤

<u>96-35</u>5 EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Sorvico Commission

FILED MAY 1 4 1009

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

CIMETRO ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 42 Cancels 1st Revised Page No. 42

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)

Change)

3.1.3.1.3 <u>Local Trunk-Basic Rates and Charges</u>: A Local Trunk -Basic Customer will be charged applicable Non-Recurring Charges, monthly Recurring Charges and usage charges as specified in Sections 3.1.3.1.3.1, 3.1.3.1.3.2 and 3.1.3.1.3.3, respectively.

3.1.3.1.3.1 Non-Recurring Charges St Louis/Kansas City

Line Connection Charges \$20.00/ \$2	20.00
(per Trunk)	

Account Setup \$0.00 / \$0.00 (per account)

Account Changes (Moves, Changes Additions)(per	s 5,	<u>Louis/Kansas (</u> \$10.50		<u>Springfield</u> 310.50
Account Changes (Per Billing Re		\$7.75	Q.,	\$7.75

Line Restoral Charge \$20.00 \$20.00 N

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

Suspension of Service \$20.50 \$20.50 Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer-initiated suspension.)

Order Expedite Charge (per line or per trunk)	\$25.00	\$25.00	N
Due Date change	\$10.00	\$10.00	N
(per occurrence) T-1 Order Expedite Charge (per T-1)	\$600.00	\$600.00	Ň
T-1 Installation Charge (per T-1)	\$200.00	\$200.00	N
<u>Optional Features</u> Digital Interface Channeliz \$0.00	zation Charg	e (Per Channe	el)

Issued: March 3, 2003

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 2, 2003



MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 42 Cancels Original Page No. 42

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)
 - 3.1.3.1.3 <u>Local Trunk-Basic Rates and Charges</u>: A Local Trunk -Basic Customer will be charged applicable Non-Recurring Charges, monthly Recurring Charges and usage charges as specified in Sections 3.1.3.1.3.1, 3.1.3.1.3.2 and 3.1.3.1.3.3, respectively.

3.1.3.1.3.1 <u>Non-Recurring Charges</u>

	<u>St Louis/Kansas City</u>
Line Connection Charges (per Trunk)	\$20.00/ \$20.00 R
Account Setup (per account)	\$0.00 / \$0.00
<u>St Loui</u> Account Changes (Moves, Changes, Additions)(per change)	<u>s/Kansas City</u> <u>Springfield</u> \$10.50 \$10.50
Account Changes	\$7.75 \$7.75

_	•	
(Per Billing Record		
Change)		

Line Restoral Charge \$15.75 \$15.75

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

Suspension of Service \$20.50 \$20.50 Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer- initiated suspension.)

<u>Optional Features</u>

Digital Interface Channelization Charge (Per Channel) \$0.00

issued: November 1, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: December 1, 2002

Missouri Public Service Commission

APR 0 2 2003 mssion

Missouri Public

REC'D NOV 01 2002

Service Commission

Missouri Fud

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 42 REGENZED

SEP 17 1998

אוט. ד טטבוט טבוואוטב טטואוא

Local Trunk-Basic Rates and Charges: A Local Trunk - Basic Customer will be 3.1.3.1.3 charged applicable Non-Recurring Charges, monthly Recurring Charges and usage charges as specified in Sections 3.1.3.1.3.1, 3.1.3.1.3.2 and 3.1.3.1.3.3, respectively.

3,1.3.1.3.1 Non-Recurring Charges

St Louis/Kansas City

\$52.25 / \$52.25

Line Connection Charges (per Trunk)

Account Setup (per account)

\$0.00 / \$0.00

Account Changes (Moves, Changes, Additions) (per change)	<u>St Louis/Kansas City</u> \$10.50	<u>Springfield</u> \$10.50
Account Changes (Per Billing Record Change)	\$ 7.75	\$7.75
Line Restoral Charge	\$15.75	\$15,75

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

Suspension of Service \$20.50 \$20.50 Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer-initiated suspension.)

Optional Features Digital Interface Channelization Charge (Per Channel) \$0.00 96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission

FILED INAY 1 4 1999

adig 2/0

ISSUED : March 2, 1999

Public Service Commussion

MISSOUR

1999

-575

CANCELLED DEC 01 2002

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

З.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

-

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)
 - 3.1.3.1.3 Local Trunk-Basic Rates and Charges (Cont'd):

3.1.3.1.3.2 Monthly Recurring Charges

<u>St Loui</u>	s/Kansas City	<u>Springfield</u>			
Local Trunk Basic Charge (Per Trunk) Per Call/Per Minute Option ¹	\$18.20	\$14.40			
Minute Option	ŞI0.20	914.4V			
Unlimited Option Analog Digital	\$43.60 \$43.60	\$39.35 \$39.35			
<u>Optional Features:</u> Interim Local Number Portability	\$0.00	\$0.00			
Remote Call Forwarding(RCF) (Per each line path)	\$20.00	\$20.00			
Digital Interface Channelization					
Charge (Per Channel)	\$0.00	\$0.00			
Overflow Routing	\$20.00	\$20.00 N			
Calling Number Delive	ery Blocking				
Selective Complete ²	\$0.00 \$0.00	\$0.00 \$0.00			
3.1.3.1.3.3 Usage Rates: The rates in section 3.1.4					

¹Effective April 1, 2002, this calling option will not be available to new customers.

²Available only to Social Service and Law Enforcement Agencies (See Section1, Calling Number Deliver).

will apply.

Issued: November 1, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: December 1, 2002



ъ. Г.

LOCAL EXCHANGE SERVICE

Missouri Public

	Sar	vice Des	criptions		CAL EXCHANGE SERVICE		
•				<u>Service (Cont</u>	<u>='d)</u>	Rec'd A	AUG 21 2002
		3.1.3	Local Tr	unk (Cont'd)		Service	Commission
			3.1.3.1	Local Trunk	-Basic (Cont'd)		Commission
				3.1.3.1.3	Local Trunk-Basic Rat	es and Charges (C	ont'd):
					3.1.3.1.3.2 Monthly H	Recurring Charges	
					<u>St Lou</u>	<u>is/Kansas City</u>	<u>Springfield</u>
					Local Trunk Basic Charge (Per Trunk) Per Call/Per Minute Option ¹	\$18.20	\$14.40
					-	\$10.20	\$14.40
					Unlimited Option Analog Digital	\$43.60 \$43.60 (I)	\$39.35 (I) \$39.35 (I)
		ſ	ANCELL	ED	<u>Optional Features:</u> Interim Local Number Portability	\$0.00	\$0.00
		ŭ	DEC 012	002 43	Remote Call Forwarding(RCF) (Per each line path)	\$20.00	\$20.00
		Public	SONIES	Antission Ri	Digital Interface Channelization Charge (Per Channel)	\$0.00	\$0.00
					Calling Number Delive	ery Blocking	
					Selective Complete ²	\$0.00 \$0.00	\$0.00 \$0.00

3.1.3.1.3.3 <u>Usage Rates</u>: The rates in section 3.1.4 will apply.

¹Effective April 1, 2002, this calling option will not be available to new customers.

²Available only to Social Service and Law Enforcement Agencies (See Section1, Calling Number Deliver).

Issued: August 21, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective Wissertember 1, 2002

FILED SEP 01 2002

Service Commissior

MISSOURI P.S.C. TARIFF NO. 1 4th Revised Page No. 43 Cancels 3rd Revised Page No. 43

Missouri Public

3. Service Descriptions (Cont'd)

LOCAL EXCHANGE SERVICE

3.1 Local Exchange Service (Cont'd)

3.1.3 Local Trunk (Cont'd)

3.1.3.1 Local Trunk-Basic (Cont'd)

3.1.3.1.3 Local Trunk-Basic Rates and Charges (Cont'd):

3.1.3.1.3.2 Monthly Recurring Charges

<u>St_Lou:</u>	is/Kansas City	<u>Springfield</u>
Local Trunk Basic Charge (Per Trunk) Per Call/Per		
Minute Option ¹	\$18.20	\$14.40
Unlimited Option Analog Digital	\$43.60 \$37.87	\$33.15 \$33.15
<u>Optional Features:</u> Interim Local Number Portability	\$0.00	\$0.00
Remote Call Forwarding(RCF) (Per each line path)	\$20.00	N \$20.00N
Digital Interface Channelization Charge (Per Channel)	\$0.00	\$0.00
Calling Number Delive	ery Blocking	
Selective Complete ²	\$0.00 \$0.00	\$0.00 \$0.00
3.1.3.1.3.3 Usage Rat	tes: The rates in	section.3.1.4

3.1.3.1.3.3 <u>Usage Rates</u>: The rates in section.3.1.4 Will apply.

FILED APR 1 1 2002

Service Commission

¹Effective April 1, 2002, this calling option will not be available to new customers.

²Available only to Social Service and Law Enforcement Agencies (See Section1, Calling Number Deliver).

Issued: March 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chícago, IL 60601 Effective: April 11, 2002



Service Commission

MISSOURI P.S.C. TARIFF NO. 1 3RD REVISED PAGE NO. 43 CANCELS 2ND REVISED PAGE NO. 43 MISSOURI Public Service Commission

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

CANCELLED

PUENC Servi

3.1

. 3,

Local Exchange Service (Cont'd)

REC'D MAR 02 2001

Ν

D

т

N

T

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)

hission

3.1.3.1.3 Local Trunk-Basic Rates and Charges (Cont'd):

3.1.3.1.3.2 Monthly Recurring Charges

	St Louis/Kansas City	Springfield	
Local Trunk - Basic Charge			
(Per Trunk)			
Per Call/Per Minute Option 1/	\$18.20	\$14.40	
Unlimited Option			
Analog	\$43.60	\$33.15	
Digital	\$37.87	\$33.15	
	· · ·		
·			

Optional Features: Interim Local Number Portability	\$0.00	\$0.00
Digital Interface Channelization Charge (Per Channel)	\$0.00	\$0.00
Calling Number Delivery Blocking Selective Complete <u>2</u> /	\$0.00 \$0.00	\$0.00 \$0.00

3.1.3.1.3.3 Usage Rates: The rates in section 3.1.4 will apply.



EFFECTIVE: April 1, 2001

1/ Effective April 1, 2001, this calling option will not be available to new subscribers.

2/ Available only to Social Service and Law Enforcement Agencies (See Section 1, Calling Number Deliver).

ISSUED : March 2, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Ĵ

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 43 CANCELS 1ST REVISED PAGE NO. 43

Missouri Public Service Commission

RECT JUN 0 7 2000

St Louis/Kanese City Springfield

LOCAL EXCHANGE SERVICE

З. Service Descriptions (Cont'd)

MISSOURI

×.

- 3.1 Local Exchange Service (Confd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)
 - 3.1.3.1.3 Local Trunk-Basic Rates and Charges (Cont'd):
 - 3.1.3.1.3.2 Monthly Recurring Charges

		<u>St Louis/Kansas City</u>	<u>Springfield</u>	
	Local Trunk - Basic Charge			
	(Per Trunk)			
	Per Call/Per Minute Option	\$18.20	\$14.40	
	Unlimited Option		••••••	
	Алаlog	\$43.60	\$33.15	т
	•	\$37.87	\$33.15	T/R
	Digital	\$31.0 <u>1</u>	\$ 53.15	ıπ
CANCELLED				
	High Capacity Inbound Service			
	Digital			
APR 0 1 2001	(per call/per minute/			
n NRP43	Unlimited Options)	\$100.00	\$100.00	
	Analog			
Public Service Commission	(per call/per minute			
Public Service Commission MISSOURI	Unlimited Options)	\$125.00	\$125.00	
	Optional Features:			
	Interim Local Number Portability	\$0.00	\$0.00	
	Digital Interface Channelization			
	Charge (Per Channel)	\$0.00	\$0.00	
	Calling Number Delivery Blocking			
	Selective	\$0.00	\$0.00	
	Complete 1/	\$0.00	\$0.00	
O' TOLULI				
	B Usage Rates: The rates in section	n 3.1.4 will apply.		
ADG 2/0 1999				
By 14 218-515		C_MISSOL	Jri Public	
Public Service Commission		SOLAIGO C	uri Publiq ommissio	



FILED JUL 0 7 2000

1/

Available only to Social Service and Law Enforcement Agencies (See Section 1, Calling Number Deliver).

ISSUED : June 7, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECTIVE: July 7, 2000

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 43 CANCEL SORIGINAL PAGE NO. 43

RFCD NOV 01 1839

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.1 Local Trunk-Basic (Cont'd)

3.1.3.1.3 Local Trunk-Basic Rates and Charges (Cont'd):

3.1.3.1.3.2 Monthly Recurring Charges

	St Louis/Kansas City	Springfield
Local Trunk - Basic Charge		
(Per Trunk)		
Per Call/Per Minute Option	\$18.20	\$14.40
Unlimited Option	\$43.60	\$33.15
High Capacity Inbound Service		
Digital		
(per call/per minute/		
Unlimited Options)	\$100.00	\$100.00
Analog		
(per call/per minute		
Unlimited Options)	\$125.00	\$125.00
Optional Features:		
Interim Local Number Portability	\$0.00	\$0.00
Digital Interface Channelization		
Charge (Per Channel)	\$0.00	\$0.00
Calling Number Delivery Blocking		
Selective	\$0.00	\$0.00
Complete 1/	\$0.00	\$0.00

EFFEC

3.1.3.1.3.3 Usage Rates: The rates in section 3.1.4 will apply.

D D D

Ν

1/

Available only to Social Service and Law Enforcement Agencies (See Section 1, Calling Number Deliver) LED JAN 10 2000

ISSUED : November 1, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

JAN 1 0 2000

CANCELLED

JUL 07 2000

By 2nd RS #43 Public Service Commission MISSOURI



MISSOURI P.S.C. TARIFF NO. 1 REGEIVED

SEP 17 1998

MU. I טשבוט טבונאוטב טעואואי

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.1 Local Trunk-Basic (Cont'd)

3.1.3.1.3 Local Trunk-Basic Rates and Charges (Cont'd):

3.1.3.1.3.2 Monthly Recurring Charges

2.2	wolinna kecnillind custdes		
		St Louis/Kansas City	Springfield
	Local Trunk - Basic Charge		
	(Per Trunk)		
	Per Call/Per Minute Option	\$18.20	\$14.40
	Unlimited Option	\$43.60	\$33.15
	High Capacity Inbound Service		
	Digital		
	(per call/per minute/		
	Unlimited Options)	\$100.00	\$100.00
	Analog		
	(per call/per minute		
	Unlimited Options)	\$125.00	\$125.00
	Optional Features:		
	Interim Local Number Portability	\$0.00	\$0.00
	(Rates for a volume of numbers		
	greater than 500 will be provided		
	on an Individual Case Basis).		
	Digital Interface Channelization		
	Charge (Per Channel)	\$0.00	\$0.00
	Calling Number Delivery Blocking		
	Selective	\$0.00	\$0.00
	Complete	\$0.00	\$0.00

CANCELLED

t

3.1.3.1.3.3 Usage Rates: The rates in section 3.1.4 will apply.

JAN 1 0 2000 By 1^{5t} RP43 Public Service Commission MISSOURI

ISSUED : March 2, 1999

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission FILED MAY 14 1999

1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 44 Cancels 1st Revised Page No. 44 Missouri Public

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1Local Exchange Service (Cont'd)

REC'D MAR 1 1 2002

3.1.3 <u>Local Trunk</u> (Cont'd)

Service Commission

- 3.1.3.2 <u>Local Trunk Direct Inward Dialing (DID)</u>: Provides the Customer with an single analog or digital connection which can carry oneway, inbound traffic¹.
 - 3.1.3.2.1 <u>Direct Inward Dialing Numbers</u>: Telephone numbers can be obtained in blocks of 20 or 100 numbers. Additional monthly charges will apply, as specified in Section 3.1.3.2.3.2.
 - 3.1.3.2.2 <u>Optional Features:</u> A Local Trunk DID customer may order the following optional features at the rates specified in Section 3.1.3.2.3.2.

Interim Local Number Portability (ILNP)

Remote Call Forwarding (RCF)

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System. Digital Interface can be used to carry one-way outbound traffic, one-way Inbound or two-way traffic, Direct Inward. Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

3.1.3.2.3 <u>Direct Inward Dialing Rates and Charges</u>: A Customer who orders a Local Trunk - DID trunk will be charged applicable Non-Recurring Charges and monthly Recurring Charges as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

Missouri Public

FILED APR 1 1 2002

Service Commission

¹Effective April, 2001, analog DID and analog 2 Way Direct service will not be available to new subscibers.

Issued: March 11, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 11, 2002



Ν
MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 44 CANCELS ORIGINAL PAGE NO. 44

> Missouri Public Service Commission

RECTO MAR 02 2001

LOCAL EXCHANGE SERVICE

- Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.2 <u>Local Trunk Direct Inward Dialing (DID)</u>: Provides the Customer with an single analog or digital connection which can carry one-way, inbound traffic. <u>1</u>/
 - 3.1.3.2.1 <u>Direct Inward Dialing Numbers</u>: Telephone numbers can be obtained in blocks of 20 or 100 numbers. Additional monthly charges will apply, as specified in Section 3.1.3.2.3.2.
 - 3.1.3.2.2 <u>Optional Features:</u> A Local Trunk DID customer may order the following optional features at the rates specified in Section 3.1.3.2.3.2.

Interim Local Number Portability (ILNP)

Digital Interface-Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System. Digital Interface can be used to carry one-way outbound traffic, one-way Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

3.1.3.2.3 <u>Direct Inward Dialing Rates and Charges</u>: A Customer who orders a Local Trunk - DID trunk will be charged applicable Non-Recurring Charges and monthly Recurring Charges as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

CANCELLED APR 1 1 2002 APR 1 1 2002 APR S-44 Public Service Commission MISSOURI

Missouri Public Service Commission

Ν

Effective April 1, 2001, analog DID and analog 2 Way Direct service will not be available to new subscribers. FILED APR 01 2001

ISSUED : March 2, 2001

1/

EFFECTIVE: April 1, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

3.

Ν

RECESSOUNDERC. TARIFF NO. 1 MIDBOULTI PUBLIC Sorvino Commission

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

MC - JOLIO GÉRERUS GRANN

SFP 17 1998

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.2 Local Trunk Direct Inward Diating (DID): Provides the Customer with an single analog or digital connection which can carry one-way, inbound traffic.
 - 3.1.3.2.1 <u>Direct Inward Dialing Numbers</u>: Telephone numbers can be obtained in blocks of 20 or 100 numbers. Additional monthly charges will apply, as specified in Section 3.1.3.2.3.2.
 - 3.1.3.2.2 <u>Optional Features:</u> A Local Trunk DID customer may order the following optional features at the rates specified in Section 3.1.3.2.3.2.

Interim Local Number Portability (ILNP)

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System. Digital Interface can be used to carry one-way outbound traffic, one-way Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

3.1.3.2.3 <u>Direct Inward Dialing Rates and Charges</u>: A Customer who orders a Local Trunk - DID trunk will be charged applicable Non-Recurring Charges and monthly Recurring Charges as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

CANCELLED

AUG 2/0 1999

MISSOURI

APR 01 2001 Strp44 Public Service Commission MISSOURI

AUG 20 1999 By TA- 8-515 Public Service Commission

ISSUED : March 2, 1999

L

/ Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: May 14, 1999

Missouri Public Service Commission

96-355

FILFD MAY 1 & 1999

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

÷ *

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.2 Local Trunk (DID) (Cont'd)
- 3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):
 - 3.3.2.3.1 <u>Non-Recurring Charges</u>

1 <u>Non-Recurring Charges</u>	<u>St Louis/Kansas City</u>	<u>Springfield</u>	
Installation: Initial Block of 20 DID Numbers	\$5.00	\$5.00	R
Each Addt'l Block of 20 DID Numbers (up to 1000)	\$5.00	\$5.00	R
Initial Block of 100 DID Numbers	\$170.50	\$170.50	
Each Addt'l Block of 100 DID	\$170.50	\$170.50	
Line Connection (per DID Trunk)	\$20.00	\$20.00	
Account Setup (per account)	\$0.00	\$0.00	
Account Changes (Moves, Changes, Additio (per change)	\$10.50 ns)	\$10.50	
Account Changes (per Billing Record chan	\$7.75 ge)	\$7.75	
Order Expedite Charge (per line or per trunk)	\$25.00	\$25.00	N
Due Date change (per occurrence)	\$10.00	\$10.00	N
T-1 Order Expedite Charg (per T-1)	e \$600.00	\$600.00	N
T-1 Installation Charge (per T-1)	\$200.00	\$200.00	N

Issued: March 3, 2003

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 2, 2003



Misaouri Public

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.2 Local Trunk - (DID) (Cont'd)

3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.3.2.3.1 Non-Recurring Charges

Mon Recurring Charges	<u>St Louis/Kansas City</u>	<u>Springfield</u>
Installation: Initial Block of 20 DID Numbers	\$186.00	\$186.00
Each Addt'l Block of 20 DID Numbers (up to 1000)	\$31.00	\$31.00
Initial Block of 100 DID Numbers	\$170.50	\$170.50
Each Addt'l Block of 100 DID	\$170.50	\$170.50
Line Connection (per DID Trunk)	\$20.00 (R)	\$20.00 (R)
Account Setup (per account)	\$0.00	\$0.00
Account Changes (Moves, Changes, Addition (per change)	\$10.50 s)	\$10.50
Account Changes (per Billing Record chang	\$7.75 e)	\$7.75

CANCELLED



Issued: November 1, 2002

- - - ----

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: December 1, 2002

Missouri Publie Service Cemmission

FILED DEC 01 2002

Service Commission

RECT NOV 01 2002

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - Local Trunk (Cont'd) 3.1.3

3.1.3.2 Local Trunk - (DID) (Cont'd)

3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.1.3.2.3.1 Non-Recurring Charges

LOCAL EXCHANGE SERVICE

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 45** BEUENES

SEP 17 1998

MO. FUDLIL SERVICE LUMIN

St Louis/Kansas City Springfield

CANC

DEC Public

	Installation: Initial Block of 20 DID Numbers	\$186.00	\$186.00
ELLED	Each Addt'l Block of 20 DID Numbers (up to 1000)	\$ 31.00	\$31.00
25 45 Commission	Initial Block of 100 DID Numbers	\$170.50	\$170.50
	Each Addt'i Block of 100 DID	\$170.50	\$170.50
	Line Connection (per DID Trunk)	\$195.75	\$195.75
	Account Setup (per account)	\$0.00	\$0.00
	Account Changes (Moves, Changes, Additions) (per change)	\$10.50	\$10.50
BY TA-18-	Account Changes Billing Record change) QR 999 575	\$ 7.75	\$ 7.75
Public Service Cor MISSOUR	nmission L		9

96-355

ISSUED : March 2, 1999

L

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Sorvico Commission

FILED MAY 1 4 1009

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

•

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 46 Cancels ORIGINAL Page No. 46

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

3.1.3 Local Trunk (Cont'd)

3.1.3.2 Local Trunk - DID (Cont'd)

3.1.3.2.3 <u>Direct Inward Dialing Rates and Charges (Cont'd)</u>:

3.1.3.2.3.1 Non-Recurring Charges (Con't)

St Louis/Kansas CitySpringfieldLine Restoral Charge\$20.00\$20.00(per Trunk)

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily Interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently reestablished, charges apply as for a new installation of service.)

Suspension of Service \$20.50 Restoral Charge (Per Trunk) (Applies for trunk restoral after Customer-initiated suspension.)

Optional Features

Digital Interface Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System.

Digital Interface can be used to carry one-way outbound traffic, one-way Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601





LOCAL EXCHANGE SERVICE

- 3. Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.2 Local Trunk DID (Cont'd)
 - 3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):
 - 3.1.3.2.3.1 Non-Recurring Charges (Con't)

 St Louis/Kensas City
 Springfield

 Line Restoral Charge
 \$15.75
 \$15.75

 (per Trunk)
 \$15.75
 \$15.75

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily Interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

 Suspension of Service
 \$20.50
 \$20.50

 Restoral Charge (Per Trunk)
 (Applies for trunk restoral after

 Customer-initiated suspension.)

Optional Features

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System.

Digital Interface can be used to carry one-way outbound traffic, oneway Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

AUG 2/0 1999 8-575 Public Service Commission ISSUED : March 2, 1999

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission

[1] 印 way 主义 1999

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 46

マクロシィアン RECEIVED SEP 17 1998

MO. FUDLIL JERVILL LUNIN

APR 0 2 2003

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

÷

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

3.50

p :

MISSOURI P.S.C. TARIFF NO. 1 4th Revised Page No. 47 Cancels 3rd Revised Page No. 47

.

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

3.1.3 Local Trunk (Cont'd)

3.1.3.2 Local Trunk - DID (Cont'd)

3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.1.3.2.3.2 Monthly Recurring Charges

	<u>St. Louis/Kansas City</u>	<u>Springfield</u>
Local Trunk DID Charge (per trunk)		
Analog ¹ Digital	\$90.70 \$37.87	\$80.25 \$37.87
DID number charge (per block of 20 numbers)	\$10.00	\$10.00
Per Block of 100 numbers	\$23.50	\$23.50
initial purchase	\$2,000 00 numbers after an of 2000 numbers in a tistical Area (MSA) from switch.) ²	\$2,000
<u>Optional Feature</u>	<u>s</u> :	
Interim Local Num Portability (Rates for a voluthan 500 will be Case Basis)	mber \$0.00 ume of numbers greater provided on an Individua	\$0.00 al
Remote Call Forwarding(RCF) (Per each line pa	\$20.00 ath)	\$20.00
Digital Interfaco Channelization Charge (Per Channel) \$0		\$0.00
Overflow Routing	\$20.00	\$20.00 N

¹Effective April 1, 2001, analog DID and analog 2Way Direct service will not be available to new subscribers.

²This charge applies to orders placed on or after April 1, 2001.

Issued: November 1, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Filed

Effective: December 1, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 3rd Revised Page No. 47 Cancels 2nd Revised Page No. 47 Missouri Public

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

CANCEL

Publ

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.2 Local Trunk DID (Cont'd)

3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.1.3.2.3.2 Monthly Recurring Charges

	<u>S</u>	t. Louis/	Kansas City	Springfield
ANCELLED	Local Trunk DID Charge (per trunk)			
DEC 01 2002	Analog ^ı Digital	\$90.70 \$37.87		\$80.25 \$37.87
244 RS 417 244 RS Commission 6 Sen South	DID number charge (per block of 20 number	\$10.00 s)		\$10.00
la com	Per Block of 100 numbers	\$23.50		\$23.50
	initial purch	ase of 20 Statistic	nbers after an 000 numbers in a cal Area (MSA) from ch.) ²	\$2,000
	<u>Optional Feat</u>	ures:		
		\$0.00 volume og	f numbers greater ided on an Individual	\$0.00
	Remote Call Forwarding(RC (Per each lin	F) \$20.00 e path)	Missouri Public	N \$20.00 N
	Digital Inter Channelizatic Charge (Per Channel)	on	FILED APR 1 1 200	
		S	ervice Commissi	ion

¹Effective April 1, 2001, analog DID and analog 2Way Direct service will not be available to new subscribers.

²This charge applies to orders placed on or after April 1, 2001.

Issued: March 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: April 11, 2002



Service Commission

3.

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 47 CANCELS 1ST REVISED PAGE NO. 47

LOCAL EXCHANGE SERVICE

Missouri Public Service Commission

RECD MAR 02 2001

D D D

3.1 Local Exchange Service (Cont'd)

Service Descriptions (Cont'd)

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.2 Local Trunk DID (Cont'd)

3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.1.3.2.3.2 Monthly Recurring Charges

	St Louis/Kansas City	Springfield	
Local Trunk - DID Charge			
(per trunk)			
Analog <u>1</u> /	\$90.70	\$80.25	N
Digital	\$37.87	\$37.87	
DID number charge (per block of 20 numbers)	\$10.00	\$10.00	
Per Block of 100 numbers	\$23.50	\$23.50	



			D D D
DID Number Charge (Per Block of 1000 numbers after an initial purchase of 2000 numbers in a Metropolitan Statistical Area (MSA) from a single Company switch.) <u>2/</u>	\$2,000	\$2,00	-
<u>Optional Features:</u> Interim Local Number Portability (Rates for a volume of numbers greater than 500 will be provided on an Individual Case Basis)	\$0.00	\$0.00)
Digital Interface Channelization Charge (Per Channel)	^{\$0.00} M Serv	issouff ice Cor	Public nmlasion
Direct service will not be available to new subscrib 2001.	^{ers.} FIL	ED APR	01 2001 ^N

1/

2/

Effective April 1, 2001, analog DID and analog 2 Way Direct service will not be available to new subscribers. This charge applies to orders placed on or after April 1, 2001.

ISSUED : March 2, 2001

EFFECTIVE: April 1, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

MISSOURI P.S.C. TARIFF NO. 1 **1ST REVISED PAGE NO. 47** CANCELS ORIGINAL PAGE NO. 47

St Louis/Kansas City Springfield

LOCAL EXCHANGE SERVICE

- Service Descriptions (Cont'd)
 - Local Exchange Service (Cont'd) 3.1

- Local Trunk (Cont'd) 3.1.3
 - 3.1.3.2 Local Trunk - DID (Cont'd)
 - 3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.1.3.2.3.2 Monthly Recurring Charges

				•
	Local Trunk - DID Charge			
	(per trunk)			
	Analog	\$90.70	\$80.25	Т
	Digital	\$37.87	\$37.87	T/R
CANCELLED	DID number charge	\$10.00	\$10.00	
OANOLLED	(per block of 20 numbers)			
APR ,0 1 2001	Per Block of 100 numbers	\$23.50	\$23.50	
2NS RP 47 Public Service Commission MISSOURI	High Capacity Inbound Service			
Public Service Commission	Digital	\$100.00	\$100.00	
MISSOURI	Analog	\$125.00	\$125.00	
	(Rates for a volume of Numbers			
	greater than 1000 will be provided			
	on an Individual Case Basis)			
	Optional Features:			
	Interim Local Number Portability	\$0.00	\$0.00	
λ.	(Rates for a volume of numbers greater			
CANCELLED	than 500 will be provided on an Individual			
CANCELLED	Case Basis)			
	Digital Interface Channelization Charge			
AUG 2×0 1999	(Per Channel)	\$0.00	\$0.00	
By TA-98-575	S.M.	issouri Pu	blic	
Public Service Commission M/SSOURI	3 8 /V	ra Comu	1881or	r
	t-11			

FILED JUL 0 7 2000



Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECTIVE: July 7, 2000

Miasouri Public Service Commission

RECD JUN 07 2000

3.

ł

i

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 47

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

> 3.1.3 Local Trunk (Cont'd)

3.1.3.2 Local Trunk - DID (Cont'd)

3.1.3.2.3 Direct Inward Dialing Rates and Charges (Cont'd):

3.1.3.2.3.2 Monthly Recurring Charges

		St Louis/Kansas City	<u>Springfield</u>
Local Trunk - DID Cha (per trunk)	arge	\$90.70	\$80.25
DID number charge (per block of 20 numbe	ers)	\$10.00	\$ 10.00
Per Block of 100 numb	ers	\$23.50	\$23.50
High Capacity Inbound Digital Analog	l Service	\$100.00 \$125.00	\$100.00 \$125.00
(Rates for a volume of greater than 1000 will on an Individual Case	be provided		
<u>Optional Features</u> : Interim Local Number (Rates for a volume of than 500 will be provid Case Basis)	numbers greater	\$0.00	\$0.00
Digital Interface Chanr (Per Channel)	nelization Charge	\$0.00	\$0.00

CANCELLED

JUL 0 7 2000 By 1^{5t} RP 47 Public Service Commission MISSOURI

96-355

ISSUED : March 2, 1999

L

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Missouri Public Service Commission

FILED MAY 1 4 1999

Mis. : JULIU SERVICE CUMM

SEP 17 1998

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missouri Public

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1

REC'D MAR 1 1 2002 Cancels and Revised Page No. 48 2nd Revised Page No. 48

3.

LOCAL EXCHANGE SERVICE <u>Service Descriptions (Cont'd)</u> <u>3.1 Local Exchange Service (Cont'd)</u>

Service Commission

3.1.3 Local Trunk (Cont'd)

3.1.3.3

Local Trunk-2 Way Direct Provides the customer with a two-way direct dial digital or analog connection which can carry both inbound and outbound traffic and the ability to route a block of numbers to a trunk group, receive out pulsed digits on calls incoming through that trunk group and make outgoing calls using the same trunks.¹

3.1.3.3.1 <u>2 Way Direct Dialing Numbers</u> Telephone numbers can be obtained in blocks of 20 or 100 numbers. Non-Recurring Charges and monthly recurring charges will apply, as specified Sections 3.1.3.3.2.1 and 3.1.3.3.2.2.

- 3.1.3.3.2. <u>2 Way Direct Rates and Charges</u> A Customer who orders a Local Trunk will be charged applicable Non-Recurring Charges, monthly Recurring Charges, and usage rates as specified in Sections 3.1.3.3.2.1, 3.1.3.3.2.2 and 3.1.4.
- 3.1.3.3.3 <u>Features:</u> The following features are available:
 - 3.1.3.3.3.1 Standard Features

Touchtone Calling Number Delivery Blocking Selective Hunting (Circular, Sequential, and Uniforn Call Distribution)

3.1.3.3.3.2.2 Optional Features: Interim Local Number Portability Calling Number Delivery Blocking (Complete)² Remote Call Forwarding(RCF) Digital Interface Channelization Charge(Per Channel)

> Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trank-capable Key System. Digital Interface can be used to carry one-way outbound traffic, one-way Imbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.3.2.1 and 3.1.3.3.2.2.

> > Missouri Public

FILED APR 1 1 2002

Service Commission

¹Effective April 1, 2001, analog DID and analog 2 Way Direct Service will not be available to new subscribers.

²Available only to Social Service and Law Enforcement Agencies (See Section1, Calling Number Delivery).

led: March 11, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 11, 2002



N

3.

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 48 **CANCELS 1ST REVISED PAGE NO. 48**

Missouri Public Service Commission

REC'D MAR 02 2001

- Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)

3.1.3 Local Trunk (Cont'd)

Local Trunk-2 Way Direct 3.1.3.3

Provides the customer with a two-way direct dial digital or analog connection which can carry both inbound and outbound traffic and the ability to route a block of numbers to a trunk group, receive out pulsed digits on calls incoming through that trunk group and make outgoing calls using the same trunks. 1/

3.1.3.3.1 2 Way Direct Dialing Numbers Telephone numbers can be obtained in blocks of 20 or 100 numbers. Non-Recurring Charges and monthly recurring charges will apply, as specified Sections 3.1.3.3.2.1 and 3.1.3.3.2.2.

.3.1.3.3.2 2 Way Direct Rates and Charges A Customer who orders a Local Trunk will be charged applicable Non-Recurring Charges, monthly Recurring Charges, and usage rates as specified in Sections 3.1.3.3.2.1, 3.1.3.3.2.2 and 3.1.4.

3.1.3.3.3 Features: The following features are available:

LOCAL EXCHANGE SERVICE

3.1.3.3.3.1 Standard Features Touchtone Calling Number Delivery Blocking - Selective Hunting (Circular, Sequential, and Uniforn Call Distribution)

3.1.3.3.3.2 Optional Features: Interim Local Number Portability Calling Number Delivery Blocking (Complete) 2/

> Digital Interface Channelization Charge (Per Channel)

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System.

Digital Interface can be used to carry one-way outbound traffic, one-way Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.3.2.1 and 3.1.3.3.2.2.

EFFECTIVE: April 1, 2001

^d Missouri Public Service Commission

Effective April 1, 2001, analog DID and analog 2 Way Direct service will not be available to new subscribers. FILED APR 01 2001 Available only to Social Service and Law Enforcement Agencies (See Section 1, Calling Number Delivery).

ISSUED : March 2, 2001

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta GA 30328

APR 11 2002 3 eVRS 48 minission



<u>1</u>/

2/

Ν

LOCAL EXCHANGE SERVICE

CANCELS ORIGINAL PAGE NO. 48

RFCT NOV 01 1999

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 48

- 3. <u>Service Descriptions (Cont'd)</u>
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.3 Local Trunk-2 Way Direct

Provides the customer with a two-way direct dial digital or analog connection which can carry both inbound and outbound traffic and the ability to route a block of numbers to a trunk group, receive out pulsed digits on calls incoming through that trunk group and make outgoing calls using the same trunks.

3.1.3.3.1 2 Way Direct Dialing Numbers Telephone numbers can be obtained in blocks of 20 or 100 numbers. Non-Recurring Charges and monthly recurring charges will apply, as specified Sections Т 3.1.3.3.2.1 and 3.1.3.3.2.2. 3.1.3.3.2 2 Way Direct Rates and Charges A Customer who orders a Local Trunk will be charged applicable Non-Recurring Charges, monthly Recurring Charges, and usage rates as specified in Sections 3.1.3.3.2.1, 3.1.3.3.2.2 and 3.1.4. Т 3.1.3.3.3 Features: The following features are available: N 3.1.3.3.3.1 Standard Features Touchtone **CANCELLED** Calling Number Delivery Blocking - Selective Hunting (Circular, Sequential, and Uniform Call Distribution) 3.1.3.3.3.2 Optional Features: APR 01 2001 Interim Local Number Portability 2NO RP43 Calling Number Delivery Blocking (Complete) 1/ Public Service Commission MISSOURI Digital Interface Channelization Charge (Per Channel) Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 ANCEI/I F channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System. Digital Interface can be used to carry one-way outbound traffic, oneway Inbound or two-way traffic, Direct Inward Dialing, or a combination



/	A Law Enforcement Agencies (See Section 1, Calling Number Delivery). FILED JAN 1 0 2000	
Available only to Social Service an	J Law Enforcement Agencies (See Section 1, Calling Number Delivery). アニエク ひつれ エレン とう	
- 1		

in Sections 3.1.3.3.2.1 and 3.1.3.3.2.2.

ISSUED : November 1, 1999

1/

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

JAN 1 0 2000

Ser Lannellon

thereof. Applicable Non-Recurring and Recurring charges can be found PUTATE

EFFECTI

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.3 Local Trunk-2 Way Direct

Provides the customer with a two-way direct dial digital or analog connection which can carry both inbound and outbound traffic and the ability to route a block of numbers to a trunk group, receive out pulsed digits on calls incoming through that trunk group and make outgoing calls using the same trunks.

3.1.3.3.1 2 Way Direct Dialing Numbers

Telephone numbers can be obtained in blocks of 20 or 100 numbers. Non-Recurring Charges and monthly recurring charges will apply, as specified Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

3.1.3.3.2 2 Way Direct Rates and Charges

A Customer who orders a Local Trunk will be charged applicable Non-Recurring Charges, monthly Recurring Charges, and usage rates as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2.

CANCELLED

JAN 1 0 2000 By リジアスタリタ Public Service Commission MISSOURI

ISSUED : March 2, 1999

L

<u>96-3</u>55 EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission

FILED MAY 1 4 1999

RECEIVED NO. 48 SEP 17 1998

MISSOURI P.S.C. TARIFF NO. 1

MU. FULLIU SETTINE UCININ

Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 48.1 Cancels 2nd Revised Page No. 48.1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.3 Local Trunk-2 Way Direct (Cont'd)

3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

3.1.3.3.2.1 Non-Recurring Charges

<u>St_Louis/Ka</u>	<u>nsas City</u>	Springfield	1
Installation: Initial Block of 20 DID Numbers	\$5.00	\$5.00	R
Each Addt'l Block of 20 DID Numbers (up to 1000)	\$5.00	\$5.00	R
Initial Block of 100 DID Numbers	\$170.50	\$170.50	
Each Addt'l Block of 100 DID	\$170.50	\$170.50	
Line Connection (per DID Trunk)	\$20.00	\$20.00	
Account Setup (per account)	\$0.00	\$0.00	
Account Changes (Moves, Changes, Additions) (per change)	\$10.50	\$10.50	
Account Changes (per Billing Record change)	\$7.75	\$7.75	
Order Expedite Charge (per line or per trur		\$25.00	N
Due Date change (per occurrence)	\$10.00	\$10.00	N
T-1 Order Expedite Charge (per T-1)	\$600.00	\$600.00	N
T-1 Installation Charge (per T-1)	\$200.00	\$200.00	N

CANCELLED September 1, 2008 Missouri Public Service Commission

Filed Missouri Public Service Commission

Missouri Public

RECD NOV 01 2002

Service Commission

Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

- 3.1.3 Local Trunk (Cont'd)
 - Local Trunk-2 Way Direct (Cont'd) 3.1.3.3

3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

LOCAL EXCHANGE SERVICE

3.1.3.3.2.1 Non-Recurring Charges

St Louis/Kansas City Springfield

Installation: Initial Block of 20 DID Numbers	\$186.00	\$186.00
Each Addt'l Block of 20 DID Numbers (up to 1000)	\$31.00	\$31.00
Initial Block of 100 DID Numbers	\$170.50	\$170.50
Each Addt'l Block of 100 DID	\$170.50	\$170.50
Line Connection (per DID Trunk)	\$20.00 (R)	\$20.00(R)
Account Setup (per account)	\$0.00	\$0.00
Account Changes (Moves, Changes, Additions) (per change)	\$10.50	\$10.50
Account Changes (per Billing Record change)	\$7.75	\$7.75

CANCELLED



Issued: November 1, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: December 1, 2002

Missouri Public Service Commission

FILED DEC 01 2002

. •

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd) 3.

PUDIIC

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.3 Local Trunk-2 Way Direct (Cont'd)

3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

3.1.3.3.2.1 Non-Recurring Charges

REC'D NOV 01 1999

Ν

Ν

MISSOURI P.S.C. TARIFF NO. 1

ORIGINAL PAGE NO. 48.1 Microuri Fublic Service Conmission

		St Louis/Kansas City	Springfield
	Installation:		
	Initial Block of		
TIEN	20 DID Numbers	\$186.00	\$186.00
CANCELLED	Each Addt'l Block of		
UT W	20 DID	\$ 31.00	\$31.00
DEC 01 2002	Numbers (up to 1000)	,	••••
July S Tomssion	Initial Block of		
DEC 548.1 14R Commission	100 DID Numbers	\$170.50	\$170.50
UDIIC SUMSSOON	Each Addťl Block of		
	100 DID	\$170.50	\$170.50
	Line Connection	\$195.75	\$195.75
	(per DID Trunk)		
	Account Setup	\$0.00	\$0.00
	(per account)		
	Account Changes	\$10.50	\$10.50
	(Moves, Changes, Additions)		
	(per change)		
\mathbf{x}	Account Changes	\$7.75	\$7.75
CANCELLED	(per Billing Record change)		
$/$ b_k			



FILED JAN 1 0 2000

EFFECTI

JAN 1 0 2000

CANCELLI AUG 2/0 1999 -18-575 TA έy Fublic Service Commission

ISSUED : November 1, 1999

_---

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 48.2

Micround Public

RECT NOV 01 1999

Ν

Ν

LOCAL EXCHANGE SERVICE

- 3. <u>Service Descriptions (Cont'd)</u>
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.3 Local Trunk-2 Way Direct (Cont'd)
 - 3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

3.1.3.3.2.1 Non-Recurring Charges (Cont'd)

Line Restoral Charge (per Trunk) St Louis/Kansas CitySpringfield\$15.75\$15.75

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily Interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

 Suspension of Service
 \$20.50
 \$20.50

 Restoral Charge (Per Trunk)
 (Applies for trunk restoral after

 Customer-initiated suspension.)

Optional Features

Digital Interface- Digital Interface provides a DS-1 digital transmission facility operating at 1.544 Mbps and time division multiplexed into 24 channels for the connection of Basic [DID] trunks to the Customer's PBX or trunk-capable Key System.

Digital Interface can be used to carry one-way outbound traffic, oneway Inbound or two-way traffic, Direct Inward Dialing, or a combination thereof. Applicable Non-Recurring and Recurring charges can be found in Sections 3.1.3.3.2.1 and 3.1.3.3.2.2.



0U.1 **E**G

FILED JAN 1 0 2000

EFFECTIV

JAN 1 0 2000

ISSUED : November 1, 1999

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

**

\$64.20

\$30.00

St Louis/Kansas City Springfield

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

3.1.3 Local Trunk (Cont'd)

3.1.3.3 Local Trunk-2 Way Direct (Cont'd)

3.1.3.3.2 <u>2 Way Direct Rates and Charges (Cont'd)</u>

3.1.3.3.2.2 <u>Monthly Recurring Charges</u>	5
--	---

St Local Trunk-DID Charge (per trunk) Per Minute/ Per Call Options¹ Analog² Digital

Flat Rate Option		
Analog ² Digital	\$90.70 \$37.87	\$80.25 \$37.87
DID number charge (per block of 20 numbers)	\$10.00	\$10.00
Per Block of 100 numbers	\$23.50	\$23.50

\$72.56

\$30.00

<u>Optional Features</u> : Interim Local Number Portability \$0.00		\$0.00	
Remote Call Forwarding (RCF)	\$20 00	¢20.00	

Forwarding (RCF) \$2 (Per each line path)	30.00	\$20.00
Digital Interface Channelization Charge (Per Channel) \$6	2.00	\$0.00
Calling Number Delivery		
Blocking (Complete) ³	\$0.00	\$0.00
Overflow Routing	\$20.00	\$20.00

¹Effective April 1, 2001, this calling option will not be available to new subscribers.

²Effective April 1, 2002 analog DID and analog 2 Way Direct service will not be available to new subscribers.

³Available only to Social Service and Law Enforcement Agencies (Section 1, Calling Number Delivery).

Issued: November 1, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: December 1, 2002



Ν

MISSOURI P.S.C. TARIFF NO. 1 3rd Revised Page No. 48.3 Cancels 2nd Revised Page No. 48.3 Missouri Public

LOCAL EXCHANGE SERVICE

- 3. Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

Service Commission

\$10.00 Missouri Public

FILED APR 1 1 2002

Service Commission

Ν

Ν

3.1.3.3 Local Trunk-2 Way Direct (Cont'd)

3.1.3.3.2 <u>2 Way Direct Rates and Charges (Cont'd)</u>

3.1.3.3.2.2 Monthly Recurring Charges

\$72.56

\$23.50

St Louis/Kansas City Springfield

\$64.20

\$30.00

\$80.25

\$37.87

\$23.50

\$0.00

Local Trunk-DID Charge (per trunk)

Analog²

Per Minute/ Per Call Options¹

Digital \$30.00 Flat Rate Option Analog² \$90.70 Digital \$37.87

DID number charge \$10.00 (per block of 20 numbers)

Per Block of 100 numbers

Optional Features: Interim Local Number Portability \$0.00

Remote Call Forwarding (RCF) \$20.00 \$20.00 (Per each line path)

Digital Interface Channelization Charge (Per Channel) \$0.00 \$0.00

Calling Number Delivery Blocking (Complete)³ \$0.00 \$0.00

¹Effective April 1, 2001, this calling option will not be available to new subscribers.

²Effective April 1, 2002 analog DID and analog 2 Way Direct service will not be available to new subscribers.

³Available only to Social Service and Law Enforcement Agencies (Section 1, Calling Number Delivery).

Issued: March 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601



REC'D MAR 1 1 2002

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 48.3 CANCELS 1ST REVISED PAGE NO. 48.3

> Missouri Public Service Commission

REC'D MAR 0 2 2001

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

3.

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)
 - 3.1.3.3 Local Trunk-2 Way Direct (Cont'd)
 - 3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

3.1.3.3.2.2 Monthly Recurring Charges

		St Louis/Kansas City	Springfield
	Local Trunk - DID Charge (per trunk)		
•	Per Minute/Per Call Options 1/		
	Analog 2/	\$72.56	\$64.20
	Digital	\$30.00	\$30.00
	Flat Rate Option		
	Analog 2/	\$90.70	\$80.25
	Digital	\$37.87	\$37.87
	DID number charge (per block of 20 numbers)	\$10.00	\$10.00
	Per Block of 100 numbers	\$23.50	\$23.50
	Optional Features: Interim Local Number Portability	\$0.00	\$0.00
	Digital Interface Channelization Charge (Per Channel)	\$0.00	\$0.00
	Calling Number Delivery Blocking (Complete) <u>3</u> /	\$0.00	\$0:00

CANCELLED
APR 1 1 2002 APR 1 1 2002 APR 1 1 2002 Commission
PULINC SENISSOUN

Effective April 1, 2001, this calling option will not be available to new subscribers.	Missouri Public Service Commission
Effective April 1, 2001, this calling option will not be available to new subscribers. Effective April 1, 2001, analog DID and analog 2 Way Direct service will not be available to new subs Available only to Social Service and Law Enforcement Agencies (Section 1, Calling Number Delivery)	FILED APR 01 2001

ISSUED : March 2, 2001

¥

2/

<u>3</u>/

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: April 1, 2001

N N

Ν

T

N N

Т

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 48.3 CANCELS ORIGINAL PAGE NO. 48.3

St Louis/Kansas City Springfield

LOCAL EXCHANGE SERVICE

- 3. Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.3 Local Trunk-2 Way Direct (Cont'd)

3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

3.1.3.3.2.2 Monthly Recurring Charges

Local Trunk - DID Charge

REC'D JUN 0 7 2000

CANCELLED

APR 01 2001 2ND RP 48.3 Public Sarvice Commission MISSOURI

CANCEL

ΙĤ

Public Service Commission MISSOURI

2/0 1999

AUG

By

(per trunk)			
Per Minute/Per Call Options			
Analog	\$72.56	\$64.20	т
Digital	\$30.00	\$30.00	T/R
Flat Rate Option			
Analog	\$90.70	\$80.25	Т
Digital	\$37.87	\$37.87	T/R
DID number charge (per block of 20 numbers)	\$10.00	\$10.00	
Per Block of 100 numbers	\$23.50	\$23.50	
Optional Features:			
Interim Local Number Portability	\$0.00	\$0.00	
Digital Interface Channelization Charge			
(Per Channel)	\$0.00	\$0.00	
Calling Number Delivery			
Blocking (Complete) <u>1</u> /	\$0.00	\$0.00	

Missouri Public Service Commission

FILED JUL 0 7 2000

Available only to Social Service and Law Enforcement Agencies (Section 1, Calling Number Delivery).

58-575

ISSUED : June 7, 2000

1/

ļ

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: July 7, 2000

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 48.3

Schuceun Public

RECT NOV 01 1999

Ν

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.3 Local Trunk (Cont'd)

3.1.3.3 Local Trunk-2 Way Direct (Cont'd)

3.1.3.3.2 2 Way Direct Rates and Charges (Cont'd)

3.1.3.3.2.2 Monthly Recurring Charges	St Louis/Kansas City	Springfield
Local Trunk - DID Charge (per trunk)		
Per Minute/Per Call Options Flat Rate Option	72.56 90.70	64.20 80.25
DID number charge (per block of 20 numbers)	\$10.00	\$10.00
Per Block of 100 numbers	\$23.50	\$23.50
Optional Features: Interim Local Number Portability	\$0.00	\$0.00
Digital Interface Channelization Charge (Per Channel)	\$0.00	\$0.00
Calling Number Delivery Blocking (Complete) <u>1</u> /	\$0.00	\$0.00 N

CANCELLFD

JUL 07 2000 By ドナ RP 48.3 Public Service Commission MISSOURI

School The State

______/

Available only to Social Service and Law Enforcement Agencies (Section Section 1, Calling Number Delivery) DAN 1 0 2000

ISSUED : November 1, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFEC

JAN 1 0 2000

ALL MATERIAL ON THIS PAGE WAS REMOVED

· .

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 49 CANCELS ORIGINAL PAGE NO. 49

LOCAL EXCHANGE SERVICE

Missouri Public Service Commission RECD MAR 02 2001

N

Ν

Ν

3.1 Local Exchange Service (Cont'd)

Service Descriptions (Cont'd)

- 3.1.4 <u>Usage Rates</u>: All Local Exchange Service Customers may order service on a per minute or per call usage basis as stated below. These rates will apply to all outgoing direct-dialed calls placed to Stations within the caller's local exchange area, as defined herein. If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.
 - 3.1.4.1 <u>Per Minute Usage Rate</u>: <u>1</u>/ The following rates will be applied on a per minute basis. Peak rates willapply from 8:00 a.m. through 5:00 p.m. Monday through Friday. Off Peak rates will apply at all other times.

St Louis/Kansas City/Springfield

<u>PE</u>	<u>AK</u>	<u>OFF F</u>	PEAK
1stMinute	Addl'Minutes	1stMinute	Addl' Minutes
\$ 0.044	\$0.014	\$ 0.033	\$0.011

3.1.4.2 <u>Per Call Usage Rate</u>: <u>1</u>/ The following rate will be applied on a per call basis, regardless of the duration of the call: <u>St Louis/Kansas City</u> Springfield

\$0.070

\$0.060

Missouri Public Service Commission

FILED APR 01 2001

Effective April 1, 2001, this calling option will not be available to new subscribers.

Per Call

ISSUED : March 2, 2001

CANCELLED September 1, 2008 Missouri Public Service Commission

1/

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: April 1, 2001

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 49

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

SEP 17 1998

MO. FUDLIL SERVILL ULIMM

3.1 Local Exchange Service (Confd)

- 3.1.4 <u>Usage Rates</u>: All Local Exchange Service Customers may order service on a per minute or per call usage basis as stated below. These rates will apply to all outgoing direct-dialed calls placed to Stations within the caller's local exchange area, as defined herein. If the computed charge includes a fraction of a cent, the fraction is rounded to the nearest whole cent.
 - 3.1.4.1 <u>Per Minute Usage Rate</u>: The following rates will be applied on a per minute basis. Peak rates will apply from 8:00 a.m. through 5:00 p.m. Monday through Friday. Off Peak rates will apply at all other times.

St Louis/Kansas City/Springfield

<u>PEAK</u>		<u>OFF F</u>	OFF PEAK		
1stMinute	Addl'Minutes	1stMinute	Addi' Minutes		
\$ 0.044	\$0.014	\$ 0.033	\$0.011		

3.1.4.2 <u>Per Cell Usage Rate</u>: The following rate will be applied on a per cell basis, regardless of the duration of the cell:

i	<u>St Louis/Kansas City</u>	<u>Springfield</u>
:	\$0.070	\$0.060

CANCELLED

APR 01 2001 By 1St RP49 Public Service Commission MISSOURI

CANCELLED
$/$ \rangle $\hat{\mathcal{D}}_{k}$
AUG 2 0 1999
By TA 78-575
By TA 198-575 Public Service Commission MISSOURI

Per Call

96-355

EFFECTIVE: May 14, 1999

ISSUED : March 2, 1999

L

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission

FILED MAY 1 4 1999

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 50 Cancels Original Page No. 50

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008
3.

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 50

			LOCAL EXCHANGE SEF	VICE RECEIVED
Service Descriptions (Cont'd)		SEP 17 1998		
3.1	<u>Local Ex</u> 3.1.5			MO. PUBLIC SERVICE COMM
		I Local Trunk DID. Local ISDN-PRI is prov standard of the Integrated Services Digital r with the capabilities of simultaneous acce	· · · · ·	
		3.1.5.1.	to create an Local ISDN-PRI Service Ar PRI Service Arrangements per location,	One or more Service Configurations can be combined rangement. Customers may have multiple Intellispan however for each. Service Arrangement one Service controlling D channel will always reside on Service

The ISDN Digital Facility is ordered separately and not Included as part of the Service Configuration.

The D channel is a 64 Kbps channel that carries signaling and control for the B channels. The capabilities of the D channel are contained within the customers Service Configuration.

- 3.1.5.1.1 <u>Service Configuration 1</u> The first Service Configuration for any PRI Local Service Arrangement must be a Service Configuration 1. Service Configuration 1 provides twenty-three (23) 64 Kbps B channels and one (1) primary 64 Kbps D signaling channel. The primary D channel is an-out-of band signaling channel used to control and route all of the B channel traffic within the PRI Local Service Arrangement.
- 3.1.5.1.2 <u>Service Configuration 2</u> This configuration is only available in conjunction with Service Configuration 1. It provides an additional twenty-four (24) 64 Kbps B channels which are controlled by the D channel on Service Configuration 1. Multiple Service Configuration 2s may be associated with a Service Configuration 1 subject to switching equipment limitations.



L

96-355

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

FILED MAY 1 4 1999

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 51 Cancels Original Page No. 51

Missouri Public

. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

REC'D MAR 11 2002

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI)

LOCAL EXCHANGE SERVICE

- Service Commission 3.1.5.1. Local ISDN-PRI Service Arrangement (Cont'd)
 - 3.1.5.1.3 <u>Service Configuration 3</u> This configuration is also only available to subscribers who have Service Configuration 1. It provides a backup 64 Kbps D channel and twenty-three (23) additional 64 Kbps B channels. The backup D channel will control and route all of the B channel traffic of a single PRI Service Arrangement in the event the D channel on the primary configuration (Service Configuration 1) fails. A maximum of one Service Configuration 3 may be included with each Local ISDN-PRI Service Arrangement. Service Configuration 3 is available subject to switching equipment capabilities.

The bearer (or B) channels are used for information transfer between users. B channels provide the same features as Local Trunk Basic and Local Trunk DID. To receive the DID capability, customers must order separately DID telephone numbers, as found in Section 3.1.3.2 preceding.

3.1.5.2 <u>Optional Features</u>: In addition to providing Local Exchange Services specified above, Local ISDN-PRI can be configured to support the following optional advanced ISDN features:

> <u>Call-by-Call Service Selection</u>: Allows B channels to be shared between multiple services carried over a single Local ISDN-PRI Serving Arrangement.

> <u>Calling Number Delivery:</u> Allows the ustomer to receive the originating caller's working telephone number provided this information is forwarded to the Company by the caller's Local Exchange and/or Long Distance Carrier.

<u>Remote Call Forwarding (RCF)</u>

N T

ጥ

Feature Package 1 includes call by call Feature and Calling number Delivery.

3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges:

Non-Recurring and Monthly Recurring Charges as specified below, apply per T-1 T unless otherwise noted. The Local ISDN PRI T-1 non-recurring and recurring charges include Service Configuration and B Channel Service charges. Customers subscribing to DID capability will be assessed DID number charges as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2, respectively. Missouri Public

FILED APR 1 1 2002

Service Commission

Effective: April 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Issued: March 11, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 51 CANCELS ORIGINAL PAGE NO. 51

Service Commission

REC'D MAR 2 1 2000

LOCAL EXCHANGE SERVICE

- Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI)
 - 3.1.5.1. Local ISDN-PRI Service Arrangement (Cont'd)

CANCELLED APR 1 1 2002 APR 1 1 2002 CR 51 CR 51 Comm Public Service Comm Missouri ommission

3.1.5.1.3 <u>Service Configuration 3</u> - This configuration is also only available to subscribers who have Service Configuration 1. It provides a backup 64 Kbps D channel and twenty-three (23) additional 64 Kbps B channels. The backup D channel will control and route all of the B channel traffic of a single PRI Service Arrangement in the event the D channel on the primary configuration (Service Configuration 1) fails. A maximum of one Service Configuration 3 may be included with each Local ISDN-PRI Service Arrangement. Service Configuration 3 is available subject to switching equipment capabilities.

The bearer (or B) channels are used for information transfer between users. B channels provide the same features as Local Trunk Basic and Local Trunk DID. To receive the DID capability, customers must order separately DID telephone numbers, as found in Section 3.1.3.2 preceding.

3.1.5.2 <u>Optional Features</u>: In addition to providing Local Exchange Services specified above, Local ISDN-PRI can be configured to support the following optional advanced ISDN features:

<u>Call-by-Call Service Selection</u>: Allows B channels to be shared between multiple services carried over a single Local ISDN-PRI Serving Arrangement.

<u>Calling Number Delivery</u>: Allows the customer to receive the originating caller's working telephone number provided this information is forwarded to the Company by the caller's Local Exchange and/or Long Distance Carrier.

These features can be ordered separately or combined in Feature Package 1.

Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges: Non-Recurring and Monthly Recurring Charges as specified below, apply per T-1 unlessotherwise noted. The Local ISDN PRI T-1 non-recurring and recurring charges include Service Configuration and B Channel Service charges. Customers subscribing to DID capability will be assessed DID number charges as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2, respectively.

BOWIES COMMISSION FILED APR 2 0 2000

T T

Т

EFFECTIVE: April 20, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

3.1.5.3

CANCEII AUG By Public Service Commission MISSOURI

ISSUED : March 21, 2000

LOCAL EXCHANGE SERVICE

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 51

 Service Descriptions (Correction) 	<u>nťď)</u>
---	-------------

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI)
 - 3.1.5.1. Local ISDN-PRI Service Arrangement (Cont'd)

FRECEIVED SEP 17 1998

MU. FUDLIU SERVIUL UVININ

3.1.5.1.3 <u>Service Configuration 3</u> - This configuration is also only available to subscribers who have Service Configuration 1. It provides a backup 64 Kbps D channel and twenty-three (23) additional 64 Kbps B channels. The backup D channel will control and route all of the B channel traffic of a single PRI Service Arrangement in the event the D channel on the primary configuration (Service Configuration 1) fails. A maximum of one Service Configuration 3 may be included with each Local ISDN-PRI Service Arrangement. Service Configuration 3 is available subject to switching equipment capabilities.

The bearer (or B) channels are used for information transfer between users. B channels provide the same features as Local Trunk Basic and Local Trunk DID. To receive the DID capability, customers must order separately DID telephone numbers, as found in Section 3.1.3.2 preceding.

3.1.5.2 Optional Features: In addition to providing Local Exchange Services specified above, Local ISDN-PRI can be configured to support the following optional advanced ISDN features:

<u>Call-by-Call Service Selection</u>: Allows B channels to be shared between multiple services carried over a single Local ISDN-PRI Serving Arrangement.

<u>Calling Number Delivery</u>: Allows the customer to receive the originating caller's working telephone number provided this information is forwarded to the Company by the caller's Local Exchange and/or Long Distance Carrier.

These features can be ordered separately or combined in Feature Package 1.

3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges: In addition to the Non-Recurring and Monthly Recurring Charges specified below, all applicable charges associated with Local Trunk - Basic will apply, as specified in Sections 3.1.3.1.3.1 (except for the per trunk line connection charges) and 3.1.3.1.3.2, respectively. Customers subscribing to DID capability will be assessed DID number charges as specified in Sections 3.1.3.2.3.1 and 3.1.3.2.3.2, respectively.

> Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200

> > Atlanta, GA 30328

CANCELLED APR 2 0 2000 155 RR 51 Public Service Commission MISSOURI

ISSUED : March 2, 1999

L

96-355

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILED MAY 1 4 1999

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 4th Revised Page No. 52 Cancels 3rd Revised Page No. 52

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro ACCESS TRANSMISSION

MISSOURI P.S.C. TARIFF NO. 1 3rd Revised Page No. 52 Cancels 2nd Revised Page No. 52

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):
 - 3.1.5.3.1 <u>Non-Recurring Charges</u>

	<u>St Louis/Kansas City</u>	<u>Springfiel</u>	<u>d</u>
Service Reconfiguration Charge <u>1</u> /	\$50.00	\$50.00	
Local ISDN PRI T-1 Installation (Per T-1)	\$200.00	\$200.00	
Account Setup (per account)	\$0.00	\$0.00	
Account Changes (Moves, Changes, Additions) (per change)	\$10.50	\$10.50	
Due Date change (per occurrence)	\$10.00	\$10.00	N
T-1 Order Expedite Charge (per T-1)	\$600.00	\$600.00	N

<u>1</u>/ Applies for each Service Configuration affected i.e., if the Service Reconfiguration will effect one Service Configuration 1 and three Service Configuration 2s, four (4) Service Reconfiguration Charges would apply.

Issued: March 3, 2003

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 2, 2003

Filed Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 52 Cancels NEBOUNT Public 0. 52

RECT NOV 01 2002

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

- 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 <u>Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates</u> and Charges (cont'd):
 - 3.1.5.3.1 <u>Non-Recurring Charges</u>

	<u>St Louis/</u>	<u>Kansas City</u>	<u>Springfield</u>
Service Reconfiguration Charge <u>1</u> /	\$50.00		\$50.00
Local ISDN PRI T-1 Installation (Per T-1)	\$200.00	(R)	\$200.00(R)
Account Setup (per account)	\$0.00		\$0.00
Account Changes (Moves, Changes, Additions) (per change)	\$10.50		\$10.50

CANCELLED



<u>1</u>/ Applies for each Service Configuration affected i.e., if the Service Reconfiguration will effect one Service Configuration 1 and three Service Configuration 2s, four (4) Service Reconfiguration Charges would apply.

Issued: November 1, 2002

Carmen L.Felíciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: December 1, 2002 Missouri Public Service Commission

FILED DEC 01 2002

MISSOURI P.S.C. TARIFF NO. 1 **1ST REVISED PAGE NO. 52** CANCELS ORIGINAL PAGE NO. 52

RFCT) MAR 2 1 2000

Ν Ν

D

Ð

Ν

LOCAL EXCHANGE SERVICE

- Service Descriptions (Cont'd) 3.
- Service Commission

3.1 Local Exchange Service (Cont'd)

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)

3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

3.1.5.3.1	Non-Recurring Charges				
		St Louis/Kansas City	Springfield		
	Service Reconfiguration Charge <u>1</u> /	\$ 50.00	\$50.00		
	Local ISDN PRI T-1 Installation (Per T-1)	\$2,790.00	\$2,790.00		

CANCELLED DEC_012002 mission

GANCE AUG /0 199 75 Public Service Commussion **MISSOURI**

Account Setup	\$0.00	\$0.00	N
Account Changes (Moves, Changes, Additions) (per change)	\$10.50	\$10.50	N

CERTAIN MATERIAL PREVIOUSLY LOCATED ON THIS PAGE CAN NOW BE FOUND ON PAGE NO. 52.1.

1/ Applies for each Service Configuration affected i.e., if the Service Reconfiguration will effect one Service Configuration 1 and three Service Configuration 2s, four (4) Service Reconfiguration Charges would apply.

ISSUED : March 21, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECTIVE: April 20, 2000 Missourt Public Bervice Commission



3.

MISSOURI P.S.C. TARIFF NO. 1 RECEIVED

SFP 17 1998

MU. PUDLIU ODINIUL UDINIM

Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

- 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

LOCAL EXCHANGE SERVICE

3.1.5.3.1	Non-Recurring Charges	St Louis/Kansas City	Springfield
	Service Reconfiguration Charge <u>1</u> /	\$ 50.00	\$50.00
	Local ISDN-PRI Serving Arrangement.		
	Local ISDN-PRI Service Configuration 1	<u>Per Facility</u> \$2790.00	\$2790.00
	Local ISDN-PRI Service Configuration 2	<u>Per Facility</u> \$2790.00	\$2790.00
	Local ISDN-PRI Service Configuration 3	<u>Per Facility</u> \$2790.00	\$2790.00
		<u>Per Active</u> <u>B Channel</u>	
	B Channel Service	\$ 0.00	\$0.00

Optional Features:

Non-Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

		St Louis/Kansas City	Springfield
APR 2 0 2000	Per Location		
p. 15TRP52	Call-by-Call Option	\$ 10.00	\$10.00
By St RP52 Public Service Commission MISSOURI	Calling Number Delivery	\$ 100.00	\$100.00
MISSUUNI	Feature Package 1	\$ 105.00	\$105.00
	(Includes Call-by-Call & Calling		
	Number Delivery)		

1/

Applies for each Service Configuration affected i.e., if the Service Reconfiguration will effect one Service Configuration 1 and three Service Configuration 2s, four (4) Service Reconfiguration Charges would apply. 96-355

ISSUED : March 2, 1999

L

CANCELLED

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECTIVE: May 14, 1999 Missouri Public Service Commission

FILED WAY 1 & 1999

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

3.

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 52.1

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

Mice in Public Service Commission

RECTO MAR 2 1 2000

\$20.50

Ν

Ν

М

М

3.1 Local Exchange Service (Cont'd)

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)

3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

3.1.5.3.1	Non-Recurring Charges (cont'd)	St Louis/Kansas City Springfie		
	Account Changes (Per Billing Record Change)	\$7.75	\$7.75	
	Line Restoral Charge	\$15.75	\$15.75	

(Applies for trunk restoral after temporary interruption of service initiated by the Company. If service is temporarily interrupted and payment is not received within 10 days following the interruption, the Company reserves the right to discontinue service. If the service is discontinued and subsequently re-established, charges apply as for a new installation of service.)

Suspension of Service \$20.50 Restoral Charge (Per trunk)

(Applies for trunk restoral after Customer- initiated suspension.)

Optional Features:

Non-Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

	St Louis/Kansas City	<u>Springfield</u>	
Per Location			
Call-by-Call Option	\$ 10.00	\$10.00	
Calling Number Delivery	\$ 100.00	\$100.00	
Feature Package 1	\$ 105.00	\$105.00	
(Includes Call.by.Call & Calling			

(Includes Call-by-Call & Call Number Delivery)

COLUMN PUBLIC Sai FILED APR 2 0 2000

CANCELLED

HUDIC Service Commission MISSOURI

ISSUED : March 21, 2000

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: April 20, 2000

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

2

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Business Services

Missouri P.S.C.Tariff No. 1 4th Revised Page No. 53 Cancels 3rd Revised Page No. 53

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

ب •

:

÷

3.1 Local Exchange Service (Cont'd)

- 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

3.1.5.3.2 Monthly Recurring Charges

	<u>St Loui</u> s/Kansas City	Springfield
Local ISDN PRI T-1 Flat Rate Option	$\frac{\text{Per } T-1}{\$1,760.00}$ I	\$1,760.00 I
Local ISDN-PRI T-1 Per Minute/ Per Call Options <u>1</u> /	\$440.00 I	\$440.00 I

1/Effective April 1, 2001, this calling option will not be available to new subscribers.

Issued: June 30, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano 205 N. Michigan Avenue Chicago, IL 60601 Effective: August 1, 2008

3.

MISSOURI P.S.C. TARIFF NO. 1 3RD REVISED PAGE NO. 53 CANCELS 2ND REVISED PAGE NO. 53

LOCAL EXCHANGE SERVICE

Missouri Public Service Commission

Service Descriptions (Cont'd)

*

RECD MAR 02 2001

Ν

3.1 Local Exchange Service (Confd)

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)

•

3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

3.1.5.3.2	Monthly Recurring Charges	St Louis/Kansas City	Springfield
	Local ISDN PRI T-1 Flat Rate Option	<u>Per T-1</u> \$1,600.00	\$1,600.00
	Local ISDN-PRI T-1 Per Minute/ Per Call Options <u>1</u> /	\$400.00	\$400.00

Missouri Public Servico Commission

FILED APR 01 2001 N

ISSUED : March 2, 2001

CANCELLED August 1, 2008 Missouri Public Service Commission

<u>1</u>/

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

Effective April 1, 2001, this calling option will not be available to new subscribers.

EFFECTIVE: April 1, 2001

1

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 53 CANCELS 1ST REVISED PAGE NO. 53

N

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

SEP 01 2000

RECEIVED

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd) MISSOURI Public Service Commission

3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

 3.1.5.3.2
 Monthly Recurring Charges
 St Louis/Kansas City
 Springfield

 Local ISDN PRI T-1 Flat Rate Option
 Per T-1
 \$1,600.00
 \$1,600.00

 Local ISDN-PRI T-1 Per Minute/ Per Call Options
 \$400.00
 \$400.00

CANCELLED

APR 01 2001 L, 3N RP 53 Public Service Commission MISSOURI

CANCE

16

Бy

aug 2/0 1999

Autoric Service Commission

FILED

OCT 01 2000

MISSOURI Public Service Commission

EFFECTIVE: October 1, 2000

ISSUED : September 1, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

98-575

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 53 CANCELS ORIGINAL PAGE NO. 53

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

3.1.5

Service Commission

RECT MAR 2 1 2000

D

D

Ν

Ν

- 3.1 Local Exchange Service (Cont'd)
 - Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):
 - 3.1.5.3.2 Monthly Recurring Charges

St Louis/Kansas City Springfield

Local ISDN PRI T-1 Flat Rate Option

Per T-1 \$1,600.00

\$1,600.00

CANCELLED

OCT 0 1 2000 By 2 ^{AO} RS #53 Public Service Commission MISSOURI



EFFECTIVE: April 20, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 21, 2000

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

RECEIVED SEP 17 1998

MISSOURI P.S.C. TARIFF NO. 1

ORIGINAL PAGE NO. 53

MO. FUDLIU OENVIUL UUWW

3.1 Local Exchange Service (Cont'd)

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)

Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd): 3.1.5.3

3.1.5.3.2	Monthly Recurring Charges		
		St Louis/Kansas City	Springfield
	Local ISDN-PRI Service Configuration 1	Per Facility \$100.00	\$100.00
	Local ISDN-PRI Service Configuration 2	Per Facility \$100.00	\$100.00
	Local ISDN-PRI Service Configuration 3	<u>Per Facility</u> \$100.00	\$100.00
	B Channel Service	Per active <u>B channel</u> \$ 21.00	\$32.00

CANCELLED

APR 2 0 2000 By 15± RP 53 Public Service Commission **MISSOURI**

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 2, 1999

Ł

Missouri Public Service Commission

FILFD MAY 1 4 1999



i

I

ı.

ł

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

- 1. 2

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

3.1.5.3.2 Monthly Recurring Charges (cont'd)

Optional Features:

Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

	<u>St Louis/Kansas City</u>	Springfield
Call-by-Call Option	<u>Per Location</u> \$375.00	\$375.00
Calling Number Number Delivery	\$100.00	\$100.00
Feature Package 1 (Includes Call-by- Call and Calling Number Delivery)	\$425.00	\$425.00
Remote Call Forwarding (RCF)(per each line pat		\$20.00
Overflow Routing	\$20.00	\$20.00

3.1.5.3.3 <u>Usage Charges</u>: The rates and charges specified in Section 3.1.5.3.2 for FlatRate Option will apply to circuit switched voice and data calls. The rates specified in Section 3.1.4 will apply for customers selecting the per minute or per call option.

Issued: November 1, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601





Ν

Missouri Public

MISSOURI P.S.C. TARIFF NO. 1 3rd Revised Page No. 54

RECD MAR 1 1 2002 2nd Revised Page No. 54

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u> Service Commission

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):

3.1.5.3.2 Monthly Recurring Charges (cont'd)

Optional Features:

Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

	<u>St Louis/Kansas City</u>	Springfield
Call-by-Call Option	<u>Per Location</u> \$375.00	\$375.00
Calling Number Number Delivery	\$100.00	\$100.00
Feature Package 1 (Includes Call-by- Call and Calling Number Delivery)	\$425.00	\$425.00
Remote Call Forwarding (RCF)(per each line pat		\$20.00

3.1.5.3.3 <u>Usage Charges</u>: The rates and charges specified in Section 3.1.5.3.2 for FlatRate Option will apply to circuit switched voice and data calls. The rates specified in Section 3.1.4 will apply for customers selecting the per minute or per call option.

CANCELLED



Missouri Public

FILED APR 1 1 2002

Service Commission

Issued: March 11, 2002

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 11, 2002

Ν

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 54 CANCELS 1ST REVISED PAGE NO. 54

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

SEP 01 2000

RECEIVED

3.1 Local Exchange Service (Cont'd)

MISSOURI

- 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd) Public Service Commission
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):
 - 3.1.5.3.2 Monthly Recurring Charges (cont'd)

Optional Features:

Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

	St Louis/Kansas City	Springfield
Call-by-Call Option	Per Location \$ 375.00	\$375.00
Calling Number Number Delivery	\$ 100.00	\$100.00
Feature Package 1 (Includes Call-by-Call and Calling	\$425.00 Number Delivery)	\$425.00

 3.1.5.3.3
 Usage Charges: The rates and charges specified in Section 3.1.5.3.2 for FlatRate

 Option will apply to circuit switched voice and data calls. The rates specified in Section
 N

 3.1.4 will apply for customers selecting the per minute or per call option.
 N



CANCELLED
AUG 2 8 1999
Ey 7A-78-575 Public Service Commission

MISSOURI

ISSUED : September 1, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

FILED

OCT 01 2000

MISSOURI Public Service Commission

EFFECTIVE: October 1, 2000

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 54 CANCELS ORIGINAL PAGE NO. 54

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

RECT MAR 2 1 2000

3.1 Local Exchange Service (Cont'd)

3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)

- 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):
 - 3.1.5.3.2 Monthly Recurring Charges (cont'd)

Optional Features:

Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

	St Louis/Kansas City	<u>Springfield</u>
Call-by-Call Option	Per Location \$ 375.00	\$375.00
Calling Number		
Number Delivery	\$ 100.00	\$100.00
Feature Package 1	\$425.00	\$425.00
(Includes Call-by-Call and Calling	Number Delivery)	

3.1.5.3.3 <u>Usage Charges</u>: The rates and charges specified in Section 3.1.5.3.2 for FlatRate Option will apply to circuit switched voice and data calls.

T T

CANCELLED

OCT 012000 By AND RP=454 Public Service Commission MISSOURI



FILED APR 2 0 2000

EFFECTIVE: April 20, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 21, 2000

MISSOURI P.S.C. TARIFF NO. 1

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

SEP 17 1999

MU. MUDLIU OLINVIUL UUMM

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.5 Local ISDN-Primary Rate Interface (Local ISDN-PRI) (cont'd)
 - 3.1.5.3 Local ISDN-Primary Rate Interfaced (Local ISDN-PRI) Rates and Charges (cont'd):
 - 3.1.5.3.2 Monthly Recurring Charges (cont'd)

Optional Features:

Recurring Optional Feature charges are assessed once per customer location regardless of the number of arrangements or Service Configurations installed.

	St Louis/Kansas City	Springfield
Call-by-Call Option	<u>Per Location</u> \$ 375.00	\$375.00
Calling Number Number Delivery	\$ 100.00	\$100.00
Feature Package 1 (Includes Call-by-Call and Calling	\$425.00 Number Delivery)	\$425.00

3.1.5.3.3 <u>Usage Charges</u>: The rates and charges specified in Section 3.1.3.1.3.2 for unlimited option will apply to circuit switched voice calls.

CANCELLED

APR 2 0 2000 By 155 RP 54 Public Service Commission MISSOURI



ISSUED : March 2, 1999

L

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission FILFD MAY 1 4 1999

96-355

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

÷

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 55 CANCELS 1ST REVISED PAGE NO. 55

Ð

D

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)



REC'D JUN 0 7 2000

CANCÉ AUG 0 1999 By TA 98-575 Public Service Commission MISSOURI

ISSUED : June 7, 2000

٤

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission

FILED JUL 07 2000

EFFECTIVE: July 7, 2000

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 55 **CANCELS ORIGINAL PAGE NO. 55** MICOURI PURILIÀ

RFCT NOV 01 1999

Т

Т

Т

Т

Ν

D

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.6 Digital T-1 Credit Program

> Digital T-1 Credit Program provides a credit for customers who purchase Digital Local Trunk DID or Digital Local Trunk -2 way Direct trunks (Trunk Types) in blocks of 24 (an entire T-1). Customers who order an entire T-1 will be charged applicable non-recurring trunk and monthly recurring trunk charges as specified below for Local Trunk depending on service for which customer subscribes. Terms, conditions, and other charges, as described elsewhere in this Tariff, apply as appropriate.

Non-Recurring Charges

Applicable rates for non-recurring charges for Local Trunk-DID and Local Trunk -2 Way Direct will apply. The rates in section 3.1.3.2.3.1 and 3.1.3.3.2.1 will apply for non-recurring charges.

Monthly Recurring Charges

Applicable rates for monthly recurring charges for Local Trunk-DID and Local Trunk -2 Way Direct will apply. The rates in section 3.1.3.2.3.2 and 3.1.3.3.2.2 will apply for monthly recurring charges.

Customers enrolling in this service will receive the following monthly credit per trunk per T-1 ordered:

	Credit Per Trunk		
	Springfield	St Louis/Kansas City	
Local Trunk – DID (Digital)	\$21.00	\$22.00	Т
Local Trunk – 2 Way Direct	\$21.00	\$22.00	N
(Digital Flat Rate Option)			Ν

Credits earned during a partial billing period will be given in their entirety and will not be pro-rated.

In addition to the above charges, rates for usage will apply, if applicable, as specified in Section 3.1.4.

Customers must retain their original T-1 configuration. Customers who disconnect trunks from the original T-1 configuration will no longer receive the credits for that T-1. Customers may enroll more than one T-1 in this service.

CANCELLED

JUL 0 7 2000 By 2ND RP 55 Public Service Commission MISSOURI

Survice Commission

FILED JAN 1 0 2300

ISSUED : November 1, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

EFFECT

2 . Sacher da Sacher

JAN 1 0 2000

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55

LOCAL EXCHANGE SERVICE

RÉCEIVÉD

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd)

3.1.6 Digital T-1 Service

MU. FUDLIN DERVICE USIMM

SFP 17 1998

Digital T-1 Service provides a credit for customers who purchase Digital Local Trunk DID or Digital Local Trunk -2 way Direct trunks (Trunk Types) in blocks of 24 (an entire T-1). Customers who order an entire T-1 will be charged applicable non-recurring trunk and monthly recurring trunk charges as specified below for Local Trunk depending on service for which customer subscribes. Terms, conditions, and other charges, as described elsewhere in this Tariff, apply as appropriate.

Non-Recurning Charges

Applicable rates for non-recurring charges for Local Trunk-DID and Local Trunk -2 Way Direct will apply. The rates in section 3.1.3.2.3.1 will apply for non-recurring charges.

Monthly Recurring Charges

Applicable rates for monthly recurring charges for Local Trunk-DID and Local Trunk -2 Way Direct will apply. The rates in section 3.1.3.2.3.2 will apply for monthly recurring charges.

Customers enrolling in this service will receive the following monthly credit per trunk per T-1 ordered:

	Credit Per Trunk		
	Springfield	St Louis/Kansas City	
Local Trunk – DID/2 Way Direct (Digital)	\$21.00	\$22.00	

Credits earned during a partial billing period will be given in their entirety and will not be pro-rated.

Customers must retain their original T-1 configuration. Customers who disconnect trunks from the original T-1 configuration will no longer receive the credits for that T-1. Customers may enroll more than one T-1 in this service. T-1 configurations which combine trunk types are not eligible for the monthly credit.

CANCELLED

JAN 1 0 2000 By ^{|55} RP 55 Public Service Commission MISSOURI

96-355

EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : March 2, 1999

£



Missouri Public Service Commission

FILED MAY 1 4 1999



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 4th Revised Page No. 55.1 Cancels 3rd Revised Page No. 55.1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MISSOURI P.S.C. TARIFF NO. 1 3RD REVISED PAGE NO. 55.1 CANCELS 2ND REVISED PAGE NO. 55.1 Missouri Public

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

REC'D MAR 1 1 2002

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan

Service Commission

N

3.1.7.1 <u>On-Net Term Plan</u>

The On-Net Term Plan is a term plan, in lieu of all other tariffed Local Exchange Service term plans for customers who subscribe to MCI WorldCom On-Net interstate service. Customers who enroll in this term plan must have their facilities based MCI Local Exchange service usage billed to a single On-Net invoice.

Customers subscribing to this term plan will receive discounts as set forth below, in lieu of all other tariffed term plans. The discounts will be applied to the following recurring charges: Optional Features, Ν Local ISDN-PRI Optional Features, Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, and Local ISDN-PRI T-1 Charge as well as local service usage charges.

The Volume Commitment will be determined based upon the Qualifying Volume received by the Company, or any affiliate of the Company, from customer's facilities based Local Exchange service, as defined in Section 3.1.7.2 and the Qualifying Volume of customer's there telecommunications services.

Missouri Public

FILED APR 1 1 2002

Service Commission

Issued: March 11, 2002

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: April 11, 2002

MISSOURI P.S.C. TARIFF NO. 1 3RD REVISED PAGE NO. 55.1 CANCELS 2ND REVISED PAGE NO. 55.1 RECEIVED

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd) 3.1.7 <u>Term Plan</u>

SEP 01 2000

3.1.7.1 **On-Net Term Plan**

MISSOURI Public Service Commission

The On-Net Term Plan is a term plan, in lieu of all other tariffed Local Exchange Service term plans for customers who subscribe to MCI WorldCorn On-Net interstate service. Customers who enroll in this term plan must have their facilities based MCI Local Exchange service usage billed to a single On-Net invoice.

Customers subscribing to this term plan will receive discounts as set forth below, in lieu of all other tariffed term plans. The discounts will be applied to the following recurring charges: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, and Local ISDN-PRI T-1 Charge as well as local service usage charges.

The Volume Commitment will be determined based upon the Qualifying Volume received by the Company, or any affiliate of the Company, from customer's facilities based Local Exchange service, as defined in Section 3.1.7.2 and the Qualifying Volume of customer's other telecommunications services.

CANCELLED

APR 1 1 2002 4hRS 55.1 Service Commission MISSOURI

FILED

OCT 01 2000

By TA 198-573 Public Service Commission

MISSOURI CERTAIN MATERIAL PREVIOUSLY LOCATED ON THIS PAGE CANNOW BE FOUND ON PAGE NO. 55.1.1 Public Service Commission

ISSUED : September 1, 2000

EFFECTIVE: October 1, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 55.1 CANCELS 1ST REVISED PAGE NO. 55.1

T

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

3.1 Local Exchange Service (Cont'd) 3.1.7 Term Plan

Misseum 1-400 Nice Commis REC'D MAR 2 1 2000

3.1.7.1 On-Net Term Plan

The On-Net Term Plan is a term plan, in lieu of all other tariffed Local Exchange Service term plans for customers who subscribe to MCI WorldCom On-Net interstate service. Customers who enroll in this term plan must have their facilities based MCI Local Exchange service usage billed to a single On-Net invoice.

Customers subscribing to this term plan will receive discounts as set forth below, in lieu of all other tariffed term plans. The discounts will be applied to the following recurring charges: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DiD Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, and Local ISDN-PRI T-1 Charge as well as local service usage charges.

The Volume Commitment will be determined based upon the Qualifying Volume received by the Company, or any affiliate of the Company, from customer's facilities based Local Exchange service, as defined in Section 3.1.7.2 and the Qualifying Volume of customer's other telecommunications services.

Discounts: Customers will receive the following discounts applied to Eligible Volume charges and usage.

	Term Commitment/Discount				
Volume Commitment	<u>1 Year</u>	<u>2 Years</u>	<u>3 Years</u>	<u>4 Years</u>	<u>5 Years</u>
\$100 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$250 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$500 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$1,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$2,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$3,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$4,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$5,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$7,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$120,000 Jannual	14.0%	17.0%	20.0%	23.0%	26.0%
\$180,000 /annuai	14.0%	17.0%	20.0%	23.0%	26.0%
\$300,000 /annual	16.0%	19.0%	22.0%	25.0%	28.0%
\$600,000 /annual	18.0%	21.0%	24.0%	27.0%	30.0%
\$900,000/annual	21.0%	24.0%	27.0%	30.0%	30.0%
\$1.2M/annual	24.0%	27.0%	30.0%	30.0%	30.0%
\$1.8M/annual	27.0%	30.0%	30.0%	30.0%	30.0%
2.4M/annual	30.0%	30.0%	30.0%	30.0%	30.0%

CANCELLED

OCT 012000 By 3ra RPH55, 1 Public Service Commission MISSOURI

ISSUED : March 21, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 FILED APR 2 0 2000



Missouri Public Sorvice Commission

REC'D FEB 29 2000

MCImetro ACCESS TRANSMISSION SERVICES, LLC MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 55.1 CANCELS ORIGINAL PAGE NO. 55.1

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

3.1.7 <u>Term Plan</u>

3.1.7.1 On-Net Term Plan

The On-Net Term Plan is a term plan, in lieu of all other tariffed Local Exchange Service term plans for customers who subscribe to MCI WorldCom On-Net interstate service. Customers who enroll in this term plan must have their facilities based MCI Local Exchange service usage billed to a single On-Net invoice.

Customers subscribing to this term plan will receive discounts as set forth below, in lieu of all other tariffed term plans. The discounts will be applied to the following recurring charges: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, Local ISDN-PRI Local Service Configuration Charge, Local ISDN-PRI B Channel Service Charge as well as local service usage charges.

The Volume Commitment will be determined based upon the Qualifying Volume received by the Company, or any affiliate of the Company, from customer's facilities based Local Exchange service, as defined in Section 3.1.7.2 and the Qualifying Volume of customer's other telecommunications services.

Discounts: Customers will receive the following discounts applied to Eligible Volume charges and usage.

	Term Commitment/Discount				
Volume Commitment	<u>1 Year</u>	2 Years	<u>3 Years</u>	4 Years	<u>5 Years</u>
\$100 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$250 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$500 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$1,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$2,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$3,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$4,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$5,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$7,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$120,000 /annual	14.0%	17.0%	20.0%	23.0%	26.0%
\$180,000 /annual	14.0%	17.0%	20.0%	23.0%	26.0%
\$300,000 /annual	16.0%	19.0%	22.0%	25.0%	28.0%
\$600,000 /annual	18.0%	21.0%	24.0%	27.0%	30.0%
\$900,000/anriual	21.0%	24.0%	27.0%	30.0%	30.0%
\$1.2M/annual	24.0%	27.0%	30.0%	30.0%	30.0%
\$1.8M/annual	27.0%	30.0%	30.0%	30.0%	30.0%
2.4M/annual	30.0%	30.0%	30.0%	30.0%	30.0%

CANCELLED

APR 2 0 2000 By 2^{MD} RP 55.1 Public Service Commission MISSOURI



FILED MAR 3 0 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 T/D T/D

Ν

Ν

N

Ν

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.1

Missouri Public Sorvice Commission

RFCD MAY 28 1999

N

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions (Cont'd)</u>

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan

3.1.7.1 On-Net Term Plan

The On-Net Term Plan is a term plan, in lieu of all other tariffed Local Exchange Service term plans. Customers who subscribe to facilities based Local Exchange Service offered by the Company through the On-Net Term Plan are subject to the terms and conditions set forth in MCI Telecommunications F.C.C Tariff No. 1, section C-3.464, and this tariff. Customers who enroll in this term plan must have their facilities based MCI Local Exchange service usage billed to a single On-Net invoice.

Customers subscribing to this term plan will receive discounts as set forth below, in lieu of all other tariffed term plans. The discounts will be applied to the following recurring charges: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, Local ISDN-PRI Local Service Configuration Charge, Local ISDN-PRI B Channel Service Charge as well as local service usage charges.

<u>Discounts</u>: Customers will receive the following discounts applied to Eligible Volume charges and usage.

	Term Commitment/Discount				
Volume Commitment	<u>1 Year</u>	<u>2 Years</u>	<u>3 Years</u>	4 Years	<u>5 Years</u>
6400 h	£ 0.04				4 7 - 1 /
\$100 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$250 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$500 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$1,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$2,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$3,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$4,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$5,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$7,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$120,000 /annual	14.0%	17.0%	20.0%	23.0%	26.0%
\$180,000 /annual	14.0%	17.0%	20.0%	23.0%	26.0%
\$300,000 /annuai	16.0%	19.0%	22.0%	25.0%	28.0%
\$600,000 /annua)	18.0%	21.0%	24.0%	27.0%	30.0%



FILED JUK 27 1999

EFFECTIVE: June 27, 1999

By 1^{5±} RP55.1 Public Service Commission MISSOURI

ISSUED : May 28, 1999

CANCELLED

MAR 3 0 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlantø, GA 30328

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan

3.1.7.1 On-Net Term Plan

Discounts: Customers will receive the following discounts applied to Eligible Volume charges and

Term Commitment/Discount 1/					
Volume Commitment	<u>1 Year</u>	2 Years	<u>3 Years</u>	<u>4 Years</u>	<u>5 Years</u>
\$100 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$250 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$500 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$1,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$2,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$3,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$4,000 /month	10.0%	13.0%	16.0%	19.0%	22.0%
\$5,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$7,000 /month	12.0%	15.0%	18.0%	21.0%	24.0%
\$1,200 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$3,000 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$6,000 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$12,000 /annual	10.0%	13.0%	16.0%	19.0%	22.0%
\$24,000 /annual	10.0%	13.0%	16.0%	19.0%	22.0%
\$36,000 /annual	10.0%	13.0%	16.0%	19.0%	22.0%
\$48,000 /annual	10.0%	13.0%	16.0%	19.0%	22.0%
\$60,000 /annual	12.0%	15.0%	18.0%	21.0%	24.0%
\$84,000 /annual	12.0%	15.0%	18.0%	21.0%	24.0%
\$120,000 /annual	14.0%	17.0%	20.0%	23.0%	26.0%
\$180,000 /annual	14.0%	17.0%	20.0%	23.0%	26.0%
\$300,000 /annual	16.0%	19.0%	22.0%	25.0%	28.0%
\$600,000 /annual	18.0%	21.0%	24.0%	27.0%	30.0%
\$900,000/annual	21.0%	24.0%	27.0%	30.0%	30.0%
\$1.2M/annual	24.0%	27.0%	30.0%	30.0%	30.0%
\$1.8M/annual	27.0%	30.0%	30.0%	30.0%	30.0%
2.4M/annual	30.0%	30.0%	30.0%	30.0%	30.0%

CANCELLED AÙG 2 0 1939 Public Service Commission MISSOURI

Beginning October 1, 2000, monthly volume commitments will no longer be available to new subscribers.

ISSUED : September 1, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

FILED

MISSOURI Public Service Commission

EFFECTIVE: October 1, 2000 2000

CANCELLED September 1, 2008 **Missouri Public** Service Commission

1/

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.1.1

RECEIVED

SEP 01 2000

MISSOURI Public Service Commission

М

Ν

Μ

N
LOCAL EXCHANGE SERVICE

Missouri Public Service Commission

REC'D MAR 11 2002

<u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

3.1.7 Term Plan (Cont'd)

3.1.7.2 Local On-Net Term Plan

The Local On-Net Term Plan is a term plan, in lieu of all other tariffed term plans, available to facilities based standalone local exchange service customers. Customers who subscribe to Local On-Net Term Plan are subject to the following conditions:

Definition of Terms:

Qualifying Volume is the customer's total monthly recurring charges and usage of the following, after the application of promotional and other discounts: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, Standard and Optional Features, Local ISDN-PRI T-1 Charge, and Local ISDN-PRI Optional Features as well as local service usage charges. Charges for the following are not included as Qualifying Volume and are not calculated in satisfaction of the Local On-Net Term Plan volume commitment: Non-Recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, and Local ISDN-PRI; non-Recurring for Local ISDN-PRI Optional Features; Directory Assistance usage; non-recurring and recurring charges for Directory Listings; Operator Assisted Surcharges (including Busy Line Verification and Busy Line Interrupt), and taxes.

Eligible Volume is the customer's total monthly recurring charges and usage of the following, after the application of promotional and other discounts: recurring charges for Ν Standard and Optional Features; recurring charges for Local ISDN-PRI Optional Features Local Line Charge, Local Trunk-Ν Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, Local ISDN-PRI T-1 Charge as well as local service usage charges. Charges for the following are not included as Eligible Volume and will not receive Local On-Net Term Plan volume discounts: Non-Recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, and Local ISDN-PRI; non-recurring charges for Standard and Optional Features; non-recurring charges for Local ISDN-PRI Optional Features; Directory Assistance usage; non-recurring charges for Directory Listings; Operator Assisted Surcharges (including Busy Line Verification and Busy Line Interrupt), and taxes.

CERTAIN MATERIAL FROM THIS PAGE HAS BEEN MOVED TO PAGE 55.2.1

Missouri Public Service Commission

FILED APR 11 2002

Issued: March 11, 2002

Effective: April 11, 2002

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.2

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont'd)

3.1.7.2 Local On-Net Term Plan

Missouri Public Sorvico Commissión

RECTO MAY 28 1999

Ν

The Local On-Net Term Plan is a term plan, in lieu of all other tariffed term plans, available to facilities based standalone local exchange service customers. Customers who subscribe to Local On-Net Term Plan are subject to the following conditions:

Definition of Terms:

Qualifying Volume is the customer's total monthly recurring charges and usage of the following, after the application of promotional and other discounts: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, Standard and Optional Features, Local ISDN-PRI Local Service Configuration Charge, Local ISDN-PRI B Channel Service Charge, and Local ISDN-PRI Optional Features as well as local service usage charges. Charges for the following are not included as Qualifying Volume and are not

calculated in satisfaction of the Local On-Net Term Plan volume commitment: Non-Recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DiD, Local Trunk-2 Way Direct, and Local ISDN-PRI; non-Recurring for Local (SDN-PRI Optional Features; Directory Assistance usage; non-recurring and recurring charges for Directory Listings; Operator Assisted Surcharges (including Busy Line Verification and Busy Line Interrupt), and taxes.

Eligible Volume is the customer's total monthly recurring charges and usage of the following, after the application of promotional and other discounts: Local Line Charge, Local Trunk-Basic Charge, Local Trunk-DID Charge, Local Trunk-2 Way Direct Charge, DID/2 Way Direct Number Charges, Local ISDN-PRI Local Service Configuration Charge, Local ISDN-PRI B Channel Service Charge as well as local service usage charges. Charges for the following are not included as Eligible Volume and will not receive Local On-Net Term Plan volume discounts: Non-Recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, and Local ISDN-PRI; recurring and non-recurring charges for Standard and Optional Features; non-recurring and recurring charges for Local ISDN-PRI Optional Features; Directory Assistance usage; non-recurring and recurring charges for Directory Listings; Operator Assisted Surcharges (including Busy Line Verification and Busy Line Interrupt), and taxes.

<u>Term Commitment and Renewal Options</u>: A customer must commit to service for a term of either one, two, three, four, or five years. The term of service will commence no earlier than the first day of the next billing month in which the customer subscribes to the plan. A plan will automatically renew for an equivalent term and volume commitment upon expiration of its term unless the customer provides written notification to cancel the Local On-Net Term Plan, which must be received by the Company no more than 30 days after expiration of the existing term. If the customer cancels the existing term plan within 30 days after expiration of the existing term, the customer will receive the discounts for which the customer qualifies during the 30day period following the expiration of the existing term of service.

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: June 27, 1999 Missouri Public Somico Commission

FILED JUN 27 1999

Ν

CANCELLED

APR 2 0 2000 JST RP 55. 2 Public Service Commission MISSOURI

ISSUED : May 28 , 1999

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.2.1 Cancels Original Page No. 55.2.1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008



2...

LOCAL EXCHANGE SERVICE

Missouri Public

3.1 Local Exchange Service (Contid)

Service Descriptions (Cont'd)

- 3.1.7 <u>Term Plan (Cont'd)</u>
 - 3.1.7.2 Local On-Net Term Plan(Cont'd)

REC'D MAR 1 1 2002

Service Commission

Term Commitment and Renewal Options: A customer must commit to service for a term of either one, two, three, four, or five years. The term of service will commence no earlier than the first day of the next billing month in which the customer subscribes to the plan. A plan will automatically renew for an equivalent term and volume commitment upon expiration of its term unless the customer provides written notification to cancel the Local On-Net Term Plan, which must be received by the Company no more than 30 days after expiration of the existing term. If the customer cancels the existing term plan within 30 days after expiration of the existing term, the customer will receive the discounts for which the customer qualifies during the 30-day period following the expiration of the existing term of service.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE 55.2

Missouri Public

. . . .

FILED APR 1 1 2002

Service Commission

Effective: April 11, 2002

Issued: March 11, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.3 Cancels 2nd Revised Page No. 55.3

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont'd)

3.1.7.2 Local On-Net Term Plan (Cont'd)

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 55.3 CANCELS 1ST REVISED PAGE NO. 55.3

RECEIVED

SEP 01 2000

N

Ν

MISSOURI Public Service Commission Volume commitment: A customer may elect a Local On-Net Term Plan monthly volume

CANCELLED AUG 2 0 1999 TA - 8-575 Cy TA 18-575 Public Service Commission MISSOURI

MISSOURI Public Service Commission

other discounts: \$100; \$250; \$500; \$1,000; \$2,000; \$3,000; \$4,000; \$5,000; \$7,000, or an annual volume commitment of one of the following amounts, calculated after the application of promotional and other discounts: \$1,200, \$3,000, \$6,000, \$12,000, \$24,000, \$36,000, \$48,000, \$60,000, \$84,000, \$120,000; \$180,000; \$300,000; \$600,000, \$900,000, \$1.2M, \$1.8M, and \$2.4M. The customer's volume commitment will be based upon the customer's Qualifying Volume. At any time during the term of service, a customer may elect a higher equivalent Annualized volume commitment. If a customer so elects, the term of service expiration date will not change. The new volume commitment will apply beginning in the following month for monthly commitments and as of the beginning of the commitment year in which the new volume commitment is elected for annual commitments. The discounts associated with the new commitment volume level apply to charges and usage only from the new election date forward.

commitment of one of the following amounts, calculated after the application of promotional and

Underutilization Charges: If at the end of any year of the term of service, a customer fails to satisfy its annual volume commitment, the customer must pay the difference between the customer's actual Qualifying Volume in that year and the annual volume commitment. If at the end of any month of the term of service, a customer fails to satisfy its monthly volume commitment, the customer must pay the difference between the customer's actual Qualifying Volume in that month and the monthly volume commitment.

Early Termination Charges:

Cancellation of Discontinuance without Liability: If: (I) the customer's use of Local Exchange Service under a Local On-Net Term Plan equals or exceeds the Customer's equivalent annualized minimum volume commitment or monthly volume commitment and (ii) at the time of termination the customer is enrolled in a new Local On-Net Term Plan with a volume commitment which equals or exceeds the Customer's existing volume commitment, the customer may terminate service under Term Plan without liability as follows: (I) the customer may terminate service at any time during the last three months of the term of service if the customer's Local On-Net Term Plan's term commitment is one year: or (ii) the customer may terminate service at any time during the last six months of the term of service if the customer's new Local On-Net Term Plan's term commitment is equal to or greater than two years.

Cancellation or Discontinuance with Liability: Discontinuance of all services furnished under the Local On-Net Term Plan prior to the expiration of the committed term of service constitutes discontinuance of the plan and the customer will be billed and required to pay an early termination charge equal to the Underutilization Charge for the year of termination plus 50% of all of each annual or monthly volume commitment for each year remaining in the unfulfilled term of service.

EFFECTIVE: October 1, 2000

ISSUED : September 1, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

CANCELLED September 1, 2008 **Missouri Public** Service Commission

FILED OCT 01 2000

Missouri Public Sorvice Commission

REC'D FEB 29 2000

MCImetro ACCESS TRANSMISSION SERVICES, LLC

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 55.3 CANCELS ORIGINAL PAGE NO. 55.3

> T T

LOCAL EXCHANGE SERVICE

- Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont'd)

3.1.7.2 Local On-Net Term Plan (Cont'd)

Volume commitment: A customer may elect a Local On-Net Term Plan monthly volume commitment of one of the following amounts, calculated after the application of promotional and other discounts: \$100; \$250; \$500; \$1,000; \$2,000; \$3,000; \$4,000; \$5,000; \$7,000, or an annual volume commitment of one of the following amounts, calculated after the application of promotional and other discounts: \$120,000; \$180,000: \$300,000; \$600,000, \$900,000, \$1.2M, \$1.8M, and \$2.4M. The customer's volume commitment will be based upon the customer's Qualifying Volume. At any time during the term of service, a customer may elect a higher equivalent Annualized volume commitment. If a customer so elects, the term of service expiration date will not change. The new volume commitment will apply beginning in the following month for monthly commitments and as of the beginning of the commitment year in which the new volume commitment volume level apply to charges and usage only from the new election date forward.

<u>Underutilization Charges</u>: If at the end of any year of the term of service, a customer fails to satisfy its annual volume commitment, the customer must pay the difference between the customer's actual Qualifying Volume in that year and the annual volume commitment. If at the end of any month of the term of service, a customer fails to satisfy its monthly volume commitment, the customer must pay the difference between the customer's actual Qualifying Volume in that monthly volume commitment.

Early Termination Charges:

Cancellation of Discontinuance without Liability: If: (I) the customer's use of Local Exchange Service under a Local On-Net Term Plan equals or exceeds the Customer's equivalent annualized minimum volume commitment or monthly volume commitment and (ii) at the time of termination the customer is enrolled in a new Local On-Net Term Plan with a volume commitment which equals or exceeds the Customer's existing volume commitment, the customer may terminate service under Term Plan without liability as follows: (I) the customer may terminate service at any time during the last three months of the term of service if the customer's Local On-Net Term Plan's term commitment is one year: or (ii) the customer may terminate service at any time during the last six months of the term of service if the customer's new Local On-Net Term Plan's term commitment is equal to or greater than two years.

<u>Cancellation or Discontinuance with Liability</u>: Discontinuance of all services furnished under the Local On-Net Term Plan prior to the expiration of the committed term of service constitutes discontinuance of the plan and the customer will be billed and required to pay an early termination charge equal to the Underutilization Charge for the year of termination plus 50% of all of each annual or monthly volume commitment for each year remaining in the unfulfilled term of service.

CANCELLED

OCT 012000 By 200 RPH55.3 Public Service Commission MISSOURI

Hissouri Publig

FILED MAR 3 0 2000

ISSUED : February 29, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: March 30, 2000



LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont'd)

3.1.7.2 Local On-Net Term Plan (Cont'd)

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.3

Missouri Public. Sorvico Commissión

REC'D MAY 28 1999

Ν

<u>Volume commitment</u>: A customer may elect a Local On-Net Term Plan monthly volume commitment of one of the following amounts, calculated after the application of promotional and other discounts: \$100; \$250; \$500; \$1,000; \$2,000; \$3,000; \$4,000; \$5,000; \$7,000, or an annual volume commitment of one of the following amounts, calculated after the application of promotional and other discounts: \$120,000; \$180,000: \$300,000; \$600,000. The customer's volume commitment will be based upon the customer's Qualifying Volume. At any time during the term of service, a customer may elect a higher equivalent Annualized volume commitment. If a customer so elects, the term of service expiration date will not change. The new volume commitment will apply beginning in the following month for monthly commitments and as of the beginning of the commitment year in which the new volume commitment is elected for annual commitments. The discounts associated with the new commitment volume level apply to charges and usage only from the new election date forward.

<u>Underutilization Charges</u>: If at the end of any year of the term of service, a customer fails to satisfy its annual volume commitment, the customer must pay the difference between the customer's actual Qualifying Volume in that year and the annual volume commitment. If at the end of any month of the term of service, a customer fails to satisfy its monthly volume commitment, the customer must pay the difference between the customer's actual Qualifying Volume in that monthly volume commitment.

Early Termination Charges:

CANCELLED

MAR 3 0 2000 By 15t R.P 55.3 Public Service Commission MISSOURI <u>Cancellation of Discontinuance without Liability</u>: If: (I) the customer's use of Local Exchange Service under a Local On-Net Term Plan equals or exceeds the Customer's equivalent annualized minimum volume commitment or monthly volume commitment and (ii) at the time of termination the customer is enrolled in a new Local On-Net Term Plan with a volume commitment which equals or exceeds the Customer's existing volume commitment, the customer may terminate service under Term Plan without liability as follows: (I) the customer may terminate service at any time during the last three months of the term of service if the customer's Local On-Net Term Plan's term commitment is one year: or (ii) the customer may terminate service at any time during the last six months of the term of service if the customer's new Local On-Net Term Plan's term commitment is equal to or greater than two years.

<u>Cancellation or Discontinuance with Liability</u>: Discontinuance of all services furnished under the Local On-Net Term Plan prior to the expiration of the committed term of service constitutes discontinuance of the plan and the customer will be billed and required to pay an early termination charge equal to the Underutilization Charge for the year of termination plus 50% of all of each annual or monthly volume commitment for each year remaining in the unfulfilled term of service.



ISSUED : May 28 , 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: June 27, 1999

Mipsouri Public Sorvien Commission

FILED JUN 27 1999

Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.4 Cancels 2nd Revised Page No. 55.4

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 2ND REVISED PAGE NO. 55.4 CANCELS 1ST REVISED PAGE NO. 55.4

LOCAL EXCHANGE SERVICE

- 3. Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont'd)
 - 3.1.7.2 Local On-Net Term Plan (Cont'd)

RECEIVED

SEP 01 2000

MISSOURI Public Service Commission

Ν

Ν

Ν

Discounts: Customers will receive the following discounts applied to Eligible Volume charges and usage.

	Term Commitment/Discount 1/				
Volume Commitment	<u>1 year</u>	2 years	<u>3 years</u>	<u>4 years</u>	<u>5 years</u>
\$ 100 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$250 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$500 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$1,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$2,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$3,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$4,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$5,000 /month	7.0%	10.0%	13.0%	16.0%	19.0%
\$7,000 /month	7.0%	10.0%	13.0%	16.0 %	19.0%
\$1,200 /annual	0.0%	3.0%	6.0%	9.0%	12.0%
\$3,000 /annual	0.0%	3.0%	6.0%	9.0%	12.0%
\$6,000 /annual	0.0%	3.0%	6.0%	9.0%	12.0%
\$12,000 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$24,000 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$36,000 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$48,000 /annual	5.0%	8.0%	11.0%	14.0%	17.0%
\$60,000 /annual	7.0%	10.0%	13.0%	16.0%	19.0%
\$84,000 /annual	7.0%	10.0%	13.0%	16.0 %	19.0%
\$120,000 /annual	9.0%	12.0%	15.0%	18.0 %	21.0%
\$180,000 /annua!	9.0%	12.0%	15.0%	18.0%	21.0%
\$300,000 /annual	11.0%	14.0%	17.0%	20.0%	23.0%
\$600,000 /annual	13.0%	16.0%	19.0%	22.0%	25.0%
\$900,000/annual	16.0%	19.0%	22.0%	25.0%	25.0%
\$1.2M/annual	19.0%	22.0%	25.0%	25.0%	25.0%
\$1.8M/annual	22.0%	25.0%	25.0%	25.0%	25.0%
\$2.4M/annual	25.0%	25.0%	25.0%	25.0%	25.0%



OCT 01 2000

FILED

1/

Beginning October 1, 2000, monthly volume commitments will no longer be available to new subscribers.

MISSOURI New York Commission

EFFECTIVE: October 1, 2000

ISSUED : September 1, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Bervice Commission

MCImetro ACCESS TRANSMISSION SERVICES, LLC

I

REC'D FEB 29 2000

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 55.4 CANCELS ORIGINAL PAGE NO. 55.4

LOCAL EXCHANGE SERVICE

- 3. Service Descriptions (Cont'd)
 - 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont'd)

3.1.7.2 Local On-Net Term Plan (Cont'd)

Discounts: Customers will receive the following discounts applied to Eligible Volume charges and usage.

	Term Commitment/Discount				
Volume Commitment	1 year	2 years	<u>3 years</u>	4 years	<u>5 years</u>
\$ 100 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$250 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$500 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$1,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$2,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$3,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$4,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$5,000 /month	7.0%	10.0%	13.0%	16.0%	19.0%
\$7,000 /month	7.0%	10. 0%	13.0%	16.0 %	19.0%
\$120,000 /annual	9.0%	12.0%	15.0%	18.0 %	21.0%
\$180,000 /annual	9.0%	12.0%	15.0%	18.0%	21.0%
\$300,000 /annual	11.0%	14.0%	17.0%	20.0%	23.0%
\$600,000 /annual	13.0%	16.0%	19.0%	22.0%	25.0%
\$900,000/annual	16.0%	19.0%	22.0%	25.0%	25.0%
\$1.2M/annual	19.0%	22.0%	25.0%	25.0%	25.0%
\$1.8M/annual	22.0%	25.0%	25.0%	25.0%	25.0%
\$2.4M/annual	25.0%	25.0%	25.0%	25.0%	25.0%

CANCELLED

Ν

Ν

OCT 0 1 2000 By 2 NO RPH 55.4 Public Service Commission MISSOURI

Sorvice Commission

FILED MAR 3 0 2000

ISSUED : February 29, 2000

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: March 30, 2000

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.4

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 Term Plan (Cont d)

3.1.7.2 Local On-Net Term Plan (Cont'd)

Missouri Public Sorvigo Commission

RECT) WAY 28 1999

Discounts: Customers will receive the following discounts applied to Eligible Volume charges and usage.

Term Commitment/Discount					
Volume Commitment	<u>1 year</u>	<u>2 years</u>	<u>3 years</u>	<u>4 years</u>	<u>5 years</u>
\$ 100 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$250 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$500 /month	0.0%	3.0%	6.0%	9.0%	12.0%
\$1,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$2,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$3,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$4,000 /month	5.0%	8.0%	11.0%	14.0%	17.0%
\$5,000 /month	7.0%	10.0%	13.0%	16.0%	19.0%
\$7,000 /month	7.0%	10.0%	13.0%	16.0 %	19.0%
\$120,000 /annual	9.0%	12.0%	15.0%	18.0 %	21.0%
\$180,000 /annual	9.0%	12.0%	15.0%	18.0%	21.0%
\$300,000 /annual	11.0%	14.0%	17.0%	20.0%	23.0%
\$600,000 /annual	13.0%	16.0%	19.0%	22.0%	25.0%

CANCELLED

MAR 3 0 2000 By 15t RP 55.4 Public Service Commission MISSOURI

Missouri Public Borvico Commission

FILED JUN 27 1999

EFFECTIVE: June 27, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ISSUED : May 28, 1999

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

1

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

LOCAL EXCHANGE SERVICE



<u>Service Descriptions (Cont'd)</u>

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7 <u>Term Plan (Cont'd)</u>
 - 3.1.7.2 Local On-Net Term Plan (Cont'd)

Service Commission

Missouri Public

RECD JUL 12 2002

Agent Program:

<u>Eligibility:</u> To be eligible for this program, the Customer: must subscribe to this-program through a Company-designated agent; may not receive any discounts or the benefits of any promotional offering except for the Local Satisfaction Guarantee Promotion, Local Nationwide One Promotion and the Local Install Waiver Promotion; may not receive service under any other term plan;

<u>Term Commitment and Renewal Options</u>: The Customer must commit to service for a term of one years (Initial Term). The term of service will automatically renew on a month-to-month basis (Extended Term) upon expiration of its Initial Term and Extended Term(s), unless either the Customer or the Company provides written notification to terminate the term, which must be received by the other party no more than 60 days prior to the expiration of the existing term.

<u>Credits:</u> During each monthly period of the term of service, the Customer will receive a credit equal to 10 percent of the standard tariffed monthly recurring charges (excluding features) and usage rates for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct and ISDN-PRI service.

<u>Optional Feature Package:</u> The Customer will be charged \$10.00 monthly recurring charge for the following feature package:

Includes Standard Features plus Call Waiting/Cancel Call Waiting Call Transfer or 3 Way Calling (mutually exclusive) Call Forward Busy Call Forward No Answer Caller Id-Name and Number Calling Party Number Delivery Speed Dialing - 8

Classifications, Practices and Regulations:

<u>Termination</u>: Either the Customer or the Company may terminate this program for Cause. For purposes of this program only: (i) as to payment of invoices, "Cause" shall mean the Customer's failure to pay any invoice within 30 days after the date of the invoice; and, (ii) for all other matters, "Cause" shall mean a breach by the other party of any material provision of this program, provided that written notice of the breach has been given to the breaching party, and the breach has not been cured within 30 days after delivery of such notice.

ALL MATERIAL ON THIS PAGE IS NEW .

Missouri Public Service Commission

FILED AUG 12 2002

Effective: August 12, 2002

Issued: July 12, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

•

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

5. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange Service (Cont'd)

- 3.1.7 Term Plan (Cont'd)
 - 3.1.7.3 <u>Save Program</u>

<u>Definitions</u>:

An existing customer is a customer who is receiving service under this tariff for at least 12 months prior to enrollment in this program.

Eligibility: To be eligible for this program, the customer: - must be an existing facilities based business customer;

- must be in good standing;
- must bill at least \$50 in current charges in any one month prior to enrollment in this program;
- must represent to the Company's satisfaction, as determined with the Company's sole discretion, that it will enter into another carrier's offer in absence of any further inducement from the Company to remain subscribed to Company services.

<u>Benefits</u>: After enrollment in this program, the Company will provide a one-time credit, not to exceed \$5,000, equal to the net monthly recurring charges billed during the first full invoice after the Company receives notification from the Customer that it wishes to discontinue all service (Program Credit).

Other Conditions:

A Customer who discontinues all service within three months after receiving the Program Credit will be billed and required to repay the Program Credit.

ALL MATERIAL ON THIS PAGE IS NEW.

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: April 1, 2003



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 5th Revised Page No. 55.7 Cancels 4th Revised Page No. 55.7

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

Missouri PSC Tariff No. 1 4th Revised Page No. 55.7 Cancels 3rd Revised Page No. 55.7

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange (Cont'd)

3.1.7 <u>Term Plan (Cont'd)</u>

3.1.7.4 <u>Local Plus Program¹</u>

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Special Customer Arrangement (SCA) Guide Type 1, 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15 and 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com
- must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

<u>Features</u>:

The following optional features are available for Local Line Service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in 3.1. except monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Vanity Number

Features and applicable feature charges for Local Trunk Basic, Local Trunk DID, Local Trunk 2 Way Direct and ISDN-PRI are available as described in Sections 3.1.3.1, 3.1.3.2, 3.1.3.3, 3.1.5.3 Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for local Trunk-DID and Local Trunk 2-Way Direct Service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Sections 3.1.3.3.2.1, 3.1.3.2.3.2.

¹Effective January 16, 2004 this service will no longer be available to new customers

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective March 8, 2007



MCImetro Access Transmission Services, LLC Missouri PSC Tariff No. 1 3rd Revised Page No. 55.7 Cancels 2nd Revised Page No. 55.7

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange (Cont'd)

- 3.1.7 Term Plan (Cont'd)
 - 3.1.7.4 <u>Local Plus Program¹</u>

Eligibility: To be eligible for this program, customers:

must subscribe to service under Special Customer Arrangement (SCA) Guide Type 1, 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15 and 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com

must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in 3.1. except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features

Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Features and applicable feature charges for Local Trunk Basic, Local Trunk DID, Local Trunk 2 Way Direct and ISDN-PRI are available as described in Sections 3.1.3.1, 3.1.3.2, 3.1.3.3, 3.1.5.3 Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for local Trunk-DID and Local Trunk 2-Way Direct Service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Sections 3.1.3.3.2.1, 3.1.3.2.3.2.

MATERIAL ON THIS PAGE WAS MOVED TO PAGE NO. 55.8

¹Effective January 16, 2004 this service will no longer be available to new customers

Issued April 13, 2004

Cancelled March 8, 2007 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Suite 1100 205 N. Michigan Ave. Chicago, IL 60601 Effective May 14, 2004



MISSOURI P.S.C. TARIFF NO. 1 2nd Revised Page No. 55.7 Cancels 1st Revised Page No. 55.7 Missouri Public

RFC'D DEC 1 6 2003

LOCAL EXCHANGE SERVICE Service Descriptions (Cont'd)

З.

3.1 Local Exchange Service (Cont'd)

- 3.1.7 Term Plan (Cont'd)
 - 3.1.7.4 Local Plus Program¹

Service Commission Eliqibility: To be eligible for this program, customers: must subscribe to service under Special Customer Arrangement (SCA) Guide Type 1, 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15 and 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com;

Must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

Definitions: Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

<u>Features</u>

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in section 3.1 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 all Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Features and applicable feature charges for Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct and ISDN-PRI are available as described in Sections 3.1.3.1, 3.1.3.2, 3.1.3.3, 3.1.5.3. Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID numbers as specified in Section 3.1.3.3.2.1, 3.1.3.2.3.2. Section

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (I Plan 1: Plan 2:	Per line)					\$35.00 \$40.00
Local Trunks (Per trunk)	(Basic,	DID	and	2	Way	Direct)	\$40.00
Local Trunks (Per T-1)	(Basic,	DID	and	2	Way	Direct)	\$672.00

Local ISDN-PRI (Per T-1)	\$672.00
Feature Package 1	\$3.50
Feature Package 2	\$6.50

Missouri Public Service Commission

FILED JAN 80 2004

Effective January 16, 2004 this service will no longer be available to ew customers.

Issued: December 16, 2004

Effective:

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

JAN 2 0 2004



D

N Ν

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 55.7 Cancels Original Page No. 55.7

LOCAL EXCHANGE SERVICE

Missouri Public

3.1 Local Exchange Service (Cont'd)

Service Descriptions (Cont'd)

3.1.7 <u>Term Plan (Cont'd)</u>

3.1.7.4 Local Plus Program

Service Commission

RFC'D OCT 03 2003

<u>Eligibility</u>: To be eligible for this program, customers: must subscribe to service under Special Customer Arrangement (SCA) Guide Type 1, 2, 3, 4, 5, 6, 7, or 8 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com;

Must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

<u>Features:</u>

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in section 3.1 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 all Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Features and applicable feature charges for Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct and ISDN-PRI are available as described in Sections 3.1.3.1, 3.1.3.2, 3.1.3.3, 3.1.5.3. Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID numbers as specified in Section 3.1.3.3.2.1, 3.1.3.2.3.2.

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per line)		
Plan 1:	\$35,00	Т
Plan 2: a) for 1-3 lines	\$40.00	N
b) for 4 or more lines		1
(rates apply to all lines)	\$40.00	N
Local Trunks (Basic, DID and 2 Way Direct)	\$40.00	
(Per trunk)		
Local Trunks (Basic, DID and 2 Way Direct)	\$672.00	
(Per T-1)		
Local ISDN-PRI (Per T-1)	\$672.00	
Feature Package 1 Nicoouri Public	\$3.50	
Feature Package 1 Feature Package 2 Service Commission	\$6.50	
2014160 Gallingerett		

FILED NOV 03 2003

Issued: October 3, 2003

Effective: November 3, 2003



L

3.

LOCAL EXCHANGE SERVICE

Missouri Public

REC'D AUG 2 9 2003

3.1 Local Exchange Service (Cont'd)

Service Descriptions (Cont'd)

Term Plan (Cont'd) 3.1.7

3.1.7.4 Local Plus Program

Service Commission

Eligibility: To be eligible for this program, customers: must subscribe to service under Special Customer Arrangement (SCA) Guide Type 1, 2, 3, 4, 5, 6, 7, or 8 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com;

Must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

<u>Features:</u>

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in section 3.1 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this progranCELLET

Local Line Optional Features Feature Package 1 Feature Package 2 all Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

NOV 0 3 2003 25 55.7 . uplic service Commission MISSOURI

Features and applicable feature charges for Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct and ISDN-PRI are available as described in Sections 3.1.3.1, 3.1.3.2, 3.1.3.3, 3.1.5.3. Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID numbers as specified in Section 3.1.3.3.2.1, 3.1.3.2.3.2.

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.1.

Monthly Charges: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per line)	\$35.00
Local Trunks (Basic, DID and 2 Way Direct)	\$40.00
(Per trunk) •	
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$672.00
Local ISDN-PRI (Per T-1)	\$672.00
Feature Package 1	\$3.50
Feature Package 2	\$6.50
reature raewaye z	~0.J0

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 63.15.1

Issued: August 29, 2003

Enversering Cubble, 2003 Service Commission



Missouri P.S.C. Tariff No. 1 6th Revised Page No. 55.8 Cancels 5th Revised Page No. 55.8

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

÷

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

3. Service Descriptions (Cont'd)

3.1 Local Exchange (Cont'd)

3.1.7 Term Plan (Cont'd)

3.1.7.4 Local Plus Program¹ (Cont'd)

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified Section 3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per line) Plan 1	\$38.50 I
Plan 2:	\$44.001
Local Trunks (Basic, DID and 2 Way Direct) (per Trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$739.00
Local ISDN-PRI (Per T-1)	\$739.00 I
Feature Package 1	\$3.50
Feature Package 2 Remote Access to Call forwarding (Plan 2)	\$6.50 \$7.00
Remote Access to Gai forwarding (Flan 2)	ψ1.00

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Discounts:2

These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount Customer must; 1) demonstrate to the company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 155 discount and 3 or more years for up to a 15% discount.

³Effective January 16, 2004 this service will no longer be available to new customers

²Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: June 30, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange (Cont'd)

3.1.7 <u>Term Plan (Cont'd)</u>

3.1.7.4 Local Plus Program¹ (Cont'd)

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified Section 3.1.

Monthly Charges: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per line) Plan 1 Plan 2:	\$35.00 \$40.00
Local Trunks (Basic, DID and 2 Way Direct) (per Trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$672.00
Local ISDN-PRI (Per T-1) Feature Package 1 Feature Package 2 Remote Access to Call forwarding (Plan 2)	\$672.00 \$3.50 \$6.50 \$7.00

<u>Benefits</u>:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Discounts:2

These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount Customer must;1) demonstrate to the company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 155 discount and 3 or more years for up to a 15% discount.

¹Effective January 16, 2004 this service will no longer be available to new customers ²Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: February 6, 2007

CANCELLED August 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective March 8, 2007

MISSOURI P.S.C. TARIFF NO. 1 3rd Revised Page No. 55.8 Cancels 2nd Revised Page No. 55.8

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange (Cont'd)

- 3.1.7 <u>Term Plan (Cont'd)</u>
 - 3.1.7.5 Local Plus Program¹ (Cont'd)

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified in Section 3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per line) Plan 1: Plan 2:	\$35.00 \$40.00
Local Trunks (Basic, DID and 2 Way Direct)	\$40.00
(Per Trunk)	
Local Trunks (Basic, DID and 2 Way Direct)	\$672.00
(Per T-1)	
Local ISDN-PRI (Per T-1)	\$672.00
Feature Package 1	\$3.50
Feature Package 2	\$6.50

<u>Benefits</u>: Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Discounts</u>:² These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Discounts</u>: These discounts are identical to, and shall not be N in addition to, discounts applicable to the Local Plus Program I for intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, I in lieu of all other discounts, in response to competitive I marketplace conditions. To be eligible for this discount, I Customer must: 1) demonstrate to the Company's reasonable I satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year I for up to a 20% discount, or 2 or more years for up to a 25% I discount.

¹Effective January 16, 2004, this service will no longer be available to new customers. ²Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: June 30, 2006

Cancelled March 8, 2007 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: August 1, 2006



N

MCImetro Access Transmission Services, LLC

Cancels 1st Revised Page No. 55.8 Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1 Local Exchange (Cont'd)

- 3.1.7 Term Plan (Cont'd)
 - 3.1.7.4 Local Plus Program (Cont'd)

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified Section 3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per line) Plan 1 Plan 2:	\$35.00 \$40.00
Local Trunks (Basic, DID and 2 Way Direct) (per Trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$672.00
Local ISDN-PRI (Per T-1) Feature Package 1 Feature Package 2	\$672.00 \$3.50 \$6.50

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Discounts:

These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount Customer must;1) demonstrate to the company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 155 discount and 3 or more years for up to a 15% discount.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.7

Issued April 13, 2004

Cancelled August 1, 2006



Missouri Public

Service Commission

Ν

1

Ν

MISSOURI P.S.C TARIFF NO.1 1st Revised Page NO. 55.8 Cancels Original Page No. 55.8

Missouri Public

Local Exchange Service

Service Descriptions (Cont'd)

Local Plus Program¹ (cont'd) 3.1.7.5

These discounts are identical to, and shall not be in RECD then to $162003_{\rm N}$ discounts applicable to Companion Intrastate Service and Companion Interstate Service. Service Commisside

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount. Ν

D

t

Ì

D

CANCELLED 1 4 2004

Missourt Public Borvies Commission

FILED JAN BO 2004

¹Effective January 16, 2004 this service will no longer be available to new customers.

Ν Ν

Issued: December 16, 2004

Effective:

Missouri Public

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C TARIFF NO.1 Original Page No. 55.8 Local Experies 2003

Service Descriptions(Cont'd) з.

Discounts:

3.1.7.5 Local Plus Program (cont'd)

Service Commission

Customer with a one-year contract who subscribes to service under SCA Guide Type 1, 2, 3, 4, 5, 6, 7 8, and 9 and who, at the expiration of the initial term, renew their contract for a second one-year period will receive a 5 percent discount applied to Eligible Charges incurred during the second year, in lieu of all other discounts. The 5 percent discount will remain in place for each subsequent year that the customer renews service.

In lieu of the 5 percent discount above a customer with a one-year contract who subscribes to service under SCA Guide Type 2, 3, 4, 5, 6, 7, 8, 9 and who, at the expiration of the initial term, renew their contract for a second one-year period will receive a 10 percent discount applied to Eligible Charges incurred during the second year, in lieu of all other discounts. This discount is in response to competitive marketplace conditions and to be eligible for this discount the existing or prospective Customer must demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement to subscribe, or remain subscribed to the Company's exchange service. The 10 percent discount will remain in place for each subsequent year that the customer renews service. customer renews service.

The following disclaimers apply to Stand Alone Local Plus Program Line-based Service in addition to those set forth in the Service Attachment, Customer understands that use of the Service is restricted in Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THEIMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges."

3.1.7.6

Local Nationwide One Program the Local Nationwide One Program is available to existing facilities-based business customers who order a new T-1 of Digital Local Trunk-Basic, Digital Local Trunk-DID, Digital Local Trunk-2 Way Direct and/or Local ISDN-PRI service (Program Service). To be eligible for this program, an existing customer must be an existing subscriber to an On-Net Term Plan or Local On-Net Term Plan (Term Plan).

<u>Benefits</u>: Enrolled Customers may select a metered plan or flat rate plan per each location. The following monthly recurring charge (Program Charge) per T-1 will apply for the length of the customers term commitment based on the plan selected:

Padii Boxotelai		
Monthly_Recurring_(<u>Charge (Per T-1)</u>	
Local Trunk-Basic,	Local Trunk-DID and Loca	al Trunk-2 Way
Metered Plan	\$280.60	Trunk-2 Way

Flat Plan

Local ISDN-PRI Metered Plan Flat Plan

\$241.00 \$596.00

\$635.60

JAN 2 0 2004 JST RS 53.8 Ve By following program Public Service Commission MISSOURI Customers selecting the Metered Plan will receive monthly usage rates:

<u>1st Minute</u> Each Additional Minute \$0.0158 \$0.0095

The Program Charge is in lieu of the standard tariffed monthly recurring charges for Program Service, usage charges, and any other local promotions or programs.

Other Conditions: Customers enrolled in the Metered Plan sho have the than 70% of their traffic carried via inbound local and the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorting in the per call of more than ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with ten minutes are sorted with receive the benefits of this program.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 63.15.2.

FILED OCT 01 2003

Issued: August 29, 2003





MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 4th Revised Page No. 55.9 Cancels 3rd Revised Page No. 55.9

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED

Effective: September 1, 2008

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1.7.7 Local and Long Distance-Line Solutions

<u>Eliqibility</u>: To be eligible for this plan, the customer: must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance-Line Solutions as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and must subscribe to the Local and Long Distance-Line Solutions offered in the MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.

must subscribe to service under Special Customer Arrangement SCA Guide Types 2, 3, 4, 5, 6, 7, 8, 9, or 10, as described in The Guide.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified Section 3.1 in this tariff.

<u>Monthly Recurring Charges</u>: A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Line basis. The following Monthly recurring charges apply:

<u>Offering</u>	Monthly Recurring Charge (Unlimited)
Plan 1:	\$60
Plan 2:	\$60

<u>Benefits</u>: Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Features</u>: The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Sections 3.1.2.3.1 and 3.1.2.3.2, except that the Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

MATERIAL ON THIS PAGE WAS MOVED TO PAGE NO. 55.9.1.

¹Effective January 16, 2004, this service will no longer be available to new customers.

Issued: June 30, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: August 1, 2006



MISSOURI P.S.C TARIFF NO.1 2nd Revised Page No. 55.9 Cancels 3rd Revised Page No. 55.9 Local Exchange Service

3. Service Descriptions(Cont'd)

Local and Long Distance-Line Solutions 1 3.1.7.7

<u>Eligibility:</u> To be eligible for this plan, the customer: must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance-Line Solutions as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and must subscribe to the Local and Long Distance-Line Solutions offered in the MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.

must subscribe to service under Special Customer Arrangement SCA Guide Types 2, 3, 4, 5, 6, 7, 8, 9, or 10, as described in The Guide.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified in section 3.1 in this tariff.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Line basis. The following Service. The Offer is available Monthly recurring charges apply:

Offering	<u>Monthly Recurring Charge (Unlimited)</u>	
Plan 1	\$60.00	
		Л

Plan 2:

\$60.00

Benefits: Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Features</u>

<u>Features:</u> The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified sections 3.1.2.3.1 and 3.1.2.3.2, except that the Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge Feature Package 1 \$3.50 \$6.50 Feature Package 2

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Ν Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

¹Effective January 16, 2004 this service will no longer be available to new customers.

January 20, 2004

Ν

N

Effective: January 16, 2004

Cancelled August 1, 2006 Missouri Public Service Commission

Issued: December 16, 2004

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Missouri Public Service Commission

Filed

Ν D

Missouri Public

MISSOURI P.S.C TARIFF NO.1

1st Revised Page No. 55.9

Local meriange Service 2003 Local meriange Service 2003

- 3. Service Descriptions (Cont'd)
 - 3.1.7.7

Local and Long Distance-Line Solutions Service Commission Eligibility: To be eligible for this plan, the customer: must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance-Line Solutions as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and must subscribe to the Local and Long Distance-Line Solutions offered in the MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.

must subscribe to service under Special Customer Arrangement SCA Guide Types 2, 3, 4, 5, 6, 7, 8, 9, or 10 as described in The Guide.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified in section 3.1 in this tariff.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Line basis. The following Monthly recurring charges apply:

<u>Offering</u> Plan 1:		Monthly Recurring Charge (Unlimited) \$60	ጥ
Plan 2:	a) for 1-3 lines	\$60	ิ่ง
	b) for 4 or more lines:	\$60	
	rate applies to all lines)	•	N

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Features:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified sections 3.1.2.3.1 and 3.1.2.3.2, except that the Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features

Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge \$3.50 Feature Package 1 Feature Package 2 \$6.50

JAN 2 0 2004 55.9 B٧ Public Service Commission

MISSOURI

CANCELLED

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

Customers committing to a new term of service that equals or exceeds lyear for 5 percent discount, 2 years for a 10 percent discount, or 3 years for a 15 percent discount will receive applicable discounts applied to eligible Charges, in lieu of all other discounts.

Issued: October 3, 2003

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: November 3, 2003

Missouri Public Service Commission

FILED NOV 03 2003

Local Exchange Service

MISSOURI P.S.C TARIFF NO.1 Original Page No. 55.9 Missouri Public

3. Service Descriptions (Cont'd)

Local and Long Distance-Line Solutions 3.1.7.7

REC'D AUG 2 9 2003

Eligibility: To be eligible for this plan, the customer:

must designate the Company as its local exchange service Correspondent the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance-Line Solutions as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and must subscribe to the Local and Long Distance-Line Solutions offered in the MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.

must subscribe to service under Special Customer Arrangement SCA Guide Types 2, 3, 4, 5, 6, 7, 8, 9, or 10 as described in The Guide.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified in section 3.1 in this tariff.

<u>Monthly Recurring Charges:</u> A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Line basis. The following Monthly recurring charges apply:

Offering Monthly Recurring Charge (Unlimited)

Unlimited

<u>Benefits</u>:

Denerits. Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Features:</u> The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified sections 3.1.2.3.1 and 3.1.2.3.2, except that the Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

\$60

Monthly Recurring Charge

\$3.50 \$6.50 Feature Package 1 Feature Package 2

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A Customer with a one-year contract who subscribes to service under SCA Guide Types 6, 7, 8, 9, or 10 and who, at the expiration of the initial term, renew their contract for a second one-year period will receive a 5 percent discount applied to monthly recurring plan charges, monthly recurring feature and feature package charges incurred during the second year, in lieu of all other discounts. The 5 percent discount will remain in place for each subsequent year that the customer renews service.

CANCELLED

NOV 0 3 2003

Audilic Service Commission MISSOURI

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 63.15.3 FILED OBT 01 2003

Issued: August 29, 2003

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

Ν

N

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1.7.7 <u>Local and Long Distance Line Solutions¹</u>

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with Name and Number Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Vanity Number

Monthly Recurring ChargeFeature Package 1\$3.50Feature Package 2\$6.50Remote Access to Call Forwarding (Plan 2)\$7.00

<u>Discounts</u>:² These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 25% discount, and 3 years for up to a 15% discount

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 20% discount, 2 years for up to a 25 % discount, and 3 years for up to a 25% discount.

 1 Effective January 16, 2004 this service will no longer be available to new customers 2 Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective March 8, 2007
Local Exchange Service

3. Service Descriptions(Cont'd)

3.1.7.7 Local and Long Distance-Line Solutions 1

> Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge Feature Package 1 \$3.50 Feature Package 2 \$6.50

²Discounts: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in addition to, discounts applicable to the Local and long Distance-Line Solutions intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount. 1 Ν

²Effective August 1, 2006 these discounts will no longer be available to new Ν ¹Effective January 16, 2004, this service will no longer be available to new customers.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.9

Issued: June 30, 2006

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: August 1, 2006



Ν

Ν

Cancelled March 8, 2007 Missouri Public Service Commission

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: September 1, 2008

MCImetro ACCESS TRANSMISSION SERVICES, LLC MISSOURI P.S.C TARIFF NO.1 1st Revised Page No. 55.10 Cancels Original Page No. 55.10

Local Exchange Service

3. Service Descriptions(Cont'd)

3.1.7.7 Local and Long Distance-Line Solutions¹ Cont'd)

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C. Tariff No. 3., will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3., Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

For new customers who disconnect Companion Local Service under this tariff, Companion Interstate Service under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3. will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide and MCI WorldCom On-Net Service-Voice under MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3.

For new customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3., Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide.

Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts or promotions including any term plan discounts.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to Local and Long Distance-Line Solution Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

¹Effective January 16, 2004 this service will no longer be available to new customers.

N N

January 20, 2004

Effective: January 16,

MCImetro ACCESS TRANSMISSION SERVICES, LLC

Local EREPORAGE 2003

MISSOURI P.S.C TARIFF NO.1 Original Page No. 55.10

3. Service Descriptions (Cont'd)

-

Local and Long Distance-Line Solutions Contracts 3.1.7.7

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C. Tariff No. 3., will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3., Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

For new customers who disconnect Companion Local Service under this tariff, Companion Interstate Service under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3. will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide and MCI WorldCom On-Net Service-Voice under MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3.

For new customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO P.S.C Tariff No. 3., Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to WorldCom On Net Voice Services Option 1 under The Guide.

<u>Other Conditions:</u> Services under this plan are not eligible to receive the benefits of any discounts or promotions including any term plan discounts.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to Local and Long Distance-Line Solution Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges. and special construction charges.

CANCELLED

JAN 2 0 2004 By 1St RS 55.10 Public Service Commission MISSOURI

Missouri Public Service Commission FILED OGT 01 2003

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 63.15.4

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 Original Page No. 55.10.1

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

.

3.1 Local Exchange Service(Cont'd)

3.1.7.7.1 Multi-State Local Service Program I

Multi-State Local Service Program I is available to any Qualifying Customer. A Qualifying Customer is one that meets all of the following requirements:

- Customer must bill at least \$2.5 million per year in Total Service Charges.
- Customer must demonstrate prior spending of \$2 million in Total Service Charges for the past 12 months. Customer must agree to a new three-year term of at least \$2
- million annual spending on Total Service Charges.
- -Customer must subscribe to Qualifying Service in a minimum of 30 states.

The following rates/charges are available to Qualifying Multi-State Local Service Program I Customers:

Other Charges Lines - Flat Monthly Rate Lines - Measured Monthly Rate Lines - Usage Rate Per Minute Lines - EUCL/LNP Trunks - Flat Monthly Rate PRI/Full T1 - Flat Monthly		/ <u>Springfield</u> \$15.00 \$13.20 \$0.0090 \$6.50 \$24.00	Rest of <u>State</u> \$19.80 \$19.20 \$0.0090 \$6.50 N/A
Rate PRI/Full T1 - Measured	\$403.20 \$324.00	\$403.20 \$324.00	N/A N/A
Monthly Rate PRI/Full T1 - Usage Rate Per Minute	\$0.0030	\$0.0030	N/A
Initial 20 DID Blocks Additional 20 DID Blocks Initial 100 DID Blocks Additional 100 DID Blocks	\$6.00 \$6.00 \$14.10 \$14.10	\$6.00 \$6.00 \$14.10 \$14.10	N/A N/A N/A N/A
<u>Operator Assisted Charges</u> Station to Station Person to Person Third Number Billing Busy Line Verification Busy Line Interrupt Collect Calling	\$1.10 \$2.40 \$2.40 \$1.20 \$1.85 \$1.00	\$1.10 \$2.40 \$2.40 \$1.20 \$1.85 \$1.00	\$1.10 \$2.40 \$2.40 \$1.20 \$1.85 \$1.00
Other One-Time Charges Business Line Expedite Local Trunk Expedite Local T1/PRI Expedite Service Restorable	\$25.00 \$25.00 \$600.00	\$25.00 \$25.00 \$600.00	\$25.00 \$25.00 \$600.00
per Account Due Date Change	\$20.00	\$20.00	\$20.00
per Occurrence	\$10.00	\$10.00	\$10.00

ALL MATERIAL ON THIS PAGE IS NEW

Issued: September 15, 2006

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601

Effective: October 15, 2006



ALL MATERIAL ON THIS PAGE WAS REMOVED

,

З.

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 Original Page No. 55.10.2

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont)</u>

- 3.1 Local Exchange Service(Cont'd)
 - 3.1.7.7.1 <u>Multi-State Local Service Program I</u> (cont'd)

<u>Other Charges</u> Call Assistant 411 - Directory Assistance Additional Listing Alternative Call Listing Non-Listed Number Non-Published Number Message Referral	Kansas City/ St.Louis \$10.00 \$0.48 \$2.45 \$2.45 \$1.20 \$1.60 N/C	Springfield \$10.00 \$0.30 \$2.45 \$2.45 \$1.20 \$1.60 N/C	Rest of <u>State</u> \$10.00 \$0.48 \$2.45 \$2.45 \$1.20 \$1.60 N/C
<u>Install/Set-Up Charges</u> Account Set-Up Charge Record Change Charge Service Move Charge Service Add Charge Service Change Charge Trunk Group Rearrangement Call Assistant Install Additional Listing Install Non-List Number Install Non-Publish Number Install Company Restoral Customer Restoral PIC Charge Business Line	Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived	Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived	Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived Waived
Local Trunk Local T1/PRI Initial 20, 100 DID Blocks Additional 20, 100 DID Blocks	Waived Waived Waived Waived	Waived Waived Waived Waived	Waived Waived Waived Waived

"Total Service Charges" shall mean all charges, after application of all discounts and credits, incurred by Qualifying Customer for Qualifying Services provided, specifically excluding: Taxes; Charges for equipment (unless otherwise agreed upon); Verizon Wireless charges; Charges incurred for goods or services where Verizon acts as agent for Qualifying Customer in its acquisition of goods or services; Non-recurring charges; Governmental Charges; International pass-through access charges (i.e., Type 3/PTT) and charges for international access provided by Verizon (i.e., Type 1); and Other charges expressly excluded by this tariff.

"Qualifying Services" shall include, but are not limited to PRI/Full T1, Local Lines Flat Rated, Local Lines Measured.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: September 15, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: October 15, 2006



ALL MATERIAL ON THIS PAGE WAS REMOVED

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 Original Page No. 55.10.3

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont)</u>

•

- 3.1 Local Exchange Service (Cont'd)
 - 3.1.7.7.1 <u>Multi-State Local Service Program I</u> (cont'd)

In order to purchase Qualifying Services under the Multi-State Local Service Program I, Qualifying Customers must enter into an agreement with Verizon. The agreement may set forth certain terms and conditions not otherwise as described in this tariff, including but not limited to liability limitation, indemnification, termination, limitation period, service commitment and equipment required for service, and those terms and conditions shall be given full force and effect.

Qualifying Customers who enter into an agreement for the Multi-State Local Service Program will receive certain performance credits from Verizon, including credits if Verizon fails to meet certain Service Levels; such credits and service levels are to be defined by agreement between the Qualifying Customer and Verizon.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: September 15, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: October 15, 2006

ŧ



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

3. <u>Service Descriptions (Cont'd)</u>

3.1.7.8 <u>Business Advantage Program</u>¹

Eliqibility: To be eligible for this program, customers:

must subscribe to service under Special Customer Arrangement (SCA) Guide Types 6, 7, 8, 9, or 10 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com;

must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in sections 3.1.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with Name and Number Remote Call Forwarding Vanity Number

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified in section 3.1 in this tariff.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per Plan 1: Plan 2:	line):	\$40.00 \$40.00
Feature Package Feature Package		\$3.50 \$6.50

MATERIAL ON THIS PAGE WAS MOVED TO PAGE NO. 55.11.1.

Issued: June 30, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601 Effective: August 1, 2006



Local Exchange Service

3. Service Descriptions (Cont'd)

3.1.7.8 Business Advantage Program¹

Eligibility

To be eligible for this program, customers:

must subscribe to service under Special Customer Arrangement (SCA) Guide Types 6, 7, 8, 9, or 10 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com;

must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u> Eligible Charges: Monthly recurring charges for Local Line, and optional features.

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in sections 3.1.1.2.3.1 3.1.2.3.2 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

<u>Non-Recurring Charges:</u> Applicable non-recurring charges apply to services under this program as specified section 3.1 in this tariff.

Monthly Charges: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per Plan 1: Plan 2:	line)	\$40.00 \$40.00
Feature Package Feature Package	12	\$3.50 \$6.50

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years or up to a 15% discount and 3 or more years for up to a 15% discount. Ń

Other Conditions:

Other Conditions: The following disclaimers apply to Business Advantage Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges. construction charges.

¹Effective January 16, 2004 this service will no longer be available to new customers.

January 20, 2004

Ν

Ν

Issued: December 16, 2004 Cancelled August 1, 2006 Missouri Public

Service Commission

Effective: January 16, 2004



D

Ν

D

MISSOURI P.S.C TARIFF NO.1 1st Revised Page No. 55.11 Missouri Putancels Original Page No. 55.11

Local Exchange Service

- 3. Service Descriptions (Cont'd)
 - 3.1.7.8 Business Advantage Program

RECTD OCT 03 2003

Eligibility:

To be eligible for this programservice Commission

must subscribe to service under Special Customer Arrangement (SCA) Guide Types 6, 7, 8, 9, or 10 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com;

must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in sections 3.1.1.2.3.1 3.1.2.3.2 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.1 in this tariff.

Monthly Charges: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

	\$40.00	CANCELLED	Т
Plan 2: a) for 1-3 lines b) for 4~or more lines (rate applies to all lines)	\$40.00 \$40.00	JAN 2 0 2004	N N
Feature Package 1 Feature Package 2	\$3.50 \$6.50	By 2nd RS 55.11 Public Service Commission MISSOURI	

Discounts:

A customer who subscribes to Service under SCA Guide Types 6,7,8,9, or 10 and who commits to a new term of service that equals or exceeds 1 year for a 5 percent discount, 2 years for a 10 percent discount, or 3 years foe a 15 percent discount will receive applicable discounts applied to Eligible Charges, in lieu of all other discounts.

Other Conditions:

The following disclaimers apply to Business Advantage Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE

FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.





Local Exchange Service

3. Service Descriptions (Cont'd)

3.1.7.8 Business Advantage Program

<u>Eligibility:</u> To be eligible for this program, customers:

Missouri Public

REC'D AUG 2 9 2003

must subscribe to service under Special Customer Afrancement (SCAN is signifypes 6, 7, 8, 9, or 10 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.worldcom.com; must be a new facilities based business customer or an existing facilities based business customer who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u> Eligible Charges: Monthly recurring charges for Local Line, and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in sections 3.1.1.2.3.1 3.1.2.3.2 except for Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

CANCELLPH

NOV 0 3 2003

15t RS 55.11

Non-Recurring Charges: Applicable non-recurring charges apply Mission under this program as specified section 3.1 in this tariff.

Monthly Charges: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

\$40.00 \$3.50 \$6.50 Local Line (Per line) Feature Package 1 Feature Package 2

Discounts: Customer with a one-year contract who subscribes to service under SCA Type 6, 7, 8, 9 or 10 and who, at the expiration of the initial term, renew their contract for a second one-year period will receive a 5 percent discount applied to Eligible Charges incurred during the second year, in lieu of all other discounts. The 5 percent discount will remain in place for each subsequent year that the customer renews service.

Other Conditions:

The following disclaimers apply to Business Advantage Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges. construction charges.



FILED OCT 01 2003

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 63.15.5

ALL MATERIAL ON THIS PAGE WAS REMOVED

з.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

3. Service Descriptions(Cont'd)

3.1.7.8 <u>Business Advantage Program</u>¹

 $^{2}\underline{\text{Discounts}}$: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in	71
addition to, discounts applicable to the Business Advantage Program	N
Pricing plan intrastate and interatate land interations Advantage Program	
Pricing plan intrastate and interstate long distance service. A discount	1
will be provided on the monthly recurring charges, in lieu of all other	ł
discounts, in response to competitive marketplace conditions.	
To be eligible for this discount, Customer must: 1) demonstrate	1
to the Company's reasonable satisfaction that it will	1
accept another company's offer in the absence of any further	[
inducement and 2) solid in the absence of any further	
inducement, and 2) commit to a new minimum term of	1
service that equals or exceeds 1 year for up to a 20% discount,	I
or 2 or more years for up to a 25% discount.	N
	N

Other Conditions:

The following disclaimers apply to Business Advantage Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.11

 $^{2}\mbox{Effective}$ August 1, 2006 these discounts will no longer be available to new customers.

Ν

¹Effective January 16, 2004, this service will no longer be available to new customers.

Issued: June 30, 2006

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission Effective: August 1, 2006



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

З.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MISSOURI P.S.C TARIFF NO.1 lst Revised Page No. 55.12 Cancels Original Page No. 55.12

Local Exchange Service

3. Service Descriptions(Cont'd)

3.1.7.9 Local and Long Distance Service Plus Plan/Local and Long Distance Service Trunk Solution/Local and Long Distance Service-Line Solution II¹

Eligibility: To be eligible for this plan, the customer:

must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance Service Plus Plan as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and must subscribe to the Local and Long Distance Service Plus Plan offered in the MCI WorldCom Communications PSC Tariff No.3("Companion Intrastate Service").

must subscribe to service under Special Customer Arrangement SCA Types 1, 2,3,4,5,6,7,8,9,10,11,12, as described in The Guide.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified 21.1

Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per- T-1 or ISDN -PRI basis. The following Monthly recurring charges apply:

Offering

Offering	Monthly Recurring Charge
A	(per_line, trunk T-lor ISDN -PRI)
Plan 1: (Springfield Areas)	\$49.50 I
(St. Louis/Kansas City)	\$60.50 I
Plan 2:	\$66.00 I
B	\$65.00
C	\$1,540 I

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Features:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable nonrecurring and monthly recurring charges will apply to optional features as specified in 21.1, except that the Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Features and applicable feature charges for Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct and ISDN-PRI are available as described in Sections 21.2.1, 21.2.2, 21.2.3, 21.5.3. Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service.

¹Effective January 16, 2004 this service will no longer be available to new customers.

Local Exchange Service

3. Service Descriptions (Cont'd)

3.1.7.9 <u>Local and Long Distance Service Plus Plan/Local and Long Distance</u> Service Trunk Solution/Local and Long Distance Service-Line Solution II¹

Eligibility: To be eligible for this plan, the customer:

must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Local and Long Distance Service Plus Plan as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.worldcom.com; ("Companion Interstate Service") and must subscribe to the Local and Long Distance Service Plus Plan offered in the MCI WorldCom Communications PSC Tariff No.3("Companion Intrastate Service").

must subscribe to service under Special Customer Arrangement SCA Types 1, 2, 3, 4, 5, 6, 7, 8, 9, 10, 11, 12, as described in The Guide.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified 21.1

Monthly Recurring Charges:

A monthly recurring charge will apply for each Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per- T-1 or ISDN -PRI basis. The following Monthly recurring charges apply:

<u>Offering</u>	Monthly Recurring Charge (per_line, trunk T-lor ISDNPRI)
A Plan 1: (Springfield Areas) (St. Louis/Kansas City)	\$45.00 \$55.00
Plan 2:	\$60.00
B C	\$65.00 \$1,400

<u>Benefits</u>:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Features:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in 21.1, except that the Feature Package 1 and Feature Package 2 monthly recurring charges which are specified within this program.

Features and applicable feature charges for Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct and ISDN-PRI are available as described in Sections 21.2.1, 21.2.2, 21.2.3, 21.5.3. Blocks of telephone numbers can be obtained in blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service.

¹Effective January 16, 2004 this service will no longer be available to new customers.

January 20, 2004

Effective: Januarv 16, 2004

Ν

Ν

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

FIIC

ALL MATERIAL ON THIS PAGE WAS REMOVED

.

:

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. <u>Service Description (Cont'd)</u>

3.1.7.9 Local and Long distance Service Plus Plan/Local and Long Distance Service Trunk Solution/Local and long Distance Service-Line Solution II Cont'd)

> Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Caller ID with Name and Number Vanity

Monthly Recurring Charge:Feature Package 1\$3.50Feature Package 2\$6.50Remote Access to Call Forwarding (Plan 2)\$7.00DID/2 Way Direct Numbers (per each block of 20 numbers)\$6.25

These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

Discounts:² A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 years for up to a 25% discount and 3 years for up to a 25% discount.

 1 Effective January 16, 2004, this services will no longer be available to new customers. 2 Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007



Ν

MCImetro Access Transmission Services, LLC d/b/a Verizon Access Transmission Services MISSOURI P.S.C TARIFF NO.1 1st Revised Page No. 55.13 Cancels Original Page No. 55.13

Local Exchange Service

3. Service Descriptions(Cont'd)

3.1.7.9 <u>Local and Long Distance Service Plus Plan/Local and Long Distance</u> <u>Service Trunk Solution/Local and Long Distance Service-Line</u> <u>Solution II</u>¹ (cont.)

> Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring ChargeFeature Package 1\$3.50Feature Package 2\$6.50DID/2 Way Direct Numbers\$6.25(per each block of 20 numbers)

These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

²<u>Discount</u>: A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in Ν addition to, discounts applicable to the Local and Long Distance Service Plus Plan/Local and Long Distance Service Trunk Solution/Local and Long Distance Service-Line Solution II intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, 1 Ν or 2 or more years for up to a 25% discount.

MATERIAL ON THIS WAS MOVED TO PAGE NO. 55.14

¹Effective January 16, 2004, this service will no longer be available to new customers. ²Effective August 1, 2006 these discounts will no longer be available to new customers. N

Issued: June 30, 2006

Effective: August 1, 2006

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601



March 8, 2007 Missouri Public Service Commission

Cancelled

MCImetro ACCESS TRANSMISSION SERVICES, LLC

- 3. Service Descriptions(Cont'd)
 - 3.1.7.9 <u>Local and Long Distance Service Plus Plan/Local and Long Distance</u> <u>Service Trunk Solution/Local and Long Distance Service-Line</u> Solution II¹ (cont.)

Local Exchange Service

Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge

Feature Package 1\$3.50Feature Package 2\$6.50DID/2 Way Direct Numbers\$6.25(per each block of 20 numbers)

These discounts are identical tc, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carriers offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications PSC Tariff No.3 will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

¹Effective January 16, 2004 this service will no longer be available to new customers.

Issued: December 16, 2004

Cancelled August 1, 2006 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: January 16, 2004 Filed Missouri Public

N

N January 20, 2004

Service Commission

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

3. Service Descriptions (Cont'd)

3.1.7.9

Local and Long Distance Service Plus Plan/Local and Long Distance Service Trunk Solution/Local and Long Distance Service-Line Solution II¹ (cont.).

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI Communications PSC Tariff No.3 will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

The following disclaimers apply to Stand Alone Local Plus Program Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY Verizon. Verizon will install the Line-based Service from the point of the local exchange carrier's smart jack to the Customer's premises. Customer will be responsible for all in side wiring and special construction charges.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.14

¹Effective January 16, 2004 this service will no longer be available to new customers. Issued: June 30, 2006

> Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: August 1, 2006

Filed

Missouri Public

Service Commission

CANCELLED September 1, 2008 Missouri Public Service Commission

MCImetro ACCESS TRANSMISSION SERVICES, LLC

· ·

Local Exchange Service

3. Service Descriptions(Cont'd)

3.1.7.9 <u>Local and Long Distance Service Plus Plan/Local and Long Distance</u> <u>Service Trunk Solution/Local and Long Distance Service-Line</u> <u>Solution II¹ (cont.)</u>.

The following disclaimers apply to Stand Alone Local Plus Program Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THEIMMEDIATE TERMINATION OF THE SERVICE BY WORLDCOM. WorldCom will install the Line-based Service from the point of the local exchange carrier's smart jack to the Customer's premises. Customer will be responsible for all in side wiring and special construction charges.

 1 Effective January 16, 2004 this service will no longer be available to new customers.

Issued: December 16, 2004 Cancelled August 1, 2006 Missouri Public Service Commission

Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 January 20, 2004



Ν

Ν

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

--

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. <u>Service Description (Cont'd)</u>

3.1.7.10 <u>Verizon Business Services I Local Line Solution</u>

Eligibility: To be eligible for this program, customers:

must subscribe to service under Product Package Guide Types 13, 14, 15 or 16 as described in the Company's Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness/guide.com;

must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.3 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features

Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Remote Call Forwarding (Plan 2)

Monthly Recurring Charge: Feature Package 1 \$3.50 Remote Call Forwarding (Plan 2) \$7.00

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified in Section 3.1- 3.1.2.3.2.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge (unlimited) applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

<u>Benefits</u>: Upon Installation of Local Service, customers will receive unlimited local exchange service usage

Local Line (Per Line)

Plan 1: \$35.00 Plan 2: \$40.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Issued: February 6, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission

Filed Missouri Public Service Commission

Ν

Ν

Т

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.10 <u>Verizon Business Services I Local Line Solution</u>

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Types 13, 14, 15 or 16 as described in the Company's Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness/guide.com; T
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definítions:</u>

Eligible Charges: Monthly recurring charges for Local Line, and optional features.

<u>Features</u>:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.3 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features

Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding

<u>Monthly Recurring Charge</u>: Feature Package 1

\$3.50

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified in Section 3.1- 3.1.2.3.2.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge (unlimited) applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

<u>Benefits</u>: Upon Installation of Local Service, customers will receive unlimited local exchange service usage

Local Line (Per Line)

Plan 1: \$35.00 Plan 2: \$40.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

-ilea

Missouri Public

Service Commission

Effective: May 1, 2006

Cancelled March 8, 2007 Missouri Public Service Commission MCImetro ACCESS TRANSMISSION SERVICES, LLC

 ϵ

MISSOURI P.S.C. TARIFF NO. 1 1st REVISED PAGE NO. 55.15 Cancels ORIGINAL PAGE NO. 55.15

LOCAL EXCHANGE SERVICE

з. Service Description (Cont)

MCI Business Services I Local Line Solution 3.1.7.10

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Types 13, 14, 15 or 16 as described in the Company's Service Publication and Price Guide located on the Company's Internet site at www.mci.com;
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, and optional features.

<u>Features</u>: The following optional features are available for Local Line service in The following optional features. Applicable non-recurring addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.3 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features

Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding

Monthly Recurring Charge: Feature Package 1

\$3.50

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified in Section 3.1- 3.1.2.3.2.

Monthly Charges: The following flat rate monthly recurring charge (unlimited) applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Benefits: Upon Installation of Local Service, customers will receive unlimited local exchange service usage

Local Line (Per Line)

Plan 1: \$35.00 Plan 2: \$40.00 (R)

N

Ν

D

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Issued: April 13, 2004

Cancelled

May 1, 2006 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 14, 2004



MCImetro ACCESS TRANSMISSION SERVICES, LLC

l

LOCAL EXCHANGE SERVICE

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.15

Missouri Public

Service Description (Cont)

3.1.7.10 MCI Business Services I Local Line Solution

RECT DEC 1 6 2003

Eligibility: To be eligible for this program, customers: Service Commission

- must subscribe to service under Product Package Guide Types 13, 14, 15 or 16 as described in the Company's Service Publication and Price Guide located on the Company's Internet site at www.mci.com;
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.3 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features

Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

CANCELLED

Monthly Recurring Charge

Feature Package 1

MAY 1 4 2004 By15+8555.15 \$3.50 Public Service Commission MISSOURI

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified in Section 3.1- 3.1.2.3.2.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge (unlimited) applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Local Line (Per Line)

Plan 1: \$40.00 Plan 2: \$40.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

ALL MATERIAL ON THIS PAGE IS NEW.

Missouri Public Service Commission

FILED JAN 20 2004

Effective:

ssued: December 16, 2003

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

JAN 2 0 2004

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. Service Description (Cont)

3.1.7.10 <u>Verizon Business Services I Local Line Solution (Cont.)</u>

¹<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in N addition to, discounts applicable to the Verizon Business Services I Local Line Solution intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, | Customer must: 1) demonstrate to the Company's reasonable satisfaction | that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, N

Other Conditions:

The following disclaimers apply to Verizon Business Service I Local Line Solution in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY Verizon. Verizon will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be services under this plan are not eligible to receive the benefits of any discounts, or promotions including any term discounts other than those stated above.

 $^1\!\text{Effective}$ August 1, 2006 , these discounts will no longer be available to new customers.

N N

Issued: June 30, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

> **Filed** Missouri Public Service Commission

Effective: August 1, 2006

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.16 Cancels Original Page No. 55.16

Т

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.10 <u>Verizon Business Services I Local Line Solution (Cont.)</u>

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Other Conditions:

The following disclaimers apply to Verizon Business Service I Local T Line Solution in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY Verizon. Verizon will install the Line-based Service from the T point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges. Services under this plan are not eligible to receive the benefits of any discounts, or promotions including any term discounts other than those stated above.

Issued: April 1, 2006

Cancelled August 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601





MCImetro ACCESS TRANSMISSION SERVICES, LLC

į,

1

Ţ.

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.10 MCI Business Services I Local Line Solution (Cont.)

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Other Conditions:

The following disclaimers apply to MCI Business Service I Local Line Solution in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges. Services under this plan are not eligible to receive the benefits of any discounts, or promotions including any term discounts other than those stated above.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: December 16, 2003

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 January 20, 2004 Effective: January 16, 2004

> **Filed** Missouri Public Service Commission
ALL MATERIAL ON THIS PAGE WAS REMOVED

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.17 Cancels Original Page No. 55.17

Т

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.11 Verizon Business Services I Local & Long Distance

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the Verizon Business Services I Local and T Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusinewss/guide.com T ("Companion Interstate Service") and must subscribe to the Verizon Business Services I Local and Long Distance T Line Solution Service offered in the Verizon Communications, T Inc. MO PSC Tariff No. 3.
- must subscribe to service under Product Package Guide Types 13, 14, 15, or 16 as described in The Guide.
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Non-recurring Charges:</u> Applicable non-recurring charges apply to services under this program as specified in Section 3.1 - 3.1.2.3.2.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply:

Offe	erinq	Monthly Recurring Charge
		(per line, trunk, T-1 or ISDN-PRI)
А	Plan 1 Plan 2	\$40.00 \$40.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

В	\$65
С	\$1,400

Issued: April 1, 2006

Effective: May 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission

Filed Missouri Public Service Commission MCImetro ACCESS TRANSMISSION SERVICES, LLC

LOCAL EXCHANGE SERVICE

- 3. Service Description (Cont)
 - 3.1.7.11 MCI Business Services I Local & Long Distance

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the MCI Business Services I Local and Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.mci.com ("Companion Interstate Service") and must subscribe to the MCI Business Services I Local and Long Distance Line Solution Service offered in the MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.
- must subscribe to service under Product Package Guide Types 13, 14, 15, or 16 as described in The Guide.
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified in Section 3.1 - 3.1.2.3.2.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply:

<u>Off</u> e	erinq	<u>Monthly_Recurring_Charge</u>
		(per line, trunk, T-1 or ISDN-PRI)
A	Plan 1 Plan 2	\$40.00 \$40.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

В	\$65
С	\$1,400

Issued: December 16, 2003

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 January 20, 2004 Effective: January 16, 2004



ī.

i

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 5th Revised Page No. 55.18 Cancels 4th Revised Page No. 55.18

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. Service Description (Cont'd)

3.1.7.11 <u>Verizon Business Services I Local & Long Distance</u>

> Benefits: Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Features</u>: The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified within this program.

Features and applicable features charges for Local Trunk Basis, Local Trunk DID, Local Trunk 2 Way Direct and ISDN-PRI are available as described in sections 3.1.1 and 3.6. Blocks of telephone numbers can be obtained in Blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Remote Access to Call Forwarding (Plan 2)

Monthly Recurring Charge Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) DID/2 Way Direct Numbers (per each block of 20 numbers)

\$3.50 \$20.00 \$7.00 \$6.25

Discounts: 1 These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services I Local and Long Distance intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount.

 1 Effective August 1, 2006 , these discounts will no longer be available to new customers.

Issued: February 6, 2007

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: March 8, 2007



N

Ν

3. Service Description (Cont)

3.1.7.11 Verizon Business Services I Local & Long Distance

> <u>Benefits</u> Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>'eatures</u>:

Features: The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified within this program.

Features and applicable features charges for Local Trunk Basis, Local Trunk DID, Local Trunk 2 Way Direct and ISDN-PRI are available as described in sections 3.1.1 and 3.6. Blocks of telephone numbers can be obtained in Blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding

Monthly Recurring Charge Feature Package 1 \$3.50 Remote Call Forwarding \$20.00 DID/2 Way Direct Numbers (per each block of 20 numbers) \$6.25

 $\frac{^{1}\text{Discounts}}{^{1}\text{Counts}}$: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services I Local and Long Distance intrastate and interstate long distance service. A discount Ν intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount. Ň

MATERIAL ON THIS PAGE WAS MOVED TO PAGE NO. 55.18.1

¹Effective August 1, 2006 , these discounts will no longer be available to new customers.

Ν Ν

Issued: June 30, 2006

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Effective: August 1, 2006



March 8, 2007 Missouri Public Service Commission

Cancelled

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missourri P.S.C. Tariff No. 1 2nd Revised Page No. 55.18 Cancels 1st Revised Page No. 55.18

T

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

Verizon Business Services I Local & Long Distance 3.1.7.11

> <u>Benefits</u> Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Features</u>:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified with Feature Package 1 monthly recurring charges which are specified within this program.

Features and applicable features charges for Local Trunk Basis, Local Trunk DID, Local Trunk 2 Way Direct and ISDN-PRI are available as described in sections 3.1.1 and 3.6. Blocks of telephone numbers can be obtained in Blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service

<u>Local Line Optional Features</u> Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding

Monthly Recurring Charge Feature Package 1 Remote Call Forwarding DID/2 Way Direct Numbers (per each block of 20 numbers) \$3.50 \$20.00 \$6.25

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan. T

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 1, 2006



Cancelled August 1, 2006

MCImetro ACCESS TRANSMISSION SERVICES, LLC

5

MISSOURI P.S.C. TARIFF NO. 1 1st REVISED PAGE NO. 55.18 Cancels ORIGINAL PAGE NO. 55.18

LOCAL EXCHANGE SERVICE

Service Description (Cont) 3.

3.1.7.11 MCI Business Services I Local & Long Distance

Benefits

Upon Installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified within this program.

Features and applicable features charges for Local Trunk Basis, Local Trunk DID, Local Trunk 2 Way Direct and ISDN-PRI are available as described in sections 3.1.1 and 3.6. Blocks of telephone numbers can be obtained in Blocks of 20 DID numbers for Local Trunk-DID and Local Trunk-2 Way Direct service

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding

<u>Monthly Recurring Charge</u> Feature Package 1 Remote Call Forwarding DID/2 Way Direct Numbers (per each block of 20 numbers)

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

\$3.50 \$20.00 \$6.25

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or T more years for up to a 15% discount.

Termination of Service: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan subscription to this plan.

Issued: April 13, 2004

Cancelled

May 1, 2006 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 14, 2004



N

Ň

D

Ν

Ν

Service Description (Cont)

3.1.7.11 MCI Business Services I Local & Long Distance

<u>Benefits:</u>

Service Commission

RECD DEC 1 6 2003

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

<u>Features</u>:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

<u>Monthly Recurring Charge</u> Feature Package 1 MAY 1 4 2004 By 1541555 1 0 Ublic Service Commission

CANCELLED

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

\$3.50

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: December 16, 2003

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Missouri Public Sarvice Commission

Effective:

JAN 2 0 2004

. A.,.



Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.18.1 Cancels Original Page No. 55.18.1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. Service Description (Cont)

3.1.7.11 <u>Verizon Business Services I Local & Long Distance</u> (cont)

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.18

Issued: June 30, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601



Effective: August 1, 2006

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missourri P.S.C. Tariff No. 1 1ST Revised Page No. 55.19 Cancels Original Page No. 55.19

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.12 Verizon Business Services I Local & Long Distance Line Solution T

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the Verizon Business Services I Local
 T and Long Distance Line Solution Service as described in
 the Company's "Service Publication and Price Guide"
 (The Guide) located on the Company's website at
 www.Verizon.com ("Companion Interstate Service")
 and must subscribe to the Verizon Business Services
 T Local and Long Distance Line Solution Service offered
 in the Verizon Communications, Inc. MO PSC Tariff No. 3.
- must subscribe to service under Product Package Guide Types 13,
 14, 15, or 16 as described in The Guide.
 - must be a new business services customer provisioned via UNE-Platform (UNE-P) or existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Non-recurring Charges: Applicable non-recurring charges apply to services under this program as specified in Section 3.1-3.1.2.3.2.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Line basis. The following Monthly Recurring Charges apply:

<u>Offering</u>	Monthly Recurring Charge (unlimited)
Plan 1	\$60.00
Plan 2	\$60.00

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Features:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified within this program.

Issued: April 1, 2006

Effective: May 1, 2006

Filed Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission MCImetro ACCESS TRANSMISSION SERVICES, LLC

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.12 MCI Business Services I Local & Long Distance Line Solution

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the MCI Business Services I Local and Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.mci.com ("Companion Interstate Service") and must subscribe to the MCI Business Services I Local and Long Distance Line Solution Service offered in the MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.
- must subscribe to service under Product Package Guide Types 13, 14, 15, or 16 as described in The Guide.
- must be a new business services customer provisioned via UNE-Platform (UNE-P) or existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Non-recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified in Section 3.1-3.1.2.3.2.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Line basis. The following Monthly Recurring Charges apply:

<u>Offerina</u>	Monthly Recurring Charge (unlimited)
Plan 1	\$60.00
Plan 2	\$60.00

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Features:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3 except that the Feature Package 1 monthly recurring charges which are specified within this program.

ALL MATERIAL ON THIS PAGE IS NEW.





ALL MATERIAL ON THIS PAGE WAS REMOVED

1

-

÷

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

•

Ν

Ν

Local Exchange Service

3. Service Descriptions (Cont'd)

3.1.7.12 Verizon Business Services I Local & Long Distance Line Solution (Cont'd)

> Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding¹ Remote Access to Call Forwarding (Plan 2) Vanity Number

Monthly Recurring Charge
Feature Package 1\$3.50Remote Access to Call Forwarding (Plan 2)\$7.00

<u>Discounts</u>:¹ These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount, and 3 years or more years for up to a 15% discount.

These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services I Local and Long Distance Line Solution intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts in response to competitive marketplace conditions.

To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25 % discount.

Termination of Service: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

- For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Business Services MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this pl
- For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Business Services MO PSC Tariff No. 3, companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

'Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007



- 3. Service Description (Cont)
 - 3.1.7.12 <u>Verizon Business Services I Local & Long Distance</u> <u>Line Solution (Cont)</u>

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge Feature Package 1 \$3.50

¹<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in N addition to, discounts applicable to the Verizon Business Services I Local and Long Distance Line Solution intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, N

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

 $^{1}\!Effective$ August 1, 2006, these discounts will no longer be available to new customers.

N N

Issued: June 30, 2006

Effective: August 1, 2006

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Filed Missouri Public Service Commission

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.20 Cancels Original Page No. 55.20

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.12 <u>Verizon Business Services I Local & Long Distance</u> <u>Line Solution</u> (Cont)

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge 53.50

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Communications, Inc. T MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3, T companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



August 1, 2006 Missouri Public Service Commission

Cancelled

T

3. Service Description (Cont)

3.1.7.12 MCI Business Services I Local & Long Distance Line Solution (Cont)

Local Line Optional Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

Monthly Recurring Charge 53.50

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: December 16, 2003

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 January 20, 2004

Effective: January 16, 2004



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

2

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 1ST Revised Page No. 55.21 Cancels Original Page No. 55.21

Т

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.12 <u>Verizon Business Services I Local & Long Distance</u> <u>Line Solution</u> (Cont)

For new customers who disconnect Companion Local Service under this tariff, Companion Interstate Service under The Guide and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC T Tariff No. 3, will terminate and the customer will be automatically subscribed to Verizon Business Services I Voice Services Option T 1 under The Guide and Verizon Business Services I Voice Service under Verizon Communications, Inc. MO PSC Tariff No. 3.

For new customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Communications, Inc. MO PSC Tariff No. 3. Companion T Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to Verizon Business Services I Voice Services Option T 1 under The Guide.

Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts other than those stated above or promotions including any term plan discounts.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to Verizon Business Services I Local and Long Distance Line Solution based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY VERIZON. Verizon will install the T Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission

Filed Missouri Public Service Commission MCImetro ACCESS TRANSMISSION SERVICES, LLC

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.12 MCI Business Services I Local & Long Distance Line Solution (Cont)

For new customers who disconnect Companion Local Service under this tariff, Companion Interstate Service under The Guide and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3, will terminate and the customer will be automatically subscribed to MCI Business Services I Voice Services Option 1 under The Guide and MCI Business Services I Voice Service under MCI WorldCom Communications, Inc. MO PSC Tariff No. 3.

For new customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI WorldCom Communications, Inc. MO PSC Tariff No. 3. Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to MCI Business Services I Voice Services Option 1 under The Guide.

Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts other than those stated above or promotions including any term plan discounts.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to MCI Business Services I Local and Long Distance Line Solution based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: December 16, 2003

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 January 20, 2004 Effective: January 16, 2004

> **Filed** Missouri Public Service Commission

ī.

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. <u>Service Description (Cont'd)</u>

3.1.7.13 <u>Verizon Business Services I Local</u>

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Type 13, 14, 15 or 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness/guide.com;
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk DID, Local Trunk -2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

<u>Features:</u>

Flat Rate Service Line Features: Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.2 except for feature package I monthly recurring charges which are specified within this program. The following optional features are available for Flat-Rate Local Line service in addition to the Local Line Standard Features.

<u>Local Line Standard Features</u> Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2)

<u>Metered Service Line Features</u>: The following features are included in metered services feature package.

Calling Party Number delivery (outbound) with Caller ID Blocking Selective, or Caller ID Blocking-Complete (mutually exclusive to CPN Delivery), Call Forward Variable Caller Transfer, or Three-way Conference Calling Speed Dial (8 codes), Caller ID w/Name (inbound) Call Waiting/ Cancel Call Waiting.

<u>Flat Rate and Metered Service ISDN Features:</u> ISDN features and applicable feature charges for ISDN-PRI are available as described in Sections 3.1.5.3.2. Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.2.3.2

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007



Ν

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missourri P.S.C. Tariff No. 1 2nd Revised Page No. 55.22 Cancels 1st Revised Page No. 55.22

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.13 <u>Verizon Business Services I Local</u>

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Type 13, 14, 15 or 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness/guide.com; T
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk DID, Local Trunk -2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

Flat Rate Service Line Features:

Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.2 except for feature package I monthly recurring charges which are specified within this program. The following optional features are available for Flat-Rate Local Line service in addition to the Local Line Standard Features.

Local Line Standard Features Feature Package 1 Remote Call Forwarding

<u>Metered Service Line Features</u>: The following features are included in metered services feature package.

Calling Party Number delivery (outbound) with Caller ID Blocking Selective, or Caller ID Blocking-Complete (mutually exclusive to CPN Delivery), Call Forward Variable Caller Transfer, or Three-way Conference Calling Speed Dial (8 codes), Caller ID w/Name (inbound) Call Waiting/ Cancel Call Waiting.

Flat Rate and Metered Service ISDN Features: ISDN features and applicable feature charges for ISDN-PRI are available as described in Sections 3.1.5.3.2. Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.2.3.2

Issued: April 1, 2006

Cancelled March 8, 2007 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



 \mathbf{T}

MCImetro ACCESS TRANSMISSION SERVICES, LLC

÷

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.13 <u>MCI Business Services I Local</u>

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Type 13, 14, 15 or 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com;
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local T/N Trunk-Basic, Local Trunk DID, Local Trunk -2 Way Direct, Local | ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge | and optional features. T/N

Features:

Flat Rate Service Line Features: Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.2 except for feature package I monthly recurring charges which are specified within this program. The following optional features are available for Flat-Rate Local Line service in addition to the Local Line Standard Features.

<u>Local Line Standard Features</u> Feature Package 1 Remote Call Forwarding

<u>Metered Service Line Features</u>: The following features are included in metered services feature package.

Calling Party Number delivery (outbound) with Caller ID Blocking Selective, or Caller ID Blocking-Complete (mutually exclusive to CPN Delivery), Call Forward Variable Caller Transfer, or Three-way Conference Calling Speed Dial (8 codes), Caller ID w/Name (inbound) Call Waiting/ Cancel Call Waiting.

Flat Rate and Metered Service ISDN Features: ISDN features and applicable feature charges for ISDN-PRI are available as described in Sections 3.1.5.3.2. Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.2.3.2

MATERIAL ON THIS PAGE WAS MOVED TO PAGE NO. 55.23

Issued: April 13, 2004

-- - - -

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 14, 2004



T/N

N/T

N/T

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.22 Missouri Public

LOCAL EXCHANGE SERVICE

Service Description (Cont)

RECD DEC 1 6 2003

- 3.1.7.13 MCI Business Services I Local
 - Eligipility: To be eligible for this program, customers:
 - must subscribe to service under Product Package Guide Type 13, 14, 15 or 16 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com;
 - must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

<u>Eligible Charges</u>: Monthly recurring charges for Local Line and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.3 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Call Waiting/Cancel Call Waiting Caller ID with name and Number Remote Call Forwarding Vanity Number

CANCELLED

MAY 1 4 2004

MISSOURI

By STRS 55.42 Public Service Commission

Monthly Recurring Charge

Feature Package 1

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified Section 3.1-3.1.2.3.2.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

\$3.50

Local Line (per line)

Plan 1: \$35.00 Plan 2: \$40.00

ALL MATERIAL ON THIS PAGE IS NEW.



Effective:

FILED JAN 20 2004

ssued: December 16, 2003

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

JAN 2 0 2004

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 4th Revised Page No. 55.23 Cancels 3rd Revised Page No. 55.23

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. <u>Service Description (Cont'd)</u>

3.1.7.13 <u>Verizon Business Services I Local (Cont'd)</u>

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified Section 3.1-3.1.2.3.2.

<u>Monthly Charges</u>: The following monthly charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Charge	es	for	Flat	t Rate	Service:	
Local					· · · · · · · · · · · · · · · · · · ·	
	Pl	an	1:			

Plan 1:	\$35.00
Plan 2:	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (per Trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (per T-1)	\$672.00
Local ISDN-PRI (Per T-1)	\$672.00
Feature Package 1	\$3.50
Remote Call Forwarding	\$20.00
Remote Access to Call Forwarding (Plan 2)	\$7.00

Charges for Metered Service:

Local Line (per line) Plan 1: Plan 2:	\$35.00 \$40.00
Regional Metered Line	\$23.00
Regional Metered Line Usage	\$0.0150
Nationwide Metered Line	\$28.00
Nationwide Metered Line Local Usage	\$0.025
Local Metered T1/ISDN-PRI	\$540.00
Per T1 ISDN-PRI Local Metered T1/ ISDN-PRI usage	\$0.005

Metered Feature Package (optional) \$19

Benefits:

Unless the state has established a different requirement, for Customers who reside within the most denset zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007



Ν

Т

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.13 <u>Verizon Business Services I Local</u> (CONT)

<u>Non-Recurring Charges:</u> Applicable non-recurring charges apply to services under this program as specified Section 3.1-3.1.2.3.2.

<u>Monthly Charges</u>: The following monthly charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

<u>Charges for Flat Rate Service</u> : Local Line (per line)	
Plan 1: Plan 2:	\$35.00 \$40.00
Local Trunks (Basic, DID and 2 Way Direct) (per Trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (per T-1)	\$672.00
Local ISDN-PRI (Per T-1)	\$672.00
Feature Package 1	\$3.50
Remote Call Forwarding	\$20.00
Charges for Metered Service:	
Local Line (per line) Plan 1: Plan 2:	\$35.00 \$40.00
Regional Metered Line Regional Metered Line Usage	\$23.00 \$0.0150
Nationwide Metered Line Nationwide Metered Line Local Usage	\$28.00 \$0.025
Local Metered T1/ISDN-PRI Per T1 ISDN-PRI Local Metered T1/ ISDN-PRI usage	\$540.00 \$0.005

Metered Feature Package (optional) \$19

<u>Benefits</u>:

Unless the state has established a different requirement, for Customers who reside within the most denset zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601



Cancelled March 8, 2007

3. Service Description (Cont)

3.1.7.13 MCI Business Services I Local (CONT)

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified Section 3.1-3.1.2.3.2.

<u>Monthly Charges</u>: The following monthly charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

<u>Charqes for Flat Rate Service</u> : Local Line (per line)		T
Plan 1: Plan 2:	\$35.00 \$40.00	
Local Trunks (Basic, DID and 2 Way Direct) (per Trunk)	\$40.00	N
Local Trunks (Basic, DID and 2 Way Direct) (per T-1)	\$672.00	
Local ISDN-PRI (Per T-1)	\$672.00	
Feature Package 1	\$3.50	
Remote Call Forwarding	\$20.00	N
<u>Charges for Metered Service</u> :		N
Local Line (per line)	¢25.00	N I I
	\$35.00 \$40.00	N
Local Line (per line) Plan 1:		N
Local Line (per line) Plan 1: Plan 2: Regional Metered Line	\$40.00 \$23.00	N
Local Line (per line) Plan 1: Plan 2: Regional Metered Line Regional Metered Line Usage Nationwide Metered Line	\$40.00 \$23.00 \$0.0150 \$28.00 \$0.025 \$540.00	N

Benefits:

Unless the state has established a different requirement, for Customers who reside within the most denset zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.22

Issued: April 13, 2004

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 14, 2004



Т

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 55.23 Missouri Public

LOCAL EXCHANGE SERVICE

REC'D DEC 1 6 2003

Service Description (Cont)

3.1.7.13 MCI Business Services I Local (CONT)

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

The following disclaimers apply to Stand Alone MCI Business Services I Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

ALL MATERIAL ON THIS PAGE IS NEW.



Effective:

Missouri Public Service Commission

FILED JAN 20 2004

ssued: December 16, 2003

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 JAN 2 0 2004

Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.23.1 Cancels 2nd Revised Page No. 55.23.1

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. Service Description (Cont)

3.1.7.13 <u>Verizon Business Services I Local</u> (CONT)

¹<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge, per minute usage charges, and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in N addition to, discounts applicable to the Verizon Business Services I Local intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, | Customer must: 1) demonstrate to the Company's reasonable satisfaction | that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, N

The following disclaimers apply to Stand Alone Verizon Business Services I Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY Verizon. Verizon will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

¹Effective August 1, 2006, these discounts will no longer be available to new customers.

N N

Issued: June 30, 2006

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: August 1, 2006



CANCELLED September 1, 2008 Missouri Public Service Commission

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.23.1 Cancels Original Page No. 55.23.1

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.13 <u>Verizon Business Services I Local</u> (CONT)

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge, per minute usage charges, and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

The following disclaimers apply to Stand Alone Verizon Business T Services I Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY Verizon. Verizon will install the T Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



August 1, 2006 Missouri Public Service Commission

Cancelled

MCImetro ACCESS TRANSMISSION SERVICES, LLC

LOCAL EXCHANGE SERVICE

. Service Description (Cont)

3.1.7.13 <u>MCI Business Services I Local</u> (CONT)

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge, per minute usage charges, and monthly recurring charges for N optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or T more years for up to a 15% discount.

The following disclaimers apply to Stand Alone MCI Business Services I Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE NO. 55.23.

ssued: April 13, 2004

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 14, 2004


Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.24 Cancels 2nd Revised Page No. 55.24

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. Service Description (Cont)

3.1.7.14 Verizon Business Services I Flex T1¹ 中 The Verizon Business Services I Flex T1 is a flexible, Т integrated solution for small and medium businesses. The Flex T1 must be configured with a minimum of 8 circuits (of which 4 must be Local trunks). Beyond the 4 required Local trunks, the additional 4 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local, Local Trunks and Private Internet Protocol Services. Customers will be charged a per circuit/trunk charge for the additional 4 circuits and any circuits/trunks added on top of the required 8 circuits/trunks charges. Eliqibility Customer must subscribe to service under a Verizon Business ጥ Services Agreement or renew under a Verizon Business Services T Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.verizonbusiness/quide.com Т Customer must also subscribe to the Verizon Business Services т I Flex T1 offering as described in Verizon Communications Т Inc, MO PSC Tariff No. 3. The Verizon Business Services I Flex T1 cannot be used Т in conjunction with the Verizon Business Services I Local and Long Distance Plan or the Verizon Business Services I Local and Long Distance Line Solution. Monthly Recurring Charges: A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Trunk basis. The following Monthly Recurring Charges apply: Monthly Recurring Charge Flex T1: \$290 (\$240 in a Local Lit Building) (for 4 trunks of Local) Additional Circuit Fee: \$40 per each Local Trunk, Frame Relay DS-0, Private IP DS-0, and/or Integrated Internet Access DS-0 Т Customers are eligible to receive Verizon Business Services I Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, and Private IP service.

¹Effective April 1, 2005 this plan will no longer be available to new customers.

Issued: April 1, 2006

CANCELLED

September 1, 2008 Missouri Public

Service Commission

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006

Filed Missouri Public Service Commission

3. Service Description (Cont)

3.1.7.14 MCI Business Services I Flex T1¹

The MCI Business Services I Flex T1 is a flexible, integrated solution for small and medium businesses. The Flex T1 must be configured with a minimum of 8 circuits (of which 4 must be Local trunks). Beyond the 4 required Local trunks, the additional 4 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local, Local Trunks and Private Internet Protocol Services. Customers will be charged a per circuit/trunk charge for the additional 4 circuits and any circuits/trunks added on top of the required 8 circuits/trunks charges.

Eligibility

Customer must subscribe to service under a MCI Business Services Agreement or renew under a MCI Business Services Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.mci.com">http://www.mci.com.

Customer must also subscribe to the MCI Business Services I Flex T1 offering as described in MCI WorldCom Communications Inc, MO PSC Tariff No. 3.

The MCI Business Services I Flex T1 cannot be used in conjunction with the MCI Business Services I Local and Long Distance Plan or the MCI Business Services I Local and Long Distance Line Solution.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Trunk basis. The following Monthly Recurring Charges apply:

Flex T1:	Monthly Recurring Charge \$290 (\$240 in a Local Lit Building) (for 4 trunks of Local)
Additional Circuit Fee:	\$40 per each Local Trunk, Frame Relay DS-0, Private IP DS-0, and/or Integrated Internet Access DS-0

Customers are eligible to receive MCI Business Services I Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, and Private IP service.

¹Effective April 1, 2005 this plan will no longer be available to new customers.

Ν

Issued: March 1, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: April 1, 2005



MCImetro ACCESS TRANSMISSION SERVICES, LLC

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.14 MCI Business Services I Flex T1

The MCI Business Services I Flex T1 is a flexible, integrated solution for small and medium businesses. The Flex T1 must be configured with a minimum of 8 circuits (of which 4 must be Local trunks). Beyond the 4 required Local trunks, the additional 4 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local, Local Trunks and Private Internet Protocol Services. Customers will be charged a per circuit/trunk charge for the additional 4 circuits and any circuits/trunks added on top of the required 8 circuits/trunks charges.

<u>Eliqibility</u>

Customer must subscribe to service under a MCI Business Services Agreement or renew under a MCI Business Services Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at <http://www.mci.com/>www.mci.com.

Customer must also subscribe to the MCI Business Services I Flex T1 offering as described in MCI WorldCom Communications Inc, MO PSC Tariff No. 3.

The MCI Business Services I Flex T1 cannot be used in conjunction with the MCI Business Services I Local and Long Distance Plan or the MCI Business Services I Local and Long Distance Line Solution.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Trunk basis. The following Monthly Recurring Charges apply:

Flex Tl:	Monthly Recurring Charge \$290 (\$240 in a Local Lit Building) (for 4 trunks of Local)
Additional Circuit Fee:	\$40 per each Local Trunk, Frame Relay DS-0, Private IP DS-0, and/or Integrated Internet Access DS-0

Customers are eligible to receive MCI Business Services I Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, and Private IP service.

ALL MATERIAL ON THIS PAGE IS NEW.

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

T

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.14 <u>Verizon Business Services I Flex T1¹</u>

Additional benefits:

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in Verizon WorldCom Communications Tariff No. 2.

Non Recurring Charges: Applicable non-recurring charges apply to services under the Verizon Business Services I Flex T1 as specified T in section 3.1.3.1.3.1, 3.3.2.3.1, 3.1.3.3.2.1 Customers are eligible to utilize the Local Installation Waiver Promotion. Verizon T will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

¹Effective April 1, 2005 this plan will no longer be available to new customers.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006

Filed Missouri Public Service Commission

CANCELLED September 1, 2008 Missouri Public Service Commission

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 55.25 Cancels Original Page No. 55.25

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.14 MCI Business Services I Flex T1¹

Additional benefits:

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI WorldCom Communications Tariff No. 2.

<u>Non Recurring Charges:</u> Applicable non-recurring charges apply to services under the MCI Business Services I Flex T1 as specified in section 3.1.3.1.3.1, 3.3.2.3.1, 3.1.3.3.2.1 Customers are eligible to utilize the Local Installation Waiver Promotion. MCI will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

¹Effective April 1, 2005 this plan will no longer be available to new customers.

Issued: March 1, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: April 1, 2005



Ν

MCImetro ACCESS TRANSMISSION SERVICES, LLC

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.14 MCI Business Services I Flex T1

Additional benefits:

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI WorldCom Communications Tariff No. 2.

Non Recurring Charges: Applicable non-recurring charges apply to services under the MCI Business Services I Flex T1 as specified in section 3.1.3.1.3.1, 3.3.2.3.1, 3.1.3.3.2.1 Customers are eligible to utilize the Local Installation Waiver Promotion. MCI will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

ALL MATERIAL ON THIS PAGE IS NEW.

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. Service Description (Cont)

3.1.7.15 <u>Verizon Business Services Versatile T1</u>¹ T Verizon Business Services Versatile T1 ("Versatile T1") is an T integrated solution for small businesses. Customer must order a minimum of 8 circuits, 4 of which must be local lines or trunks, and one of which must be Internet Dedicated Access, and a maximum of 24 circuits based on Packages A, B, or C as detailed below.

<u>Eliqibility</u>

Customer must subscribe to service under , Verizon Guide Type 17, T with a minimum l-year term , as described in the Company's Service Publication and Price Guide" located on the Company's Internet site at www.verizonbusiness/guide.

Customer must also subscribe to the Verizon Business Services T Versatile Tl offering as described in Verizon Missouri T PSC Tariff No. 3.

Customer is not eligible to receive the benefits of any other program or promotion related to local, long distance, conferencing or internet dedicated access services.

Benefits:

- Package A: Customers who subscribe to Versatile T1 service with a minimum 8 circuits and a maximum 12 circuits will receive the monthly recurring charges detailed below.
- Package B: Customers who subscribe to Versatile Tl service with a minimum 13 circuits and a maximum 18 circuits will receive the monthly recurring charges detailed below.
- Package C: Customers who subscribe to Versatile Tl service with a minimum 19 circuits or a maximum 24 circuits will receive the monthly recurring charges detailed below.

Monthly Recurring Charges:

A monthly recurring charge applies to the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The following Monthly Recurring Charges apply:

Service	Monthly Recurring Charge:
Package A	\$640.00
Package B	\$775.00
Package C	\$970.00

¹Effective October 15, 2005, Versatile will no longer be available to new subscribers and existing customers will no longer be able to place orders.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



CANCELLED September 1, 2008 Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 55.26 Cancels Original Page No. 55.26

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.15 <u>MCI Business Services Versatile T1¹</u>

MCI Business Services Versatile T1 ("Versatile T1") is an integrated solution for small businesses. Customer must order a minimum of 8 circuits, 4 of which must be local lines or trunks, and one of which must be Internet Dedicated Access, and a maximum of 24 circuits based on Packages A, B, or C as detailed below.

<u>Eligibility</u>

Customer must subscribe to service under , MCI Guide Type 17, with a minimum 1-year term , as described in the Company's Service Publication and Price Guide" located on the Company's Internet site at <u>www.mci.com</u>.

Customer must also subscribe to the MCI Business Services Versatile T1 offering as described in MCI WorldCom Missouri PSC Tariff No. 3.

Customer is not eligible to receive the benefits of any other program or promotion related to local, long distance, conferencing or internet dedicated access services.

<u>Benefits</u>:

- Package A: Customers who subscribe to Versatile T1 service with a minimum 8 circuits and a maximum 12 circuits will receive the monthly recurring charges detailed below.
- Package B: Customers who subscribe to Versatile T1 service with a minimum 13 circuits and a maximum 18 circuits will receive the monthly recurring charges detailed below.
- Package C: Customers who subscribe to Versatile T1 service with a minimum 19 circuits or a maximum 24 circuits will receive the monthly recurring charges detailed below.

Monthly Recurring Charges:

A monthly recurring charge applies to the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The following Monthly Recurring Charges apply:

Service	Monthly Recurring Charge:
Package A	\$640.00
Package B	\$775.00
Package C	\$970.00

¹Effective October 15, 2005, Versatile will no longer be available to new subscribers and existing customers will no longer be able to place orders.

N N

Issued: September 15, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: October 15, 2005



MCImetro ACCESS TRANSMISSION SERVICES, LLC

-

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.15

MCI Business Services Versatile T1 MCI Business Services Versatile T1 ("Versatile T1") is an integrated solution for small businesses. Customer must order a minimum of 8 circuits, 4 of which must be local lines or trunks, and one of which must be Internet Dedicated Access, and a maximum of 24 circuits based on Packages A, B, or C as detailed below.

Eligibility

Customer must subscribe to service under , MCI Guide Type 17, with a minimum 1-year term , as described in the Company's Service Publication and Price Guide" located on the Company's Internet site at www.mci.com.

Customer must also subscribe to the MCI Business Services Versatile T1 offering as described in MCI WorldCom Missouri PSC Tariff No. 3.

Customer is not eligible to receive the benefits of any other program or promotion related to local, long distance, conferencing or internet dedicated access services.

Benefits:

- Package A: Customers who subscribe to Versatile T1 service with a minimum 8 circuits and a maximum 12 circuits will receive the monthly recurring charges detailed below.
- Package B: Customers who subscribe to Versatile T1 service with a minimum 13 circuits and a maximum 18 circuits will receive the monthly recurring charges detailed below.
- Package C: Customers who subscribe to Versatile T1 service with a minimum 19 circuits or a maximum 24 circuits will receive the monthly recurring charges detailed below.

Monthly Recurring Charges:

A monthly recurring charge applies to the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The following Monthly Recurring Charges apply:

<u>Serviçe</u>	Monthly Recurring Charge:
Package A	\$640.00
Package B	\$775.00
Package C	\$970.00

CANCELLED

October 15, 2005

ALL MATERIAL ON THIS PAGE IS NEW.

MISSOURI PUBLIC SERVICE COMMISSION

Issued: January 1, 2005

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.27 Cancels 2nd Revised Page No. 55.27

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

1

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 2nd Revised Page No. 55.27 Cancels 1st Revised Page No. 55.27

Т

Т

T

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.15 Verizon Business Services Versatile T1¹ (cont'd)

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in Verizon Business Missouri PSC Tariff No. 3.

Non Recurring Charges:

Applicable non-recurring charges apply to services under the Verizon Business Services Versatile T1 as specified in section 3.3.1.3.1, 3.3.2.3.1, 3.103.3.2.1.

Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress,jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

<u>Features</u>

DID blocks (block of 20) Nationwide rate of \$6.25 per month

The following local calling features are included at no additional charge:

Caller ID - Name & Number Complete Blocking for Caller ID Select Blocking for Caller ID Call Forwarding Variable Speed Dial 8 or 30 Call Waiting Three-way Calling Call Transfer 900/976 Blocking Toll Blocking (per line) Call Forwarding-busy Call Forwarding- Don't answer Voicemail

¹Effective October 15, 2005, Versatile will no longer be available to new subscribers and existing customers will no longer be able to place orders.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



CANCELLED September 1, 2008 Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 55.27 Cancels Original Page No. 55.27

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.15 <u>MCI Business Services Versatile T1¹</u> (cont'd)

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI WorldCom Missouri PSC Tariff No. 3.

Non Recurring Charges:

Applicable non-recurring charges apply to services under the MCI Business Services Versatile T1 as specified in section 3.3.1.3.1, 3.3.2.3.1, 3.103.3.2.1.

Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

<u>Features</u>

DID blocks (block of 20) Nationwide rate of \$6.25 per month

The following local calling features are included at no additional charge:

Caller ID - Name & Number Complete Blocking for Caller ID Select Blocking for Caller ID Call Forwarding Variable Speed Dial 8 or 30 Call Waiting Three-way Calling Call Transfer 900/976 Blocking Toll Blocking (per line) Call Forwarding-busy Call Forwarding- Don't answer Voicemail

¹Effective October 15, 2005, Versatile will no longer be available to new subscribers and existing customers will no longer be able to place orders.

N N

Issued: September 15, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: October 15, 2005



72

MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 55.27

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.15 MCI Business Services Versatile T1 (cont'd)

<u>Additional benefits</u>

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI WorldCom Missouri PSC Tariff No. 3.

Non Recurring Charges:

Applicable non-recurring charges apply to services under the MCI Business Services Versatile T1 as specified in section 3.3.1.3.1, 3.3.2.3.1, 3.103.3.2.1.

Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress,jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

<u>Features</u>

DID blocks (block of 20) Nationwide rate of \$6.25 per month

The following local calling features are included at no additional charge:

Caller ID - Name & Number Complete Blocking for Caller ID Select Blocking for Caller ID Call Forwarding Variable Speed Dial 8 or 30 Call Waiting Three-way Calling Call Transfer 900/976 Blocking Toll Blocking (per line) Call Forwarding-busy Call Forwarding- Don't answer Voicemail

CANCELLED

October 15, 2005

MISSOURI PUBLIC SERVICE COMMISSION

ALL MATERIAL ON THIS PAGE IS NEW.

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.28 Cancels 2nd Revised Page No. 55.28

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

ŧ

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 2nd Revised Page No. 55.28 Cancels 1st Revised Page No. 55.28

Т

т

Т T

Т

т

LOCAL EXCHANGE SERVICE

З. Service Description (Cont)

3.1.7.16 Verizon Business Services Flex Tl¹

Product Description: The Verizon Business Services Flex T1 is a flexible, integrated T solution for small and medium businesses. The Flex T1 must be configured with a minimum of 8 circuits (of which 4 must be Local trunks). Beyond the 4 required Local trunks, the additional 4 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local, Local Trunks and Private Internet Protocol Services. Customers will be charged a per circuit/trunk charge for the additional 4 circuits and any circuits/trunks added on top of the required 8 circuits/trunks charges.

Eligibility

Customer must subscribe to service under a Verizon Business Т Services Agreement or renew under a Verizon Business Services Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.verizonbusiness/quide.com.

Customer must also subscribe to the Verizon Business Services Flex T1 offering as described in Verizon Business MO PSC Tariff No. 3 ("Companion Intrastate Service").

The Verizon Business Services Flex T1 cannot be used in conjunction with the Verizon Business Services I Local and Long Distance Plan, the Verizon Business Services I Local and Long Distance Line Solution, or the Verizon Business Service II Local and Long Distance Plan.

<u>Monthly Recurring Charges</u>: A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Trunk basis. The following Monthly Recurring Charges apply:

Service	Monthly Recurring Charge		
Flex TI	\$290 (\$240 in a Local Lit Local	Building)	for 4 trunks of

Additional circuit fee \$40 each per Local Trunk, Frame Relay DS-0, Private IP DS-0, and/or Integrated Internet Access DS-0

Customers are eligible to receive Verizon Business Services I Local discounts, or Verizon Business Services II Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, and Private IP service.

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in Verizon Business MO PSC Tariff No. 3.

Non Recurring Charges: Applicable non-recurring charges apply to services under the Verizon Business Services Flex T1 as specified T in section 3.1.3.1.3.1, 3.3.2.3.1, 3.1.3.3.2.1. Customers are eligible to utilize the Local Installation Waiver Promotion or the Verizon Business Services II Local Installation Waiver Promotion. Verizon will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived. waived.

¹Effective April 1, 2005 this plan will no longer be available to new customers.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 1, 2006



CANCELLED September 1, 2008 **Missouri Public** Service Commission s

MISSOURI P.S.C. TARIFF NO. 1 1st Revised Page No. 55.28 Cancels Original Page No. 55.28

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.16 MCI Business Services Flex T1¹

Product Description: The MCI Business Services Flex T1 is a flexible, integrated solution for small and medium businesses. The Flex T1 must be configured with a minimum of 8 circuits (of which 4 must be Local trunks). Beyond the 4 required Local trunks, the additional 4 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local, Local Trunks and Private Internet Protocol Services. Customers will be charged a per circuit/trunk charge for the additional 4 circuits and any circuits/trunks added on top of the required 8 circuits/trunks charges.

Eligibility Customer must subscribe to service under a MCI Business Services Agreement or renew under a MCI Business Services Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.mci.com.

Customer must also subscribe to the MCI Business Services Flex T1 offering as described in MCI Worldcom Communications MO PSC Tariff No. 3 ("Companion Intrastate Service").

The MCI Business Services Flex T1 cannot be used in conjunction with the MCI Business Services I Local and Long Distance Plan, the MCI Business Services I Local and Long Distance Line Solution, or the MCI Business Service II Local and Long Distance Plan.

Monthly Recurring Charges: A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Trunk basis. The following Monthly Recurring Charges apply:

<u>Service</u> Flex T1	Monthly Recurring Charge \$290 (\$240 in a Local Lit Building) for 4 trunks		
Flex TI			
	of Local		

\$40 each per Local Trunk, Frame Relay DS-0, Private IP_DS-0, and/or Integrated Internet Additional circuit fee Access DS-0

Customers are eligible to receive MCI Business Services I Local discounts, or MCI Business Services II Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, and Private IP service.

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI Worldcom Communications MO PSC Tariff No. 3.

Non Recurring Charges: Applicable non-recurring charges apply to services under the MCI Business Services Flex T1 as specified in section 3.1.3.1.3.1, 3.3.2.3.1, 3.1.3.3.2.1. Customers are eligible to utilize the Local Installation Waiver Promotion or the MCI Business Services II Local Installation Waiver Promotion. MCI will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

¹Effective April 1, 2005 this plan will no longer be available to new customers.

Issued: March 1, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: April 1, 2005



Ν

MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 55.28

з. Service Description (Cont)

3.1.7.16 MCI Business Services Flex T1

<u>Product Description:</u> The MCI Business Services Flex T1 is a flexible, integrated solution for small and medium businesses. The Flex T1 must be configured with a minimum of 8 circuits (of which 4 must be Local trunks). Beyond the 4 required Local trunks, the additional 4 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local, Local Trunks and Private Internet Protocol Services. Customers will be charged a per circuit/trunk charge for the additional 4 circuits and any circuits/trunks added on top of the required 8 circuits/trunks charges.

Eligibility

Customer must subscribe to service under a MCI Business Services Agreement or renew under a MCI Business Services Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.mci.com.

Customer must also subscribe to the MCI Business Services Flex T1 offering as described in MCI Worldcom Communications MO PSC Tariff No. 3 ("Companion Intrastate Service").

The MCI Business Services Flex Tl cannot be used in conjunction with the MCI Business Services I Local and Long Distance Plan, the MCI Business Services I Local and Long Distance Line Solution, or the MCI Business Service II Local and Long Distance Plan.

<u>Monthly Recurring Charges</u>: A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-Local Trunk basis. The following Monthly Recurring Charges apply:

<u>Service</u> Flex Tl	<u>Monthly Recurring Charge</u>		
Flex T1	\$290 (\$240 in a	Local Lit Building) for 4 trunks
	of Local	-	

Additional circuit fee \$40 each per Local Trunk, Frame Relay DS-0, Private IP DS-0, and/or Integrated Internet Access DS-0

Customers are eligible to receive MCI Business Services I Local discounts, or MCI Business Services II Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, and Private IP service.

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI Worldcom Communications MO PSC Tariff No. 3.

Non Recurring Charges: Applicable non-recurring charges apply to services under the MCI Business Services Flex T1 as specified in section 3.1.3.1.3.1, 3.3.2.3.1, 3.1.3.3.2.1. Customers are eligible to utilize the Local Installation Waiver Promotion or the MCI Business Services II Local Installation Waiver Promotion. MCI will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

ļ

3.1.7.17 Verizon Business Services II Local¹

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Type 18, 19, 20 or 21 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com;
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Voice Mail

Features and applicable feature charges for ISDN-PRI are available as described in Section 3.1.5.3.1 and 3.1.5.3.2 Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.0.3.2

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.3.2.3.1 except for non-recurring charges which are specific within this program.

Order Expedite Charge (per line or trunk)	\$60.00
T-1 Order Expedite Charge (per T-1)	\$1400.00

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

¹Effective January 1, 2008, Verizon Business Service II, including all plans under N this service, is grandfathered and is no longer available to new customers. N

Issued: November 39 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.17 Verizon Business Services II Local

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Type 18, 19, 20 or 21 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com;
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Voice Mail

Features and applicable feature charges for ISDN-PRI are available as described in Section 3.1.5.3.1 and 3.1.5.3.2 Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.0.3.2

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.3.2.3.1 except for N non-recurring charges which are specific within this program. N

Order Expedite Charge (per line or trunk)	\$60.00	I/T
T-1 Order Expedite Charge (per T-1)	\$1400.00	I/T

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Issued: May 31, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: July 1, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission

Filed Missouri Public Service Commission

3. <u>Service Description (Cont'd)</u>

3.1.7.17 <u>Verizon Business Services II Local</u>

Eligibility: To be eligible for this program, customers:

must subscribe to service under Product Package Guide Type 18, 19, 20 or 21 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness/guide.com;

must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

<u>Features</u>:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Voice Mail

Features and applicable feature charges for ISDN-PRI are available as described in Sections 3.1.5.3.2 and 3.1.5.3.1. Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.2.3.2

<u>Non-Recurring Charges</u>: Applicable non-recurring charges apply to services under this program as specified Section 3.3.2.3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Issued: February 6, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007



Ν

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.29 Cancels Original Page No. 55.29

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.17 <u>Verizon Business Services II Local</u>

Eligibility: To be eligible for this program, customers:

must subscribe to service under Product Package Guide Type 18, 19, 20 or 21 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness/guide.com;

must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Remote Call Forwarding Voice Mail

Features and applicable feature charges for ISDN-PRI are available as described in Sections 3.1.5.3.2 and 3.1.5.3.1. Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.2.3.2

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified Section 3.3.2.3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



Cancelled March 8, 2007 Missouri Public Service Commission Т

3. <u>Service Description (Cont)</u>

3.1.7.17 MCI Business Services II Local

Eligibility: To be eligible for this program, customers:

must subscribe to service under Product Package Guide Type 18, 19, 20 or 21 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.mci.com;

must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

<u>Definitions:</u>

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Remote Call Forwarding Voice Mail

Features and applicable feature charges for ISDN-PRI are available as described in Sections 3.1.5.3.2 and 3.1.5.3.1. Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.2.3.2

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified Section 3.3.2.3.1.

<u>Monthly Charges</u>: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: March 1, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

- 3.1.7.17 Verizon Business Services II Local (Cont'd)³
 - Monthly Recurring Charge (Cont'd)

Flat Rate Pricing Structure

Local	Line (per line) ¹ Plan 1: Plan 2:	\$25.00 \$35.00
Local	Line (per line) Plan 1: Plan 2:	\$30.00 \$30.00
	Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) ¹	\$40.00
	Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) ¹	\$35.00
	Local Trunks (Basic, DID and 2 Way Direct) (Per T-1) 4	\$672.00
	Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$655.00
	Local ISDN-PRI (Per T-1) ¹ Local ISDN-PRI (Per T-1)	\$672.00 \$655.00
	Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Voice Mail Local ISDN-PRI Caller ID With Name (Per T-1)	\$5.00 \$20.00 \$7.00 \$5.00 \$60.00

¹This is rate is grandfathered as of July 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing lines, trunks, or T1/PRIs (Per T1) only. If additional new lines, trunks or T1/PRIs (Per T1) are added, the customer will pay the new rate, effective July 1, 2007. This rate will not be offered to renewals or new customers.

³Effective January 1, 2008, Verizon Business Service II, including all plans under this service, is grandfathered and is no longer available to new customers.

⁴This rate is grandfathered as of July 1, 2008. Current customer will continue to N receive this rate until their current agreement expires on existing trunks (Per T-1), + effective June 20, 2008, this rate will not be offered to renewals or new customers. N

CERTAIN MATERIAL ON THIS PAGE WAS MOVED TO PAGE 55.30.1

Issued: June 2, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: July 1, 2008

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

÷

- Verizon Business Services II Local (Cont'd)³ 3.1.7.17
 - Monthly Recurring Charge (Cont'd) Flat Rate Pricing Structure

Local Line (per line)¹ \$25.00 \$35.00 Plan 1: Plan 2: Local Line (per line) Plan \$30.00 Plan 1: Plan 2: Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk)¹ Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk)¹ Local Trunks (Basic, DID and 2 Way Direct) (Per T-1) Local ISDN-PRI (Per T-1)¹ Local ISDN-PRI (Per T-1) Feature Package 1 \$30.00 \$40.00 \$35.00 \$672.00 \$672.00 \$655.00 Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) \$5.00 \$20.00 \$7.00 \$5.00 Voice Mail Local ISDN-PRI Caller ID With Name (Per T-1)\$60.00

Metered Rate Pricing Structure²

Regional Metered Line Rate ² Plan 1: Plan 2:	\$22.00 \$32.00
Regional Metered Line ² Local Per Minute Rate: Plan 1: Plan 2:	\$0.0150 \$0.0150
Nationwide Metered Line	\$28.00
Nationwide Metered Line Local Per Minute	\$0.025
Local Metered T1/ISDN-PRI Local Metered T1/ ISDN- PRI	\$540 per T1/ISDN-PRI
Per Minute Rate	\$0.005/minute

Per Minute Rate Metered Feature Package (optional)² *\$19

> *Metered Feature Package includes: - Calling Party Number delivery (outbound) with Caller ID Blocking Selective OR Caller ID Blocking-Complete (mutually exclusive to CPN

- Delivery)
- Call Forward Variable
- Caller Transfer OR Three-way Conference Calling Speed Dial (8 codes)
- Caller ID w/Name (inbound)
- Call Waiting/ Cancel Call Waiting

¹This rate is grandfathered as of July 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing lines, trunks, or Tl/PRIs (Per Tl) only. If additional new lines, trunks or Tl/PRIs (Per Tl) are added, the customer will pay the new rate, effective July 1, 2007. This rate will not be offered to renewals or new customers.

²These rates are grandfathered as of July 1, 2007. Current customers will continue to receive these rates until their current agreement expires on existing lines or T1/PRIs only. These rates will not be offered to renewals or new customers.

¹Effective January 1, 2008, Verizon Business Service II, including all plans under Ν this service, is grandfathered and is no longer available to new customers. Ν

Issued: November 39, 2007

CANCELLED July 1, 2008 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: January 1, 2008

Missouri P.S.C Tariff No. 1 4th Revised Page No. 55.30 Cancels 3rd Revised Page No. 55.30

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.17	Verizon Business Services II Local (Cont'd)		
	Monthly Recurring Charge (Cont'd)		
	Flat Rate Pricing Structure		
	Local Line (per line) ¹ Plan 1: Plan 2: Local Line (per line) Plan 1: Plan 2: Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) ¹ Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) ¹ Local Trunks (Basic, DID and 2 Way Direct) (Per T-1) Local ISDN-PRI (Per T-1) ¹ Local ISDN-PRI (Per T-1) Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Pla Voice Mail Local ISDN-PRI Caller ID With Name (Pla	\$5.00	
	Metered Rate Pricing Structure ²		
	Regional Metered Line Rate ² Plan 1: Plan 2:	\$22.00 \$32.00	
	Regional Metered Line ² Local Per Minute Rate: Plan 1: Plan 2:	\$0.0150 \$0.0150	
	Nationwide Metered Line Nationwide Metered Line Local Per Minute	\$28.00 \$0.025	
	Local Metered T1/ISDN-PRI	\$540 per T1/ISDN-PRI	
	Local Metered T1/ ISDN- PRI Per Minute Rate Metered Feature Package (optional) ²	\$0.005/minute *\$19	
	 *Metered Feature Package includes: Calling Party Number delivery (outbound) with Caller ID Blocking Selective OR Caller ID Blocking-Complete (mutually exclusive to CPN Delivery) Call Forward Variable Caller Transfer OR Three-way Conference Calling Speed Dial (8 codes) Caller ID w/Name (inbound) Call Waiting/ Cancel Call Waiting 		

¹This rate is grandfathered as of July 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing lines, trunks, or T1/PRIs (Per T1) only. If additional new lines, trunks or T1/PRIs (Per T1) are added, the customer will pay the new rate, effective July 1, 2007. This rate will not be offered to renewals or new customers.

²These rates are grandfathered as of July 1, 2007. Current customers will continue to receive these rates until their current agreement expires on existing lines or T1/PRIs only. These rates will not be offered to renewals or new customers.

Issued; August 1, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2007

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.17 Verizon Business Services II Local Cont'd)

Lat Rate Pricing Structure		
Local Line (per line) ¹ Plan 1: Plan 2:	\$25.00 \$35.00	
Local Line (per line) Plan 1: Plan 2:	\$30.00 \$30.00	N/I N/R
Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk)	\$40.00	
Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) Local Trunks (Basic, DID and 2 Way	\$35.00	N/R
Direct) (Per T-1) Local ISDN-PRI (Per T-1) ¹ Local ISDN-PRI (Per T-1) Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Voice Mail	\$672.00 \$672.00 \$655.00 \$5.00 \$20.00 \$7.00 \$5.00	N/R
Metered Rate Pricing Structure ²		
Regional Metered Line Rate ² Plan 1: Plan 2:	\$22.00 \$32.00	
Regional Metered Line ² Local Per Minute Rate: Plan 1: Plan 2:	\$0.0150 \$0.0150	
Nationwide Metered Line Nationwide Metered Line Local	\$28.00	
Per Minute	\$0.025	
Local Metered T1/ISDN-PRI Local Metered T1/ ISDN- PRI	\$540 per T1	L/ISDN-PR
Per Minute Rate Metered Feature Package (optional) ²	\$0.005/minu *\$19	ıte
*Meterod Festure Package includes		

- *Metered Feature Package includes: Calling Party Number delivery (outbound) with Caller ID Blocking Selective OR Caller ID Blocking-Complete (mutually exclusive to CPN Delivery) Caller TD Blocking-Complete (matually chorasive Call Forward Variable Caller Transfer OR Three-way Conference Calling Speed Dial (8 codes) Caller ID w/Name (inbound) Call Waiting/ Cancel Call Waiting
- _

¹This rate is grandfathered as of July 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing lines, trunks, or T1/PRIs (Per T1) only. If additional new lines, trunks or T1/PRIs (Per T1) are added, the customer will pay the new rate, effective July 1, 2007. This rate will not be offered to renewals or new customers.

²These rates are grandfathered as of July 1, 2007. Current customers will continue to receive these rates until their current agreement expires on existing lines or T1/PRIs only. These rates will not be offered to renewals or new customers.

Issued: May 31, 2007

CANCELLED September 1, 2007 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: July 1, 2007

Filed Missouri Public Service Commission Ν

Ň

З. Service Description (Cont'd)

3.1.7.17 Verizon Business Services II Local (Cont'd)

<u>Monthly Recurring Charge</u> : Flat Rate Pricing Structure Local Line (per line)	
Plan 1: Plan 2:	\$25.00 \$33.00
Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) (Per trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1) Local ISDN-PRI (Per T-1) Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2)	\$672.00 \$672.00 \$5.00 \$20.00 \$7.00
Voice Mail	\$5.00
Regional Metered Line Rate: Plan 1: Plan 2:	\$22.00 \$32.00
Regional Metered Line <u>Local Per Minute Rate</u> :	
Plan 1: Plan 2:	\$0.0150 \$0.0150
Nationwide Metered Line Local	\$28.00
	\$0.025
Local Metered T1/ISDN-PRI Local Metered T1/ ISDN- PRI Per Minute Rate	\$540 per T1/ISDN-PRI
	\$0.005/minute
Metered Feature Package (optional)*	\$19
<pre>*Metered Feature Package includes: - Calling Party Number delivery (outbound Blocking Selective OR - Caller ID Blocking-Complete (mutually description) - Call Forward Variable - Caller Transfer OB Three-way Conference</pre>	exclusive to CPN Deliver

Caller Transfer OR Three-way Conference Calling -

- _ Speed Dial (8 codes)
- Caller ID w/Name (inbound)
 Call Waiting/ Cancel Call Waiting

Issued: February 6, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: March 8, 2007



•

Ν

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.30 Cancels Original Page No. 55.30

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.17 <u>Verizon Business Services II Local(Cont'd)</u>

Monthly Recurring Charge:	
Flat Rate Pricing Structure	
Local Line (per line)	

Plan 1:	\$25.00
Plan 2:	\$33.00
Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) (Per trunk)	\$40.00

Local Trunks (Basic, DID and 2 Way Direct) (Per T-1) \$672.00

Local ISDN-PRI (Per T-1) \$672.00

Feature Package 1 \$5.00

Remote Call Forwarding \$20.00

Voice Mail

<u>Regional Metered Line Rate:</u>	
Plan 1:	\$22.00
Plan 2:	\$32.00

Regional Metered Line Local Per Minute Rate:

Plan 1:	\$0.0150
Plan 2:	\$0.0150

\$5.00

\$28.00

\$0.025

\$19

\$540 per T1/ISDN-PRI

\$0.005/minute

Nationwide Metered Line Nationwide Metered Line Local Per Minute

Local Metered T1/ISDN-PRI Local Metered T1/ ISDN- PRI Per Minute Rate

<u>Metered Feature Fackage (optional)*</u>

*Metered Feature Package includes: - Calling Party Number delivery (outbound) with Caller ID Blocking Selective OR

- Caller ID Blocking-Complete (mutually exclusive to CPN Delivery)

- Call Forward Variable

Caller Transfer OR Three-way Conference Calling

- Speed Dial (8 codes)

- Caller ID w/Name (inbound)

- Call Waiting/ Cancel Call Waiting

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



Cancelled March 8, 2007 Missouri Public Service Commission Т

ì

3. <u>Service Description (Cont</u>)

3.1.7.17 <u>MCI Business Services II Local(Cont'd)</u>

Flat Rate Pricing Structure Local Line (per line)	
Plan 1: Plan 2:	\$25.00 \$33.00
Local Trunks (Basic, DID and 2 Way Direct) (Per Trunk) (Per trunk)	\$40.00
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$672.00
Local ISDN-PRI (Per T-1)	\$672.00
Feature Package 1	\$5.00
Remote Call Forwarding	\$20.00
Voice Mail	\$5.00
Regional Metered Line Rate: Plan 1: Plan 2: Regional Metered Line Local Per Minute Rate:	\$22.00 \$32.00
Plan 1: Plan 2:	\$0.0150 \$0.0150
Nationwide Metered Line	\$28.00
Nationwide Metered Line Local Per Minute	\$0.025
Local Metered T1/ISDN-PRI	\$540 per T1/ISDN-PRI
Local Metered T1/ ISDN- PRI Per Minute Rate	\$0.005/minute
<u>Metered Feature Package (optional)*</u>	\$19
 *Metered Feature Package includes: Calling Party Number delivery (outbon Blocking Selective OR Caller ID Blocking-Complete (mutuall Call Forward Variable Caller Transfer OR Three-way Confere 	y exclusive to CPN Deliv

- Caller ID w/Name (inbound)
- Call Waiting/ Cancel Call Waiting

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: March 1, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

- 3.1.7.17 Verizon Business Services II Local (Cont'd)³
 - Monthly Recurring Charge (Cont'd)

Metered Rate Pricing Structure²

Regional Metered Line $Rate^2$ Plan 1: \$22.00 Plan 2: \$32.00 Regional Metered Line² Local Per Minute Rate: Plan 1: \$0.0150 Plan 2: \$0.0150 Nationwide Metered Line \$28.00 Nationwide Metered Line Local Per Minute \$0.025 Local Metered T1/ISDN-PRI \$540 per T1/ISDN-PRI Local Metered T1/ ISDN- PRT Per Minute Rate \$0.005/minute Metered Feature Package (optional)² *\$19

*Metered Feature Package includes:

- Calling Party Number delivery (outbound) with Caller ID Blocking Selective OR
- Caller ID Blocking-Complete (mutually exclusive to CPN Delivery)
- Call Forward Variable
- Caller Transfer OR Three-way Conference Calling
- Speed Dial (8 codes)
- Caller ID w/Name (inbound)
- Call Waiting/ Cancel Call Waiting

¹This is rate is grandfathered as of July 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing lines, trunks, or T1/PRIs (Per T1) only. If additional new lines, trunks or T1/PRIs (Per T1) are added, the customer will pay the new rate, effective July 1, 2007. This rate will not be offered to renewals or new customers.

²These rates are grandfathered as of July 1, 2007. Current customers will continue to receive these rates until their current agreement expires on existing lines or T1/PRIs only. These rates will not be offered to renewals or new customers.

MATERIAL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE 55.30

Issued: June 2, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: July 1, 2008
ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 55.31 Cancels 2nd Revised Page No. 55.31

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont'd</u>)

3.1.7.17 Verizon Business Services II Local(Cont'd)

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Unless the state has established a different requirement, for Customers who reside within the densest zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

¹<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount.

Discounts:

These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services II Local intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount.

The following disclaimers apply to Stand Alone Verizon Business Services II Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY VERIZON. Verizon will install the Line-based Service from the point of the local exchange carrier's smartjack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

¹Effective August 1, 2006, these discounts will no longer be available to new customers.

¹Effective January 1, 2008, Verizon Business Services II, including all plans under N this service is grandfathered and is no longer available to new customers. N

Issued: November 29, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

З. Service Description (Cont)

3.1.7.17 Verizon Business Services II Local (Cont'd)

<u>Benefits:</u> Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange

Unless the state has established a different requirement, for Customers who reside within the densest zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

¹<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in Addition to, discounts applicable to the Verizon Business Services II Local intrastate and interstate long distance service. A discount N intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount. Ν

The following disclaimers apply to Stand Alone Verizon Business Services II Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY VERIZON. Verizon will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

¹Effective August 1, 2006, these discounts will no longer be available to new customers.

Ν Ν

Issued: June 30, 2006

CANCELLED January 1, 2008 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: August 1, 2006



Т

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.17 <u>Verizon Business Services II Local (Cont'd)</u>

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Unless the state has established a different requirement, for Customers who reside within the densest zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

The following disclaimers apply to Stand Alone Verizon Business T Services II Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize autodialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY VERIZON. Verizon will install the Line-based Service T from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

Issued: April 1, 2006

Cancelled

Missouri Public

Service Commission

August 1, 2006

Effective: May 1, 2006



Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 55.31

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.17 <u>MCI Business Services II Local (Cont'd)</u>

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Unless the state has established a different requirement, for Customers who reside within the densest zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

The following disclaimers apply to Stand Alone MCI Business Services II Local Line-based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize autodialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Cancelled

ł

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: March 1, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 2nd Revised Page No. 55.31.1 Cancels 1st Revised Page No. 55.31.1

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.17 Verizon Business Services II Local (Cont'd)

Optional Calling Plans

LD Voice Package

Customer will pay the following flat rate monthly recurring charge ("MRC"), fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Verizon.

Agreement TermMRC*One Year\$725.00Two Years\$685.00Three Years\$650.00

*Overage Rate:

Customers will pay a per-minute charge of \$.05 for each minute in excess of 20,000 in a month

Customer understands that the LD Voice Package is restricted in the following manner:

- (i) International long distance, Inbound (toll free), and calling card minutes are NOT included. .
- (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the LD Voice Package.
- (iii) Customer may not utilize the LD Voice Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE LD VOICE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

¹Effective January 1, 2008, Verizon Business Services II, including all plans under this service is grandfathered and is no longer available to new customers.

Issued: November 29, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

N N

CANCELLED September 1, 2008 Missouri Public Service Commission MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.31.1 Cancels Original Page No. 55.31.1

Т

Т

Т

Т

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

, **y**

3.1.7.17 <u>Verizon Business Services II Local (Cont'd)</u>

Optional Calling Plans

LD Voice Package. Customer will pay the following flat rate monthly recurring charge ("MRC"), fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Verizon.

<u>Agreement Term</u>	MRC*
One Year	\$725.00
Two Years	\$685.00
Three Years	\$650.00

<u>*Overage Rate</u>: Customers will pay a per-minute charge of \$.05 for each minute in excess of 20,000 in a month

Customer understands that the LD Voice Package is restricted T in the following manner:

- (i) International long distance, Inbound (toll free), and calling card minutes are NOT included. .
- (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the LD Voice Package.
- (iii) Customer may not utilize the LD Voice Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE LD VOICE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

Issued: October 1, 2006



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missourri P.S.C. Tariff No. 1 Original Page No. 55.31.1

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

, •

4

3.1.7.17 <u>Verizon Business Services II Local (Cont'd)</u>

Optional Calling Plans

<u>All Distance Package.</u> Customer will pay the following flat rate monthly recurring charge ("MRC"), fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Verizon.

<u>Agreement Term</u>	<u>MRC*</u>
One Year	\$725.00
Two Years	\$685.00
Three Years	\$650.00

<u>*Overage Rate</u>: Customers will pay a per-minute charge of \$.05 for each minute in excess of \$20,000 in a month

Customer understands that the All Distance Package is restricted in the following manner:

- (i) International long distance, Inbound (toll free), and calling card minutes are NOT included. .
- (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the All Distance Package.
- (iii) Customer may not utilize the All Distance Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE ALL DISTANCE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW



ALL MATERIAL ON THIS PAGE WAS REMOVED

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

4. Service Descriptions (Cont'd)

3.1.7.17 Verizon Business Services II Local (Cont'd)¹

Optional Calling Plans (Cont'd)

LD Voice 800 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 800 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 800 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 800 Minute Package.

Agreement_Term	MRC*
One Year	\$28.00
Two years	\$26.00
Three Years	\$24.00

*Overage Rate; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 800 in a month

Customer understands that the LD Voice 800 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 800 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 800 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

¹Effective January 1, 2008, Verizon Business Services II, including all plans under this service is grandfathered and is no longer available to new customers.

Issued: November 29, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

Ν

Ν

Missouri PSC Tariff No. 1 Original Page No. 55.31.2

Local Exchange Service

4. Service Descriptions (Cont'd)

3.1.7.17 Verizon Business Services II Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice 800 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 800 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 800 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 800 Minute Package.

Agreement Term	MR <u>C</u> *
One Year	\$28.00
Two years	\$26.00
Three Years	\$24.00

*Overage Rate; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 800 in a month

Customer understands that the LD Voice 800 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 800 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 800 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: March 30, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: May 1, 2007



Missouri P.S.C. Tariff No. 1 2nd Revised Page No. 55.31.3 Cancels 1st Revised Page No. 55.31.3

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

...

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

5. <u>Service Descriptions (Cont'd)</u>

3.1.7.17 <u>Verizon Business Services II Local (Cont'd)</u>¹

Optional Calling Plans (Cont'd)

LD Voice 500 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 500 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 500 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 500 Minute Package.

Agreement Term	MRC*
One Year	\$18.00
Two years	\$17.00
Three Years	\$16.00

* <u>Overage Rate</u>; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 500 in a month

Customer understands that the LD Voice 500 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 500 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 500 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

¹Effective January 1, 2008, Verizon Business Services II, including all plans under N this service is grandfathered and is no longer available to new customers. N

Issued: November 29, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

 MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri PSC Tariff No. 1 Original Page No. 55.31.3

Local Exchange Service

5. Service Descriptions (Cont'd)

3.1.7.17 Verizon Business Services II Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice 500 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 500 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 500 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 500 Minute Package.

Agreement Term	MRC*
One Year	\$18.00
Two years	\$17.00
Three Years	\$16.00

* <u>Overage Rate</u>; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 500 in a month

Customer understands that the LD Voice 500 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 500 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 500 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: March 30, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: May 1, 2007



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Local Exchange Service

6. Service Descriptions (Cont'd)

|

3.1.7.17 Verizon Business Services II Local (Cont'd)¹

Optional Calling Plans (Cont'd)

LD Voice 300 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 300 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 300 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 300 Minute Package.

Agreement Term	MRC*
One Year	\$12.00
Two years	\$11.00
Three Years	\$10.00

* <u>Overage Rate</u>; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 300 in a month

Customer understands that the LD Voice 300 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 300 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 300 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

¹Effective January 1, 2008, Verizon Business Services II, including all plans under this service is grandfathered and is no longer available to new customers.

Issued: November 29, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

N

N

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

. _____

Missouri PSC Tariff No. 1 Original Page No. 55.31.4

Local Exchange Service

6. Service Descriptions (Cont'd)

____

3.1.7.17 Verizon Business Services II Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice 300 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 300 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 300 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 300 Minute Package.

Agreement Term	MRC*
One Year	\$12.00
Two years	\$11.00
Three Years	\$10.00

* Overage Rate; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 300 in a month

Customer understands that the LD Voice 300 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 300 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services II Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 300 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: March 30, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: May 1, 2007



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

з.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C Tariff No. 1 3rd Revised Page No. 55.32 Cancels 2nd Revised Page No. 55.32

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.18 Verizon Business Services II Local & Long Distance

Eligibility: To be eligible for this plan, the customer:

must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;

must subscribe to the Verizon Business Services II Local and Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.mci.com ("Companion Interstate Service") and must subscribe to the Verizon Business Services I Local and Long Distance Line Solution Service offered in the Verizon Business Services Missouri Tariff No. 3.must subscribe to service under Product Package Guide Types 18, 19, 20, or 21 as described in The Guide.

must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Non-recurring Charges:

Applicable non-recurring charges apply to services under this program as specified in Section 18.1.3.3 except for non-recurring charges which are specified within this program.

Order Expedite Charge (per line or trunk)	\$60.00	I/T
T-1 Order Expedite Charge (per T1)	\$1400.00	I/T

MATERIAL ON THIS PAGE WAS MOVED TO PAGE 55.32.1

Issued: May 31, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Local Exchange Service

3. Service Descriptions and Rates (Cont'd)

i

1

3.1.7.18 Verizon Business Services II Local & Long Distance (Cont'd)

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the Verizon Business Services II Local and Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness/guide.com ("Companion Interstate Service") and must subscribe to the Verizon Business Services I Local and Long Distance Line Solution Service offered in the MCI Communications Services, Inc. d/b/a Verizon Business Service MO. P.S.C. Tariff No. 3.
- must subscribe to service under Product Package Guide Types 18, 19, 20, or 21 as described in The Guide.
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Non-recurring Charges:

Applicable non-recurring charges apply to services under this program as specified in Section 3.1 - 3.1.2.3.1

Monthly Recurring Charges:

A monthly recurring charges. A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply:

Offering (per line,	trunk,	T-1 or	ISDN-PRI)	Monthly Recurring Charge
Plan 1: Plan 2:				\$44.00 \$79.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

B C	\$65 \$1400 ²	
C	Flat with Cap \$1325	Ν

Benefits: Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

²This rate is grandfathered as of April 1, 2007, Current customers will continue to receive this rate until their current agreement expires on existing line only. If additional new T1/PRI (Per T1) are added, the customer will pay the new Flat with Cap Rate on the new lines, effective April 1, 2007. The rate will not be offered to renewals or new customers.

Issued: March 1, 2007

Cancelled July 1, 2007 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago IL, 60601

Effective: April 1, 2007



Ν

Т

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.18 Verizon Business Services II Local & Long Distance

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the Verizon Business Services II Local and Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <u>www.verizonbusiness/quide.com</u> T ("Companion Interstate Service") and must subscribe to the Verizon Business Services I Local and Long Distance Line T Solution Service offered in the Verizon Business MO. PSC T Tariff No. 3.
- must subscribe to service under Product Package Guide Types 18, 19, 20, or 21 as described in The Guide.
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Non-recurring Charges:

Applicable non-recurring charges apply to services under this program as specified in Section 3.1-.3.1.2.3.2.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply:

<u>Offering</u> (per line, trunk, T-1 or ISDN-PRI)	Monthly Recurring Charge
Plan 1:	\$44.00
Plan 2:	\$79.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines. B \$65

B \$65 C \$1,400

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



Cancelled April 1, 2007 Missouri Public Service Commission

3. <u>Service Description (Cont</u>)

3.1.7.18 <u>MCI Business Services II Local & Long Distance</u>

Eligibility: To be eligible for this plan, the customer:

- must designate the Company as its local exchange service carrier and the Company both as its interexchange service carrier for interstate and intrastate calling and as its carrier for intraLATA toll calling;
- must subscribe to the MCI Business Services II Local and Long Distance Line Solution Service as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at <u>www.mci.com</u> ("Companion Interstate Service") and must subscribe to the MCI Business Services I Local and Long Distance Line Solution Service offered in the MCI Worldcom Communications MO. PSC Tariff No. 3.
- must subscribe to service under Product Package Guide Types 18, 19, 20, or 21 as described in The Guide.
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Non-recurring Charges:

Applicable non-recurring charges apply to services under this program as specified in Section 3.1-.3.1.2.3.2.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local Line basis, Offering B is available on a per-Local Trunk basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply:

Offering	<u>Monthly_Recurring</u>
(per line, trunk, T-1 or ISDN-PRI)	
Plan 1:	\$44.00
Plan 2:	\$79.00

Unless the state has established a different requirement, for Customers who reside within the most dense zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

В	\$65
С	\$1,400

<u>Benefits</u>:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: March 1, 2005

<u>Charge</u>



ALL MATERIAL ON THIS PAGE WAS REMOVED

÷

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missouri P.S.C Tariff No. 1 1st Revised Page No. 55.32.1 Cancels Original Page No. 55.32.1

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.18 Verizon Business Services II Local & Long Distance (Cont'd)

<u>Monthly Recurring Charges</u>: A monthly recurring charge will apply for the offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local line basis, Offering B is available on a per local Trunk Basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply.

Offering A ² Plan 1: Plan 2:	Monthly Recurring Charge (per <u>line, trunk, T-1 or ISDN-PRI)</u> \$44.00 \$79.00
Offering A: Flat with Cap* Plan 1: Plan 2:	\$56.00 \$56.00

* Offering A Flat with Cap : Any customer who exceeds the equivalent N of 800 minutes during a billing cycle will be billed for each | additional minute at a rate of \$0.05. N

Unless the state has established a different requirement, for customers who reside within the most dense zone of the top 50MSA, as defined by the FCC in Docket 99-98, service is limited to maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

В				\$65°
В	Flat	with	Cap**	\$61
С				\$1400 ¹
С	Flat	with	Cap	\$1325

**Offering B Flat with Cap: Any customer who exceeds the equivalent N of 800 minutes during a billing cycle will be billed for each | additional minute at a rate of \$0.05. N

Benefits

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive Unlimited local exchange service usage.

^TThis rate is grandfathered as of April 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing line only. If additional new T1/PRI (Per T1) are added, the customer will pay the new Flat with cap Rate on the new lines effective April 1, 2007. The rate will not be offered for renewals or new customers.

²This rate is grandfathered as of July 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing line/trunks only. If additional new lines/trunks are added, the customer will pay the new rate on the new lines/trunks, effective July 1, 2007.

Issued; June 29, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: August 1, 2007

Ν

Ν

 MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C Tariff No. 1 Original Page No. 55.32.1

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.18 Verizon Business Services II Local & Long Distance (Cont'd)

<u>Monthly Recurring Charges</u>: A monthly recurring charge will apply for the offering under this plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. Offering A is available on a per-Local line basis, Offering B is available on a per local Trunk Basis and Offering C is available on a per T-1 or ISDN-PRI basis. The following Monthly Recurring Charges apply.

Offering A ² Plan 1:	Monthly Recurring Charge (per line, trunk, T-1 or ISDN-PRI) S44.00	
Plan 2:	\$79.00	
Plan 1: Plan 2:	\$56.00 \$56.00	N/I N/R

Unless the state has established a different requirement, for customers who reside within the most dense zone of the top 50MSA, as defined by the FCC in Docket 99-98, service is limited to maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

В	\$65 ²	
В	\$61	N/R
C	\$1400 ¹	
C (Flat with Cap)	\$1325	

Benefits

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive Unlimited local exchange service usage.

MATERAIL ON THIS PAGE WAS PREVIOUSLY LOCATED ON PAGE 55.32

This rate is gradfathered as of April 1, 2007. Current customers will continue to receive this rate until their current agreement expires on existing line only. If additional new T1/PRI (Per T1) are added, the customer will pay the new Flat with cap Rate on the new lines effective April 1, 2007. The rate will not be offered ro renewals or new customers.

²This rate is grandfathered as of July 1, 2007. Current customers will continue to N receive this rate until their current agreement expires on existing line/trunks only. | If additional new lines/trunks are added, the customer will pay the new rate on the N new lines/trunks, effective July 1, 2007. N

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago IL, 60601 Effective: July 1, 2007

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. <u>Service Description (Cont'd)</u>

3.1.7.18 Verizon Business Services II Local & Long Distance (Cont'd)

<u>Features</u>: The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. <u>Applicable non-recurring and monthly recurring charges will apply to</u> optional features as specified in Section 3.1.2.3.3, except that the Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Remote Call Forwarding

Monthly Recurring Charge

Feature Package 1\$5.00Remote Call Forwarding\$20.00Remote Access to Call Forwarding\$7.00Voice Mail\$5.00

<u>Discounts</u>:¹ These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services II Local and Long Distance intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount.

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Business MO. PSC Tariff No. 3 will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Business MO. PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

¹Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: February 6, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 8, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission

Filed Missouri Public Service Commission

Ν

3. Service Description (Cont)

3.1.7.18 Verizon Business Services II Local & Long Distance (Cont'd)

Features:

<u>Features</u>: The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3, except that the Feature Package 1 monthly recurring charges which are specified within this program

Local Line Optional Features Feature Package 1 Remote Call Forwarding

Monthly Recurring Charge

Feature Package 1	\$5.00
Remote Call Forwarding	\$20.00
Voice Mail	\$5.00
	4J.UU

¹Discounts: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

Discounts: These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services II Local and Long Distance intrastate and interstate long distance service. A discount intrastate and interstate long distance service. A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount. Ń

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Business MO. PSC Tariff No. 3 will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Business MO. PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan time of subscription to this plan.

'Effective August 1, 2006, these discounts will no longer be available to

Ν

Issued: June 30, 2006

Cancelled March 8, 2007 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601

Filed Missouri Public Service Commission

Effective: August 1, 2006

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.33 Cancels Original Page No. 55.33

Т

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.18 <u>Verizon Business Services II Local & Long Distance</u> (Cont'd)

Features:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3, except that the Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Remote Call Forwarding

Monthly Recurring Charge

Feature Package 1	\$5.00
Remote Call Forwarding	\$20.00
Voice Mail	\$5.00

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service</u>: The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in Verizon Business MO. T PSC Tariff No. 3 will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon T Business MO. PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



August 1, 2006 Missouri Public Service Commission

Cancelled

3. <u>Service Description (Cont</u>)

3.1.7.18 <u>MCI Business Services II Local & Long Distance</u> (Cont'd)

<u>Features</u>:

The following optional features are available for Local Line service under this plan in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges will apply to optional features as specified in Section 3.1.2.3.3, except that the Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Optional Features Feature Package 1 Remote Call Forwarding

Monthly Recurring Charge

Feature Package 1	\$5.00
Remote Call Forwarding	\$20.00
Voice Mail	\$5.00

<u>Discounts</u>: These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for a 5% discount, 2 years for up to a 15% discount and 3 or more years for up to a 15% discount.

<u>Termination of Service:</u> The following provisions will apply to customers who terminate service, continue to maintain a Company account, and do not subscribe to other service offerings under this tariff:

For existing customers who disconnect Companion Local Service only under this tariff, Companion Interstate Service offered under The Guide and Companion Intrastate Service offered in MCI Worldcom Communications MO. PSC Tariff No. 3 will terminate and the customer will be automatically re-subscribed to the service offering under this tariff and The Guide to which the customer subscribed at the time of subscription to this plan.

For existing customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI Worldcom Communications MO. PSC Tariff No. 3, Companion Interstate Service under The Guide and Companion Intrastate Service will terminate and the customer will then be automatically re-subscribed to the service offering under The Guide to which the customer subscribed at the time of subscription to this plan.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: March 1, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.34 Cancels Original Page No. 55.34

т

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.18 <u>Verizon Business Services II Local & Long Distance</u>(Cont'd)

For new customers who disconnect Companion Local Service under this tariff, Companion Interstate Service under The Guide and Companion Intrastate Service offered in Verizon Business MO. PSC Tariff No. T 3 will terminate and the customer will be automatically subscribed to Verizon Business Services II Voice Services Option 1 under The Guide and Verizon Business Services I Voice Service under T Verizon Business MO. PSC Tariff No. 3.

For new customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in Verizon Business T MO. PSC Tariff No. 3. Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to Verizon Business T Services II Voice Services Option 1 under The Guide.

Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts other than those stated above or promotions including any term plan discounts.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to Verizon Business Service II Local T and Long Distance Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines or 24 trunks per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY VERIZON. Verizon will install the Line-based Service from the T point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

Issued: April 1, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



3. <u>Service Description (Cont</u>)

3.1.7.18 MCI Business Services II Local & Long Distance (Cont'd)

For new customers who disconnect Companion Local Service under this tariff, Companion Interstate Service under The Guide and Companion Intrastate Service offered in MCI Worldcom Communications MO. PSC Tariff No. 3 will terminate and the customer will be automatically subscribed to MCI Business Services II Voice Services Option 1 under The Guide and MCI Business Services I Voice Service under MCI Worldcom Communications MO. PSC Tariff No. 3.

For new customers who disconnect Companion Local Service under this tariff and Companion Intrastate Service offered in MCI Worldcom Communications MO. PSC Tariff No. 3. Companion Interstate Service under The Guide and Companion Intrastate Service under this tariff will terminate and the customer will be automatically subscribed to MCI Business Services II Voice Services Option 1 under The Guide.

Other Conditions:

Services under this plan are not eligible to receive the benefits of any discounts other than those stated above or promotions including any term plan discounts.

Customers who subscribe to service via a company-designated Internet site will receive Electronic Billing invoicing only.

The following disclaimers apply to MCI Business Service II Local and Long Distance Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer is limited to 30 lines or 24 trunks per location, (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (iii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY MCI. MCI will install the Line-based Service from the point of the local exchange carrier's smart-jack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: January 27, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: March 1, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.18.a <u>Verizon Business Services II Flex Tl Plus</u>¹ The Verizon Business Services II Flex Tl Plus is a flexible, integrated solution for small and medium businesses. The Flex T1 Plus must be configured with a minimum of 12 circuits (of which 6 must be Local trunks or lines; but cannot be both). Beyond the 6 required Local trunks or lines, the additional 6 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local Lines, Local Trunks, Private Internet Protocol Services and Metro Private Line (Lit buildings only). Customers will be charged a per circuit/trunk charge for the required 12 circuits/trunks and any additional circuits/trunks added on top of the initial 12

Eligibility:

Customer must subscribe to service under a Verizon Business Services II Agreement or renew under a Verizon Business Services II Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.mci.com. customer must also subscribe to the Verizon Business Services II Flex T1 Plus offering as described in Verizon Business Services, Inc. Missouri PUC Tariff No. 1 ("Companion Intrastate Service"). The Verizon Business Services II Flex Tl Plus cannot be used in conjunction with the Verizon Business Services II Local and Long Distance Plan.

Monthly Recurring Charges:

A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-circuit basis. The following Monthly Recurring Charges apply:

Service Monthly Recurring Charge: Flex T1 Plus \$40(\$35 in a Local Lit Building) Per Circuit (minimum of 12 circuits and 6 must be local) Customers are eligible to receive MCI Verizon Business Services II Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, Private IP service, and Metro Private Line (Lit Buildings Only).

Additional Benefits:

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI Verizon Business Services, Inc. Missouri PSC Tariff No. 1.

Non Recurring Charges: Applicable non-recurring charges apply to services under the Verizon Business Services II Flex T1 Plus as specified in section 3.1.3.1.3.1, 3.3.2.3.1, and 3.1.3.3.2.1 Customers are eligible to utilize the Local Installation Waiver Promotion. MCI will waive the one-time installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services, Metro Private Line (Lit Buildings Only) within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

¹Effective July 1, 2007, this plan will no longer be available to new customers.

N

Issued; May 31, 2007

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: July 1, 2007
Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.35 Cancels Original Page No. 55.35

T

т

Т T

T

LOCAL EXCHANGE SERVICE

З. Service Description (Cont)

3.1.7.18 Verizon Business Services II Flex T1 Plus Τ Verizon Business Services II Flex T1 Plus Product Description The Verizon Business Services II Flex T1 Plus is a flexible, integrated solution for small and medium businesses. The Flex T1 Plus must be configured with a minimum of 12 circuits (of which 6 must be Local trunks or lines; but cannot be both). Beyond the 6 required Local trunks or lines, the additional 6 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local Lines, Local Trunks, Private Internet Protocol Services and Metro Private Line(Lit buildings only). Customers will be charged a per circuit/trunk charge for the required 12 circuits/trunks and any additional circuits/trunks added on top of the initial 12 T top of the initial 12

Eligibility Customer must subscribe to service under a Verizon Business Services II Agreement or renew under a Verizon Business Services II Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.Verizon.com.customer must also subscribe to the Verizon Business Services II Flex T1 Plus offering as described in Verizon Business, Inc. MO PSC Tariff No. 1.

The Verizon Business Services II Flex T1 Plus cannot be used in conjunction with the Verizon Business Services II Local and Long Distance Plan.

<u>Monthly Recurring Charges</u>: A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-circuit basis. The following Monthly Recurring Charges apply:

Service Monthly Recurring Charge Flex T1Plus\$40 (\$35 in a Local Lit Building) Per Circuit (minimum of 12 circuits and 6 must be local)

Customers are eligible to receive Verizon Business Services II Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, Private IP service, and Metro Private Line(Lit Buildings Only)..

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in Verizon Business, Inc. MO PSC Tariff No. 1.

Non Recurring Charges: Applicable non-recurring charges apply to services under the Verizon Business Services II Flex T1 Plus T as specified in section 3.1.3.1.3.1, 3.3.2.3.1 and 3.1.3.3.2.1 Customers are eligible to utilize the Local Installation Waiver Promotion. Verizon will waive the one-time installation charges T which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services, Metro Private Line (Lit Buildings Only) within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 1, 2006



Cancelled July 1, 2007 Missouri Public Service Commission

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 55.35

LOCAL EXCHANGE SERVICE

з. Service Description (Cont)

3.1.7.18 MCI Business Services II Flex T1 Plus

Product Description The MCI Business Services II Flex Tl Plus is a flexible, integrated solution for small and medium businesses. The Flex Tl Plus must be configured with a minimum of 12 circuits (of which 6 must be Local trunks or lines; but cannot be both). Beyond the 6 required Local trunks or lines, the additional 6 circuits (in DS-0 increments) may be Integrated Internet Access, Frame Relay, Local Lines, Local Trunks, Private Internet Protocol Services and Metro Private Line(Lit buildings only). Customers will be charged a per circuit/trunk charge for the required 12 circuits/trunks and any additional circuits/trunks added on top of the initial 12

Eligibility

Customer must subscribe to service under a MCI Business Services II Agreement or renew under a MCI Business Services II Agreement with a minimum 1 year commitment as described in the Company's "Service Publication and Price Guide" located on the Company's Internet site at www.mci.com.customer must also subscribe to the MCI Business Services II Flex T1 Plus offering as described in MCI WorldCOm Communications, Inc. MO PSC Tariff No. 1 ("Companion Intrastate Service").

The MCI Business Services II Flex T1 Plus cannot be used in conjunction with the MCI Business Services II Local and Long Distance Plan.

Monthly Recurring Charges: A monthly recurring charge will apply for the Plan. This charge is identical to, and shall not be in addition to, any monthly recurring charge applicable to Companion Interstate Service or Companion Intrastate Service. The Offer is available on a per-circuit basis. The following Monthly Recurring Charges apply:

Service Monthly Recurring Charge Flex TlPlus\$40 (\$35 in a Local Lit Building) Per Circuit (minimum of 12 circuits and 6 must be local)

Customers are eligible to receive MCI Business Services II Local discounts on top of the monthly recurring charge for: Integrated Internet Access, Frame Relay, Local, Local Trunks, Private IP service, and Metro Private Line (Lit Buildings Only) ..

Additional benefits

In addition to the monthly recurring charge, customers will receive the benefits of the Companion Intrastate Service as described in MCI WorldCOm Communications, Inc. MO PSC Tariff No. 1.

Non Recurring Charges: Applicable non-recurring charges apply to services under the MCI Business Services II Flex T1 Plus as specified in section 3.1.3.1.3.1, 3.3.2.3.1 and 3.1.3.3.2.1 Customers are eligible to utilize the Local Installation Waiver Promotion. MCI will waive the onetime installation charges which include Port charges and/or DS1 local loop access associated with the implementation of Integrated Internet Access, Frame Relay, Local, Local Trunks, Private Internet Protocol Services, Metro Private Line(Lit Buildings Only) within the 48 contiguous US States under this Agreement. Customer will receive the promotional waiver for the length of the contract term. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: March 1, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: April 1, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

3. <u>Service Description (Cont</u>)

3.1.7.19 Local T1 Rewards Plan

Offer:

Eligible customers, as defined below, who enroll in this Plan and subscribe to facilities based MCI Local T1 and/or PRI Service ("Promotional Service") will receive a waiver of monthly recurring charges for Plan Service, applied to Customer's first, second (if eligible), third (if eligible), following activation of Promotional Service. In addition, customer will receive a waiver of DID block monthly recurring charges for the duration of the Customer's MCI Services Agreement ("Agreement").

Benefit:

<u>Term</u> 1 Year	<u>Benefit</u> Waiver of one month Plan Service and Monthly Recurring charges for DID blocks for duration of Agreement.
2 Year	Waiver of two months Plan Service and Monthly Recurring charges for DID blocks for duration of Agreement.
3+year	Waiver of three months Plan Service and Monthly Recurring charges for DID blocks for duration of

Agreement.

Eligibility:

- New customers and existing customers who enroll in an Agreement and, for the first time, subscribe to MCI Business Services I Local, MCI Business Services I Local Line Solution, MCI Business Services I Local and Long Distance, MCI Business Services Local and Long Distance Line Solution, MCI Business Services II Local or MCI Business Services II Local and Long Distance in combination with MCI Full Local T1/ PRI service.
- Existing MCI Local customers enrolled in an Agreement having a three-year or greater term with a minimum of 12 months remaining who order additional Plan Service are eligible to receive a waiver of three months Plan Service.
- Existing MCI Local customers enrolled in an Agreement having a two-year term commitment with a minimum of 12 months remaining who order additional Plan Service are eligible to receive a waiver of two months Plan Service.
- Plan Benefit applies only to Plan Service ordered during the plan enrollment.
- Customer may not receive the benefits of MCI Loyalty Plus I, MCI Loyalty Plus II and MCI Loyalty Plus III promotions on Plan Service.
- Customers who subscribe to MCI Business Service Flex T1, UNE-P, Local Lines, Trunk and Metered Rate service are not eligible.

ALL MATERIAL ON THIS PAGE NEW.

Issued: September 15, 2005

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

2

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 lst Revised Page No. 55.37 Cancels Original Page No. 55.37

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.20 <u>Verizon Business Services II Install Waiver Plan</u> T The Company will offer this plan to all new business facilities based customers or UNE P delivery customers who convert existing local exchange service from another local exchange carrier to MCImetro Local Services or existing customers adding MCImetro Local Services. To receive the benefits of this plan, customer must commit, at the time of converting to MCImetro Local Service, to at least a one year term commitment. Eligible customers will have the installation charges listed below waived for new circuits implemented within the plan:

> Account Setup Account Charges (including Moves, Changes, Additions and Billing Record Changes) Line Connection Charges (Local Line, Local Trunk-Basic, Local Trunk DID, Local Trunk-2 Way Direct) Direct Inward Dialing (DID)/2 Way Direct Installation for Blocks of DID/2 Way Direct Numbers Non-Recurring Charges for Local ISDN-PRI T-1 installation and optional features Selective Call Screening Non-Recurring Charge Non-Recurring charges for Optional Features Additional Telephone Number Listing (set up charge) Alternative Call Listing (set up charge) Restoral charges (set up charge) Toll Restrictions (set up charge) Call Assistance Install (set up charge) Voice Mail

> Customers who receive service under a Special Contract Arrangement (SCA) are eligible to receive the benefits of this plan.

Issued: April 1, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



MCImetro ACCESS TRANSMISSION SERVICES, LLC

3 4

MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 55.37

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.20 <u>MCI Business Services II Install Waiver Plan</u> The Company will offer this plan to all new business facilities based customers or UNE P delivery customers who convert existing local exchange service from another local exchange carrier to MCImetro Local Services or existing customers adding MCImetro Local Services. To receive the benefits of this plan, customer must commit, at the time of converting to MCImetro Local Service, to at least a one year term commitment. Eligible customers will have the installation charges listed below waived for new circuits implemented within the plan:

> Account Setup Account Charges (including Moves, Changes, Additions and Billing Record Changes) Line Connection Charges (Local Line, Local Trunk-Basic, Local Trunk DID, Local Trunk-2 Way Direct) Direct Inward Dialing (DID)/2 Way Direct Installation for Blocks of DID/2 Way Direct Numbers Non-Recurring Charges for Local ISDN-PRI T-1 installation and optional features Selective Call Screening Non-Recurring Charge Non-Recurring charges for Optional Features Additional Telephone Number Listing (set up charge) Alternative Call Listing (set up charge) Restoral charges (set up charge) Toll Restrictions (set up charge) Call Assistance Install (set up charge) Voice Mail

> Customers who receive service under a Special Contract Arrangement (SCA) are eligible to receive the benefits of this plan.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: September 15, 2005

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: October 15, 2005



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

÷

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 lst Revised Page No. 55.38 Cancels Original Page No. 55.38

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont</u>)

3.1.7.21 <u>Verizon Business Services I Local Availability Enhancement Plan</u> T <u>Offer</u>: Verizon Customers currently enrolled in an On-Net Term Plan are T

eligible to receive Verizon Business Services I Local Pricing as found in the Companion Local Tariffs. Customers adding Local service for the first time to their On-Net Term Plan with a minimum one-year term commitment are eligible to receive the benefits described in the following offers as specified in Sections 3-3.1.7.10, 3-3.1.7.11, 3-3.3.7.12, 3-3.1.7.13:

-	Verizon Business	Services	Ι	Local			Т
-	Verizon Business	Services	Ι	Local	and Long Distance		Т
	Verizon Business	Services	Ι	Local	Line Solution		Т
	Verizon Business	Services	Ι	Local	and Long Distance	Line	T
	Solution				_		

Customers will receive the discounts under the term plan applicable to their agreement in lieu of any other term-based discount.

<u>Eliqibility</u>:

This plan applies only with respect to On-Net term plan customers who do not currently have Verizon as their Local Service Provider. In addition, the plan only applies to new circuits of Verizon Business Services I. Т

Issued: April 1, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Ĉ,

Effective: May 1, 2006



MISSOURI P.S.C. TARIFF NO. 1 Original Page No. 55.38

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont)</u>

MCI Business Services_I Local_Availability_Enhancement_Plan_

<u>Offer</u>:

MCI Customers currently enrolled in an On-Net Term Plan are eligible to receive MCI Business Services I Local Pricing as found in the Companion Local Tariffs. Customers adding Local service for the first time to their On-Net Term Plan with a minimum one-year term commitment are eligible to receive the benefits described in the following offers as specified in Sections 3-3.1.7.10, 3-3.1.7.11, 3-3.3.7.12, 3-3.1.7.13:

- MCI Business Services I Local
- MCI Business Services I Local and Long Distance
- MCI Business Services I Local Line Solution

1

- MCI Business Services I Local and Long Distance Line Solution

Customers will receive the discounts under the term plan applicable to their agreement in lieu of any other term-based discount.

Eligibility:

This plan applies only with respect to On-Net term plan customers who do not currently have MCI as their Local Service Provider. In addition, the plan only applies to new circuits of MCI Business Services I.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: February 8, 2006

Cancelled

May 1, 2006 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: March 9, 2006



ALL MATERIAL ON THIS PAGE WAS REMOVED

-

1

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont'd)</u>

. *

÷

3.1.7.22 Local T1/PRI Lit Building Plan¹

<u>Offer:</u> Customer will receive a monthly recurring charge (Plan Charge) of \$595.00 for Local T1/PRI service.

Eligibility: Customer must:

- Be a new or existing customer adding new circuits into Lit Buildings;
- Be located and provision in a building connected via Company owned fiber to the Company's network (Lit Building);
- Receive service under Verizon Business Service II Local which receives the benefits of this plan under a term of service which equals or exceeds one year;
- Represent the Company's satisfaction, as determined with the company's sole discretion, that it would not permit conversion of facilities without further inducement; and

Agree to allow the Company and its subcontractors and their respective employees and agents access to Customer or Authorized User premises at which service is being or will be provided (including access to associated equipment).

<u>Other Conditions:</u>

Customer may not receive the benefits of Verizon Loyalty Plus I, Verizon Loyalty Plus II, Verizon Loyalty Plus III, Local Tl Rewards and VBS II Local Availability Enhancement Plans or any discounts on the monthly recurring charges that receive the benefits of this plan.

¹Effective February 1, 2007, this plan will no longer be available to new customers. N

Issued: January 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Effective: February 1, 2007



3. <u>Service Description (Cont</u>)

- 3.1.7.22 Local <u>T1/PRI Lit Building Plan</u>
 - <u>Offer:</u> Customer will receive a monthly recurring charge (Plan Charge) of \$595.00 for Local T1/PRI service.

Eligibility: Customer must:

- Be a new or existing customer adding new circuits into Lit Buildings;
- Be located and provision in a building connected via Company owned fiber to the Company's network (Lit Building);
- Receive service under Verizon Business Service II Local which receives the benefits of this plan under a term of service which equals or exceeds one year;
- Represent the Company's satisfaction, as determined with the company's sole discretion, that it would not permit conversion of facilities without further inducement; and

Agree to allow the Company and its subcontractors and their respective employees and agents access to Customer or Authorized User premises at which service is being or will be provided (including access to associated equipment).

Other Conditions:

Customer may not receive the benefits of Verizon Loyalty Plus I, Verizon Loyalty Plus II, Verizon Loyalty Plus III, Local T1 Rewards and VBS II Local Availability Enhancement Plans or any discounts on the monthly recurring charges that receive the benefits of this plan.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601 Effective: May 1, 2006



Cancelled

Feburary 01, 2007 Missouri Public Service Commission

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missouri P.S.C Tariff No. 1 1st Revised Page No. 55.40 Cancels Original Page No. 55.40

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.23 Verizon Loyalty Plus I Plan¹

Offer:

Existing Company customers who simultaneously order a new eligible MCImetro Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: a one-time credit in the amount indicated in the table below, applied to the monthly recurring charges for the new MCImetro Legacy Company-provided service unit. The credit will be applied to the second full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credit is applied. Except for this one-time credit, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCImetro Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCImetro Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2,3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com on any enrolled unit.

TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group <u>Voice</u>	Sub-Product	Service Unit	Compa <u>Credit Value</u>	anion Installation Charge Waiver <u>(Yes/No)</u>
	Local Trunk Local and Long	Trunk	\$50	Y
	Distance Service Solution II Local T1 Local PRI	Trunk Trunk Tl Circuit PRI Circuit	\$50 \$500 \$500	Y Y Y

¹Effective July 1, 2007, this plan will no longer be available to new customers. N

Issued; May 31, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission Effective: July 1, 2007

Missourri P.S.C. Tariff No. Original Page No. 55.40

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.23 Verizon Loyality Plus I Plan

<u>Offer</u>

<u>Order:</u> Existing Company customers who simultaneously order a new eligible MCImetro Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: a one-time credit in the amount indicated in the table below, applied to the monthly recurring charges for the new MCImetro Legacy Company-provided service unit. The credit will be applied to the second full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer' ĥhe activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credit is applied. Except for this one-time credit, this plan does not change the terms and conditions that apply to the new plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, monthly charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCImetro Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCImetro Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com on any enrolled unit. www.verizonbusiness.com on any enrolled unit.

TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

				n Installation arge Waiver
Product Group <u>Voice</u>	<u>Sub-Product</u>	<u>Service Unit</u>	<u>Credit Value</u>	<u>(Yes/No)</u>
<u></u>	Local Trunk Local and Long Distance Service-Trunk	Trunk	\$50	Y
	Solution II	Trunk	\$50	Y
	Local Tl	T1 Circuit	\$500	Y
	Local PRI	PRI Circuit	\$500	Y

ALL MATERIAL ON THIS PAGE IS NEW

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 1, 2006



Cancelled July 1, 2007 Missouri Public Service Commission

i

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.41 Cancels Original Page No. 55.41

LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont'd)</u>

1

3.1.7.24 <u>Verizon Loyalty Plus II Plan</u>¹

Offer:

Existing Company customers who simultaneously order a new eligible MCImetro Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: two credits in the amount indicated in the table below, applied to the monthly recurring charges for the new MCImetro Legacy Company provided service unit. The credits will be applied to the second full-month's and the sixth full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credits are applied. Except for these two credits, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCImetro Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCImetro Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com on any enrolled unit.

TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group <u>Voice</u>	Sub-Product	<u>Service Unit</u>	<u>Credit Value</u>	Companion Installation Charge Waiver <u>(Yes/No)</u>
<u>0100</u>	Local Trunk Local and Long Distance Service	Trunk	\$50	Y
	Trunk Solution II	Trunk	\$50	Y
	Local T1	Tl Circuit	\$500	Y
	Local PRI	PRI Circuit	\$500	Y

¹Effective February 1, 2007, will no longer be available to new customers.

Ν

Issued: January 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: February 1, 2007



Missourri P.S.C. Tariff No. Original Page No. 55.41

LOCAL EXCHANGE SERVICE

Service Description (Cont) 3.

3.1.7.24 Verizon Loyalty Plus II Plan

Offer

Offer: Existing Company customers who simultaneously order a new eligible MCImetro Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: two credits in the amount indicated in the table below, applied to the monthly recurring charges for the new MCImetro Legacy Company provided service unit. The credits will be applied to the second full-month's and the sixth full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credits are applied. Except for these two credits, this plan does not change the terms and conditions that apply to the new service unit. that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, monthly charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCImetro Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCImetro Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan.

Customer may enroll any eligible service unit, but any individual unit may be enrolled only once. Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com on any enrolled unit.

TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group	Sub-Product	<u>Service Unit</u>	<u>Credit Value</u>	Companion Installtion Charge Waiver <u>(Yes/No)</u>
<u>Voice</u>	Local Trunk Local and Long Distance Service-	Trunk	\$50	У
	Solution II Local T1 Local PRI	Trunk T1 Circuit PRI Circuit	\$50 \$500 \$500	Y Y Y

ALL MATERIAL ON THIS PAGE IS NEW

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 1, 2006



Feburary 01, 2007

Cancelled

1

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missourri P.S.C. Tariff No. 1 1st Revised Page No. 55.42 Cancels Original Page No. 55.42

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.25 Verizon Loyalty Plus III Plan¹

Offer

Offer: Existing Company customers who simultaneously order a new eligible MCImetro Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: three credits in the amount indicated in the table below, applied to the monthly recurring charges for the new MCImetro Legacy Company service unit. The credits will be applied to the second full-month's, sixth full-month's, and twelfth full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation each eligible service unit must be active, with no pending cancellation request, at the time the credits are applied. Except for these three credits, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly recurring charges, expedite charges, change charges, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCImetro Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCImetro Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan. Customer may enroll any eligible service unit, but any individual unit may be enrolled only once.

Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com on any enrolled unit.

TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group <u>Voice</u>	Sub-Product_	<u>Service Unit</u>	Compa <u>Credit Value</u>	anion Installtion Charge Waiver <u>(Yes/No)</u>
<u></u>	Local Trunk Local and Long Distance Service-	Trunk	\$50	Y
	Solution II Local T1 Local PRI	Trunk Tl Circuit PRI Circuit	\$50 \$500 \$500	Y Y Y

¹Effective September 1, 2006, this plan will no longer be available to new customers. Ν

Issued: August 1, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Ave. Chicago, IL 60601

Effective: September 1, 2006



Missourri P.S.C. Tariff No. 1 Original Page No. 55.42

LOCAL EXCHANGE SERVICE

3. Service Description (Cont)

3.1.7.25 Verizon Loyalty Plus_III_Plan

<u>Offer</u>:

Offer: Existing Company customers who simultaneously order a new eligible MCImetro Legacy Company service "unit" (see table below), excluding upgrades and conversions on existing service and enroll that unit in this plan will receive the following benefit: three credits in the amount indicated in the table below, applied to the monthly recurring charges for the new MCImetro Legacy Company service unit. The credits will be applied to the second full-month's, sixth full-month's, and twelfth full-month's invoice following activation of the new service unit. Customer may designate that the credit be applied either as an invoice credit or a deposit in Customer's Fund account. To receive the benefits of this plan, each eligible service unit must be active, with no pending cancellation request, at the time the credits are applied. Except for these three credits, this plan does not change the terms and conditions that apply to the new service unit.

Customers will also receive a waiver of the non-recurring Company-billed installation charges for installation following plan enrollment of all eligible service units identified below. Usage charges, monthly charges imposed by third parties (including access, surcharges, any charges imposed by third parties (including access, egress, jack, or wiring charges), taxes or tax-like surcharges, or other Governmental Charges will not be waived.

This plan applies only with respect to new eligible MCImetro Legacy Company provided service units ordered for installation within 30 days of order (or by Company's quoted installation date if later), by a Customer with at least one MCImetro Legacy Company service which has been actively billing for at least 90 days prior to enrollment in this plan. Customer may enroll any eligible service unit, but any individual unit may be enrolled only once.

Customer may not receive: (i) any discounts, (ii) the benefits of: any Special Customer Arrangement (SCA) or Product Package, other than Special Customer Arrangements (SCA) Guide Types 2, 3,4,5,6,7,8 or 9 or Produce Package SCA Guide Types 13, 14, 15, 16, 18, 19, 20 and 21, or (iii) the benefits of Verizon Business Services I Flex T1 Promotion, Regional Plus Frame Relay Promotion and Competitive Voice 2 Promotion as described in the Company's "Service Publication and Price Guide" (The Guide) located on the Company's website at www.verizonbusiness.com on any enrolled unit.

TABLE OF ELIGIBLE COMPANY SERVICE UNITS AND CREDIT AMOUNTS

Product Group	Sub-Product	<u>Service_Unit_</u>	Compa <u>Credit Value</u>	nion Installtion Charge Waiver <u>(Yes/No)</u>
Voice	Local Trunk Local and Long Distance Service-	Trunk	\$50	Y
	Solution II Local T1 Local PRI	Trunk Tl Circuit PRI Circuit	\$50 \$500 \$500	Y Y Y

ALL MATERIAL ON THIS PAGE IS NEW

Issued: April 1, 2006

Carmen L. Feliciano Tariff Administrator Chicago, IL 60601

Effective: May 1, 2006



September 1, 2006 Missouri Public Service Commission

Cancelled

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

-

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missouri P.S.C Tariff No. 1 Original Page No. 55.42.1

Effective: November 1, 2007

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local

Eligibility: To be eligible for this program, customers:

- must subscribe to service under Product Package Guide Type 22 or 23 as described in the Company's "Service Publication and Price Guide located on the Company's Internet site at www.verizonbusiness.com/publications/service_guide
- must be a new business services facilities-based customer or a new business services customer provisioned via UNE-Platform (UNE-P), or an existing business services facilities-based customer or an existing business services customer provisioned via UNE-P who is eligible for renewal under their existing term plan agreement.

Definitions:

Eligible Charges: Monthly recurring charges for Local Line, Local Trunk-Basic, Local Trunk-DID, Local Trunk-2 Way Direct, Local ISDN-PRI T-1 charge, Local Trunk T-1 charge, DID number charge and optional features.

Features:

The following optional features are available for Local Line service in addition to the Local Line Standard Features. Applicable non-recurring and monthly recurring charges for optional features will apply as specified in Section 3.1.2.3.1 and 3.1.2.3.2 except for Feature Package 1 monthly recurring charges which are specified within this program.

Local Line Standard Features Feature Package 1 Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) Voice Mail

Features and applicable feature charges for ISDN-PRI are available as described in Section 3.1.5.3.1 and 3.1.5.3.2 Grouping of telephone numbers can be obtained in blocks of 20 DID numbers or 100 DID numbers, as applicable, for Local Trunk-DID and Local Trunk-2 Way Direct service. Applicable monthly recurring charges will apply for blocks of 20 DID or 100 DID numbers, as applicable, numbers as specified in Section 3.1.3.0.3.2

Non-Recurring Charges: Applicable non-recurring charges apply to services under this program as specified section 3.3.2.3.1 except for non-recurring charges which are specific within this program.

Order Expedite Charge \$60.00 (per line or trunk) T-1 Order Expedite Charge \$1400.00

(per T-1)

Monthly Charges: The following flat rate monthly recurring charge applies in lieu of monthly recurring charges for these services as specified elsewhere in this tariff:

ALL MATERIAL ON THIS PAGE IS NEW

Issued; October 1, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

CANCELLED September 1, 2008 Missouri Public Service Commission

ALL MATERIAL ON THIS PAGE WAS REMOVED

-

2

-

З.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Ν

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a	Verizon	Business	Services	III	Local	(Cont'd)	

Monthly Recurring Charge (Cont'd)

Flat Rate Pricing Structure

Local	Line (per line) Plan 1: Plan 2:	\$30.00 \$30.00
	Local Trunks (Basic, DID and 2 Way Direct) (Per T-1) 1	\$672.00
	Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$655.00
	Local ISDN-PRI (Per T-1)	\$655.00
	Feature Package 1	\$5.00
	Remote Call Forwarding	\$20.00
	Remote Access to Call Forwarding (Plan 2)	\$7.00
	Voice Mail	\$5.00
	Local ISDN-PRI Caller ID with Name (Per T-1)	\$60.00

¹This rate is grandfathered as of June 20, 2008. Current customer will continue to N receive this rate until their current agreement expires on existing trunks (Per T-1) | only. If additional new trunks (Per T-1) are added, the customer will pay the new | rate on the new trunks (Per-T-1), effective June 20, 2008. This rate will not be | offered to renewals or new customers.

Issued: June 2, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: July 1, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Monthly Recurring Charge (Cont'd)	
Flat Rate Pricing Structure	
Local Line (per line) Plan 1: Plan 2:	\$30.00 \$30.00
Local Trunks (Basic, DID and 2 Way Direct) (Per T-1)	\$672.00
Local ISDN-PRI (Per T-1)	\$655.00
Feature Package 1	\$5.00
Remote Call Forwarding	\$20.00
Remote Access to Call Forwarding (Plan 2)	\$7.00
Voice Mail	\$5.00
Local ISDN-PRI Caller ID With Name (Per T-1)	\$60.00

ALL MATERIAL ON THIS PAGE IS NEW

Issued; October 1, 2007

CANCELLED July 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.42.3 Cancels Original Page No. 55.42.3

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE WAS REMOVED

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Missouri P.S.C Tariff No. 1 Original Page No. 55.42.3

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Benefits:

Upon installation of Companion Intrastate Service, Companion Interstate Service and Local Service, customers will receive unlimited local exchange service usage.

Unless the state has established a different requirement, for Customers who reside within the densest zone of the Top 50 MSA, as defined by the FCC in Docket No. 99-98, service is limited to a maximum of three lines per location. Customers whose service location resides outside of the most dense zone of the Top 50 MSA may order an unlimited number of lines.

Discounts:

These discounts are identical to, and shall not be in addition to, discounts applicable to the Verizon Business Services III Local pricing plan for intrastate and interstate long distance service.

A discount will be provided on the monthly recurring charges, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 20% discount, or 2 or more years for up to a 25% discount.

The following disclaimers apply to Stand Alone Verizon Business Services III Local Line- based Service in addition to those set forth in the Service Attachment. Customer understands that use of the Service is restricted in the following manner: (i) Customer may not utilize auto-dialers or any similar type of device in connection with the Service; and (ii) Customer may not utilize the Service in any call center environment or in connection with any similar such application. CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS ON ITS USE OF THE SERVICE WILL RESULT IN THE IMMEDIATE TERMINATION OF THE SERVICE BY Verizon. Verizon will install the Linebased Service from the point of the local exchange carrier's smartjack to the Customer's premises. Customer will be responsible for all inside wiring and special construction charges.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

Missouri P.S.C Tariff No. 1 1st Revised Page No. 55.42.4 Cancels Original Page No. 55.42.4

- -- --

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a

••

1

D |-| | D

Issued: November 29, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

Missouri P.S.C Tariff No. 1 Original Page No. 55.42.4

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice Package

LD Voice Package Customer will pay the following flat rate monthly recurring charge ("MRC"), fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 20,000 minutes per month originating from a T1 or PRI purchased from Verizon.

Agreement TermMRC*One Year\$725.00Two Years\$685.00Three Years\$650.00

*Overage Rate: Customers will pay a per-minute charge of \$.05 for each minute in excess of 20,000 in a month

Customer understands that the LD Voice Package is restricted in the following manner:

- (i) International long distance, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer may not utilize auto-dialers or any similar type of device in connection with the LD Voice Package.
- (iii) Customer may not utilize the LD Voice Package in any call center environment or in connection with any such similar environment.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION OF THE LD VOICE PACKAGE BY VERIZON AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

Missouri P.S.C Tariff No. 1 1st Revised Page No. 55.42.5 Cancels Original Page No. 55.42.5

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a

÷ , †

ļ

D I I D

Issued: November 29, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

FILED Missouri Public Service Commission MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice 800 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 800 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 800 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 800 Minute Package.

Agreement TermMRC*One Year\$28.00Two years\$26.00Three Years\$24.00

*Overage Rate; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 800 in a month

Customer understands that the LD Voice 800 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 800 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services III Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 800 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007



Missouri P.S.C Tariff No. 1 1st Revised Page No. 55.42.6 Cancels Original Page No. 55.42.6

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a

s - *

D | | | D

Issued: November 29, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008

FILED Missouri Public Service Commission -----

Missouri P.S.C Tariff No. 1 Original Page No. 55.42.6

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice 500 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 500 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 500 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 500 Minute Package.

Agreement TermMRC*One Year\$18.00Two years\$17.00Three Years\$16.00

* Overage Rate; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 500 in a month

Customer understands that the LD Voice 500 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 500 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services III Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 500 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

CANCELLED January 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

· ··· -- ---- ······

Missouri P.S.C Tariff No. 1 1st Revised Page No. 55.42.7 Cancels Original Page No. 55.42.7

MCI ON-NET LOCAL EXCHANGE SERVICE

3. <u>Service Description (Cont'd)</u>

3.1.7.25a

. . *

ł

ł

D | | | D

Issued: November 29, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: January 1, 2008
MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Optional Calling Plans (Cont'd)

LD Voice 300 Minute Package for Verizon Business Lines or Trunks If Customer selects the Long Distance (LD) Voice 300 Minute Package, Customer will pay the following flat rate monthly recurring charge ("MRC"), which is fixed for the Term of the Agreement, for all intrastate and interstate outbound Long Distance calls from any single Customer site up to 300 minutes per month originating from a Verizon business line or Verizon trunk (via Local Service - CLEC or Local Service - ILEC) associated with the LD Voice 300 Minute Package.

Agreement Term	MRC*
One Year	\$12.00
Two years	\$11.00
Three Years	\$10.00

* <u>Overage Rate</u>; Customer will pay a per-minute charge of \$.05 for each minute of usage in excess of 300 in a month

Customer understands that the LD Voice 300 Minute Package is restricted in the following manner:

- (i) International LD, Inbound (toll free), and calling card minutes are NOT included.
- (ii) Customer must have a valid Verizon Local Voice Service contract and must not have any other Verizon Outbound Domestic LD Voice service on the business line or trunk associated with the LD Voice 300 Minute Package.
- (iii) Discounting of the MRC beyond the Term rates specified is not permitted.
- (iv) The Overage Rate may be discounted only in accordance with the current Verizon Business Services III Long Distance Voice Term and Volume Discount Schedule.

CUSTOMER EXPRESSLY ACKNOWLEDGES THAT ANY VIOLATION OF THE FOREGOING RESTRICTIONS WILL RESULT IN THE IMMEDIATE TERMINATION BY VERIZON OF THE LD VOICE 300 MINUTE PACKAGE AND THE RESUMPTION OF STANDARD RATES FOR AFFECTED SERVICES.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

3.1.7.25a Verizon Business Services III Local (Cont'd)

Verizon Business Services III Install Waiver Plan The Company will offer this plan to all new business facilities based customers or UNE P delivery customers who convert existing local exchange service from another local exchange carrier to MCImetro Local Services or existing customers adding MCImetro Local Services. To receive the benefits of this plan, customer must commit, at the time of converting to MCImetro Local Service, to at least a one year term commitment. Eligible customers will have the installation charges listed below waived for new circuits implemented within the plan:

Account Setup Account Charges (including Moves, Changes, Additions and Billing Record Changes) Line Connection Charges (Local Line, Local Trunk-Basic, Local Trunk DID, Local Trunk-2 Way Direct) Direct Inward Dialing (DID)/2 Way Direct Installation for Blocks of DID/2 Way Direct Numbers Non-Recurring Charges for Local ISDN-PRI T-1 installation and optional features Selective Call Screening Non-Recurring Charge Non-Recurring charges for Optional Features Additional Telephone Number Listing (set up charge) Alternative Call Listing (set up charge) Restoral charges (set up charge) Toll Restrictions (set up charge) Call Assistance Install (set up charge) Voice Mail

Customers who receive service under a Special Contract Arrangement (SCA) are eligible to receive the benefits of this plan.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

ALL MATERIAL ON THIS PAGE WAS REMOVED

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C Tariff No. 1 Original Page No. 55.42.9

MCI ON-NET LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

۰.

3.1.7.25a Verizon Business Services III Local (Cont'd)

Verizon Business Services III Local Availability Enhancement Plan

Offer:

Verizon Customers currently enrolled in an On-Net Term Plan are eligible to receive Verizon Business Services III Local Pricing as found in the Companion Local Tariffs. Customers adding Local service for the first time to their On-Net Term Plan with a minimum one-year term commitment are eligible to receive the benefits described in the following offers as specified in Sections 3-3.1.7.10, 3-3.1.7.11, 3-3.3.7.12, 3-3.1.7.13:

- Verizon Business Services I Local
- Verizon Business Services I Local and Long Distance
- Verizon Business Services I Local Line Solution
- Verizon Business Services I Local and Long Distance Line Solution

Customers will receive the discounts under the term plan applicable to their agreement in lieu of any other term-based discount.

Eligibility:

This plan applies only with respect to On-Net term plan customers who do not currently have Verizon as their Local Service Provider. In addition, the plan only applies to new circuits of Verizon Business Services III.

ALL MATERIAL ON THIS PAGE IS NEW

Issued: October 1, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: November 1, 2007

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 Original Page No. 55.42.10

LOCAL EXCHANGE SERVICE

3. Service Description (Cont'd)

Verizon Business Services Flex Tl Plus Plan

Verizon Business Services Flex T1 Plus is a flexible, integrated service. The Flex T1 Plus must be configured with a minimum of 12 circuits (of which 6 must be Local Trunks or Local Lines; but cannot be both). Beyond the 6 required Local Trunks or Local Lines, the additional 6 circuits (in DS-0 increments) may be Integrated Internet Access (unregulated), Frame Relay Port (unregulated), Local Lines, Local Trunks, Private Internet Protocol ("IP") Services Access (unregulated) and Metro Private Line (unregulated and available in Lit buildings only). Customers will be charged a per circuit/trunk charge for the required 12 circuits/trunks and any additional circuits/trunks.

Eligibility:

7

Customer must subscribe to or renew service under a Verizon Business Services Agreement with a minimum 1 (one) year commitment as described in the Service Publication and Price Guide. Verizon Business Services Flex T1 Plus cannot be used in conjunction with the Verizon Business Services Local and Long Distance Plan.

Rates and Rate Application: The following monthly recurring charges apply:

Monthly Recurring Charge

\$40 (\$35 in a Local Lit Building) Per Circuit (Minimum of 12 circuits and 6 must be local)

The Offer is available on a per-circuit basis. Applicable non-recurring charges apply to services under the Verizon Business Services Flex T1 Plus as specified in Local Line and Local Trunk service descriptions.

Discounts:

Customers are eligible to receive Verizon Business Services local discounts on the Flex T1 monthly recurring charge for: Integrated Internet Access (unregulated), Frame Relay Port (unregulated), Local Lines, Local Trunks, Private IP Service Access (unregulated), and Metro Private Line (unregulated, and available in Lit Buildings Only).

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: February 7, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: March 10, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.42.11 Cancels Original Page No. 55.42.11

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

-

÷

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Local Exchange Tariff

3. Service Description (Cont'd)

Local Voice - Line Rewards PLUS Plan

Offer:

Eligible customers, as defined below (individually, a "Customer"), who enroll in this plan and order Local-CLEC Local Line Service ("Plan Service") will receive the following promotional benefits as shown below for each Plan Circuit, as applicable, based upon the Term of the Customer's Verizon Business service agreement("Agreement").

Agreement Term	Benefit (applied as applicable, to Customer's first, second and third invoice(s) following activation of Promotional Service) "MRC" refers to the monthly recurring charge.
One (1) year	Credit of one (1) month's MRC for the Plan Service and one (1) free month of Voice Mail Service.
Two (2) years	Credit of two (2) months MRC for the Plan Service and two (2) free months of Voice Mail Service.
Three Plus (3+) years	Credit of three (3) months MRC for the Plan Service and three (3) free months of Voice Mail Service.

Existing:

Verizon Local-CLEC customers subscribed to an Agreement -

With a three-year or greater Term with a minimum of 12 months remaining in the Term who order additional Plan Service are eligible to receive a credit of three (3) months MRC for the Plan Service.

With a two-year Term with a minimum of 12 months remaining in the Term who order additional Plan Service are eligible to receive a credit of two (2) months MRC for the Plan Service.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: March 7, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: April 8, 2008

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Local Exchange Tariff

3. Service Description (Cont'd)

Local Voice - T1/PRI Rewards Plus Promotion

Offer:

. ·

Eligible new customers, as defined below (individually, a "Customer"), who enroll in this plan and order Local-CLEC T1 and/or PRI Service ("Plan Service") will receive the Benefits listed immediately below, applied - as applicable based on the Term of the Customer's Verizon Business service agreement (the "Agreement") - to Customer's first, second, and third invoice(s) following activation of the Plan Service.

AgreementBenefitTerm"MRC" refers to the monthly recurring charge.		
One (1) year	Credit of one (1) month's MRC for the Plan Service, all MRCs for DID blocks, and five (5) instances of Remote Call Forwarding for the duration of the Agreement.	
Two (2) years	Credit of two (2) months' MRCs for the Plan Service, all MRCs for DID blocks, and five (5) instances of Remote Call Forwarding for the duration of the Agreement.	
Three-plus (3+) years	Credit of three (3) months' MRCs for the Plan Service, all MRCs for DID blocks, and five (5) instances of Remote Call Forwarding for the duration of the Agreement.	

Existing Verizon Local-CLEC customers subscribed to an Agreement -

With a three-year or greater Term with a minimum of 12 months remaining in the Term who order additional Plan Service are eligible to receive a credit of three (3) months MRC for the Plan Service.

With a two-year Term with a minimum of 12 months remaining in the Term who order additional Plan Service are eligible to receive a credit of two (2) months MRC for the Plan Service.

Eligibility:

Plan Benefits apply only to Plan Service ordered during the Plan enrollment period.

Customer may not receive the benefits of Verizon Loyalty Plus I, Verizon Loyalty Plus II, Verizon Loyalty Plus III Plans on this Plan Service.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: March 7, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: April 8, 2008

CANCELLED September 1, 2008 Missouri Public Service Commission

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

-

2

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

. .

Missouri P.S.C. Tariff No. 1 Original Page No. 55.43

LOCAL EXCHANGE SERVICE

- -

3. <u>Service Description (Cont'd</u>)

.

2 · · ·

- 3.1.7.26 <u>Multi-State Local Service Program II</u> Multi-State Local Service Program II is available to any Qualifying Customer. A Qualifying Customer is one that meets all of the following requirements:
 - Customer must agree to subscribe to all of the following services: Local Line, Local Trunk, Local Trunk T1/PRI, Metered Line, Local Metered T1, Local and Long Distance Line Solution, Local and Long Distance Trunk Solution.
 - Customer must be a new Verizon Business customer.
 - Customer must agree to a new three-year term of at least \$2.5 million annual spending on Total Service Charges.
 - Customer must subscribe to Qualifying Service in a minimum of 49 states.

The following Monthly Recurring Charges ("MRC") are available to Qualifying Multi-State Local Service Program II Customers:

Product	MRC
Local Line (Kansas City, Springfield, St. Louis)	\$25.00
Local Line (rest of state)	\$33.00
Local Trunk	\$40.00
Local Trunk Tl/PRI	\$672.00
Metered Line (Kansas City, Springfield, St. Louis)	\$22.00
Metered Line (rest of state)	\$32.00
Local Metered Tl	\$540.00
Local and Long Distance Line Solution (Kansas City, Springfield, St. Louis)	\$44.00
Local and Long Distance Line Solution (rest of state)	\$79.00
Local and Long Distance Trunk Solution (per trunk)	\$65.00
Local and Long Distance Trunk Solution (per T1/PRI)	\$1,400.00
Qualifying Customers will receive a further discount of the MRCs listed above.	of 35% off

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: October 31, 2006

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: December 1, 2006



MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 55.44 Cancels Original Page No. 55.44

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services Missouri P.S.C. Tariff No. 1 Original Page No. 55.44

LOCAL EXCHANGE SERVICE

з. Service Description (Cont'd)

3.1.7.26 Multi-State Local Service Program II (Cont'd)

> In addition, the following per minute rates for Metered Line and Local Metered T1 are available to Qualifying Customers:

Product	<u>Per Minute Rate</u>
Metered Line (Kansas City, Springfield, St. Louis)	\$0.0150
Metered Line (rest of state)	\$0.0150

Local Metered T1 \$0.0050

"Total Service Charges" shall mean all charges, after application of all discounts and credits, incurred by Qualifying Customer for Qualifying Services provided, specifically excluding: Taxes; Charges for equipment (unless otherwise agreed upon); Verizon Wireless charges; Charges incurred for goods or services where Verizon acts as agent for Qualifying Customer in its acquisition of goods or services; Non-recurring charges; Governmental Charges; International pass-through access charges (i.e., Type 3/PTT) and charges for international access provided by Verizon (i.e., Type 1); and Other charges expressly excluded by this tariff.

"Qualifying Services" shall include, but are not limited to Local Line, Local Trunk, Local Trunk T1/PRI, Metered Line, Local Metered T1, Local and Long Distance Line Solution, Local and Long Distance Trunk Solution.

ALL MATERIAL ON THIS PAGE IS NEW.

Issued: October 31, 2006

CANCELLED September 1, 2008 **Missouri Public** Service Commission

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

Effective: December 1, 2006



Missouri P.S.C. Tariff No. 1 3rd Revised Page No. 56 Cancels 2nd Revised Page No. 56

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

-

-

3.

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 2nd REVISED PAGE NO. 56 CANCELS 1ST REVISED PAGE NO. 56 MISSOURI PUBLIC Service Commission

RECD JUL 1 2 2002

LOCAL EXCHANGE SERVICE

3.2 <u>Directory Assistance</u>

Service Descriptions (Cont'd)

A Customer may obtain Directory Assistance in determining telephone numbers within its local calling area by calling the Directory Assistance operator. This directory assistance service is provided to business customers only.

3.2.1 Each call to Directory Assistance will be charged as follows:

<u>St Louis/H</u>	Kansas City	<u>Springfield</u>	n agina sa sa -
Per Call	\$0.48	\$0.48	

The charge applies to each call regardless of whether The Directory Assistance Bureau is able to furnish the requested telephone number one request may be made on each Directory Assistance Call.

3.2.2 A credit will be given for calls to Directory Assistance as follows:

 The Customer experiences poor transmission or is cut-off during the call; or

The Customer is given an incorrect telephone number.

To obtain such a credit, the Customer must notify its Customer Service representative.

Missouri Public Service Commission

FILED AUG 1 2 2002

Issued: July 12, 2002

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L.Feliciano Tariff Administrator Suite 1100 Chicago, IL 60601 Effective: August 12, 2002

MCImetro ACCESS TRANSMISSION SERVICES, LLC

3.

MISSOURI P.S.C. TARIFF NO. 1 1ST REVISED PAGE NO. 56 CANCELS ORIGINAL PAGE NO. 56

LOCAL EXCHANGE SERVICE

Service Descriptions (Cont'd)

3.2 Directory Assistance

Service Commission

A Customer may obtain Directory Assistance in determining telephone numbers within its local calling area by calling 1999 the Directory Assistance operator. This directory assistance service is provided to business customers only.

3.2.1 Each call to Directory Assistance will be charged as follows:

	St Louis/Kansas City		Springfiel	<u>d</u>
Per Call	\$0.48	1	\$0.48	I

The Customer may request a maximum of two telephone numbers per call to Directory Assistance service.

3.2.2 A credit will be given for calls to Directory Assistance as follows:

- The Customer experiences poor transmission or is cut-off during the call; or
- The Customer is given an incorrect telephone number.

To obtain such a credit, the Customer must notify its Customer Service representative.

CANCELLED 2002 2S54 unissi**o**r - 501

CANCE AUG 98-575 £γ Public Service Commission MISSOURI

Sei

FILED JAN 13 2000

ISSUED : December 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 EFFECTIVE: January 13, 2000

MISSOURI P.S.C. TARIFF NO. 1

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

3.2 Directory Assistance

MO. FUBLIC SERVICE COMM

SEP 17 1998

A Customer may obtain Directory Assistance in determining telephone numbers within its local calling area by calling the Directory Assistance operator. This directory assistance service is provided to business customers only.

3.2.1 Each call to Directory Assistance will be charged as follows:

	St Louis/Kansas City	Springfield
Per Call	\$0.30	\$0.30

The Customer may request a maximum of two telephone numbers per call to Directory Assistance service.

3.2.2 A credit will be given for calls to Directory Assistance as follows:

The Customer experiences poor transmission or is cut-off during the call; or

The Customer is given an incorrect telephone number.

To obtain such a credit, the Customer must notify its Customer Service representative.

CANCELLED

JAN 1 3 2000 By トゲ RP56 Public Service Commission MISSOURI



ISSUED : March 2, 1999

L

<u>96-3</u>55 EFFECTIVE: May 14, 1999

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission

FILED MAY 1 4 1999

)



ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro ACCESS TRANSMISSION SERVICES, LLC

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 57

LOCAL EXCHANGE SERVICE

SFP 17 1998

3. Service Descriptions (Cont'd)

3.3 **Operator Assistance**

MO. PUDLIC ACTIVICE COMM

A Customer may obtain the assistance of a local operator to complete local exchange telephone calls in the following manner. In addition to the rates specified in Section 3.1, surcharges as specified in Section 3.3.1 will apply:

Third Number Billing: Provides the Customer with the capability to charge a local call to a third number which is different from the called or calling party. The party answering at the third number has the option to refuse acceptance of the charges in advance or when queried by the operator.

Collect Calls: Provides the Customer with the capability to charge a call to the called party. On the operator announcement of a collect call, the called party has the option to refuse acceptance of charges in advance or when queried by the operator.

Calling Cards: Provides the Customer with the capability to place a call using a calling card of an Interexchange Carrier with or without the assistance of an operator.

Person to Person: Calls completed with the assistance of an operator to a particular Station and person specified by the caller. The call may be billed to the called party.

Station to Station: Calls completed with the assistance of an operator to a particular Station. The call may be billed to the called party.

General Assistance: The Customer has the option to request general information from the operator, such as dialing instructions, country or city codes, area code information and Customer Service 800 telephone numbers, but does not request the operator to complete the call.

3.3.1 Operator Assisted Surcharges: The following surcharges will be applied on a per call basis.

	St Louis/Kansas City	<u>Springfield</u>
Third Number Billing	\$2.40	\$2.40
Collect Calling	\$1.00	\$1.00
Person to Person	\$2.40	\$2.40
Station to Station	\$1.10	\$1.10
General Assistance	N/C	N/C



Busy Line Verification and Interrupt Service: Busy Line Verification and Interrupt Service, which is furnished where and to the extent that facilities permit, provides the Customer with the following options:

3321 Busy Line Verification: Upon request of the calling party, the Company will determine if the line is clear or in use and report to the calling party.

3.3.2.2 Busy Line Verification with Interrupt. The operator will interrupt the call on the called line only if the calling party indicates an emergency and requests interruption. 96-355

ISSUED : March 2, 1999

L

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

CANCELLED September 1, 2008 **Missouri Public** Service Commission FILED MAY 1 & 1999

EFFECTIVE: May 14, 1999 Missouri Public Sorvico Commission

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

3.

-

-

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro ACCESS TRANSMISSION SERVICES, LLC

3.3

3.

MISSOURI P.S.C. TARIFF NO. 1 **ORIGINAL PAGE NO. 58**

				LOCAL EXCHANGE SERVICE	RECEIVED
<u>Service</u> 3.3	Descriptions		(0#4)		SEP 17 1999
5.5	<u>0087810</u> 3.3.2	r Assistance Busy Line		and Interrupt Service: (Cont'd)	MU. TUDLIU OLTIVIUL UNIM
		3.3.2.3	<u>Rates</u> : Ra circumsta		nterrupt Service will apply under the following
			3.3.2.3.1	The operator verifies that the line Verification charge described belo	is busy with a call in progress. The Busy Line w will apply.
			3.3.2.3.2	The operator verifies that the line Verification charge described belo	is available for incoming calls. The Busy Line w will apply.
			3.3.2.3.3	Customer requests interruption. the called party the name of the c	ed number is busy with a call in progress and the The operator will then interrupt the call, advising alling party. The Busy Line Verification charge (e, as described below, will both apply.
					St Louis/Kansas City Springfield

	E	er Request
Busy Line Verification	\$1.20	\$1.20
Busy Line Interrupt	\$1.85	\$1.85

CANCELLED AUG 2 0 1999 TA- 88-575 Public Service Commission MISSOURI

ISSUED : March 2, 1999

L

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

96-355

EFFECTIVE: May 14, 1999 Missouri Public Service Commission

FILED MAY 1 & 1999

CANCELLED September 1, 2008 **Missouri Public** Service Commission MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri P.S.C. Tariff No. 1 1st Revised Page No. 59 Cancels Original Page No. 59

LOCAL EXCHANGE SERVICE

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

3.

Issued: July 31, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

MCImetro ACCESS TRANSMISSION SERVICES, LLC MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 59

RECEIVED

SEP 17 1999

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

£

3.3 Operator Assistance (Cont'd)

- 3.3.3 Operator Service Requirements MU. FUDLIU JERVIUE UDIMIN
 - Company will not bill for incomplete calls where answer supervision is available. Company will not bill for incomplete calls and will remove any charges for incomplete calls upon (I) subscriber notification or (ii) Company's knowledge.
 - 2) The caller and billed party, if different from the caller, will be advised that Company is the operator service provider at the time of the initial contact.
 - Rate quotes will be given upon request, at no charge, including all rate components and any additional charges.
 - Only tariffed rates approved by this Commission for Company shall appear on any local exchange telephone company (LEC) billings.
 - 5) Company shall be listed on the LEC billing if the LEC has multi-company billing ability.
 - 6) Company will employ reasonable calling card verification procedures, acceptable to the telephone company issuing the calling card.
 - Company will route all 0- or 00- emergency calls in the quickest possible manner to the appropriate local emergency service provider, at no charge.
 - Upon request, Company will transfer calls to other authorized interexchange Company or to the LEC, if billing can list the caller's actual origination point.
 - Company will refuse operator services to traffic aggregators which block access to other Companies.
 - 10) Company will assure that traffic aggregators will post and display information including: (1) that Company is the operator service provider, (2) detailed complaint procedures; and (3) instructions informing the caller on procedures to reach the LEC operator and other authorized interexchange Companies.

CANCELLER AUG 2 0 1999 TA/98-575 By THY 48-575 rublic Service Commission 96-355 MISSOURY ISSUED : March 2, 1999 EFFECTIVE: May 14, 1999 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 FILED MAY 1 4 1999 Atlanta, GA 30328

CANCELLED September 1, 2008 Missouri Public Service Commission

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

FILED Missouri Public Service Commission

-

-

З.

MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

Missouri PSC Tariff No. 1 1st Revised Page No. 59.9.1 Cancels Original Page No. 59.9.1

Local Exchange Service З. Service Descriptions (Cont'd) 3.1.7.7 Local and Long Distance Line Solutions' Local Line Optional Features Feature Package 1 Feature Package 2 Call Waiting/Cancel Call Waiting Caller ID with Name and Number Remote Call Forwarding Remote Access to Call Forwarding (Plan 2) N Vanity Number Monthly Recurring Charge Feature Package 1 \$3.50 Feature Package 2 \$6.50 Remote Access to Call Forwarding (Plan 2) \$7.00 N Discounts:² These discounts are identical to, and shall not be in addition to, discounts applicable to Companion Intrastate Service and Companion Interstate Service.

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must: 1) demonstrate to the Company's reasonable satisfaction that it will accept another company's offer in the absence of any further inducement, and 2) commit to a new minimum term of service that equals or exceeds 1 year for up to a 5% discount, 2 years for up to a 25% discount, and 3 years for up to a 15% discount

A discount will be provided on the monthly recurring plan charge and monthly recurring charges for optional features and feature packages, in lieu of all other discounts, in response to competitive marketplace conditions. To be eligible for this discount, Customer must; 1) demonstrate to the Company's reasonable satisfaction that it will accept another exchange carrier's offer in absence of any further inducement, and 2) commit to a new term of service that equals or exceeds 1 year for up to a 20% discount, 2 years for up to a 25 % discount, and 3 years for up to a 25% discount.

¹Effective January 16, 2004 this service will no longer be available to new customers ²Effective August 1, 2006, these discounts will no longer be available to new customers.

Issued: February 6, 2007

CANCELLED September 1, 2008 Missouri Public Service Commission Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective March 8, 2007

Filed Missouri Public Service Commission

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 60

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions</u> (Cont d)

3.4 Directory Listings

MO. PUBLIC SERVICE COMM

SFP 17 1998

The Company shall provide for a single directory listing, termed the primary listing, in the telephone directory published by the dominant exchange service provider in the Customer's exchange area of the Station number which is designated as the Customer's main billing number. Directory listings of additional Company Station numbers, other than the Customer's main billing number, associated with a Customer's service will be provided for a monthly recurring charge per listing.

- 3.4.1 The Company reserves the right to limit the length of any listing in the directory by the use of abbreviations when, in its judgment, the clearness of the listing or the identification of the Customer is not impaired thereby. Where more than one line is required to properly list the Customer, no additional charge is made.
- 3.4.2 The Company may refuse a listing which is known not to constitute a legally authorized or adopted name, obscenities in the name, or any listing which, in the opinion of the Company, is likely to mislead or deceive calling persons as to the identity of the listed party, or is a contrived name used for advertising purposes or to secure a preferential position in the directory or is more elaborate than is reasonably necessary to identify the listed party. The Company, upon notification to the Customer, will withdraw any listing which is found to be in violation of its rules with respect thereto.
- 3.4.3 Each listing must be designated Government or Business to be placed in the appropriate section of the directory. In order to aid the user of the directory, and to avoid misleading or deceiving the calling party as to the identity of the listed party, only business listings may be placed in the Business Section and only residential listings in the Residential section. The Company, upon notification to the Customer, will withdraw any listing which is found to be in violation of its rules with respect thereto.
- 3.4.4 In order for listings to appear in an upcoming directory, the Customer must furnish the listing to the Company in time to meet the directory publishing schedule.

GANCELLED TAF NR-575 . uplic Service Commission MISSOURI

ISSUED : March 2, 1999

t.

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

<u>96-355</u>

FILFD MAY 1.4 1999

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

-

3.

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 60

LOCAL EXCHANGE SERVICE

3. <u>Service Descriptions</u> (Cont d)

3.4 Directory Listings

MO. PUBLIC SERVICE COMM

SFP 17 1998

The Company shall provide for a single directory listing, termed the primary listing, in the telephone directory published by the dominant exchange service provider in the Customer's exchange area of the Station number which is designated as the Customer's main billing number. Directory listings of additional Company Station numbers, other than the Customer's main billing number, associated with a Customer's service will be provided for a monthly recurring charge per listing.

- 3.4.1 The Company reserves the right to limit the length of any listing in the directory by the use of abbreviations when, in its judgment, the clearness of the listing or the identification of the Customer is not impaired thereby. Where more than one line is required to properly list the Customer, no additional charge is made.
- 3.4.2 The Company may refuse a listing which is known not to constitute a legally authorized or adopted name, obscenities in the name, or any listing which, in the opinion of the Company, is likely to mislead or deceive calling persons as to the identity of the listed party, or is a contrived name used for advertising purposes or to secure a preferential position in the directory or is more elaborate than is reasonably necessary to identify the listed party. The Company, upon notification to the Customer, will withdraw any listing which is found to be in violation of its rules with respect thereto.
- 3.4.3 Each listing must be designated Government or Business to be placed in the appropriate section of the directory. In order to aid the user of the directory, and to avoid misleading or deceiving the calling party as to the identity of the listed party, only business listings may be placed in the Business Section and only residential listings in the Residential section. The Company, upon notification to the Customer, will withdraw any listing which is found to be in violation of its rules with respect thereto.
- 3.4.4 In order for listings to appear in an upcoming directory, the Customer must furnish the listing to the Company in time to meet the directory publishing schedule.

GANCELLED TAF NR-575 . uplic Service Commission MISSOURI

ISSUED : March 2, 1999

t.

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

<u>96-355</u>

FILFD MAY 1.4 1999

CANCELLED September 1, 2008 Missouri Public Service Commission Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

ALL MATERIAL ON THIS PAGE HAS BEEN REMOVED

З.

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601 Effective: September 1, 2008

ï

MCImetro ACCESS TRANSMISSION SERVICES, LLC

3.

P>	-	\sim	MISSOURI P.S.C. TARIFF NO. 1 ORIGINAL PAGE NO. 61	
r -	7	٢.	ORIGINAL PAGE NO. 61	

LOCAL EXCHANGE SERVICE

SEP 17 1998

3.4 Directory Listings (Cont'd)

Service Descriptions (Cont'd)

MO. FUDLIL DERVICE UNINIM

- 3.4.5 Directory listings are provided in connection with each Customer service as specified herein.
 - 3.4.5.1 <u>Primary Listing</u>: A primary listing contains the name of the Customer, or the name under which a business is regularly conducted, as well as the address and telephone number of the Customer. This listing is provided at no additional charge.
 - 3.4.5.2 <u>Additional Listings</u>: In connection with business service, additional listings are available only in the names of Authorized Users of the Customer's service, as defined herein. Rates for additional listings are specified in Sections 3.4.5.8 and 3.4.5.9.
 - 3.4.5.3 <u>Nonpublished Listings</u>: Listings that are not printed in directories nor available from Directory Assistance.

A Nonpublished Telephone Service will be furnished, at the Customer's request providing for the omission or deletion of the Customer's telephone listing from the telephone directory and, in addition, the Customer's telephone listing will be omitted or deleted from the directory assistance records, subject to the provisions set forth in Section 2.1.4. Rates for Nonpublished Listings are specified in Sections 3.4.5.8 and 3.4.5.9.

- 3.4.5.4 <u>Nonlisted Numbers</u>: A Nonlisted number will be furnished at the Customer's request, providing for the omission or deletion of the Customer's listing from the telephone directory. Such listings will be carried in the Company's directory assistance and other records and will be given to any calling party.
- 3.4.5.5 <u>Foreign Listings</u>: Where available, a listing in a phone directory which is not in the Customer's immediate calling area. The Customer will be charged the rates specified in the tariff published by the specific exchange carrier providing the Foreign Listing.
- 3.4.5.6 <u>Alternate Call Listings</u>: Where available, a listing which references a telephone number which is not the primary listing for the Customer. The Customer must provide written verification that the alternate telephone number is authorized to accept calls.
- 3.4.5.7 <u>Information Listings</u>: Where available, additional lines of information which may be included with a primary, additional or reference listings.



ISSUED : March 2, 1999

٤

EFFECTIVE: May 14, 1999

Missouri Public Service Commission

FILFD MAY LA 1999

96-355

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328

CANCELLED September 1, 2008 Missouri Public Service Commission MCImetro Access Transmission Services LLC d/b/a Verizon Access Transmission Services

LOCAL EXCHANGE SERVICE

CERTAIN MATERIAL ON THIS PAGE HAS BEEN REMOVED

3.4 <u>Emergency Services (Enhanced 911)</u>: Allows Customers to reach appropriate emergency services including police, fire and medical services. Enhanced 911 has the ability to selectively route an emergency call to the primary E911 provider so that it reaches the correct emergency service located closest to the caller. In addition, the Customer's address and telephone information will be provided to the primary E911 provider for display at the Public Service Answering Point (PSAP).

MCImetro is obligated to supply the E911 service provider in MCImetro service area with accurate information necessary to update the E911 database at the time MCImetro submits customer orders to the local exchange telecommunications company whose service is being resold pursuant to these tariffs. At the time MCImetro provides basic local service to a customer by means of MCImetro own cable pair, or over any other exclusively owned facility, MCImetro will be obligated to make the necessary equipment of facility additions in the E911 service provider's equipment in order to accurately and properly update the database for E911.

MCImetro will be obligated to provide facilities to route calls from the end users to the proper Public Safety Answering Point. MCImetro recognizes the authority of the E911 customer to establish service specifications and grant final approval or denial of service configurations offered by MCImetro. MCImetro will collect 911 surcharges and remit all surcharge revenue to the appropriate governmental entity pursuant to RSMo 190.310.

3.6 <u>Presubscription:</u> PIC-2 allows Customers to presubscribe to their carrier of choice for interLATA and intraLATA tolls calls, without dialing the Access Code. The following charge applies each time the Customer requests a change to their PIC. This charge applies per line or per trunk for each Local Line or Local Trunk PIC change requested, subsequent to the initial designation:

PIC-2 Change \$1.49 (per line or per trunk)

Issued: July 31, 2008

Effective: September 1, 2008

Carmen L. Feliciano Tariff Administrator 205 N. Michigan Avenue Chicago, IL 60601

CANCELLED - Missouri Public Service Commission - 08/07/2023 - LN-2024-0032 - YC-2024-0026

З.

MISSOURI P.S.C. TARIFF NO. 1

LOCAL EXCHANGE SERVICE

3. Service Descriptions (Cont'd)

SEP 17 1998

3.4 Directory Listings (Cont'd)

MO. FUDLIC SERVICE COMM

3.4.5.8 <u>Non-Recurring Charges</u>: Non- Recurring Charges associated with Directory Listings are as follows:

	<u>St Louis/Kansas City</u>	<u>Springfield</u>
Primary Listing	N/C	N/C
Addt'l Listing (Per Listing)	\$9.50	\$9.50
Non-Listed Number (Per Number)	\$6.00	\$6.00
Non-Published Number (Per Number)	\$6.00	\$6.00

3.4.5.9 <u>Recurring Charges</u>: Monthly Recurring Charges associated with Directory Listings are as follows:

Primary Listing	N/C	N/C
Addt'i Listing (Per Listing)	\$2.45	\$2.45
Non-Listed Number (Per Number)	\$1.20	\$1.20
Non-Published Number (Per Number)	\$1.60	\$1.60
Alternate Call Listing	\$2.45	\$2.45

3.4 Emergency Services (Enhanced 911): Allows Customers to reach appropriate emergency services including police, fire and medical services. Enhanced 911 has the ability to selectively route an emergency call to the primary E911 provider so that it reaches the correct emergency service located closest to the caller. In addition, the Customer's address and telephone information will be provided to the primary E911 provider for display at the Public Service Answering Point (PSAP).

MCImetro is obligated to supply the E911 service provider in MCImetro service area with accurate information necessary to update the E911 database at the time MCImetro submits customer orders to the local exchange telecommunications company whose service is being resold pursuant to these tariffs. At the time MCImetro provides basic local service to a customer by means of MCImetro own cable pair, or over any other exclusively owned facility, MCImetro will be obligated to make the necessary equipment of facility additions in the E911 service provider's equipment in order to accurately and properly update the database for E911.

MCImetro will be obligated to provide facilities to route calls from the end users to the proper Public Safety Answering Point. MCImetro recognizes the authority of the E911 customer to establish service specifications and grant final approval or denial of service configurations offered by MCImetro. MCImetro will collect 911 surcharges and remit all surcharge revenue to the appropriate governmental entity pursuant to RSMo 190.310.

<u>Presubscription</u>: PIC-2 allows Customers to presubscribe to their carrier of choice for interLATA and intraLATA tolls calls, without dialing the Access Code. The following charge applies each time the Customer requests a change to their PIC. This charge applies per line or per trunk for each Local Line or Local Trunk PIC change requested, subsequent to the initial designation:

PIC-2 Change \$1.49 (per line or per trunk)

CANCELLED their subsc AUG 2 0 1999 Ey TA 8-575 Public Service Commission MISSOURI ISSUED : March 2, 1999

Ĺ

3.6

EFFECTIVE: May 14, 1999

<u>96-3</u>55

Sandy Chandler Tariff Manager Six Concourse Parkway, Suite 3200 Atlanta, GA 30328 Missouri Public Service Commission FILFD MAY 1 4 1999

CANCELLED September 1, 2008 Missouri Public Service Commission